



PLAN CHANGE 34 NETWORK UTILITIES AND RENEWABLE ENERGY GENERATION HUTT CITY DISTRICT PLAN

and

PLAN CHANGE 38 NETWORK UTILITIES AND RENEWABLE ENERGY GENERATION UPPER HUTT DISTRICT PLAN

DECISION





Commissioners' Recommendation

PLAN CHANGE 34 NETWORK UTILITIES AND RENEWABLE ENERGY GENERATION HUTT CITY DISTRICT PLAN PLAN CHANGE 38 NETWORK UTILITIES AND RENEWABLE ENERGY GENERATION UPPER HUTT DISTRICT PLAN

Report and Recommendation of M St.Clair, L Bridson and G McArthur Acting as Commissioners jointly appointed by the Hutt City Council and Upper Hutt City Council pursuant to Section 34A of the Resource Management Act 1991

1. INTRODUCTION

Context

- 1.1 We were jointly appointed by the Hutt City Council and Upper Hutt City Council to hear submissions to, and to consider and make recommendations on, Plan Change 34 and Plan Change 38 (the Plan Changes), which amend the Network Utilities section of and add a new Renewable Energy Generation section to the Hutt City Council District Plan and the Upper Hutt City Council District Plan.
- 1.2 The Plan Changes have been the subject of a Council "section 32" report, consultation with infrastructure providers, land owners and occupiers, and of course the recent public notification and hearing, culminating in this report.
- 1.3 Before discussing the details of the Plan Changes and the submissions to them, there are some procedural issues that we need to address.

Report Outline

- 1.4 In terms of the above, having familiarised ourselves with the Plan Changes and the background material, read all submissions, conducted the hearing and heard from the Council officers and submitters, we hereby record our recommendations. In this respect, this report is divided into the following parts:
 - (a) Background/Plan Changes Outline:

This section includes an outline of the background to the Plan Changes, including the sequence of events leading to this report. It also outlines the main components of the Plan Changes including an overview of the purpose. This background section provides a relevant context to considering each of the submissions to the plan changes.

(b) Statutory Requirements:

This section sets out the statutory requirements under the Resource Management Act 1991 (the RMA) that govern the decision making process in regard to the Plan Changes.

(c) Assessment of Submissions:

In this section, we record the various submissions received to the Plan Changes, outline the concerns of the submitters to the Plan Changes, and, where relevant, amplify on the evidence/statements presented at the hearing. We then undertake an assessment of the aspects of each of the submissions or groups of submissions and conclude with a recommendation on each Plan Change.

2. BACKGROUND

Procedural Sequence

- 2.1 The background to the Plan Changes is set out in full in the Officer's Report and the proposed Plan Change documentation, and is held on the respective Councils' file. Hence we will not repeat that in detail here.
- 2.2 Hutt City Council, Plan Change 34 itself was publicly notified on 2 December 2014 with the submission period closing on 30 January 2015. There were no late submissions. The summary of those submissions was notified on 8 April 2015, with further submissions closing on 21 April 2015. Eleven submissions were received and two further submissions were received. We note that part of the submission by Wellington Electricity Lines Limited (HS9) was withdrawn on 7 April 2015. Hence we have not considered those parts of their submission.
- 2.3 Upper Hutt City Council, Plan Change 38 itself was publicly notified on 3 December 2014 with the submission period closing on 30 January 2015. There were no late submissions. The summary of those submissions was notified on 8 April 2015, with further submissions closing on 22 April 2015. Nine submissions were received and three further submissions were received. We note that part of the submission by Wellington Electricity Lines Limited (HS9) was withdrawn on 7 April 2015. Hence we have not considered those parts of their submission.
- 2.4 Prior to the hearing commencing on 20 August 2015, we issued a series of minutes. The first minute, dated 26 May 2015, set out directions for the pre-circulation of the section 42A report, submitter's expert evidence, and conferencing between experts. The second minute, dated 3 August 2015, drew to the attention of the parties two preliminary procedural issues relating to the hearing date and an extension in the timetable for the lodging of expert evidence. The third minute, dated 5 August 2015, was in response to a request by Greater Wellington Regional Council for an extension in the timetable to lodge expert evidence. These extension requests were granted. Copies of these minutes were circulated to all parties and are held on Council's file.
- 2.5 As a result of the conferencing between experts the range of matters still in contention at the commencement of the hearing had been significantly reduced and the Commissioners recognize the efforts of the parties in that regard.

The Hearing

2.6 The hearing was convened on 20 August 2015 in The Dowse Art Museum (James COE 1 Room) at 45 Laings Road, Lower Hutt. We heard from the following people during the course of the hearing:

Submitters

 Ms. Caroline Ammundsen – Policy Advisor - Greater Wellington Regional Council

- Ms. Sharyn Westlake Senior Engineer, Flood Protection Department, Greater Wellington Regional Council
- Ms. Mary Barton Senior Environmental Planner Chorus New Zealand Ltd
- Mr. Graeme McCarrison Spark New Zealand Trading Limited
- Mr. Thomas Anderson Senior Resource Management Consultant at Incite for Chorus New Zealand Limited and Spark New Zealand Trading Limited
- Ms. Frances Wedde Counsel Transpower New Zealand Ltd
- Mr. Dougall Campbell Group Manager Environmental Policy Transpower New Zealand Ltd
- Ms. Pauline Whitney Associate Principal and Senior Planner at Boffa Miskell Ltd – consultant planner to Transpower New Zealand Ltd
- Ms. Julie Silvester Submitter
- Mr. Warren Thessman- Submitter.

Council Officers

- Ms. Gina Sweetman, Consultant Planner for Hutt City Council and Upper Hutt City Council s42A Reporting officer
- Mr. Andrew Cumming Divisional Manager Environmental Policy, Hutt City Council
- Ms. Nicola Etheridge Policy Manager at Upper Hutt City Council
- Ms. Felicity Boyd Policy Planner at Upper Hutt City Council
- Mrs. Emma Manohar, Senior Solicitor, DLA Piper for Hutt City Council
- 2.7 The hearing commenced with a presentation by the reporting officer for both Plan Changes, Ms. Sweetman. Ms. Sweetman outlined the matters that had changed since the receipt and distribution of the submitter's expert evidence. We then heard from the submitters, identified above. We record that representatives of the Royal Forest and Bird Society were not able to attend the hearing and that Ms. Sweetman read out a statement on behalf of that submitter.
- 2.8 Ms. Sweetman then responded to issues that had arisen during the hearing. We exercised the opportunity to question all persons present.
- 2.9 At that point, we adjourned the hearing at 3:56pm on Thursday 20 August 2015 indicating that we wished to consider whether or not we had all the information required to make our recommendations. On the 25th of August 2015, we issued a Minute (Minute #4) advising the parties that we had no further information requirements. As such the hearing was officially closed as at 25 August 2015.

Outline of Plan Changes

- 2.10 As mentioned above, the purpose of the Plan Changes is set out fully in the reporting officer's report which is held on the Council file. In summary, the policy aims of the Plan Changes are to:
 - *a) review and update existing network utility provisions to ensure that they:*
 - reflect best practice;
 - provide greater consistency between district plans across the region and

• best serve and reflect public and stakeholder expectations and requirements;

b) avoid unnecessary duplication between District Plan provisions and other legislation or regulations;

c) give effect to the National Policy Statement on Electricity Transmission (NPSET) (Hutt City only);

d) give effect to the National Policy Statement on Renewable Electricity Generation (NPSREG);

e) give effect to the Regional Policy Statement for the Wellington Region (RPS); and *f*) amend the District Plan as required so as not to duplicate or conflict with the National Environmental Standard on Electricity Transmission Activities (NESETA) and the National Environmental Standard on Telecommunication Facilities (NESTF).¹

3. STATUTORY PROVISIONS

The Resource Management Act 1991

3.1 In this section of the report we set the statutory provisions we are required to take account of in reaching our recommendation.

Schedule 1

- 3.2 Part 1 of Schedule 1 to the RMA sets out the procedure for council initiated plan changes.
- 3.3 Directions on decisions are set out in clause 10 of Schedule 1 of the RMA, which states:

10 Decisions on provisions and matters raised in submissions

- (1) A local authority must give a decision on the provisions and matters raised in submissions, whether or not a hearing is held on the proposed policy statement or plan concerned.
- (2) The decision—
 - (a) must include the reasons for accepting or rejecting the submissions and, for that purpose, may address the submissions by grouping them according to—
 - *(i)* the provisions of the proposed statement or plan to which they relate; or
 - (ii) the matters to which they relate; and
 - (b) may include—
 - *(i)* matters relating to any consequential alterations necessary to the proposed statement or plan arising from the submissions; and
 - *(ii) any other matter relevant to the proposed statement or plan arising from the submissions.*
- (3) To avoid doubt, the local authority is not required to give a decision that addresses each submission individually.
- (4) The local authority must—

¹ Section 42A (S42) Report, Page 4, Para 4

(aaa) have particular regard to the further evaluation undertaken in accordance with subclause (2)(ab) when making its decision; and

- (a) give its decision no later than 2 years after notifying the proposed policy statement or plan under clause 5; and
- (b) publicly notify the decision within the same time.
- 3.4 Matters to be considered in any plan changes are set out in section 74 of the RMA as follows:

74 Matters to be considered by territorial authority

- (1) A territorial authority must prepare and change its district plan in accordance with—
 - (a) its functions under <u>section 31</u>; and
 - (b) the provisions of <u>Part 2</u>; and
 - (c) a direction given under <u>section 25A(2)</u>; and
 - (d) *its obligation (if any) to prepare an evaluation report in accordance with <u>section 32</u>; and*
 - (e) its obligation to have particular regard to an evaluation report prepared in accordance with <u>section 32</u>; and
 - (f) any regulations.
- (2) In addition to the requirements of <u>section 75(3) and (4)</u>, when preparing or changing a district plan, a territorial authority shall have regard to—
 - (a) **any**
 - (i) proposed regional policy statement; or
 - (ii) proposed regional plan of its region in regard to any matter of regional significance or for which the regional council has primary responsibility under <u>Part 4</u>; and
 - (b) any—
 - *(i)* management plans and strategies prepared under other Acts; and
 - (ii) [Repealed]
 - (iia) relevant entry on the New Zealand Heritage List/Rārangi Kōrero required by the <u>Heritage New Zealand Pouhere Taonga Act 2014</u>; and
 - (iii) regulations relating to ensuring sustainability, or the conservation, management, or sustainability of fisheries resources (including regulations or bylaws relating to taiapure, mahinga mataitai, or other non-commercial Maori customary fishing),—

to the extent that their content has a bearing on resource management issues of the district; and

- (c) the extent to which the district plan needs to be consistent with the plans or proposed plans of adjacent territorial authorities.
- (2A)A territorial authority, when preparing or changing a district plan, must take into account any relevant planning document recognised by an iwi authority and lodged with the territorial authority, to the extent that its content has a bearing on the resource management issues of the district.
- (3) In preparing or changing any district plan, a territorial authority must not have regard to trade competition or the effects of trade competition.

Commissioners' Report and Recommendation

- 3.5 Finally, section 75 of the RMA states that:
 - (3) A district plan must give effect to—
 - (a) any national policy statement; and
 - (b) any New Zealand coastal policy statement; and
 - (c) any regional policy statement.
 - (4) A district plan must not be inconsistent with—
 - (a) a water conservation order; or
 - (b) a regional plan for any matter specified in section 30(1).

Part 2 Matters

- 3.6 The purpose of the RMA is to promote the sustainable management of natural and physical resources (Section 5). This means managing the use of natural and physical resources in a way that enables people and communities to provide for their social, cultural and economic well-being while sustaining those resources for future generations, protecting the life-supporting capacity of ecosystems and avoiding, remedying or mitigating adverse effects on the environment.
- 3.7 Section 6 contains a list of matters of national importance that all persons exercising functions and powers under shall recognise and provide for.
- 3.8 Section 7 addresses 'other matters' which, in achieving the purpose of the RMA, persons exercising functions and powers under the RMA shall have particular regard to. Those matters of particular relevance to the Plan Changes, as noted in the reporting officer's report are:
 - (b) the efficient use and development of natural and physical resources
 - (ba) the efficiency of the end use of energy
 - (c) the maintenance and enhancement of amenity values
 - *(f) maintenance and enhancement of the quality of the environment*
- 3.9 Section 8 provides that in achieving the purpose of the RMA, all persons exercising functions and powers under it, in relation to managing the use, development and protection of natural and physical resources, shall take into account the principles of the Treaty of Waitangi (Te Tiriti O Waitangi).
- 3.10 We note that the Wellington Tenths Trust, the Port Nicholson Settlement Trust, Orongomai Marae, Te Runanganui O Taranaki Whanui ke te Upoko o te Ika a Maui (Waiwhetu Marae) and Te Rūnanga o Toa Rangātira Inc and were consulted in preparation of the Plan Changes. We were advised that Wellington Tenths Trust, the Port Nicholson Settlement Trust, and Orongomai Marae, had no particular comment on the pre-notification draft Plan Changes². We also note that none of

² S42A, Para 16, Page 7

those parties lodged submissions to the proposed Plan Changes and that no submissions were received from any other iwi authority.

Sections 31, 32, 72 & 76 of the RMA

- 3.11 Section 31 sets out the Councils' functions for the purpose of giving effect to the RMA. The Councils' functions are stated in section 31 of the RMA and include:
 - 31(1)(b) the control of any actual or potential effects of the use, development or protection of land.
- 3.12 Section 32 of the RMA requires a report which summarises the process of evaluation undertaken in the preparation of a Plan Change. A Section 32 evaluation must examine the following:
 - (1) An evaluation report required under this Act must—
 - (a) examine the extent to which the objectives of the proposal being evaluated are the most appropriate way to achieve the purpose of this Act; and
 - (b) examine whether the provisions in the proposal are the most appropriate way to achieve the objectives by—
 - *(i) identifying other reasonably practicable options for achieving the objectives; and*
 - (ii) assessing the efficiency and effectiveness of the provisions in achieving the objectives; and
 - (iii) summarising the reasons for deciding on the provisions; and
 - (c) contain a level of detail that corresponds to the scale and significance of the environmental, economic, social, and cultural effects that are anticipated from the implementation of the proposal.
- 3.13 An evaluation must also take into account:
 - (2) An assessment under subsection (1)(b)(ii) must—
 - (a) identify and assess the benefits and costs of the environmental, economic, social, and cultural effects that are anticipated from the implementation of the provisions, including the opportunities for—
 - (i) economic growth that are anticipated to be provided or reduced; and
 - (ii) employment that are anticipated to be provided or reduced; and
 - (b) if practicable, quantify the benefits and costs referred to in paragraph (a); and
 - (c) assess the risk of acting or not acting if there is uncertain or insufficient information about the subject matter of the provisions.
- 3.14 Section 32 requires that a report must be prepared, summarising the evaluation and giving reasons. The section 32 requirements of the RMA were addressed in the officer's report and the changes to the documentation provided at the hearing.

- 3.15 In addition, we note that as the further submission process closed after the 21st of April (PC34) and the 22nd of April 2015 (PC38), Section 32AA requires that we undertake a further evaluation of any changes to the Plan Changes and refer to that evaluation in this recommendation report³.
- 3.16 We note that in response to questioning, Ms. Sweetman was of the view that the section 32 evaluation met the requirements of the RMA. In addition, Ms. Whitney addressed the issue of section 32 in her evidence as it related to aspects of Transpower's issues⁴. We did not hear any evidence to the contrary. We record that Ms. R Brown commented on the s32 report in her submission (US9.1-US9.7). For completeness we address those submission points now and agree with Ms. Sweetman⁵ recommending that they be accepted in part insofar as they are noted.
- 3.17 In reaching our recommendations, we record that we have considered the changes in accordance with section 32(1) to (4) of the RMA and have done so in a level of detail that correspondences to the scale and significance of the changes.
- 3.18 Section 72 states as follows:

The purpose of the preparation, implementation, and administration of district plans is to assist territorial authorities to carry out their functions in order to achieve the purpose of this Act.

- 3.19 The following provisions of section 76 are also relevant:
 - (1) A territorial authority may, for the purpose of –
 (a) Carrying out its functions under this Act; and
 - (b) Achieving the objectives and policies of the plan, include rules in a district plan.
 - (3) In making a rule, the territorial authority shall have regard to the actual or potential effect on the environment of activities including, in particular, any adverse effect...
- 3.20 In relation to the statutory requirements, and the evidence and reports presented, our findings and recommendations are set out below.

4. OVERVIEW OF SUBMISSIONS TO PLAN CHANGES 34(HCC) and 38 (UHCC)

Numbers and Categories

4.1 For Plan Change 34 (Hutt City) there were 11 submissions and two further submissions received within the statutory time period. As we noted in paragraph 2.2 above, parts of one submission were later withdrawn.

³ Resource Management Act 1991, Schedule 12, Clause 2

⁴ Ms. P Whitney, Evidence in Chief (EIC), Pages 27 - 29, Paras 66 - 77

⁵ S42A Report, Page 18, Para 79

- 4.2 The following list sets out the submitters and further submitters for PC34:
 - New Zealand Transport Agency (NZTA) (HS1)
 - Greater Wellington Regional Council (GWRC) (HS2);
 - KiwiRail Holdings Limited (KiwiRail) (HS3);
 - Mary Beachen (HS4);
 - Warren Thessman (HS5);
 - Transpower New Zealand Limited (Transpower)(HS6);
 - Spark New Zealand Trading Limited (Spark) (HS7);
 - Chorus New Zealand Limited (Chorus) (HS8);
 - Wellington Electricity Lines Limited (WELL) (HS9);
 - Powerco Limited (Powerco) (HS10);
 - Julie Sylvester (HS11).

Further submissions were received from:

- Transpower New Zealand (HFS1):
- Powerco Limited (HFS2).
- 4.3 For Plan Change 38 (Upper Hutt City) there were 9 submissions and three further submissions received within the statutory time period. As we noted in paragraph 2.3 above, one submission was later partly withdrawn.
- 4.4 The following list sets out the submitters and further submitters for PC34:
 - GWRC (US1);
 - KiwiRail (US2);
 - Spark (US3);
 - Transpower (US4);
 - Chorus (US5);
 - WELL (US6)
 - Powerco (US7);
 - Royal Forest and Bird Protection Society of New Zealand Incorporated Upper Hutt Branch (US8);
 - Roz Brown (US9).

Further submissions were received from:

- Powerco Limited (UFS3);
- Transpower New Zealand Limited (UFS2);
- WEL (UFS3).
- 4.5 In reviewing the submissions we have found it useful to follow the format set out by Ms. Sweetman in her Section 42A(S42A) hearing's report. We have categorised the submissions to PC34 and PC 38 in the following manner noting that this list includes matters that may no longer be in contention:
 - A. Plan Changes in its entirety/ general submissions points

Commissioners' Report and Recommendation

- i. Plan Change 34
- ii. Plan Change 38 Section 32
- iii. Plan Change 34 Renewable Energy
- B. Joint Matters between Plan Changes New Network Utilities Chapter General
 - i. Definitions Maintenance, Minor Upgrading, Upgrading
 - ii. Regionally significant network utilities
 - iii. Stopbanks / Flood Hazard Area
 - iv. Transmission Line Rules and Standards
 - v. Masts Rules and Standards
 - vi. Cabinets Rules and Standards
 - vii. Matters of Discretion
- C. Plan Change 34 Network Utilities Chapter and the National Grid
 - i. Definitions
 - ii. Issues Objectives and Policies
 - iii. Rules and Standards
- D. Plan Change 38 Network Utilities
 - i. Definitions
 - ii. Issues, objectives, policies, methods and anticipated environmental results
 - iii. Zone Rules
 - iv. Network Utilities Rules and Standards
- E. Plan Change 38 Renewable Energy Generation
 - i. Issues, objectives and policies
 - ii. Rules and Standards
- 4.6 We have used these categories as the basis for the assessment that follows. In addition, we also record that through the hearing process, both in the pre-circulated hearing report and evidence, and at the hearing itself, amendments to the Plan Changes were agreed to by the parties. We have recorded above that we helpfully received from Ms. Sweetman a copy of the matters that were still in contention as the final piece of information before closing the hearing. It is these matters in contention that we have focused our deliberations on and set out in the recommendations. That said, where the parties agreed on matters, we record that we have considered those matters fully in reaching our recommendations and adopt as our reasons those provided in the hearings report and the evidence presented to us. Regardless, our recommendations address all the submissions and further submissions made on the Plan Changes and are set out in Table 1 for Hutt City Council and Table 2 for Upper Hutt City Council and are attached as Appendix 1 and Appendix 2 respectively.

Preliminary Issues

4.7 The preliminary issues relate to a number of matters "of scope" of aspects of the submissions to the plan changes, and therefore the scope of our jurisdiction to assess them. The four relevant submissions are Powerco, Wellington Electricity Limited (WELL), Transpower and Ms. Sylvester. We note that prior to the hearing Powerco and WELL reached agreement with the Councils in regard to the scope issue related to their submissions and are no longer pursuing that relief⁶.

⁶ Reporting Officer Speaking Notes dated 20 August 2015, Section 2.1 and 3.1, Page1

- 4.8 The Transpower submissions identified in the s42A report as being out of scope (Submission Nos. 4.7 4.13 on PC38) were withdrawn at the hearing⁷. However, in relation to PC34, submission 6.45, Ms. Wedde, counsel for Transpower argued that Amendments 61-64 which related to earthworks in the National Grid Yard were within scope of PC34. HCC had sought legal advice on this matter and Mrs. Manohar, on behalf of HCC, presented arguments to us at the hearing that aspects of the submission were not within scope.
- 4.9 Firstly we note the submission point was in two parts, and Ms. Sweetman, in responding to matters raised in the hearing, rightly drew our attention to the fact that in the s42A Report, she had left out that part of the submission point which related to a request for a change from a restricted discretionary activity status for building and structures to that of non-complying. Ms. Sweetman recommended that this part of submission be accepted. We concur with that recommendation.
- 4.10 The next matter is the remainder of the submission 6.45 which we note primarily relates to earthworks around transmission towers. We note that Ms. Wedde and Mrs. Manohar were in agreement as to the applicable case law⁸. Ms. Wedde set out in submissions arguments to show that remainder of submission 6.45 met both limbs of the High Court's bipartite test as established in the Clearwater case. Mrs. Manohar agreed with Ms. Wedde that both limbs of the test needed to be meet. However, Mrs. Manohar at the hearing and in the advice to Council included in the s42A Report, was of the view that neither limb had been met. In particular we note Mrs. Manohar's argument that, "there is a real risk that people who may be directly affected by the proposed rules may have been denied an opportunity to respond to the changes."9 We questioned both Ms. Wedde and Mrs. Manohar on the details of their respective positions in some detail. In conclusion we find Mrs. Manohar's argument persuasive. We are not convinced that all parties affected by the change had the opportunity to participate in this matter.
- 4.11 As such we find that least one limb of the test has not been met and therefore those aspects of the submission are beyond the scope of the Plan Change. In determining this it follows that we have not addressed that relief sought in the substantive part of our discussion.
- 4.12 Ms. Sylvester, in submission HS11.1, questioned what consideration had been given to the effects on animals in rural areas where the Plan Change has effects. At the hearing, Ms. Sylvester clarified that her concern was in regard to the use of helicopters used by Transpower to check transmission lines and lack of notice given to land owners. The situation Ms. Sylvester described resulted in horses being scared by the helicopter and putting riders at risk. This being an operational issue and Transpower representatives being present, we asked that the parties meet outside of the hearing process to address the matter. We note that the parties agreed to this. We also explained to Ms. Sylvester that as an operational issue it

⁷ Ms F Wedde, Legal Submissions, Page 3, Para 10.

⁸ Palmerston North City Council v Motor Machinists Ltd [2013] NZHC 1290, [2014] NZRMA 519 and Clearwater Resort ltd v Christchurch City Council HC Christchurch AP 34/02.

⁹ Appendix 4, Section 42 A Report – Legal Opinion – DLA Piper Letter, dated 2 July 2015

was not matter that the Plan Change could really address and was therefore out of the scope of the Plan Change. We understood that Ms. Sylvester accepted this position. For completeness, we record that we recommend that that part of Submission 11.1 be rejected.

Plan Change 34 in its entirety

- 4.13 Ms. M Beachen (HS4.1) sought that Transpower purchase her property (137 Tirohanga Rd) as the building restrictions beneath the power lines would prevent building a dwelling on the site. The Officer's report notes that in a further submission Transpower (HFS1.1) sought that HS4.1 be rejected as provisions did not prohibit the building of a dwelling and that the request to purchase the property was beyond the scope of the Plan Change¹⁰.
- 4.14 Ms. Beachen did not attend the hearing.
- 4.15 The reporting officer's assessment of these submissions notes in summary, that:
 - Under s85 of the RMA, Council cannot require the purchase of the property;
 - PC34 would not prevent reasonable use of the property; and
 - Transmission lines had been in place since 1977 and restriction on development has been in place since that time.
- 4.16 Ms. Whitney in her evidence on behalf of Transpower addressed the matters raised by Ms. Beachen, noting that the restrictions on development already apply under NZECP34.¹¹
- 4.17 Having considered the material put before us, we prefer the evidence of Ms. Sweetman and Ms. Whitney for the reasons set out above. As such we recommend that the submission of Ms. Beachen (HS4.1) be rejected and the further submission point of Transpower (HFS1.1) be accepted.
- 4.18 Mr. W Thessman HS5.1 noted that there was *"too much struck data, exclusions, dimensions etc."* and was concerned that the implications of the Plan Change had not been understood by the Council. Mr. Thessman reinforced his concerns in his presentation at the hearing. In the hearing report, Ms. Sweetman expressed the view that the rationale for the changes had been fully set out in the S32 report and reflected the careful consideration of matters by Council. We concur with Ms. Sweetman's analysis and therefore recommend that HS5.1 be rejected.

Plan Change 34 - Renewable Energy

- 4.19 Mr. Thessman's submission HS5.2 was concerned that the provisions on renewable energy were not specific enough and Mr. Thessman had particular concerns around;
 - The use of sun trapped electronics is missing
 - What happens when small home solar panels are above the roof;
 - There is no comment on sun reflection;

¹⁰ S42A Report, Page 16, Paras 66-67

¹¹ Ms. P Whitney, EIC, Page 41, Paras 164-168

- Should vertical axis wind generators be allowed on top of houses?¹²
- 4.20 Mr. Thessman's submission (HS5.3), also raised concerns that PC34 may not be consistent with the Council's 30 year Environmental Sustainability Strategy.
- 4.21 At the hearing Mr. Thessman reiterated his concerns drawing our attention to developments in renewable energy, an increase in house generation and expressing the view that the Plan Change did not provide for these types of generation.
- 4.22 The s42A report set out the provisions in PC 34 regarding home based renewable generation devices (e.g. solar panels) and noting the matters of discretion for instances where consent is required including visual and amenity effects.¹³ In regard to the Environmental Sustainability Strategy, Ms. Sweetman was of the view that, "While it is important that the plan change and the Strategy do not conflict with each other, there is no direct relationship between each initiative. The key focus of PC34 is to address legislative requirements under the RMA."¹⁴
- 4.23 We find that PC34 does provide for home based renewable generation devices and appropriately has provisions that deal with the potential adverse effects of the installation of such devices rather than just considering the benefits of such devices. In regard to the link between the Environmental Sustainability Strategy and PC34 we are persuaded by the reasons set out in the s42A Report. We therefore recommend that submissions HS5.2 and HS5.3 be rejected.

Regionally significant network utilities – PC34 and PC38

Definition

- 4.24 Spark and Chorus (HS7.4 and HS8.7 on PC34, US3.15 and US5.19 on PC38) "seek that telecommunications and radiocommunications are included within the definition of regionally significant network utilities as follows: <u>Facilities for provision of 'telecommunication' as defined in section 5 of the Telecommunications Act 2001 and facilities for the provision of 'radio communication' as defined in section 2(1) of the Radiocommunications Act <u>1989.</u>"¹⁵
 </u>
- 4.25 Spark and Chorus supported the specific provision for regionally significant network utilities, but considered the exclusion of radiocommunications and telecommunications as inconsistent and unreasonable.
- 4.26 Spark and Chorus (HS7.6 and HS8.9 on PC34 and US3.7, US3.8, US5.7 and US5.8 on PC38) also seek consequential changes to issues, objectives, polices

¹² S42A Report, Page 19, Para 81

¹³ S42A Report, Page 19, Para 83

¹⁴ S42A Report, Page19, Para 84

¹⁵ S42A Report, Page 29, Para 130

and explanations of both Plan Changes which would be consequential to any change to the definition of Regional Significant Network Utilities.

- 4.27 We note that there is some detail around this issue and we find it necessary to repeat some of that material from the s42A report and evidence in order to provide the context for our recommendations and reasoning.
- 4.28 The section 42A report identified the issue of the inclusion of telecommunications and radiocommunications as regionally significant network utilities by reference to the s32 reports which state:

"..., in considering how to give effect to the RPS, an issue that has arisen is in respect of the definition of regionally significant infrastructure, which in the RPS includes:

- *"strategic telecommunications facilities, as defined in section 5 of the Telecommunications Act 2001*
- strategic radio communications facilities, as defined in section 2(1) of the Radio Communications Act 1989"

A review of both pieces of legislation has identified that neither contain the definitions referred to. The explanation to the relevant policies provides no guidance as to what may or may not constitute a strategic telecommunication facility or radio-communications facility. Discussions with the Regional Council have not provided clarification.

In the absence of certainty or clarification of what may or may not be a strategic telecommunications or radio-communications facility, the PC38 does not contain these two matters as being regional significant network utilities. Without definitions of what the Regional Council considers are "strategic" telecommunication and radio communication facilities, it is not appropriate for the Council to presume the exact scope and nature of what these are. If the RPS is amended in the future to clarify what is meant by these terms, the District Plan can be updated to give effect to the amended RPS."

- 4.29 We note that the full definition from the RPS is as follows: *"Regionally significant infrastructure includes:*
 - pipelines for the distribution or transmission of natural or manufactured gas or petroleum
 - strategic telecommunications facilities, as defined in section 5 of the Telecommunications Act 2001
 - strategic radio communications facilities, as defined in section 2(1) of the Radio Communications Act 1989
 - the national electricity grid, as defined by the Electricity Governance Rules 2003
 - facilities for the generation and transmission of electricity where it is supplied to the network, as defined by the Electricity Governance Rules 2003
 - the local authority water supply network and water treatment plants
 - the local authority wastewater and stormwater networks, systems and wastewater treatment plants

- the Strategic Transport Network, as defined in the Wellington Regional Land Transport Strategy 2007-2016
- Wellington City bus terminal and Wellington Railway Station terminus
- Wellington International Airport
- Masterton Hood Aerodrome
- Paraparaumu Airport
- Commercial Port Areas within Wellington Harbour and adjacent land used in association with the movement of cargo and passengers and including bulk fuel supply infrastructure, and storage tanks for bulk liquids, and associated wharflines".
- 4.30 The s42A Report set out the reporting officer's reasons for recommending that the Spark and Chorus submissions be rejected. We summarise those reasons as follows:
 - Specificity of what was included in RPS definition list inclusion of parts of networks (e.g. transmission but not distribution networks, strategic roads not all roads), concluding that the Regional Council considered some more regionally significant than others¹⁶
 - If the definition in the RPS was a drafting error then it was not Council's role to correct it.¹⁷
 - Importance and benefits of telecommunications and radiocommunications are recognised and provided for in the Plan Changes¹⁸
- 4.31 Ms. Sweetman's conclusion on this matter was that the RPS needed be changed and that was the role of the Regional Council. We also note that Ms. Sweetman recorded that this issue was addressed in a similar Plan Change for Porirua City Council and tabled a report from Greater Wellington Regional Council addressing the "regional significance" issue. We also record that there was no objection from the parties to the tabling of that document.
- 4.32 We received evidence for Spark and Chorus, from Ms. Barton from Chorus NZ Ltd, Mr. McCarrison from Spark New Zealand Trading Limited and Mr. Thomas Anderson, a resource management consultant for both Spark and Chorus.
- 4.33 Mr. McCarrison drew our attention to the integrated nature of telecommunication networks, the role of telecommunication networks as a lifeline utility and the rapidly changing way in which people access networks.
- 4.34 Ms. Barton noted that Spark were currently engaging with the Regional Council to promote a Plan Change to the RPS to amend the definition. Nonetheless Ms. Barton was of the view that the relief sought by Chorus and Spark was not inconsistent or contrary to the RPS.¹⁹

¹⁶ S42A Report, Page 32, Para 140

¹⁷ S42A Report, Page 32, Para 142

¹⁸ S42A Report, Page 32, Para 143

¹⁹ Ms Barton, EIC, Page 8, Para 32

4.35 We also heard from Mr. Anderson, a senior resource management consultant at Incite, on behalf of Chorus, in regard to this matter. Mr. Anderson considered that there was little similarity between different networks (e.g. electricity distribution and telecommunications). This meant that making comparisons in regard to the strategic component difficult and concluded that it was impossible to divide the telecommunications network into strategic and other components²⁰. Mr. Anderson also noted that this hearing was not the correct place to debate the regional significance of the telecommunications network and that the industry was in discussions with the Regional Council on amending the RPS. Mr. Anderson considered that the definition in PC34 and PC38 should be amended so that any change to RPS definition would not require consequential amendments to the Hutt City and Upper Hutt District Plans.²¹

That relief being;

Regionally Significant Network Utilities means has the same meaning as "Regionally Significant Infrastructure as defined in Appendix 3 of the Regional Policy Statement for the Wellington Region, where that infrastructure is located within [Hutt City/Upper Hutt].

(a) Pipelines for the distribution or transmission of natural or manufactured gas or petroleum ...

- 4.36 In response to matters raised in the hearing, Ms. Sweetman was of the view that there was some merit in this amendment.
- 4.37 We considered this proposed amendment very carefully. The first point we would note is that we agree with Mr. Anderson that it not the role of this panel to determine the regional significance of the telecommunications network. Rather that this is a role for the Regional Council through the RPS. We commend the telecommunications industry for engaging with the Regional Council on that matter.
- 4.38 In regard to the proposed amendment we agree that there would be efficiencies gained in that the District Plans would not need to be amended, should definition in the RPS be changed. However, the definition problem in the RPS would remain. There may be other changes required to the District Plans as a result of RPS definition change (e.g. Objectives, Policies, Methods, Rules) in order to give effect to the RPS. We find that that would be appropriately done at the time the final detail of the change is known.
- 4.39 Having considered the evidence and information placed before us in regard to the definition of Regionally Significant Network Utilities, we recommend that the submissions of Spark and Chorus be rejected. It therefore follows that we recommend that the consequential amendments sought by Spark and Chorus, also be rejected.

²⁰ Mr Anderson. EIC Page 12, Pars 46-47

²¹ Mr Anderson, EIC, Page 13-14, Paras 49-51

Stopbanks / Flood Hazard Area – PC34 and PC38

- 4.40 The submissions and further submissions relating to these provisions in both Plan Changes are set out in full in the s42A Report²². They relate to submissions by the GWRC which in summary seek:
 - A new policy requiring the location of utilities outside Flood Hazard Areas (FHA) or appropriate design (withstand flood event) if required to locate within them.
 - Amend policy 13.1.4(d) (PC34) and the explanation in 16.4.14 (PC38) to recognise that co-location was not always possible due to flood flow capacity and stopbank integrity
 - Amend policy 13.1.4(d) (PC34) and Policy 16.4.15 (PC38) to address the negative impacts of undergrounding services on or through flood protection stop-banks.
 - Amend policy 13.1.4(F) (PC34) to encourage network utility operators to consult with regional council regarding design and location of new infrastructure.
 - Amend PC34 and 38 to require consent for placement, maintenance or removal of underground services as restricted discretionary activities and works on transmission lines as controlled activities
 - That earthworks with the potential to affect stopbanks to be a restricted discretionary activity.
- 4.41 Powerco, WEL and Transpower opposed various aspects of the GWRC submissions in regard to potential impactions for their respective networks.²³
- 4.42 In the s42A Report, Ms. Sweetman provided an assessment of the above submission points and in summary Ms. Sweetman was of the view that:
 - Effects on stopbanks is not limited to network utility activities
 - The policy direction proposed by the Regional Council is not the most appropriate or efficient
 - Best planning practice would be to address natural hazards comprehensively
 - One of Regional Council's concerns is asset protection
 - Other mechanisms are available to the Regional Council
 - HCC, UHCC and GWRC are working together on how best to address this issue.
- 4.43 We received evidence and heard from Ms. Sharyn Westlake, a Senior Engineer in Flood Protection Department of GWRC. Ms. Westlake set out the role and responsibilities of GWRC in regard to river and floodplain management²⁴ as well as responding to the recommendations in the s42A Report. Ms. Westlake highlighted examples of network utility operations in stopbanks, the lack of

²² S42A Report, Pages 33-39, Paras 148 - 181

²³ S42A Report, Page 35, Paras 160 - 162

²⁴ Ms. S Westlake, EIC, Pages 2-3, Paras 2.1-2.6

consultation by network utility operators with GWRC and the issues the Council has in regard to designations and easements.²⁵

- 4.44 In regard to this matter we also received evidence and heard from Ms. Caroline Ammundsen, a policy advisor at GWRC. Ms. Ammundsen was of the view that the Plan Changes did not give effect to the RPS in regard to the risk and consequences of natural hazards.²⁶ Ms. Ammundsen did not agree with the s42A Report that the requested change in the activity status for infrastructure around stopbanks resulted in HCC and UHCC becoming third part managers of a GWRC asset and noting that all three Councils were signatories to the Hutt River Floodplain Management Plan (HRFMP) with joint responsibility in regard to flood management.²⁷ Similarly, Ms. Ammundsen disagreed with the S42A Report where it recommended waiting for the outcome of the Regional Hazards Management Strategy.²⁸ Ms. Ammundsen's view being that the issue should be addressed now.
- 4.45 Ms. Ammundsen drew our attention of the natural hazards chapters in the HCC and UHCC operative district plans and expressed the view that the proposed Plan Changes should give effect to those operative provisions.²⁹ In summary, Ms. Ammundsen concluded that the approach put forward by the Regional Council in regard to its submission on the policies and rules was an effective and efficient method to mitigate the adverse effects of flood hazard risk.³⁰
- 4.46 In responding to matters raised in the hearing, Ms. Sweetman reiterated that she was still of the overall view that the submissions should be rejected as set out in the s42A Report and that a more comprehensive approach to the matter was appropriate. Ms. Sweetman conceded that in regard to HS2.3 that the policy 13.1.4(F) of PC34 should be amended to, *"To encourage network utility providers to consult with local communities and the regional council (where relevant) on the appropriate placement, location and design of new network utilities."* Ms. Sweetman also tabled legal advice received by HCC and UHCC that a comprehensive approach to stopbank protection is beyond the scope of the Plan Changes relating to network utilities.³¹
- 4.47 In considering this matter, we prefer the evidence of Ms. Sweetman for the reasons set out in the s42A Report and for her responses to matters raised in the hearing. We consider that a more comprehensive approach to activities in hazard areas to be a more appropriate and efficient approach.
- 4.48 We therefore recommended that;

²⁵ Ms. S Westlake, EIC, Pages 3-6, Paras 3.1-3.15

²⁶ Ms. C Ammundsen, EIC, Page 3, Paras 6.1- 6.7

²⁷ Ms. C Ammundsen, EIC, Page 5, paras 8.6-8.8

²⁸ Ms. C Ammundsen, EIC, Page 6, Paras 8.9-8.11

²⁹ Ms. C Ammundsen, EIC, Page 7, Paras 9.1-9.4

 $^{^{30}}$ Ms. Ammundsen, EIC, Pages 7-11, Paras 10.1-25.0

³¹ Mrs. E Manohar, DLA Piper letter dated 19 Aug 2015

- HS 2.1, HS 2.2, HS2.4, HS2.5, HFS2.1, HFS2.2 HFS2.3 and HFS2.4 (PC34) and US1.1, US1.3, US1.4, US1.5, US1.6, UFS1.1, UFS1.3 and UFS1.4 (PC38), be rejected.
- US1.5 be accepted in part, insofar as it is made clear that the rules in Chapter 33 also applies to Chapter 30 (PC38).
- HS2.3 (PC34) be accepted in part.
- UFS3.1 be accepted.

Masts – PC34 and PC38

4.49 The submission points and reporting officer's recommendation relating to these provisions are set out in full in the s42A Report³². In general, the submissions opposed the lack of a permitted activity status for masts and associated antenna in the residential, rural and open space/recreation zones in the respective Plan Changes, as well as, seeking increased heights in the rural zones of both Plan Changes. The reporting officer recommended rejection the permitted activity status in the urban residential zones and Mr. Anderson concurred with that recommendation.³³ We understand that the remainder of the issues around mast height in the rural and open space/recreation zones still remain in contention.

Hutt City Council					
Standard	<u>Utility</u>	Ru	ıral	Recreation	
		<u>General</u>	<u>Residential</u>		
<u>13.3.2.2.1</u>	Masts, antennas, lines and single- pole support structures.	<u>15m</u>	<u>12m</u>	<u>12m</u>	
	<u>Spark & Chorus</u> seek	<u>20m</u>	<u>15m</u>	<u>15m</u>	
13.3.2.2.2	Masts and antennas (involving two or more providers).	<u>20m</u>	<u>12m</u>	<u>12m</u>	
	Spark & Chorus seek	<u>25m</u>	<u>15m</u>	<u>15m</u>	

4.50 In summary the remaining submissions from Spark and Chorus essentially sought increases in the mast heights as follows:

³² S42A Report, Pages 38 – 46, Paras 168 - 208

³³ Mr Anderson, Page 17, Para 65

Upper Hutt City PC 38					
Network Utility	<u>Rural</u>	Open Space			
<u>Masts, antennas, lines and</u> <u>single-pole support structures</u>	<u>15m</u>	<u>n/a</u>			
Spark and Chorus seek	<u>20m</u>	<u>15m</u>			
<u>Masts and antennas (where</u> <u>there are two or more providers)</u>	<u>20m</u>	<u>n/a</u>			
Spark and Chorus seek	<u>25m</u>	<u>15m</u>			

- 4.51 Ms. Sweetman's assessment of these requested amendments to the provisions led her to recommend rejecting the submission points. In summary, Ms. Sweetman considered;³⁴
 - Character and amenity values in rural residential area
 - Potential visual impacts
 - The wide range of environmental settings of recreation/open space zones
 - Inability to rely on public ownership as a mechanism to control effects
 - Permitted building height in rural zones
- 4.52 Mr. Anderson in evidence and in his presentation at the hearing, set out a contrary view to that of the reporting officer, in regard to the rural residential and recreation/open space zones being larger in areas and less developed than urban zones; therefore reducing the visual impact of masts and antennas.³⁵ Mr. Anderson also considered in evidence that³⁶:
 - 20m height in the recreation zone could reduce the need for masts in residential areas
 - Benefits of co-location of antenna on masts as incentive to telecommunication companies by way of increased height
- 4.53 On balance, we prefer the evidence of Ms. Sweetman for the reasons set out in her evidence. We therefore recommend that HS7.12, HS7.13, HS7.14, HS8.15, HS8.16, HS8.17, US3.11, US5.13, HS7.15, HS7.16, HS8.20 and HS8.21 be rejected.
- 4.54 We further recommended that US3.12 and US5.14 be accepted in part, and rejected insofar as they seek to amend 30.4 in respect of the Rural Zone.

³⁴ S42A Report, Pages -43-46, Paras 203-217

³⁵ Mr T Anderson, EIC, Page 18, Para 67

³⁶ Mr T Anderson, EIC, Pages 18-19, Paras 68-72

Policy Framework – Effects Management PC34 and PC38

- 4.55 Spark (HS7.9) and Chorus (HS8.12) seek that "policy 13.1.4(a)[PC34] be amended, so that the words "as far as practicable" be added to the end of the policy,..." as well as an amendment to the associated explanation in recognition of residual effects³⁷.
- 4.56 Spark and Chorus similarly seek amendments to Policy 16.4.12 (PC38) so that the words, " *"to the extent practicable"* or *"to the extent possible"* are added, so that it acknowledges that not all adverse effects will be able to be avoided, remedied or mitigated."³⁸
- 4.57 Ms. Sweetman agreed that amendments to the explanation for policy 13.1.4(a) (PC34) may assist decision makers.³⁹ However, in regard to the amendments requested to the policies in both Plan Changes noted above, Ms. Sweetman was of the view that the amendments would *"weaken the policy framework and lessen the onus in the first instance for applicants to avoid, remedy or mitigate adverse effects."*⁴⁰
- 4.58 Mr. Anderson saw this differently. Mr. Anderson's view was that the policy wording should be consistent with the objective and explanation and that "by including the words 'as far as practicable' within the policy resolves any potential interpretation conflicts between the objective and which recognizes constraints may cause effects and that these can be managed, and its policy which states that all actual and potential effects must be avoided, remedied or mitigated."⁴¹
- 4.59 On this we are persuaded by the evidence of Ms. Sweetman for the reasons set out in her evidence.
- 4.60 We recommend that HS7.9, HS8.12, HSF2.10, HFS1.4 be accepted in part. We further recommend that US4.17, US5.6 and US6.18, and UFS3.9, UFS1.5, UFS3.18 and UFS3.30 also be accepted in part, insofar as there is a consequential amendment proposed to 16.3.4, as well as, the matters of discretion.

Minor Upgrading PC34 and PC38

4.61 We note that at beginning hearing, the definition of minor upgrading was still in contention.⁴² However, in response to matters raised in the hearing, Ms. Sweetman reconsidered her position and set out a revised definition in some detail.⁴³ We understood that this addressed the matters raised in submissions by Transpower and Chorus.

³⁷ S42A Report, Page 57, Para 267

³⁸ S4A Report, Page 75 Para 360

³⁹ S42A Report, Page 58, Para 274

⁴⁰ S42A Report, Page 58, Para 273 and Page 75, Para 361

⁴¹ Mr T Anderson, EIC, Page 16, Para 59

⁴² S42A Report, Pages 21-25, Paras 96- 112

⁴³ Ms G Sweetman, Response to matter raised in the hearing, Pages 8-11

- 4.62 Mr. Thessman (HS5.1) queried the size of the bundling of cables included in the definition and reiterated his concerns at the hearing.
- 4.63 In the s42A report, Ms. Sweetman noted, *"the increased area of bundling cables is a relatively insignificant increase in width and provides for the continued effective use of existing network utilities."*⁴⁴ We agree. There was agreement amongst the planning experts as to the appropriateness of the definition as revised. We therefore recommend that HS8.6, US5.18, UFS3.36, HS6.6, HS9.8, US4.37, US6.7 and UFS3.27 be accepted in part and that HS5.1 be rejected.

PC38 – Issue 16.2.2

4.64 In US8.2, Forest and Bird opposed the deletion of wording in the Issue statement relating to adverse effects on natural and physical resources and on amenity values. Transpower (USF2.4) opposed the request. In the material Forest and Bird had tabled at the hearing, Dr. Wards presented arguments relating to the particular wording in the issue statement.⁴⁵ Ms. Sweetman, was of the view that adverse effects were already adequately addressed in Issue 61.2.1 and including it in Issue 16.2.2 would be unnecessary duplication.⁴⁶ We agree. We therefore recommend that US8.2 be rejected and USF2.4 be accepted.

PC38 – Policy 16.4.7 and Method 16.5.9

4.65 In US8.5, Forest and Bird sought rewording of this policy to strengthen the word 'encourage' and identify situations where consultation is required.⁴⁷ In US8.6 Forest and Bird sought that this also be reflected in the associated method. Transpower and WELL sought that the requests be rejected. In the S42A Report, Ms. Sweetman noted that the District Plan could not require consultation on particular matters as this would be contrary to s36 of the RMA, which sets out that there is no duty on applicants to undertake consultation in regard to resource consent applications. In the material presented at the hearing for Forest and Bird, Dr. B Wards considered it unreasonable that such provision could not be required and wanted the matter to be noted for future consideration.⁴⁸ We note that this is not within our jurisdiction to do so. We accept the evidence of Ms. Sweetman and recommend that US8.5 and US8.6 be rejected and that UFS2.14, USF3.47, and USF2.14 be accepted.

⁴⁴ S42A Report, Page 25, Para 109

⁴⁵ Forest and Bird, Material Presented at the Hearing, Item 5

⁴⁶ S42A Report, Page 70, Para 335

⁴⁷ S42A Report, Page 79, Para 79

⁴⁸ Forest and Bird, Material Presented at the Hearing, Item 4

Proposed Plan Change 34 - Hutt City Council District Plan: Network Utilities and Renewable Energy Generation Proposed Plan Change 38 – Upper Hutt City Council District Plan: Network Utilities and Renewable Energy Generation Commissioners' Report and Recommendation

5. RECOMMENDATION AND OVERALL REASONS

- 5.1 We recommend to the Councils that, pursuant to the First Schedule to the Resource Management Act 1991, Proposed Plan Change 34 to the Hutt City Council District Plan and Proposed Plan Change 38 to the Upper Hutt City Council District Plan be approved for all of the reasons set out in this report.
- 5.2 In terms of Part 2 of the RMA the proposal does not contravene any of the matters of national importance (Section 6), 'other matters' (Section 7), nor the principles of the Treaty of Waitangi (Section 8).
- 5.3 The Plan Changes are an appropriate way of achieving the purpose of the RMA.
- 5.4 For all of the reasons given above the Plan Changes meets the statutory requirements of the RMA, the Plan Changes satisfy Part 2 of the Act and therefore will promote the sustainable management of natural and physical resources as required by the RMA.
- 5.5 We recommend that the submissions on Plan Change 34 and Plan Change 38 be accepted, accepted in part or rejected, as outlined below.

6. RECOMMENDATIONS ON THE SUBMISSIONS

6.1 That the following submissions be <u>accepted</u>, <u>accepted in part or rejected as set</u> <u>out in Appendix One</u>. <u>That the text of Plan Change 34 be amended as set out in</u> <u>Appendix Two</u>. <u>That the text of Plan Change 38 be amended as set out in</u> <u>Appendix Three</u>.

L Bridson Panel - Hearings Commissioner

G McArthur Panel - Hearings Commissioner

M L St.Clair Chair - Hearings Commissioner

17 December 2015

APPENDIX ONE – DECISIONS ON SUBMISSIONS

Any new text that is proposed to be added is <u>underlined</u>, while any text proposed to be deleted has been struck through.

There are some submission points which the Commissioners recommend are not in scope of the Plan Change or on the Plan Change.

These are identified in *italics* and in **bold**.

GENERAL – N	NETWORK UT	ILITIES		
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Mary Beechen	4.1	Transpower to purchase submitter's property as the proposed restrictions prevent the building of any dwelling.	Oppose. The restrictions will prevent building a dwelling on the submitter's property.	Reject
Transpower	FS1.1	Reject	The provisions do not provide for prohibited activity status and therefore do not prevent development. The submitter's request that Transpower purchase the property is beyond the scope of this plan change.	Accept
Warren Thessman	5.1	Unclear that there is a reason to change the District Plan.	PC34 has too much struck out much data, exclusions and dimensions and HCC may be unaware of the consequences. Query over the bundling allowed by telcos.	Reject
Transpower New Zealand	6.1	 Give full effect to the NPSET Effect is given to the policies of the RPS Recognition of the NESETA and ensure that there are no conflicts with provisions of the District Plan and the NESETA (s44A of the RMA) The protection of the National Grid from issues of reverse sensitivity and the direct effects from others' activities 	Supports the general intent of the plan change subject to appropriate amendments.	Accept in part

		 through the provision of appropriate National Grid Yard and National Grid Corridor provisions Provide for the ongoing operation, maintenance, upgrading and development of transmission lines, support structures and substations. 		
Transpower New Zealand	6.2	Approve the plan change subject to the amendments requested.	To ensure that policies 10 and 11 of the NPSET are given effect to.	Accept in part
Spark New Zealand Trading Limited	7.1	Either (i) amend the proposed plan change as per the relief set out in the specific submissions by Spark. Or (ii) such other relief to like effect to remedy the concerns outlined in the submissions. And in relation to both (i) and (ii) above, (iii) any consequential amendments necessary as a result of the amendments to grant the relief sought above.	Spark's general relief sought in this submission point (7.1) applies to submission points 7.2 to 7.21.	Accept in part
Chorus New Zealand Limited	8.1	Either (i) amend the proposed plan change as per the relief set out in the specific submissions by Chorus Or (ii) such other relief to like effect to remedy the concerns outlined in the submissions And in relation to both (i) and (ii) above, (iii) any consequential amendments necessary as a result of the amendments to grant the relief sought above.	Chorus' general relief sought in this submission point (8.1) applies to submission points 8.2 to 8.26.	Accept in part
Powerco	10.1	Amend the relevant	Support the general	Accept in part

Limited		 provisions of the District Plan to: Address the relevant provisions in sections 5-8 of the RMA, including with respect to the sustainable management of Powerco's assets as a physical resource; Implement the statutory tests in section 32 and the requirements in the First Schedule of the RMA; Recognise and provide for the ongoing operation, maintenance, upgrade and development of Powerco's gas and electricity networks. 	intent. Powerco supports the move by Hutt, Porirua and Upper Hutt City Councils to provide a consistent approach to the management of network utilities across the three districts. Powerco has submitted on all three plan changes. Reliable and constant energy supply is critical to sustaining the regional economy, population and way of life and energy demand is constantly increasing. Powerco faces a number of constraints in terms of providing a secure and reliable supply of gas. It is critical that the planning documents adequately provide for the core strategic infrastructure, its ongoing operation and maintenance and that required to support growth. Some amendments are sought to further clarify the specific regulatory approach for gas distribution infrastructure.	
	10.2	Adopt any other such relief, including additions, deletions or consequential amendments necessary as a result of the matters raised in Powerco's submissions, as necessary.	To give effect to Powerco's submission.	Accept in part
Julie Sylvester	11.1	Not stated.	The submitter questions what consideration has been given to animals that live in the rural area where the plan change has effects. The submitter also questions what consideration do owners of rural	Reject

		properties get.	
FS1.2 Transpower	Reject	Transpower considers that, subject to the amendments it seeks, the provisions of the Plan Change are appropriately drafted to a) minimise impacts on rural land owners, and b) address the potential adverse effects of all land activities on the National Grid, which is a nationally significant physical resource.	Accept
11.2	Not stated.	The submitter comments that the topic of Renewable Energy Generation should be very high on the Hutt City Council's to do list and that some councils have this system already working in their areas, and generating additional energy for their rate payers. The submitter states that care must be taken not to destroy the environment when and if this activity takes place.	Reject

GENERAL -	GENERAL – RENEWABLE ENERGY GENERATION				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Warren Thessman	5.2	Unclear.	 The provisions for Renewable Energy Generation are not very specific and do not address some matters; in particular: The use of sun trapped electronics is missing What happens when small home solar panels are above the roof; There is no comment on sun reflection; 	Reject	

			 Should vertical axis wind generators be allowed on top of houses? 	
Warren Thessman	5.3	Unclear.	The Environmental Sustainability Strategy has been tabled with no discussion. Has Proposal 34 been ticked off to allow the so called 30 year environmental strategy to be achieved?	Reject
Julie Sylvester	11.2	Not stated.	The submitter comments that the topic of Renewable Energy Generation should be very high on the Hutt City Council's to do list and that some councils have this system already working in their areas, and generating additional energy for their rate payers. The submitter states that care must be taken not to destroy the environment when and if this activity takes place.	Reject

Amendment	Amendment 1 – Definition of Antenna			
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Spark New Zealand Trading Limited	7.2	Retain the exemption of (c) and the two 'notes' as notified.	Support in part. The definition adequately provides for antennas and provides certainty of how the provisions will be applied.	Accept
Chorus New Zealand Limited	8.2	Retain the exemption of (c) and the two 'notes' as notified.	Support in part. The definition adequately provides for antennas and provides certainty of how the provisions will be applied.	Accept

Amendment 3 – Definition of Building				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation

				on Submission Point
Wellington Electricity Lines Limited (WELL)	9.2	Amend the definition to include the following exemption: <i>"(a) A network system</i> <i>owned or operated</i> <i>by a network utility</i> <i>operator whose</i> <i>purpose is to</i> <i>provide reticulation</i> <i>from a network</i> <i>system to and from</i> <i>individual</i> <i>properties and</i> <i>structures,</i> <i>including all</i> <i>structures and</i> <i>equipment owned</i> <i>or used by a</i> <i>network utility</i> <i>operator, provided</i> <i>that a system</i> <i>including any</i> <i>structure or</i> <i>equipment does</i> <i>not exceed 1.5m in</i> <i>height and has 3m</i> ² <i>or less ground</i> <i>coverage."</i>	Oppose. The current definition used for 'Building' does not suitably reflect the intent of the 2004 Building Act when identifying what does not include a building. In particular section 9 states that a building does not include a network utility operator system and specifically excludes elements that are excluded from the definition of a building.	Accept in part
Powerco Limited	10.19	Retain as notified.	Support.	

Amendment 4 – Definition of Cabinet				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Wellington Electricity Lines Limited (WELL)	9.3	Amend as follows: "means a box-shaped structure which houses radio and telecommunication equipment, electrical equipment, equipment associated with the continued operation of network utilities, which includes single transformers and associated switching gear distributing electricity at a voltage	Support in part. 110kV, not 100kV, is the industry standard.	Accept

	up to, and including, 100kV-<u>1</u>10kV .	
	"	

Amendment 10 – Definition of Height					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Transpower New Zealand	6.3	Insert the following note after the definition of height: <i>"Note: Compliance with</i> the New Zealand Electrical Code of Practice for Electrical Safe Distances (NZECP34:2001) is mandatory under the Electricity Act 1992. All activities regulated by NZECP34, including buildings, structures, earthworks and the operation of mobile plant, must comply with that regulation. Activities should be checked for compliance even if they are permitted by the District Plan."	Oppose in part. The exclusions in clause (b) of both the maximum height and maximum overall height would mean that structures may be established too close to National Grid lines. An advice note should be attached to the definition which clearly states that compliance with the New Zealand Electrical Code of Practice for Electrical Safe Distances (NZECP34:2001) is required.	Accept	
Spark New Zealand Trading Limited	7.3	Retain exclusion for lightning rods attached to network utilities from the definition of Height and the associated definition of Maximum Overall Height as notified.	Support in part. Exclusion of lightning rods is supported as they have negligible visual effects.	Accept	
Chorus New Zealand Limited	8.3	Retain exclusion for lightning rods attached to network utilities from the definition of Height and the associated definition of Maximum Overall Height as notified.	Support in part. Exclusion of lightning rods is supported as they have negligible visual effects.	Accept	

Amendment 11 – Definition of Line					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation	

				on Submission Point
Transpower New Zealand	6.4	Amend as follows: "Means 'line' as defined in section 5 of the Telecommunications Act 2001 and includes the definition of line or in section 2 of the Electricity Act 1992."	Support in part. Clarify that the activity does not need to meet the definition of line under both Acts referenced.	Accept
Chorus New Zealand Limited	8.4	Retain definition of Line in so far as it relates to the Telecommunications Act 2001.	Support in part.	Accept
Wellington Electricity Lines Limited (WELL)	9.5	Retain as notified.	Support.	Accept in part

Amendment 12 – Definition of Maintenance					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
NZ Transport Agency	1.1	Retain as notified.	Support.	Accept	
KiwiRail	3.1	Retain as notified.	Support.	Accept	
Transpower New Zealand	6.5	Replace the definition of maintenance with one that is clear and which provides for standard National Grid maintenance activities; or alternatively delete the definition of maintenance.	Oppose. The definition is too restrictive and does not appropriately 'provide for' standard maintenance activities that Transpower regularly undertakes on the National Grid. The definition as drafted provides no greater right than what is provided for through existing use rights. The definition also lacks certainty and it is unclear which of Transpower's maintenance activities are covered by it.	Reject	
Powerco	FS2.5	Reject in part	The submission is opposed to the extent that it is uncertain what alternative definition of	Accept	

			'maintenance' is proposed.	
Wellington Electricity Lines Limited (WELL)	9.6	Retain as notified.	Support.	Accept

Amendment 14 – Definition of Minor Above Ground Line					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Chorus New Zealand Limited	8.5	Retain as notified.	Support.	Accept	
Wellington Electricity Lines Limited (WELL)	9.7	Retain as notified.	Support.	Accept	

Amendment	Amendment 15 – Definition of Minor Upgrading					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point		
Transpower New Zealand	6.6	Amend as follows: "Minor Upgrading: (d) the addition of any new circuits, lines or utility structures, <u>where this results</u> in an increase in <u>the number of new</u> <u>lines or utility</u> <u>structures</u> Note: The Resource Management (National Environmental Standards for Electricity Transmission Activities) Regulations 2009 applies to <u>all</u> the existing n <u>National g-G</u> rid , and applies to all Transmission Lines that were operational, or able to be operated, on 14 January 2010."	Oppose in part. Exclusion (d) of the definition requires clarification to ensure that in stating the addition of it means an increase in the number of, rather than the replacement of existing circuits, lines and utility structures, which should be considered a minor upgrade. In addition, the note associated with the definition is not clearly worded.	Accept in part		
Chorus New Zealand Limited	8.6	Amend as follows: "Minor Upgrading. means an increase in the carrying capacity, efficiency or security of electricity and telecommunication lines, which utilise existing or replacement support structures and includes: (j) an increase in the height of replacement poles in road reserve by a maximum of 1m, for the purpose of achieving road controlling authority clearance requirements, provided the permitted height in 13.4.2.1 is not exceeded <u>and/or the</u> replacement of an existing pole in road reserve is within 2m horizontal distance of that existing pole. (1) the addition of a new overhead telecommunication fibre optic line provided that: (i) the maximum number of fibre optic lines on existing support structures does not exceed two lines; (ii) the diameter of new fibre optic lines does not exceed 25mm; and (iii) the location of the new fibre optic line is	Oppose in part. Chorus seeks amendments to allow for: (a) the provision of up to two fibre optic lines utilising existing or replacement support structures subject to specific controls; and (b) the replacement of an existing support poles in road reserve within 2 metres of the existing pole which it is replacing. These changes are sought to allow for the efficient and effective use of existing infrastructure for activities that have less than minor adverse effects.	Accept		
----------------------------------	-----	---	--	--------		
		<u>exceed 25mm;</u> <u>and</u> (iii) the location of <u>the new fibre</u> <u>optic line is</u> <u>consistent with</u> <u>the following</u> <u>fiqure:</u> [please refer to the full				

		submission which contains the figure] Minor upgrading shall not include: (d) the addition of any new circuits, lines (with the exception of additional overhead telecommunication lines utilising existing support poles as provided for in item (I) above) or utility structures."		
Wellington Electricity Lines Limited (WELL)	9.8	Amend the definition for Minor Upgrading by including a new provision for the addition of circuits and conductors on existing support structures.	Support in part. Minor activities such as circuits and conductors should be added to the definition of Minor Upgrading as it will enable WELL to provide for growth and future demand for electricity across the city. This would also improve adherence with proposed Policy 13.1.4(c), as it would provide for co-location, mitigating visual impacts.	Accept in part

Amendment 16 – Definition of National Grid				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Transpower New Zealand	6.7	Amend as follows: "National Grid: means ' n <u>N</u> ational gG rid' as defined in the National Policy Statement on Electricity Transmission."	Support in part. Prefer that references to the National Grid are capitalised.	Accept

Amendment	Amendment 17 – Definition of National Grid Corridor				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission	

				Point
Transpower New Zealand	6.8	Amend as follows: "measured either side of the centreline of the transmission line, as depicted in Diagram 1, below-of the definition of <u>National Grid Yard</u> . The measurement of setback distances"	Support in part. Clarify that Diagram 1 is located within the definition of National Grid Yard, rather than simply stating 'below', which implies it is immediately below the text.	Accept

Amendment 18 – Definition of National Grid Yard					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Transpower New Zealand	6.9	Amend as follows: "National Grid Yard: means (a) the area located 12m either side of the centreline of an above ground National Grid transmission line which is 110kV or greater;"	Support in part. Clarify in relation to (a), to show there is no need to refer to 'above ground' since the Note under the definition states that the National Grid Yard does not apply to underground cables or any transmission lines (or sections of line) that are designated.	Accept	

Amendment 19 – Definition of Network Utility					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
NZ Transport Agency	1.2	Retain as notified.	Support.	Accept	
KiwiRail	3.2	Retain as notified.	Support.	Accept	
Transpower New Zealand	6.10	Replace clause (c) of the definition as follows: "(c) the provision, operation and maintenance of works (as defined in section 2 of the Electricity Act 1992) for the conveyancing of electricity, as defined in section 2 of the Electricity	Oppose in part. The definition could be clearer.	Accept	

		Act 1992 ;"		
Wellington Electricity Lines Limited (WELL)	9.9	Retain as notified.	Support.	Accept

Amendment 20 – Definition of Network Utility Structure					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
NZ Transport Agency	1.3	Retain as notified.	Support.	Accept	
KiwiRail	3.3	Retain as notified.	Support.	Accept	
Transpower New Zealand	6.11	Retain as notified.	Support.	Accept	
Wellington Electricity Lines Limited (WELL)	9.10	Retain as notified.	Support.	Accept	

Amendment	Amendment 22 – Definition of Regionally Significant Network Utilities				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
KiwiRail	3.4	Amend the definition as follows: <i>"Regionally Significant</i> <i>Network Utilities:</i> <i>means:</i> (f) the Strategic <i>Transport Network,</i> <i>as detailed in</i> <i>Appendix 1 to</i> the <i>Wellington Regional</i> <i>Land Transport</i> <i>Strategy 2010–2040"</i>	Supports in part. The references to the RLTS may not remain relevant when it is updated.	Reject	
Transpower New Zealand	6.12	Amend as follows: "(b) the National Grid , as defined by the National Policy Statement on Electricity Transmission "	Support in part. Reference to the NPSET is not required as Amendment 16 which contains a definition of the National Grid already makes the connection.	Accept	

Spark New Zealand Trading Limited	7.4	Add additional item (g) to the definition as follows: " (a) facilities for provision of 'telecommunication' as defined in section 5 of the Telecommunications Act 2001 and facilities for the provision of 'radio communication' as defined in section 2(1) of the Radiocommunication s Act 1989."	Oppose in part. Inclusion of telecommunications and radiocommunications in the definition will recognise the essential nature of these services to the economic, health, safety and wellbeing of the Hutt City community and the wider region.	Reject
Chorus New Zealand Limited	8.7	Add item (g) to the definition as follows: <u>"(q) facilities for</u> <u>provision of</u> <u>'telecommunicatio</u> <u>n' as defined in</u> <u>section 5 of the</u> <u>Telecommunication</u> <u>s Act 2001 and</u> <u>facilities for the</u> <u>provision of 'radio</u> <u>communication' as</u> <u>defined in section</u> <u>2(1) of the</u> <u>Radiocommunicati</u> <u>ons Act 1989."</u>	Oppose in part. Chorus seeks recognition of the essential nature of these services to the economic, health, safety and wellbeing of the Hutt City community and the wider region.	Reject
Powerco Limited	10.26	Retain as notified.	Support.	Accept

Amendment 26 – Definition of Sensitive Activity					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
KiwiRail	3.5	Delete proposed definition if not required.	Supports, however question how the existing definition of 'Noise Sensitive Activity' in the District Plan is different from the proposed definition of 'Sensitive Activity'	Accept in part	
Transpower New Zealand	FS1.3	Reject	The submitter seeks that the definition be deleted as they are not clear	Accept	

			about its purpose. Transpower considers that inclusion of the term is critical to ensuring that the District Plan gives effect to the NPS on Electricity Transmission. The term is taken from that NPS and subsequently used in rules relating to the National Grid Yard.	
Transpower New Zealand	6.13	Amend as follows: "Sensitive Activity: means the following activities: (a residential <u>buildings</u> (b) the accommodation or care of people <u></u> <u>including hospitals</u> (c) childcare facilities, kohanga reo, primary, intermediate and secondary schools."	Support in part. Prefer to see explicit reference to hospitals within the definition and also notes that the definition should apply to residential buildings and not all residential activities. This would be consistent with the definition of sensitive activities in the NPSET.	Accept

Amendment	Amendment 31 – Definition of Transmission Line				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Transpower New Zealand	6.14	Amend as follows: "means the facilities and structures used for, or associated with, the overhead or underground transmission of electricity <u>in the National</u> <u>Grid</u> ; and"	Support in part. The phrase 'in the National Grid', which is part of the definition in the NESETA, should be included in the District Plan definition. The use of the phrase in the Plan Change appears to only relate to the National Grid.	Accept	
Wellington Electricity Lines Limited (WELL)	9.11	Retain as notified.	Support.	Accept in part	

Amendment 32 – Definition of Upgrading				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point

NZ Transport Agency	1.4	Clarity is required around the scope of this definition.	Support in part. Not all upgrading results in physical changes on the ground and/or any increase in capacity	Accept in part
Transpower	6.15	Amend the definition as follows: Upgrading: As it applies to network utilities, upgrading means the replacement, repair, renewal or improvement or physical works that result in an increase in carrying capacity, operational efficiency, security or safety of existing network utilities but excludes: (a) 'maintenance' (as it relates to network utilities); (b) 'minor upgrading'; and (c) any activity specifically provided for under Rules 13.3.1.9 to 13.3.1.41; and any increase in height or size or change in location, unless such increase or change is specifically provided for and would comply with the applicable permitted activity standard specified in 13.3.2.2, 13.3.2.3 or 13.3.2.4.	The definition of 'upgrading' as proposed includes 'replacement, repair, renewal' It is unclear whether replacement, repair or renewal activities would be maintenance or upgrading purely based on both definitions covering the same activity. Some clarity is required. The improvement in carrying capacity and operational efficiency is not something that necessarily results in physical changes, or one that would necessarily be an upgrade. Changing engine type for example, or putting an extra carriage on a train, are improvements in operational efficiency and carrying capacity respectively. Therefore in some instances, some activities will be technically an 'upgrade' where they are in effect operational activities and needing to get Outline Plan or resource consent approval each time seems outside the intention of the provision or the RMA in relation to designations. KiwiRail would support wording that restricted upgrading to physical works on the network utility itself, rather than the operation of the utility.	Accept in part
New		"Upgrading:	the difference between	•

Zealand		(c) any activity specifically provided for under Rules 13.3.1.9 to 13.3.1.41; and any increase in height or size or change in location, unless such increase or change is specifically provided for and would comply with the applicable permitted activity standard specified in 13.3.2.2, 13.3.2.3 or 13.3.2.4."	minor upgrading, upgrading and 'new' facilities and unable to determine what activities would fall under the 'upgrade' category of activities. It remains unclear what is the difference between minor upgrading, upgrading and 'new' facilities. Transpower is unable to determine what activities would fall under the 'upgrade' category of activities. In relation to transmission lines, including their support structures, the definition should provide for all upgrading that is not 'minor upgrading' and which is not the establishment of new transmission lines. The definition also needs to allow for upgrades to substations. In addition Transpower is concerned specifically with clause (c) which references permitted activity standards. Standards relevant to 'upgrading' activities rules should be listed in the rule itself, and not in	
			rules should be listed in the rule itself, and not in both the definition and in the rule. Including standards in multiple locations simply adds to the complexity of the plan change.	
Spark New Zealand Trading Limited	7.5	Amend the wording of the definition to make it clear that it applies to all telecommunications and radiocommunications, other than lines which are provided for separately.	Oppose in part. The currently drafted definition is unclear and does not reflect the intended approach of providing for all network utilities under the 'Upgrading' provisions, other than telecommunication and electricity lines which are provided for in the	Accept in part

			'Minor Upgrading' provisions.	
Powerco	FS2.13	Accept in part	Powerco agrees there would be benefit in amending the definition of 'upgrading' to clarify the distinction between network utilities covered by the definition of 'minor upgrading' and those covered by the definition of 'upgrading'. It should be clear from any such amendment that the definition of 'upgrading' will apply to Powerco's gas distribution network.	Accept in part
Chorus New Zealand Limited	8.8	Amend the wording of the definition of "Upgrading" to make it clear that it applies to all telecommunications and radiocommunications, other than lines which are provided for separately.	Oppose in part. Chorus considers the currently drafted definition is unclear and does not reflect the intended approach of providing for all network utilities under the 'Upgrading' provisions, other than telecommunication and electricity lines which are provided for in the 'Minor Upgrading' provisions.	Accept in part
Wellington Electricity Lines Limited (WELL)	9.12	Retain as notified.	Support.	Accept in part

Amendment 34 – Rule 11.2.2.2 – Matters in which Council seeks to Control				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
NZ Transport Agency	1.5	Retain as notified.	Support.	Accept
KiwiRail	3.7	Retain as notified.	Support. For subdivisions, the consideration of the impact of the design and layout of the subdivision on the rail network as a	Accept

regionally significant network utility, and the outcome of consultation with KiwiRail, are supported as being matters Council seek to	
control.	

Amendmen	t 35 – Rule 1	1.2.2.3 – Assessment Crite	ria	
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
NZ Transport Agency	1.6	Retain as notified.	Support.	Accept
KiwiRail	3.8	Retain as notified.	Support that potential reverse sensitivity effects on regionally significant network utilities are considered through the subdivision process. Consideration of elements such as setbacks, design standards and mitigation requirements for development on a site at the time of subdivision, will ensure that future occupants are fully aware of the potential effects from residing adjoining a railway, and that these effects are mitigated so that the railway is not subject to reverse sensitivity effects.	Accept

Amendment 36 – Rule 11.2.3(b) and associated Standards and Terms 11.2.3.2				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Transpower New Zealand	6.16	Retain the proposed Restricted Discretionary Rule 11.2.3(b) without amending except for the following: <i>"11.2.3</i>	Oppose in part. The reference in rule 11.2.3 (b) to 'standards and terms under Rule 11.2.3.1' appears incorrect and should be	Accept

		(b) Any subdivision located within a National Grid Corridor that complies with the standard and terms under Rule 11.2.3.1 <u>11.2.3.2</u> "	amended.	
Transpower New	6.17	Retain the standards and terms (11.2.3.2) without	Oppose in part. At present, the District Plan	Accept
Zealand		amending except for the	only requires the	
		following:	demonstration of shape	
		11.2.3.2 (a)	factors where residential	
		(ii) Demonstrate that	activities are proposed:	
		each new <u>residential</u>	it is not required for	
		allotment can	commercial and	
		provide a complying shape factor as	industrial activities.	
		required under rule	A subdivision should be	
		11.2.2.1(a) <u>or in the</u>	required to demonstrate	
		<u>case of industrial or</u>	that a building platform	
		<u>commercial</u>	can be provided for all	
		<u>activities, a suitable</u>	activities outside of the	
		building platform	National Grid Yard	
		which is fully located		
		outside of the		
		national grid yard."		

Amendment	Amendment 38 – Matters of Discretion				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Transpower	6.18	Amend as follows: "The Council has restricted its discretion to the following matters when considering an application for resource consent under Rule 11.2.3.1 (i) the extent to which the design, construction and layout of the subdivision (including landscaping) demonstrates that a suitable building platform(s) for a dwelling can be located outside of the National Grid	Support in part. Transpower states that in relation to clause (i) it is not clear how landscaping relates to the demonstration that a suitable building platform can be located outside of the National Grid yard. Having noted this Transpower does consider that the potential benefits and adverse effects of landscaping should be within the discretion of Council, and seeks changes accordingly. In relation to proposed clause (b) it is noted that this refers to 'electricity	Accept	

	Vard for each new lat	transmission lines/	
	rara jor each new lot	transmission lines',	
	to ensure adverse	whereas the term	
	effects on and from	defined in the proposed	
	the National Grid	plan change is	
	and on public health	'transmission lines'.	
	and safety are	Transpower considers	
	appropriately	that the defined term	
	avoided, remedied or	should be used.	
	mitigated;		
	(ib) the extent to which	Transpower considers	
	landscaping will	that clause (a) is	
	impact on the	inappropriately limited	
	operation.	to dwellings and should	
	maintenance.	also include commercial	
	uparade and	and industrial buildings.	
	development		
	(including access) of	Transnower submit that	
	the National Grid	the second naragraph of	
	(ii) The provision for the	the note includes some	
	on-aging operation	minor inaccuracies which	
	maintenance	should be amended and	
	(including access)	that the second	
	and planned upgrade	naragraph implies	
	of Transmission	paragraph implies	
	oj transmission	vegetation under the	
	Lines;	Vegetation under the	
	(III) The risk to the	Electricity (Hazards from	
	structural integrity of	Trees) Regulations 2003	
	the National Grid;	is restricted to the	
	(iv) The extent to which	National Grid Yard, when	
	the subdivision	they are not.	
	design and		
	consequential		
	development will		
	minimise the risk of		
	injury and/or		
	property damage		
	from such lines;		
	(v) The extent to which		
	the subdivision		
	design and		
	consequential		
	development will		
	minimise the		
	potential reverse		
	sensitivity on and		
	amenity and		
	nuisance effects of		
	the transmission		
	asset.		
	Advice Note:		
	Compliance with the		
	New Zealand Electrical		
	Code of Practice for		
	Electrical Safe Distances		
	(NZECP34:2001) is		
	mandatory under the		

a N b e o p t t A C c e P F	Activities regulated by NZECP34, including buildings, structures, earthworks and the operation of mobile olant, must comply with that regulation. Activities should be checked for compliance even if they are opermitted by the District Plan.	
ע <u>ע</u> ד א א ש א ש א ש ג ג ג ת ז ת ג ע ב ג ג ג ת ת א ש ג ג ג ג ת ש א ג ג ש ג ש ג ש ג ש ג ש ג ג ג ג ג ג ג	Vegetation to be planted <u>within proximity to</u> <u>Transmission Lines</u> within the National Grid Yard as shown on the planning maps should be selected and/or managed to ensure that it will not result in that vegetation breaching the Electricity (Hazards from Trees) Regulations 2003 or prevent access to support structures. To discuss works, including tree planting near any electricity <u>Transmission</u> <u>it</u> ine especially works within the transmission	

Amendment 41 – Rule 11.2.5				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Transpower New Zealand	6.19	Amend as follows: "(a) Any subdivision of land within the National Grid Corridor that does not comply with the standards and terms under Rule <u>11.2.3.1</u> <u>11.2.3.2</u> ."	Support. The reference to Rule 11.2.3.1 appears to be incorrect and should be amended.	Accept

Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Transpower New Zealand	6.20	Retain as notified.	Support.	Accept

Amendment	: 43 – Chapte	r 13 Introduction		
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
KiwiRail	3.9	Retain as notified.	Support. In particular the paragraph which recognises that utilities are vital for a community and the ability to construct, maintain, upgrade and operate these effectively is very important.	Accept
Transpower New Zealand	6.21	Amend as follows: " The Council is required to give effect to any National Policy Statement. The National Policy Statement on Electricity Transmission came into force in 2008 and applies to effects on and effects of the transmission network National Grid. The National Grid. The National Policy Statement on Electricity Transmission's objective is to recognise the national significance of the electricity transmission network by facilitating the operation, maintenance and upgrade of the National Grid and the establishment of new transmission resources to meet the needs of present and future generations while: managing the adverse environmental effects of	Support in part. There should be minor terminology changes to ensure consistency with other parts of the Plan Change.	Accept

	the network; and	
	managing the adverse	
	effects of other activities	
	on the network.	
	For example, network	
	utilities may typically	
	include buildings, poles,	
	overnead wires, pylons,	
	pipes or antennas, which	
	may nave an aaverse	
	visual impact depending	
	on their location and	
	proximity to other land	
	use activities. The	
	Installation and	
	upgrading of network	
	utilities will also typically	
	Involve earthworks.	
	However, network	
	utilities may also involve	
	few structures and have	
	limited visual impact,	
	such as underground	
	power <u>electricity</u> and	
	telecommunication lines.	
	Network utilities are also	
	often seen as a	
	necessary and normal	
	part of the environment,	
	such as a road.	
	"	

Amendment 44 – 13.1.1 Issues, Objectives and Policies: Regionally Significant Network Utilities				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	
NZ Transport Agency	1.7	Retain as notified.	Support.	Accept
KiwiRail	3.10	Retain as notified.	Support.	Accept
Transpower New Zealand	6.22	Amend as follows: <i>"Objective To recognise <u>and protect</u> the benefits of regionally significant network utilities.</i>	Oppose. The provisions fail to 'protect' regionally significant infrastructure as required under the RPS (eg: Objective 10 and Policy 8 of the RPS).	Accept in part
		Policies (a) To identify regionally significant network utilities within the City on Council planning maps, as practicable.	Paragraph 3 of the explanation and reasons which identifies the benefits of regionally significant network utilities, does not recognise the key	

		 (b) To recognise <u>and</u> <u>protect</u> the national, regional and local benefits of regionally significant network utilities. Explanation and Reasons Policy (b) recognises that regionally significant network utilities provide benefits within the City, as well as regionally and nationally. These benefits need to be protected and considered in respect of any matter relating to regionally significant network utilities. Some of these benefits are: That people and goods can travel to, and from and around the City and Region efficiently and safely; That community well-being and public health and safety is maintained through the provision of essential services including supply of potable water, the collection and transfer of sewage and stormwater, and the provision of emergency services; People have access to electricity and gas to meet their needs, <u>and location of</u> <u>supply and demand for these services can be effectively and efficiently</u> 	national benefit arising from electricity being able to be transmitted from sources of generation to the centres of demand. This has particular relevance in the Hutt City case given that the Haywards substation is a key hub in the National Grid.	
Powerco	FS2.6	Accept	Powerco supports the amendments sought. It is appropriate to protect as well as simply recognise the benefits of regionally significant network utilities.	Accept

Spark New Zealand Trading Limited	7.6	Amend the second paragraph of the Explanation and Reasons as follows: "Policy (a) requires the Council to identify regionally significant network utilities within the City on its planning maps, <u>as practicable</u> Due to the scale of the planning maps and the extensive nature of some regionally significant network utilities, it is however not feasible to identify all regionally significant network utilities on Council planning maps, particularly the local gas distribution lines <u>and</u> <u>telecommunication</u> <u>networks</u> ." Amend item (iii) in paragraph 3 of the Explanation and Reasons to 13.1.1 Regionally Significant Network Utilities as follows: <i>"iii. People have access</i> <i>to electricity,</i> <u>telecommunication</u> <u>ns</u> and gas to meet their needs."	Oppose in part. It is inconsistent and inappropriate to exclude telecommunications and radiocommunications as there is provision for similar services expressly provided within the definition (ie: electricity and gas). It is not appropriate to map telecommunications networks due to the sheer volume of detail that would be required, therefore Spark seeks to clarify that mapping should be undertaken as practicable in paragraph two of the Explanations and Reasons.	Accept in part
Chorus New Zealand Limited	8.9	Amend the second paragraph of the Explanation and Reasons as follows: "Policy (a) requires the Council to identify regionally significant network utilities within the City on its planning maps, <u>as practicable</u> Due to the scale of the planning maps and the extensive nature of some regionally significant network utilities, it is however not feasible to identify all regionally significant network	Oppose in part. It is inconsistent and inappropriate to exclude telecommunications and radiocommunications as there is provision for similar services expressly provided within the definition (ie: electricity and gas). It is not appropriate to map telecommunications networks due to the sheer volume of detail that would be required, therefore Chorus seeks	Accept in part

		utilities on Council planning maps, particularly the local gas distribution lines <u>and</u> <u>telecommunication</u> <u>networks</u> ." Amend item (iii) in paragraph 3 of the Explanation and Reasons to 13.1.1 Regionally Significant Network Utilities as follows: <i>"iii. People have access</i> <i>to electricity,</i> <u>telecommunication</u> <u>S.</u> <u>radiocommunicatio</u> <u>ns</u> and gas to meet their needs."	to clarify that mapping should be undertaken as practicable in paragraph two of the Explanations and Reasons.	
Powerco Limited	10.3	Retain as notified	Support. Accepts it would not be practicable to show gas distribution lines on the planning maps. As an alternative, it may be possible to make information about the location of Powerco's gas lines available on the Council's GIS system.	Accept

Amendment 45 – 13.1.2 Issues, Objectives and Policies: Managing Adverse Effects, including Reverse Sensitivity Effects				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
NZ Transport Agency	1.8	Retain as notified.	Support.	Accept
KiwiRail	3.11	Retain as notified.	Support. Not opposed to development on sites adjoining the rail network, however as trains can often be intermittent and cause effects on amenity, KiwiRail support that these potential effects are mitigated.	Accept
Transpower	6.23	Amend as follows:	Oppose. The qualifier	Accept in part

New	"Objective	'unreasonably' used in	
Zealand	To ensure the operation,	the Objective of this	
	maintenance, upgrading	section is not consistent	
	and development of	with Policy 10 of the	
	regionally significant	NPSET as the Policy does	
	network utilities is not	not use that word.	
	unreasonably	The new cection fails to	
	activities	recognise the adverse	
	uctivities.	effects earthworks can	
	Policies	have on the National	
	(a) To avoid, or as	Grid and that the	
	appropriate, remedy	explanation regarding	
	or mitigate, the	Policy (b) fails to	
	potential for any	recognise that avoiding	
	adverse effects,	certain activities in the	
	including reverse	National Grid Yard is not	
	sensitivity effects on	about only protecting	
	regionally significant	access but also safety	
	incompatible new	and protecting the	
	subdivision use and	transmission lines	
	development		
	occurring under,		
	over, or adjacent to		
	regionally significant		
	network utilities.		
	(b) To ensure the safe		
	and efficient		
	maintenance,		
	and development of		
	the National Grid by		
	avoiding the		
	incompatible		
	<u>earthworks,</u>		
	establishment of or		
	changes to sensitive		
	activities and		
	huildings and		
	structures within a		
	defined National Grid		
	Yard.		
	Explanation and Reasons		
	 Dollars (a) no costano that		
	rolley (u) requires that		
	effects including reverse		
	sensitivity effects on		
	regionally significant		
	network utilities are		
	appropriately managed,		
	with priority given to		
	avoiding adverse effects,		

	where practicable, on	
	those utilities. The	
	location of inappropriate	
	new subdivision use or	
	development including	
	development, <u>mcruumy</u>	
	<u>earthworks, in proximity</u>	
	to existing regionally	
	significant network	
	utilities has the potential	
	to compromise the	
	efficient operation and	
	use of the network utility	
	including by restricting	
	access and result in the	
	benefits of that network	
	utility being reduced. In	
	addition, the safety and	
	amenity values of the	
	community may be	
	adversely affected by	
	locatina in too close	
	proximity to regionally	
	sianificant network	
	utilities The notential	
	for reverse sensitivity	
	effects may arise when	
	the nattern and density	
	of land use activities	
	changes through the	
	cubdivision or rezoning	
	of land At the time of	
	of iunu. At the time of	
	rezonny, the council will	
	seek to introduce new	
	provisions to manage	
	those potential reverse	
	sensitivity effects on	
	existing or designated	
	regionally significant	
	network utilities. Any	
	applications for	
	subdivision that involve	
	potential intensification	
	located in proximity to	
	regionally significant	
	network utilities will	
	require assessment in	
	terms of the potential	
	effects on those utilities	
	as well as consultation	
	with the relevant	
	network utility operator.	
	Policy (b) recognises the	
	importance of the	
	National Grid and seeks	
	to protect the continued	

	operation maintenance	
	operation, <u>maintenance</u>	
	una upgrade and functioning	
	<u>upgrade</u> and junctioning	
	OJ LIIUL NELWORK. THE	
	policy provides for the	
	establishment of a	
	National Gria Yara	
	activities and	
	incompatible	
	earthworks buildings	
	and structures will be	
	avoided The	
	management quoidance	
	of sensitive activities and	
	incompatible	
	earthworks of huildings	
	and structures within a	
	National Grid Yard is	
	aimed at ensurina that	
	these do not hinder	
	required access to the	
	network for its on-going	
	operation and	
	maintenance, :	
	1. maintaining access	
	to the network for its	
	on-going operation,	
	<u>maintenance,</u>	
	upgrading and	
	<u>development,</u>	
	2. mitigating safety	
	<u>risks for occupants</u>	
	<u>and users of</u>	
	properties;	
	3. protecting the	
	<u>structural integrity of</u>	
	<u>transmission lines;</u>	
	<u>and</u>	
	<u>4. maintaining the</u>	
	opportunity to	
	<u>Jurtner Optimise</u>	
	Crid lines in the	
	<u>Griu imes in the</u>	
	<u>Julure.</u> which This is a matter of	
	mational significance	
	under the National	
	Policy Statement for on	
	Flectricity Transmission	
	This Chapter contains	
	specific rules that apply	
	to the use and	
	development of land	
	within the National Grid	
	Yard throughout the	
I I	-	

		District Plan. Chapter 11 Subdivision, contains specific rules that apply to subdivision within the National Grid Corridor. The on-going operation, upgrade and maintenance of the existing National Grid is provided for by the Resource Management (National Environmental Standards for Electricity Transmission Activities) Regulations 2009. These regulations specify that existing electricity transmission activities are permitted, subject to terms and conditions to ensure that these activities do not have significant adverse effects. The standards also specify resource consent requirements for electricity transmission activities that do not meet the terms and conditions for permitted activities."		
Spark New Zealand Trading Limited	7.7	Retain as notified.	Support.	Accept
Chorus New Zealand Limited	8.10	Retain as notified.	Support.	Accept
Powerco Limited	10.4	Retain as notified.	Support.	Accept

Amendment 46 – 13.1.3 Issue, Objective and Policies: Recognising and Providing for Network Utilities				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
NZ Transport Agency	1.9	Retain as notified.	Support.	Accept
KiwiRail	3.12	Retain as notified.	Support. While new rail networks are not	Accept

Transpower	6.24	Retain as notified.	common, upgrading to ensure safety and technology adjustments are accommodated, and maintenance to ensure safety is constantly maintained, are pivotal to the operation of the network. Being able to undertake works with little impediment and restriction is supported given the nature of the network, the purpose it services for the region, and the consequences of an unmaintained network on public safety being significant. This supports the changes sought to the definition at KiwiRail's submission point 3.6. Support.	Accept
New Zealand				
Spark New Zealand Trading Limited	7.8	Retain as notified.	Support.	Accept
Chorus New Zealand Limited	8.11	Retain as notified.	Support.	Accept
Wellington Electricity Lines Limited (WELL)	9.13	Amend Policy 13.1.3 (a) as follows: "(a) To recognise and provide for the: (i) need for new and the maintenance and upgrading of existing network utilities; (ii) technical and operational requirements and constraints of network utilities in assessing their location, design, development,	Support in part. As currently worded, Policy 13.1.3(a) is not explicit enough in recognising that not all environmental effects can be avoided remedied or mitigated. The policy does not state that some adverse effects may need to be accepted.	Reject

		construction and appearance; and (iii) benefits that network utilities provide to the economic, social and cultural functioning of the City. To recognise that residual adverse environmental effects have the potential to result from the development of Network Utilities, and that such residual effects will be offset by the public good, or benefit, that will be derived from the Network Utilities development.		
Transpower New Zealand	FS1.5	Accept	Transpower supports the principle of what the submitter is seeking. It is important that the District Plan recognises that to achieve the benefits which arise from network utilities some adverse effects will be necessary. Further accepting that some adverse environmental effects may arise from network utilities is part of 'providing for' these significant physical resources.	Reject
Powerco	FS2.12	Accept / reject in part	While Powerco supports the intent of the submission, the proposed new policy is not considered to be entirely necessary as PC34 contains policies that address both the benefits and effects of network utilities and these matters must be balanced when considering an	Accept

			application in the context of all policies in Chapter 13 – Network Utilities.	
Powerco Limited	10.5	Retain as notified.	Support.	Accept

Amendment	Amendment 47 – 13.1.4 Issues, Objectives and Policies: Managing Environmental Effects					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point		
Powerco Limited	10.6	Retain Issue 13.1.4 as notified.	Support in part.	Accept		
Powerco Limited	10.7	Retain Objective 13.1.4 as notified.	Support in part.	Accept		
Transpower New Zealand	6.25	Add a new policy as follows: (aa) To take account of technical and operation requirements of the network and of the extent to which any adverse effects have been avoided remedied or mitigated by route, site and method selection when considering the adverse effects of the National Grid.	Oppose. The provisions fail to adequately recognise the constraints created by the technical and operational requirements of the National Grid as required in Policy 3 of the NPSET. The provisions fail to adequately recognise any adverse effects that may have been avoided, remedied or mitigated by the route, site or method selection as required in Policy 4 of the NPSET.	Reject		
Powerco	FS2.7	Accept in part	Powerco supports the inclusion of an additional policy requiring the technical and operational requirements of network utilities to be taken into account when considering the adverse effects of such works. Any such new policy should, however, apply in relation to all linear network utilities and not just to the National Grid.	Reject		
Wellington Electricity	9.14	Amend by adding a new policy as follows:	Support in part. The PC34 policies do not	Accept in part		

Lines Limited (WELL)		" <u>To weigh up any</u> <u>residual adverse</u> <u>environmental effect</u> <u>that cannot be otherwise</u> <u>avoided remedied or</u> <u>mitigated with the</u> <u>benefits that will be</u> <u>derived from the</u> <u>Network Utility</u> ."	suitably coalesce with the stated policy explanation. The wording used in the explanation is confusing as it reads that any effects that cannot be avoided remedied or mitigated will be required to be mitigated- which is somewhat contradictory. A more appropriate concept to be articulated in Section 13.1.4 would be the balancing of positive effects to be derived from Network Utilities, with those residual environmental adverse effects that could cannot be avoided, remedied or mitigated.	
Spark New Zealand Trading Limited	7.9	Amend 13.1.4 Policy (a) as follows: "To ensure that network utilities are designed, located, developed, constructed, upgraded, operated and maintained to avoid, remedy or mitigate any actual or potential adverse effects on the environment <u>as far as</u> <u>practicable</u> ." Amend the last sentence of the first paragraph of the Explanation and Reasons to 13.1.4 as follows: "However, in some cases, it might not be entirely possible to avoid, remedy or mitigate all adverse effects associated with a network utility due to their technical and operational constraints, meaning that there will <u>may</u> be some level of <u>residual</u> adverse effect on the surrounding environment. that	Oppose in part. Changes are sought to 13.1.4 Policy (a) and the associated Explanation and Reasons to remove inconsistencies in the current wording. In particular, changes are sought to consistently acknowledge throughout the policy framework that it is not always possible to avoid, remedy or mitigate all adverse effects of network utilities due to their technical and operational constraints.	Accept in part

		requires mitigation. In such circumstances there is a need to carefully consider both the benefits the utility will provide and the significance of the adverse effects on the surrounding environment."		
Powerco	FS2.10	Accept	The amendments proposed to Policy 13.1.4(a) and the associated Explanation and Reasons are supported for the reasons set out in the primary submission.	Accept in part
Chorus New Zealand Limited	8.12	Amend 13.1.4 Policy (a) as follows: <i>"To ensure that network utilities are designed,</i> <i>located, developed,</i> <i>constructed, upgraded,</i> <i>operated and</i> <i>maintained to avoid,</i> <i>remedy or mitigate any</i> <i>actual or potential</i> <i>adverse effects on the</i> <i>environment <u>as far as</u> <u>practicable.</u>" Amend the last sentence of the first paragraph of the Explanation and Reasons to 13.1.4 as follows: <i>"However, in some</i> <i>cases, it might not be</i> <i>entirely possible to</i> <i>avoid, remedy or</i> <i>mitigate all adverse</i> <i>effects associated with a</i> <i>network utility due to</i> <i>their technical and</i> <i>operational constraints,</i> <i>meaning that there will <u>may be some level of</u> <i>residual adverse effect</i> <i>on the surrounding</i> <i>environment. that</i> <i>requires mitigation.</i> In <i>such circumstances there</i> <i>is a need to carefully</i> <i>consider both the</i></i></i>	Oppose in part. Changes are sought to 13.1.4 Policy (a) and the associated Explanation and Reasons to remove inconsistencies in the current wording. In particular, changes are sought to consistently acknowledge throughout the policy framework that it is not always possible to avoid, remedy or mitigate all adverse effects of network utilities due to their technical and operational constraints.	Accept in part

1		1
	benefits the utility will	
	provide and the	
	, sianificance of the	
	adverse effects on the	
	surrounding	
	environment "	
	environment.	
	Amend the second	
	naragraph of the	
	Explanation and Reasons	
	to 13.1.4 as follows:	
	Policy (a) recognises the	
	importance of managing	
	the design location	
	operation upgrading	
	construction operations	
	and maintenance of	
	network utilities while	
	also acknowledging	
	that it is not always	
	nossible to and requires	
	that any notential	
	adverse effects arising	
	from network utilities	
	are avoided remedy ied	
	or mitigate d all adverse	
	effects This is due to	
	the technical and	
	operational	
	<u>operational</u> requirements of	
	<u>operational</u> <u>requirements of</u> network utilities (as	
	<u>operational</u> <u>requirements of</u> <u>network utilities (as</u> identified in Policy	
	operational requirements of network utilities (as identified in Policy 13.1.3 (a) (ii)).	
	operational requirements of network utilities (as identified in Policy 13.1.3 (a) (ii)).	
	operational requirements of network utilities (as identified in Policy 13.1.3 (a) (ii)).	
	operational requirements of network utilities (as identified in Policy 13.1.3 (a) (ii)). Amend the fifth paragraph in the	
	operational requirements of network utilities (as identified in Policy 13.1.3 (a) (ii)). Amend the fifth paragraph in the Explanation and Beasons	
	operational requirements of network utilities (as identified in Policy 13.1.3 (a) (ii)). Amend the fifth paragraph in the Explanation and Reasons to 13.1.4 (relating to	
	operational requirements of network utilities (as identified in Policy 13.1.3 (a) (iii)). Amend the fifth paragraph in the Explanation and Reasons to 13.1.4 (relating to Policy (d)) as follows:	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (iii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires the	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderaround placement	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderground placementof network utilities	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderground placementof network utilitiesunless particular	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (iii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderground placementof network utilitiesunless particularcircumstances apply	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderground placementof network utilitiesunless particularcircumstances apply.The adverse visual	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderground placementof network utilitiesunless particularcircumstances apply.The adverse visualeffects of certain	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderground placementof network utilitiesunless particularcircumstances apply.The adverse visualeffects of certainnetwork utilities can	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderground placementof network utilitiesunless particularcircumstances apply.The adverse visualeffects of certainnetwork utilities canoften be managed by	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderground placementof network utilitiesunless particularcircumstances apply.The adverse visualeffects of certainnetwork utilities canoften be managed byputting the services	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderground placementof network utilitiesunless particularcircumstances apply.The adverse visualeffects of certainnetwork utilities canoften be managed byputting the servicesunderground This is the	
	operational requirements of network utilities (as identified in Policy 13.1.3 (a) (ii)).Amend the fifth paragraph in the Explanation and Reasons to 13.1.4 (relating to Policy (d)) as follows: Policy (d) requires the underground placement of network utilities unless particular circumstances apply. The adverse visual effects of certain network utilities can often be managed by putting the services underground. This is the generally the required	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderground placementof network utilitiesunless particularcircumstances apply.The adverse visualeffects of certainnetwork utilities canoften be managed byputting the servicesunderground. This is thegenerally the requiredapproach for those	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d) requires theunderground placementof network utilitiesunless particularcircumstances apply.The adverse visualeffects of certainnetwork utilities canoften be managed byputting the servicesunderground. This is thegenerally therequiredapproach for thosenetwork utilities such as	
	operationalrequirements ofnetwork utilities (asidentified in Policy13.1.3 (a) (ii)).Amend the fifthparagraph in theExplanation and Reasonsto 13.1.4 (relating toPolicy (d)) as follows:Policy (d)) as follows:Policy (d) requires theunderground placementof network utilitiesunless particularcircumstances apply.The adverse visualeffects of certainnetwork utilities canoften be managed byputting the servicesunderground. This is thegenerally therequiredapproach for thosenetwork utilities, such ascables that can be	
	operational requirements of network utilities (as identified in Policy 13.1.3 (a) (ii)).Amend the fifth paragraph in the Explanation and Reasons to 13.1.4 (relating to Policy (d)) as follows: Policy (d) requires the underground placement of network utilities unless particular circumstances apply. The adverse visual effects of certain network utilities can often be managed by putting the services underground. This is the generally the required approach for those network utilities, such as cables that can be located underground	
	operational requirements of network utilities (as identified in Policy 13.1.3 (a) (ii)).Amend the fifth paragraph in the Explanation and Reasons to 13.1.4 (relating to Policy (d)) as follows: Policy (d) requires the underground placement of network utilities unless particular circumstances apply. The adverse visual effects of certain 	

		underground placement of electricity lines and telecommunication lines is required <u>in most</u> <u>circumstances</u> by only providing for aboveground lines in particular defined situations, such as for customer connections <u>and where permitted as</u> <u>minor upgrading.</u> and through different activity status. New above ground lines and their associated supporting structures in areas that do not have existing above ground lines are generally considered to be unacceptable within the City, except in those areas that can visually absorb new above ground lines, such as the Rural zone where they are permitted.		
Transpower New Zealand	FS1.4	Accept	Transpower supports this submission as the principle is an important part of 'providing for' the benefits of network utilities.	Accept in part
Powerco Limited	10.8	Retain Policy 13.1.4 (a) as notified.	Support in part.	Accept
Powerco Limited	10.9	Amend Policy (b) as follows: "(b) <u>To manage effects</u> <u>on health and</u> <u>safety by ensuring</u> <u>Ensure</u> network utilities, in particular those emitting electric and magnetic fields, are designed, located, upgraded, operated and maintained to comply with relevant national environmental standards and to meet other	Support in part. The explanation and reasons given to Policy (b) identify health and safety matters as being the focus; however this is not reflected in the Policy.	Accept

Transpower	FS1.6	nationally recognised standards <u>and</u> <u>guidelines</u> ." Accept	Transpower considers	Accept
New Zealand			this change is appropriate.	
Transpower New Zealand	6.26	Amend Policy (c) as follows: "(c) To enable the co- location or multiple use of network utilities where this is efficient and practicable operationally and technically feasible and assists with avoiding, remedying or mitigating adverse effects on the environment."	Oppose. The reference to 'practicable' in policy (c) should be replaced with 'operationally and technically feasible'.	Accept in part
Powerco	FS2.8	Accept	Support.	Accept
Spark New Zealand Trading Limited	7.10	Retain 13.1.4 Policy (c) as notified.	Support.	Accept in part
Chorus New Zealand Limited	8.13	Retain 13.1.4 Policy (c) as notified.	Support.	Accept in part
Powerco Limited	10.10	Retain Policy 13.1.4 (c) as notified.	Support in part.	Accept in part
Greater Wellington Regional Council	2.2	Amend policy 13.1.4 (d) as follows: "to require the underground placement of new network utilities unless (i) there are natural or physical features or structures, or technological and operational constraints <u>and</u> <u>effects on flood flow</u> <u>capacity and</u> <u>stopbank integrity,</u> that makes underground placement impractical or	Supports the encouragement of co- location services, but seeks recognition within the proposed policy or its explanation that co- location is not always appropriate, namely where its effects may be to constrict flood flows.	Reject

		unreasonable"		
Powerco	FS2.2	unreasonable" Reject in part	Powerco accepts there is a need to protect the integrity of flood protection assets and that this may not be achieved if there is a requirement to underground new network utilities under or near stopbanks. However, the policy already recognises that undergrounding of new network utilities may not be appropriate where physical features or structures are present (which would include stopbanks) and as such, Powerco does not consider the amendment is necessary. If the council is minded to include a specific reference to stopbanks and flood flow capacity, the additional wording should be amended to use 'or' rather than 'and', as follows: "to require the underground placement of new network utilities unless (i) there are natural or physical features or structures, or technological and operational constraints and or effects on flood flow capacity and or stopbank integrity, that makes underground placement impractical or unreasonable" This would ensure	Reject
			effects on flood flow capacity and stopbanks are considered separately to, rather than in conjunction with, the remainder of the	

			matters addressed in Policy 13.1.4(d).	
Transpower New Zealand	6.27	Amend Policy (d) as follows: "(d) <u>Except for</u> <u>transmission lines,</u> to require the underground placement of new network utilities unless (i) there are natural or physical features or structures, or technological and operational constraints that makes underground placement impractical or unreasonable; (ii) they are of a temporary nature and required for emergency purposes or critical events; and (iii) they are of a nature that they can only operate aboveground. "	Oppose. Policy (d) does not recognise that a key constraint associated with the undergrounding of the National Grid is the financial cost which makes it impractical for the country (it can be 10- 15 times more expensive (as a monopoly Transpower is heavily regulated and any significant expenditure must be specially approved by the Commerce Commission.)). Policy (d) should either recognise this financial constraint or specifically exempt the National Grid.	Reject
Powerco Limited	10.11	Retain Policy 13.1.4 (d) as notified.	Support in part.	Accept
Powerco Limited	10.12	Retain Policy 13.1.4 (e) as notified.	Support in part.	Accept
Greater Wellington Regional Council	2.3	Amend policy 13.1.4 (f) as follows: "(f) To encourage network utility providers to consult with local communities <u>and</u> <u>the regional council</u> on the appropriate placement, location and design of new	Support. Seek regional council be included as they are not always informed that activities and earthworks are happening on, or in, stopbanks, and uncontrolled earthworks can and do cause a real risk of stopbank breach or failure.	Accept in part

		network utilities"		
Powerco Limited	10.13	Retain Policy 13.1.4 (f) as notified.	Support in part.	Accept
Transpower New Zealand	6.28	Amend the Explanations and Reasons as follows: "Explanation and Reasons The issue and supporting objective recognise that the construction, operation, upgrading and/or maintenance of network utilities can have adverse effects and adversely affect the amenity of areas of the City, as a result of noise, emissions, and visual dominance, for example. Some network utilities are relatively large, visually prominent and capable of generating significant adverse effects on the surrounding environment. Such network utilities may also have <u>perceived</u> adverse effects on public health and safety. Adverse effects may only occur at the time of construction or installation of the utility, but in some instances may continue throughout its operation or during maintenance and/or upgrade works. For new linear network utilities, adverse effects are often best able to be mitigated through the route selection process. However, in some cases, it might not be entirely possible to avoid, remedy or mitigate all adverse effects associated with a network utility due to their technical and	Oppose. The references to public health and safety in the explanation and reasons are inaccurate and overstate the risks and the second to last sentence of the first paragraph contains some internal inconsistency.	Accept in part
		meaning there will be		

some level of adverse	
effect on the	
surrounding	
environment. that	
requires mitigation. In	
such circumstances	
there is a need to	
carefully consider both	
the henefits the utility	
will provide and the	
significance of the	
significance of the	
adverse effects on the	
surrounding	
environment.	
Policy (b) recognises that	
some network utilities	
may adversely affect	
health and safety. For	
example,	
telecommunication	
facilities generate radio	
frequency emissions	
which may have	
detrimental effects on	
health. Any potential	
health effects arising	
from radiofrequency	
emissions are addressed	
by Regulation 4 of the	
Resource Management	
(National Environmental	
Standards for	
Facilities) Regulations	
2008. Electricity	
transmission/distribution	
can generate	
ENTER which may be a	
(EVIF) WHICH MAY DE A	
nonaratas tha risk of	
generates the HSK Of	
transmission (distribution	
<u>cun generate</u>	
(FAAE) the National	
(EMF), the National	
Foncy Statement on	
<u>Electricity Transmission,</u>	
avpasures ha limited to	
the quidelines of the	
International	
<u>International</u>	
CONTINUSSION ON INON-	
IUNISING KAAIALION	
Protection (ICNIRP) to	

		provent the notential for		
		health effects		
		<u>neurin effects</u> .		
		 Policy (c) recognises that		
		the co-location and co-		
		sitina of network utilities		
		mav provide		
		environmental benefits		
		in terms of reduced		
		visual impacts and		
		consolidation of network		
		utilities in existing areas		
		thereby reducing		
		adverse effects on		
		amenity by reducing the		
		need for more network		
		utility structures. While		
		co-location is		
		encouraged it needs to		
		be understood that		
		technical requirements		
		will generally mean that		
		associated structures		
		may need to be taller or		
		bulkier to avoid		
		interference between the		
		two or more providers,		
		such as radio-frequency		
		bands. It is also		
		recognised that co-		
		location is not always		
		possible due to		
		operational issues such		
		as radiofrequency		
		interference, electrical		
		interference, lease		
		arrangements and		
		structural capacity."		
Powerco	10.14	Amend 13.1.4	Support in part. There is	Accept in part
Limited		Explanations and	concern that the	
		Reasons as follows:	wording of the	
		"Policy (b) recognises	explanation relating to	
		that some network	electromagnetic fields	
		utilities may adversely	(EMF) generated by	
		affect health and safety.	electricity	
		For example,	transmission/distribution	
		telecommunication	activities may cause	
		facilities generate radio	unnecessary concern	
		frequency emissions	about potential health	
		which may have	effects of EMF. Despite	
		detrimental effects on	numerous international	
		health. Any potential	studies, there is still no	
		health effects arising	persuasive evidence that	
		from radiofrequency	EMF poses any health	
		emissions are addressed	risks. Because of this, it	
		by Regulation 4 of the	is therefore not	

Resource Management (National Environmental Standards for Telecommunication Facilities) Regulations 2008. Electricity transmission/distribution activities can generate electromagnetic fields (EMF) which may be present a risk to public health <u>and safety, primarily though, and also generates</u> the risk of electrocution <u>from direct</u> contact with conductors or as a result of a flashover. Other possible	appropriate to suggest that EMF from electricity transmission/distribution may be a risk to health or for the policy to focus on utilities emitting electric and magnetic fields.	
health and safety risks are accidental spillage or leakage of hazardous substances from gas or petroleum pipelines, explosions from gas or petroleum pipelines, accidental overflow from sewage pump stations, and flooding from damaged/inoperative stormwater systems. Chemicals used in conjunction with some network utilities, such as water treatment plants		
for example, also pose a risk if an accidental spill occurs. There are a number of <u>relevant</u> national and international standards <u>and quidelines</u> <u>addressing health and</u> <u>safety matters</u> that are external to the District Plan. <u>Some of these but that</u> must be complied with, including the Resource Management (National Environmental Standards for Telecommunication Facilities) Regulations 2008, <u>while others</u> provide best practice		
Code of Practice and the International Commission on Non- Ionising Radiation		
---	--	
Protection (ICNIRP) Guidelines."		

Amendment 48 – How to use the Network Utility Rules				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Powerco Limited	10.15	Retain as notified.	Support.	Accept

Amendment 49 – Network Utility Rules 13.3.1				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Greater Wellington Regional Council	2.4	 Add new rules for Removal, Maintenance and Upgrading (Rules 13.3.1.1 to 13.3.1.7) Make any removal, maintenance or upgrading associated with underground services (excluding those associated with works on existing transmission lines) where they are located in or on a stopbank a restricted discretionary activity. Council's discretion should be restricted to the effects of the works on the integrity of the stop-bank. To accord with the NESETA, such works on existing electricity transmission lines in stopbanks should be a controlled activity. Council's control should be restricted to the effects of the works on the 	GWRC has concerns about the implications of permitted activity status for utilities operations where there is potential for adverse effects on flood protection assets. With regard to the National Environmental Standard on Electricity Transmission Activities (NESETA), GWRC notes that permitted activity clause 33(5) requires that earthworks related to an existing transmission line must not contribute to slope instability, drainage problems, or flooding of overland flow paths, otherwise earthworks are to be considered a controlled activity. Earthworks in stopbanks can contribute to all of these effects e.g. slope instability; therefore it is appropriate that where existing electricity	Reject

		integrity of the stop- bank.	transmission lines are concerned, the activity status should be controlled. However, for all other network utilities including new utilities, the activity status should be a restricted discretionary activity, so that GWRC retains the ability to seek that an application be refused if the flood hazard risks/effects are inappropriate.	
Powerco	FS2.3	Accept	Powerco accepts there is a need to protect the integrity of flood protection assets and can support a rule requiring restricted discretionary consent for removal, maintenance or upgrades associated with underground services where they are located in or on a stop- bank.	Reject
Transpower New Zealand	6.29	 Amend the Activity Status in 13.3.1 so that: It can be easily determined which rules apply to different transmission activities. In this regard Transpower seeks very clear rules and related definitions associated with the operation and maintenance, minor upgrading, upgrading and establishment of new transmission lines (including their support structures) and of substations National Grid activities are given the following activity status: o Permitted: The operation, 	Oppose. The rules and their associated definitions should be re- drafted to ensure that network utility operators can easily determine the activity status of their activities.	Accept in part

Transpower	6.32	 maintenance, minor-upgrading of transmission lines, transmission line support structures and substations. Restricted Discretionary: The upgrading of transmission lines, transmission line support structures and substations. Discretionary: New transmission lines, transmission line support structures and substations. Discretionary: New transmission lines, transmission line support structures and substations. Non-complying: All network utilities which do not comply with the permitted activity standards for electric and magnetic fields. Transmission lines and their support structures are covered by the same rules The status of activities which do meet standards is clearly stated Only Health and Safety Standard 13.3.2.1 applies to discretionary activities for new lines 	Oppose. The rules and	Accept in part
New Zealand		specifically provide for transmission line support structures as the standards referenced are too restrictive (particularly height). It is considered that support structures	their associated definitions should be re- drafted to ensure that network utility operators can easily determine the activity status of their activities.	

should be specifically provided for and that this would give effect to	
the NPSET and RPS.	

Amendment 49 – Network Utility Rule 13.3.1.1 Removal of existing network utilities, including any existing associated structures				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
KiwiRail	3.13	Retain as notified.	Support.	Accept
Wellington Electricity Lines Limited (WELL)	9.15	Retain as notified.	Support.	Accept
Powerco Limited	10.16	Retain as notified.	Support.	Accept

Amendment 49 – Network Utility Rule 13.3.1.2 Operation and maintenance of existing network utilities				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
KiwiRail	3.14	Amend as follows: "The operation and maintenance of existing network utilities."	Support. Question whether this should not reference 'existing' – new network utilities should also be able to be operated and maintained, with any consent or adverse effects being managed through the consent process to construct the network utility. At the time that consent is sought for construction, the operation and maintenance would not be a permitted activity as the network utility would not be existing. Certainty over whether that would also therefore require consent, or could at a future stage rely on a permitted activity rule	Accept

			would be removed with the removal of the word 'existing'.	
Transpower New Zealand	FS1.7	Accept	Transpower agrees that the term is unnecessary and may create confusion about when a network utility needed to exist to come within the scope of these rules. Transpower considers that once a network utility exists it should be subject to the permitted activity rules, regardless of when it was constructed.	Accept
Wellington Electricity Lines Limited (WELL)	9.16	Retain as notified.	Support.	Accept
Powerco Limited	10.17	Retain as notified.	Support.	Accept

Amendment 49 – Network Utility Rule 13.3.1.3 Minor upgrading of existing electricity and telecommunication lines				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
NZ Transport Agency	1.10	Remove the word existing from this rule. "The minor upgrading of existing electricity and telecommunication lines."	Support in part. All network utilities need to be able to operate and maintain their assets without the need for consent.	Accept
Transpower New Zealand	FS1.8	Accept	Transpower agrees that the term is unnecessary and may create confusion about when a network utility needed to exist to come within the scope of these rules. Transpower considers that once a network utility exists it should be subject to the permitted activity rules, regardless of when it was constructed.	Accept
Transpower	6.30	Amend Rule 13.3.1.3 as	Oppose. The rules and	Accept

New Zealand		it is unclear on whether it covers minor upgrades to the support structures associated with lines. Minor upgrade of these structure should be provided for as a permitted activity (as is anticipated by the definition of minor upgrade)	their associated definitions should be re- drafted to ensure that network utility operators can easily determine the activity status of their activities.	
Wellington Electricity Lines Limited (WELL)	9.17	Retain as notified.	Support.	Accept

Amendment 49 – Network Utility Rule 13.3.1.4 Upgrading of existing network utilities excluding electricity and telecommunication lines and gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals

Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
KiwiRail	3.15	Retain as notified.	Support.	Accept
Powerco Limited	10.18	Remove the requirement for Upgrading of Existing Network Utilities to comply with Standard 13.3.2.5.2 for Slope, Height, Depth and Area of Earthworks.	The upgrading work Powerco undertakes involves trenching which is an essential and temporary activity. Powerco submits that as currently worded, consent would be required for earthworks to enable upgrading work on existing gas distribution lines located at a depth of more than 1.5m. While the average depth of gas distribution lines is typically less than 1.5m, it may not remain constant due to topography, road or stream placement or third party actions (ie: additional filling). Powerco does not want to be in a position where upgrading is required on a section of line to find that a section is deeper than 1.5m and a consent	Accept in part

is required before the work can be carried out	
It is impractical for the Standard to set an area threshold for earthwork based on the zone in which the activity is located as linear infrastructure crosses multiple zone boundaries as well as th road reserve.	s

Amendment 49 – Network Utility Rule 13.3.1.5 Removal, operation and maintenance of network utilities and the minor upgrading of electricity and telecommunication lines that does not meet permitted activity standards

Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
KiwiRail	3.16	Retain as notified.	Support.	Accept
Transpower New Zealand	FS1.9	Accept	Transpower agrees this rule is appropriate and should be retained.	Accept

Amendment 49 – Network Utility Rule 13.3.1.6 Upgrading of existing network utilities excluding electricity and telecommunication lines and gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals that does not meet permitted activity standards

Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
KiwiRail	3.17	Retain as notified.	Support but conditional upon the definition of Upgrading and Maintenance being clarified so that repair and replacement are clearly not upgrading. Without that clarification, there is the potential that activities that would be permitted or controlled in the event the permitted standards are not complied with become restricted discretionary.	Accept
Transpower New	6.31	Amend Rule 13.3.1.6 'upgrading of network	Oppose. The rules and their associated	Accept in part

Zealand		utilities' excluding lines above 110kV that do not meet permitted activity standards - as it is unclear which rule applies if the upgrade does meet permitted activity standards	definitions should be re- drafted to ensure that network utility operators can easily determine the activity status of their activities.	
Wellington Electricity Lines Limited (WELL)	9.18	Subject to WELL's requested amended to the Definition of Minor Upgrading (submission reference 9.8) being accepted by Council, Retain Rule 13.3.1.6 as notified.	Support in part.	Accept in part

Amendment 49 – Network Utility Rule 13.3.1.8 Subdivision for the purpose of accommodating any network utility

Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Powerco Limited	10.19	Retain as notified.	Support.	Accept	
Transpower New Zealand	FS1.10	Accept	Transpower agrees this rule is appropriate and should be retained.	Accept	

Amendment 49 – Network Utility Rule 13.3.1.9 Cabinets and other network utility structures not otherwise listed (in all Activity Areas excluding Historic Residential and Landscape Protection Residential)

Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Transpower New Zealand	6.33	Amend Rules 13.3.1.9 so it is clear whether these apply to new cabinets and structures as this is not clearly stated (as compared to the rules relating to lines).	Oppose. The rules and their associated definitions should be re- drafted to ensure that network utility operators can easily determine the activity status of their activities.	Reject

Amendment	Amendment 49 – Network Utility Rule 13.3.1.10 Cabinets and other network utility structures			
not otherwis	not otherwise listed (in Historic Residential and Landscape Protection Residential Activity			
Areas)	Areas)			
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation

				on Submission Point
Transpower New Zealand	6.34	Amend Rules 13.3.1.10 so it is clear whether these apply to new cabinets and structures as this is not clearly stated (as compared to the rules relating to lines).	Oppose. The rules and their associated definitions should be re- drafted to ensure that network utility operators can easily determine the activity status of their activities.	Reject

Amendment not otherwis	Amendment 49 – Network Utility Rule 13.3.1.11 Cabinets and other network utility structures not otherwise listed that do not meet the permitted activity standards in Rule 13.3.1.9				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Transpower New Zealand	6.35	Amend Rules 13.3.1.11 so it is clear whether these apply to new cabinets and structures as this is not clearly stated (as compared to the rules relating to lines).	Oppose. The rules and their associated definitions should be re- drafted to ensure that network utility operators can easily determine the activity status of their activities.	Reject	
Spark New Zealand Trading Limited	7.11	Delete the application of the standards under Rule 13.3.1.11.	Oppose in part. It is unclear why the same standards as applied to cabinets as a permitted activity under Rule 13.3.1.9 are then again applied under the default restricted discretionary activity rule.	Accept in part	
Transpower New Zealand	FS1.11	Accept	Transpower agrees with the submitter that it is unclear why the same standards which are applied to permitted activities under Rule 13.3.1.9 are then again applied under the default restricted discretionary activity rule.	Accept in part	
Chorus New Zealand Limited	8.14	Delete the application of the standards under Rule 13.3.1.11.	Oppose in part. It is unclear why the same standards as applied to cabinets as a permitted activity under Rule 13.3.1.9 are then again	Accept in part	

	applied under the default restricted discretionary activity rule.
--	--

Amendment 49 – Network Utility Rule 13.3.1.12 Cabinets and other network utility structures not otherwise listed in this table that do not meet the restricted discretionary activity standards (in Historic Residential and Landscape Protection Residential Activity Areas)

Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Transpower New Zealand	6.36	Amend Rules 13.3.1.12 so it is clear whether these apply to new cabinets and structures as this is not clearly stated (as compared to the rules relating to lines).	Oppose. The rules and their associated definitions should be re- drafted to ensure that network utility operators can easily determine the activity status of their activities.	Reject

Amendment 49 – Network Utility Rule 13.3.1.14 Aerial crossings necessary for network utilities located on or within existing bridges and structures or across streams and including regulator stations but not compressor stations

Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
KiwiRail	3.18	Amend as follows: "Aerial crossings necessary for network utilities, located on or within existing bridges and structures or across <u>watercourses</u> streams , and including regulator stations but not compressor stations."	Support in part. The current wording of the rule could be clarified for certainty and consistency with the RMA.	Accept
Transpower New Zealand	FS1.12	Accept	Transpower agrees that amendment will help to clarify the scope of the rule, and that otherwise the rule should be retained.	Accept

Amendment 49 – Network Utility Rule 13.3.1.17 Underground Utilities				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point

Greater Wellington Regional Council	2.5	Make new underground utilities where they are located in or on a stopbank a restricted discretionary activity, not a permitted activity as currently proposed. Council's discretion should be restricted to the effects of the works on the integrity of the stop-bank.	GWRC has concerns about the implications of permitted activity status for utilities operations where there is potential for adverse effects on flood protection assets.	Reject
Powerco	FS2.4	Accept	Powerco accepts there is a need to protect the integrity of flood protection assets and can support a rule requiring restricted discretionary consent for new underground facilities where they are located in or on a stopbank.	Reject
KiwiRail	3.19	Retain as notified.	Support.	Accept

Amendment 49 – Network Utility Rule 13.3.1.18 Masts with or without associated antennas (in Commercial (all), Business (all), General Rural, Community Health and Community Iwi Activity Areas)

Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Spark New Zealand Trading Limited	7.12	Amend Rule 13.3.1.18 to provide as a permitted activity for masts with or without associated antennas as a permitted activity in the General or Hill Residential, Rural Residential and General Recreation Activity areas.	Oppose. There is no proposed permitted provision for masts and associated antennas in the General or Hill Residential, Rural Residential and General Recreation zones. The restriction on masts in these areas is unreasonable. The areas have not been specifically zoned to protect 'amenity values' and the permitted activity standards will ensure that any adverse effects are less than minor.	Reject
Chorus New	8.15	Amend Rule 13.3.1.18 to	Oppose. There is no	Reject

Zealand Limited	provide as a permitted activity for masts with or without associated antennas as a permitted activity in the General or Hill Residential, Rural Residential and General Recreation Activity areas.	proposed permitted provision for masts and associated antennas in the General or Hill Residential, Rural Residential and General Recreation zones. The restriction on masts in these areas is unreasonable. The areas have not been	
	activity in the General or	Residential, Rural	
	Hill Residential, Rural	Residential and General	
	Residential and General	Recreation zones. The	
	Recreation Activity	restriction on masts in	
	areas.	these areas is	
		unreasonable. The areas	
		have not been	
		specifically zoned to	
		protect 'amenity values'	
		and the permitted	
		activity standards will	
		ensure that any adverse	
		effects are less than	
		minor.	
	 1	1	

Amendment 49 – Network Utility Rule 13.3.1.19 Masts with or without associated antennas (in Residential (excluding Historic and Landscape Protection), Recreation (all) and Rural Residential Activity Areas)

Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Spark New Zealand Trading Limited	7.13	Consequential amendments are required to Rule 13.3.1.19 to remove the activity areas outlined in submission point 7.12 from the Rule.	Oppose. Refer to Reasons under submission point 7.12.	Reject
Chorus New Zealand Limited	8.16	Consequential amendments are required to Rule 13.3.1.19 to remove the activity areas outlined in submission point 8.15 from the Rule.	Oppose. Refer to Reasons under submission point 8.15.	Reject

Amendment 49 – Network Utility Rule 13.3.1.23 Masts, with or without associated antennas that do not meet permitted activity standards				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Spark New Zealand Trading Limited	7.14	Consequential amendments are required to 13.3.1.23 to include the activity areas outlined in submission	Oppose. Refer to Reasons under submission point 7.12.	Reject

		point 7.12.		
Chorus New 8.3 Zealand Limited	.17	Consequential amendments are required to Rule 13.3.1.23 to include these activity areas outlined in submission point 8.15.	Oppose. Refer to Reasons under submission point 8.15.	Reject

Amendment 49 – Network Utility Rule 13.3.1.26 New and additional above ground lines (in all Rural Activity Areas)					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Transpower New Zealand	6.37	Amend Rule 13.3.1.26 as: It is unclear if new support structures as well as lines are covered by these rules. Transpower considers that a single provision should cover both new lines and support structures and that this would assist the clarity of the plan. It is also unclear what activity status applies if the standards listed for either rules are not complied with.	Oppose. The rules and their associated definitions should be re- drafted to ensure that network utility operators can easily determine the activity status of their activities.	Accept in part	

Amendment 49 – Network Utility Rule 13.3.1.27 New and additional above ground lines (in all Activity Areas excluding all Rural)					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Transpower New Zealand	6.38	Amend Rule 13.3.1.27 as: It is unclear if new support structures as well as lines are covered by these rules. Transpower considers that a single provision should cover both new lines and support structures and that this would assist the clarity	Oppose. The rules and their associated definitions should be re- drafted to ensure that network utility operators can easily determine the activity status of their activities.	Accept in part	

		of the plan. It is also unclear what activity status applies if the standards listed for either rules are not complied with. As a discretionary activity the listed standards are inappropriate and effectively serve no purpose as Council has full discretion to assess effects of proposals under this rule.		
Chorus New Zealand Limited	8.18	Amend as follows: "New or additional above ground lines <u>not</u> <u>otherwise provided for</u> ."	Oppose in part. As currently drafted, the provision for a fibre optic line using existing or replacement support structures is a Discretionary Activity under Rule 13.3.1.27 in all zones except the Rural zone (where it is permitted activity under Rule 13.3.1.26). In conjunction with Chorus' submission point 8.6 to Amendment 15, an amendment is required to Rule 13.3.1.27 to remove potential conflict. Chorus state that their identified issue also applies to the permitted provision of additional lines elsewhere e.g. "minor above ground lines".	Accept

Amendment 49 – Network Utility Rule 13.3.1.28 Minor above ground lines					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Chorus New Zealand Limited	8.19	Retain as notified.	Support.	Accept	
Wellington Electricity Lines Limited	9.19	Retain as notified.	Support.	Accept	

(WELL)				
--------	--	--	--	--

Amendment 49 – Network Utility Rule 13.3.1.30 New and upgraded transformers, substations and switching stations distributing electricity and ancillary buildings				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Transpower New Zealand	6.39	Amend Rule 13.3.1.30 as it is the only specific rule applying to substations and requires consent for all but the operation and maintenance of these important components of the national grid. The rules should provide for the operation, maintenance, replacement work and minor upgrade of the substations within Hutt City without the need for resource consent.	Oppose. The rules and their associated definitions should be re- drafted to ensure that network utility operators can easily determine the activity status of their activities.	Accept in part
Wellington Electricity Lines Limited (WELL)	9.20	Amend as follows: "New and upgraded transformers, substations and switching stations (other than those encased within a cabinet and/or those that are pole mounted within the Rural Activity Area), distributing electricity and ancillary buildings."	Oppose as it includes all new and upgraded transformers as a discretionary activity in all activity areas. Transformers vary in size according to voltage. The smaller transformers will be able to meet the permitted activity standards for all zones specified in Rule 13.3.1.9 (note: subject to road reserve area standards being increased to 5m ² as sought below). The permitted activity rule for Cabinets should also apply to transformers and switch gear located within a Cabinet as the proposed definition includes these. As currently worded Rule 13.3.1.30 would require resource consent (discretionary activity) for overhead transformers in all	Accept in part

	consider that the	
	definition of "line" in	
	Section 2 of the	
	Electricity Act 1992,	
	which is inherited as the	
	definition of "line" in this	
	plan change, includes	
	overhead transformers,	
	and therefore placement	
	of an overhead	
	transformer (pole	
	mounted) on an	
	overhead line, within the	
	Rural Activity Area,	
	should be a permitted	
	activity to reflect that	
	overhead lines in the	
	rural resource area are	
	also permitted activities.	
	It is unreasonable to	
	have permitted	
	overhead lines in the	
	rural environment only	
	to then apply a	
	mandatory resource	
	consent requirement by	
	virtue of the lines	
	operational	
	requirements.	

Amendment 49 – Network Utility Rule 13.3.1.31 Underground gas distribution and transmission pipelines at a pressure not exceeding 2000 kilopascals, including aerial crossings of bridges, structures or streams, and ancillary equipment, including regulator stations but not compressor stations

Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Powerco Limited	10.20	Remove the requirement for Upgrading of Existing Network Utilities to comply with Standard 13.3.2.5.2 for Slope, Height, Depth and Area of Earthworks.	Support subject to amendment (refer to submission point 10.18). Seek consistency with Porirua City Council's Network Utility Plan Change 16.	Accept in part

Amendment decorative p artworks and located with	Amendment 49 – Network Utility Rule 13.3.1.37 Traffic control signals and devices, light and decorative poles and associated structures and fittings, post boxes, landscaped gardens, artworks and sculptures, bus stops and shelters, phone boxes, public toilets and road furniture located within road reserve				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation	

				on Submission Point
KiwiRail	3.20	Amend as follows: Traffic control signals and devices, light and decorative poles and associated structures and fittings, post boxes, landscaped gardens, artworks and sculptures, bus stops and shelters, phone boxes, public toilets and road furniture located within the road reserve and the rail <u>corridor</u> ."	Support in part. Within the rail corridor there are also traffic control signals and devices, light poles, gardens, shelters, phone boxes, etc.	Accept

Amendment	: 51 – Standa	rd 13.3.2.2.1 Height		
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Transpower New Zealand	6.40	Amend the height standard to exclude transmission lines including their associated support structures.	Oppose. This standard inappropriately applies to transmission lines and other structures. Transmission lines are commonly higher than the Standard for Height would allow. The Standard is very restrictive and would mean that the activity status rules for new and additional above ground lines (13.3.1.26 and 13.3.1.27) do not specifically provide for transmission lines. This does not give effect to the proposed objectives and policies, or the RPS and NPSET.	Accept in part
Spark New Zealand Trading Limited	7.15	Amend the permitted standard 13.3.2.2.1 to provide for a height of 20m in the General Rural zone and for a 15m height limit in the General Recreation and the Rural Residential zones for one provider	Oppose in part. The existing height provisions are too restrictive in terms of the likely level of adverse effects and the necessity of telecommunications and radiocommunication facilities, particularly in the Rural zone where the	Reject

			height (for one provider) is less than in the existing District Plan height limit, and where the environment has the ability to absorb larger structures. The provision of the same height limit for one or two providers will not encourage co- location in the Rural zone.	
Chorus New Zealand Limited	8.20	Amend the permitted standard 13.3.2.2.1 to provide for a height of 20m in the General Rural zone and for a 15m height limit in the General Recreation and the Rural Residential zones for one provider	Oppose in part. The existing height provisions are too restrictive in terms of the likely level of adverse effects and the necessity of telecommunications and radiocommunication facilities, particularly in the Rural zone where the height (for one provider) is less than in the existing District Plan height limit, and where the environment has the ability to absorb larger structures. The provision of the same height limit for one or two providers will not encourage co- location in the Rural zone.	Reject

Amendment 51 – Standard 13.3.2.2.2 Height of masts and antennas (involving two or more providers)				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Spark New Zealand Trading Limited	7.16	Amend the permitted standard 13.3.2.2.2 to provide a height of 25m for two or more providers in the General Rural Zone and to provide for a 15m height limit in the General Recreation and the Rural Residential zones.	Oppose in part. Refer to Reasons under submission point 7.15.	Reject
Chorus New Zealand	8.21	Amend the permitted standard 13.3.2.2.2 to	Oppose in part. Refer to Reasons under	Reject

Limited	provide a height of 25m for two or more providers in the General	submission point 8.20.	
	Rural Zone and to provide for a 15m height		
	limit in the General Recreation and the Rura Residential zones.		

Amendment 51 – Standard 13.3.2.2.4 Height of cabinets and other network utility structures within road reserve and not otherwise provided for				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Wellington Electricity Lines Limited (WELL)	9.21	Retain as notified.	Support.	Accept
Powerco Limited	10.21	Retain as notified.	Support.	Accept

Amendment 51 – Standard 13.3.2.2.5 Height of cabinets and network utility structures not within road reserve and not otherwise provided for				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Wellington Electricity Lines Limited (WELL)	9.22	Retain as notified.	Support.	Accept
Powerco Limited	10.22	Retain as notified.	Support.	Accept

Amendment 52 – Standard 13.3.2.3.2 Size and Diameter of masts (involving two or more providers)				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Spark New Zealand Trading Limited	7.17	Amend permitted standard 13.3.2.3.2 to provide for a diameter of 1.5m for a mast in all the Business zones involving two or more	Oppose in part. The proposed standards are too restrictive and will not encourage co- location.	Accept

		providers.		
Chorus New Zealand Limited	8.22	Amend permitted standard 13.3.2.3.2 to provide for a diameter of 1.5m for a mast in all the Business zones involving two or more providers.	Oppose in part. The proposed standards are too restrictive and will not encourage co- location.	Accept

Amendment	Amendment 52 – Standard 13.3.2.3.3 Horizontal diameter circle of antenna attached to masts					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point		
Spark New Zealand Trading Limited	7.18	Amend permitted standard 13.3.2.3.3 to provide as a permitted standard for antenna attached to a mast to be located within a horizontal circle of 5m in the General Rural zone.	Oppose in part. Refer to Reasons under submission point 7.17.	Accept		
Chorus New Zealand Limited	8.23	Amend permitted standard 13.3.2.3.3 to provide as a permitted standard for antenna attached to a mast to be located within a horizontal circle of 5m in the General Rural zone.	Oppose in part. Refer to Reasons under submission point 8.22.	Accept		

Amendment 52 – Standard 13.3.2.3.4 Horizontal diameter circle of antenna attached to masts (two or more providers)				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Spark New Zealand Trading Limited	7.19	Amend permitted standard 13.3.2.3.4 to provide as a permitted standard for antenna attached to a mast to be located within a horizontal circle of 5m in the General Rural zone.	Oppose in part. Refer to Reasons under submission point 7.17.	Accept
Chorus New Zealand Limited	8.24	Amend permitted standard 13.3.2.3.4 to provide as a permitted standard for antenna	Oppose in part. Refer to Reasons under submission point 8.22.	Accept

attached to a mast to be located within a	
horizontal circle of 5m in the General Rural zone.	

Amendment 52 – Standard 13.3.2.3.5 Antenna attached to other buildings					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Spark New Zealand Trading Limited	7.20	Amend permitted standard 13.3.2.3.5 to provide for an antenna diameter of 1.2m2 in the Suburban Commercial zone.	Oppose in part. Spark considers that the allowable area for antennas attached to buildings in the Suburban Commercial area zone is too restrictive and inconsistent with the provision in other zones elsewhere e.g. (Recreational and Rural zones 1.3m antenna diameter and 1.2m ² in area).	Accept	
Chorus New Zealand Limited	8.25	Amend permitted standard 13.3.2.3.5 to provide for an antenna diameter of 1.2m2 in the Suburban Commercial zone.	Oppose in part. The allowable area for antennas attached to buildings in the Suburban Commercial area zone is too restrictive and inconsistent with the provision in other zones elsewhere e.g. (Recreational and Rural zones 1.3m antenna diameter and 1.2m ² in area).	Accept	

Amendment 52 – Standard 13.3.2.3.6 Cabinets and other network utility structures located within road reserve (not otherwise provided for)				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Wellington Electricity Lines Limited (WELL)	9.23	That Council amend Standard 13.3.2.3.6 to enable road reserve cabinets associated with electrical distribution networks to have an	Support in part. The size and diameter standards for Cabinets located within the road reserve are too restrictive. As shown in Attachment	Accept in part

		area up to 5m ² as permitted activities adjacent to all land use zones; and	C to WELL's submission, electricity distribution cabinets are common elements within the City's road reserves and represent the only above ground structures associated with other wise permitted underground networks. The design and location of the distribution cabinet are thoroughly considered prior to their positioning, thereby ensuring that pedestrian access and safety will not be compromised by the road reserve cabinet.	
Powerco Limited	10.23	Retain as notified.	Support.	Accept

Amendment 52 – Standard 13.3.2.3.7 Cabinets and other network utility structures not otherwise provided for that are not located within road reserve				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Wellington Electricity Lines Limited (WELL)	9.24	Retain as notified.	Support.	Accept
Powerco Limited	10.24	Retain as notified.	Support.	Accept

Amendment 53 – Standard 13.3.2.4 Separation Distance and Setbacks				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
KiwiRail	3.21	Amend as follows: "Separation Distance and Setbacks With the exception of standard 13.4.4.1, which applies to all network utility structures, including lines, the following table applies to masts and antenna	Support in part. The restrictions on meeting setbacks in the road corridor by virtue of being a long narrow corridor, also apply to the rail network.	Accept

		attached to masts and any cabinet or other network utility structure that is defined as a building that are not located in the road reserve <u>or rail</u> <u>corridor</u> "		
Transpower New Zealand	6.41	Exclude transmission lines, including their associated support structures from the separation distance and setback standard.	Oppose. The standard is entirely inappropriate to apply to transmission lines, including their support structure. Transmission lines commonly cross rivers and therefore could encroach within the 20 metre riparian setback therefore is unclear what effect applying this setback standard to lines is intended to manage. The only activity status rules for new and additional above ground lines (13.3.1.26 and 13.3.1.27) do not provide for transmission lines which does not give effect to the proposed objectives and policies, the RPS or the NPSET.	Accept in part
Wellington Electricity Lines Limited (WELL)	9.25	Retain as notified but correct the drafting error relating to the exception of Standard 13.4.4.1.	Support. There is a drafting error as an item referenced as 13.4.4.1 does not exist.	Accept

Amendment 53 – Standard 13.3.2.4.3 Separation distance or setback for cabinets and other network utility structures				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Wellington Electricity Lines Limited (WELL)	9.26	Retain as notified.	Support.	Accept
Powerco Limited	10.25	Retain as notified.	Support.	Accept

Amendment	54 – Standar	d 13.3.5.2 Slope, Height, D	epth and Area of Earthwo	rks
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
NZ Transport Agency	1.11	Add new performance standard under (iv) of the rule to reflect the long, lineal and multi- zoned nature of the road network.	Support in part. The road network is a long lineal network and crosses multiple zones and the rule does not take this into account.	Accept in part
KiwiRail	3.22	Amend as follows: "Slope, Height, Depth and Area of Earthworks (iv) Area: <u>Within any Rail Corridor</u> <u>– 1000m²</u> "	Support in part. Acknowledge that the area restrictions proposed in the earthworks rules provide some certainty, however they are complicated to try and implement in the instance of rail. The rail network, like the road network, like the road network, sa long linear network, the rail network throughout the City has an underlying zone that alters consistently with the adjoining sites. If a 'site' where the works are to occur extends across multiple zones, there is uncertainty which area under 13.3.2.5.2.iv would apply. Even if the 'all other activity areas' area was used, at 500m2 that is not facilitating a large area of earthworks to occur over a long linear network. KiwiRail suggest that the earthworks in a rail corridor be referenced independent of zone recognising the unique restriction the zones and small areas place on long linear networks.	Accept in part
Transpower New Zealand	6.42	Amend as follows: "The following shall apply to all network utility activities, except to earthworks within	Oppose. The proposed standard may unnecessarily catch earthworks associated with the foundations for	Accept in part

		2.0m of the exterior walls of any network utility building or structure that is defined as a building measured in plain view and to piling undertaken for the installation of a network utility mast, where that piling is contained within 2.0m of the edge of the mast, and to earthworks within 2 metres of the outer edge of a network utility support structure without exterior walls: "	transmission support structures. Transpower is concerned that the reference to exterior walls means this exemption does not apply to transmission towers. It is appropriate that earthworks associated with these towers are exempt from the standard in the same way as earthworks for other buildings and structures are, and consistent with directions in the NPSET to provide for the effective operation, maintenance, upgrading and development of the electricity transmission network.	
Spark New Zealand Trading Limited	7.21	Amend as follows: That permitted standard 13.3.2.5.2 be amended and reformatted as follows: <i>"Slope, Height, Depth</i> <i>and Area of Earthworks</i> <i>The following shall apply</i> <i>to all network utilities</i> <i>except to:</i> <i>(a) earthworks within</i> 2.0m of the exterior walls of any network utility building or structure that is defined as a building measured in plain view; <i>(b) trenching in road</i> reserve; and to <i>(c) piling associated</i> with the installation of network utility masts."	Oppose in part. Spark submits that trenching in road reserve to provide for as a permitted activity commonly occurs throughout the country with little environmental effect. Accordingly it is considered that this activity should be excluded from compliance with the permitted activity standards. Further, the restriction on piling to within 2m of the edge of the mast is unnecessary in terms of the nature and the effects of this work.	Accept in part
Powerco	FS2.11	Accept	As set out in its own submission, Powerco considers that trenching in the road reserve should not be subject to the earthworks standards in 13.3.2.5.2 and supports the	Accept in part

			amendments proposed.	
Chorus New Zealand Limited	8.26	Amend as follows: That permitted standard 13.3.2.5.2 be amended and reformatted as follows: " <u>Slope, Height, Depth</u> <u>and Area of Earthworks</u> <u>The following shall apply</u> <u>to all network utilities</u> <u>except to:</u> (a) earthworks within <u>2.0m of the exterior</u> <u>walls of any network</u> <u>utility building or</u> <u>structure that is</u> <u>defined as a building</u> <u>measured in plain</u> <u>view;</u> (b) trenching in road <u>reserve; and to</u> (c) piling associated <u>with the installation</u> <u>of network utility</u> <u>masts.</u> "	Oppose in part. Trenching in road reserve to provide for as a permitted activity commonly occurs throughout the country with little environmental effect. Accordingly it is considered that this activity should be excluded from compliance with the permitted activity standards. Further, the restriction on piling to within 2m of the edge of the mast is unnecessary in terms of the nature and the effects of this work.	Accept in part
WELL	9.27	Amend as follows: "The following shall apply to all network utility activities, except to earthworks <u>associated</u> <u>with cable trenching</u> , within 2.0m of the exterior walls of any network utility building or structure that is defined as a building measured in plain view and to piling undertaken for the installation of a network utility mast <u>or</u> <u>support</u> <u>structure</u> , where that piling is contained within 2.0m of the edge of the mast:"	Support in part. Support the earthworks standard in so far as they do not apply to 2 meters of the external walls of any network utility building or structure. However, clarification be more explicit that the earthworks standards do not apply to Pole Structures or to cable trenching. It is unclear that the slope, height, depth, and area of earthworks do not related to support structures that are not defined as a building. As electricity support structure foundations will often be required to go deeper than 1.5m for safety purposes, a clear exclusion from this standard is necessary. As Council has determined that in all	Accept in part

			activity areas (apart from Rural) new lines are to be located underground, the associated earthwork provision should not be applicable to cable trenching. Quite often, when competing for service space within the transportation corridor, maintaining minimum clearance s from other services will require trenching depths greater than 1.5m.	
Transpower New Zealand	FS1.13	Accept	Transpower supports the submitters request and agrees with them that basic maintenance activities should not be subject to native vegetation rules.	Accept

Amendment	Amendment 55 – Standard 13.3.2.6 Native Vegetation Cover					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point		
Transpower New Zealand	6.43	Amend standard 13.3.2.6 by including an exemption for tree removal and trimming undertaken in accordance with the Electricity (Hazard and Trees) Regulations 2003.	It is unclear what is meant by the phrase 'destruction of any native vegetation'. This standard appears to catch the management of vegetation required to meet the Electricity (Hazard and Trees) Regulations 2003. If the standard does catch these activities it would mean that basic maintenance works would require resource consent. Requiring resource consent for basic maintenance works does not give effect to the proposed objectives and policies, or to the RPS and NPSET.	Accept		
Wellington Electricity	9.28	Amend as follows: "Within the Rural	Support in part. Agree with Council that	Accept in part		

I				
	Lines		Residential and General	destruction of native
	Limited		Rural Activity Areas, <u>and</u>	vegetation should be
	(WELL)		apart from damage	avoided; however WELL
			caused by a Network	are required to maintain
			Utility Operators	growth limits on
			maintenance activities,	vegetation within set
			there shall be no	distances from electricity
			destruction of any native	conductors under the
			vegetation where:"	Schedule: Growth Limit
				Zones of the Electricity
				(Hazards from Trees)
				Regulations 2009. In
				order to meet this
				obligation under the
				regulation, damage to
				native vegetation
				provisions should be
				exempt in PC34 for the
				electricity network utility
				operator.
		1	1	1

Amendment 57 – Standard 13.3.2.8 Temporary Above Ground Lines					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Wellington Electricity Lines Limited (WELL)	9.29	Retain as notified.	Support.	Accept	

Amendment 59 – 13.3.4 Matters in which Council has restricted its discretion					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Transpower New Zealand	6.44	Replace the matters of discretion with the following: " <u>(a Any positive effects</u> <u>to be derived from</u> <u>the activity.</u> (b) The degree, extent <u>and effects of the</u> <u>non-compliance with</u> <u>the Permitted</u> <u>Activity Conditions.</u> (c) Health and safety. (d) Suitability of the site <u>for the proposed</u> <u>activity, including</u>	Oppose. The proposed matters of discretion fail to allow recognition of the benefits of regional significant network utilities, do not adequately give effect to Policies 3 and 4 of the NPSET and generally are not all appropriate. The matters should be redrafted to be simpler, consistent with those in Amendment 72 relating	Accept in part	

Powerco	FS2.9	consideration of geotechnical and natural hazard constraints.(e) Layout, design and location of proposed structures.(f) Traffic and transport effects.(g) Extent of any earthworks.(h) Effects on historic heritage.(i) Visual, character and amenity effects.(j) Noise and lighting effects.(j) Effects on public access.(j) Effects on natural character and native vegetation.(m) Adequacy of the methods of mitigation/remediati on or ongoing management, including the extent to which any adverse effects have been avoided, remedied or mitigated by the route, site and method selection.(n) Any constraints arising from technical and operational requirements of the network which may limit measures to avoid, remedy or mitigate environment effects"Make consequential amendments to the references to the matters of discretion in section 13.3.1.Accept in part	to renewable energy generation. In addition: Clause (n) relating to alternatives should be removed. The Act only requires applicants to describe possible alternatives if it is likely that the activity will result in significant adverse effects on the environment. The proposed clause is unnecessary (as it duplicates the Act) and creates an undue expectation that alternatives will be assessed in every instance Clause (r) relating to consultation is not a relevant 'matter of discretion'. It is unclear how the extent of consultation with affected persons or the community should have a bearing on a resource consent application. The Act does not require consultation.	Accept in part
POWELCO	F32.9	Accept in part	discretion are considered to be appropriate.	ассерт іп рагт

Amendment	Amendments 61, 62, 63 and 64 – Rules relating to activities in the National Grid Yard					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point		
Transpower New Zealand	6.45	 Amend 13.4 as follows: a) Include the new permitted activity rule: "In all activity areas, earthworks within the National Grid Yard which: a) are for a Network Utility or undertaken as part of agricultural or domestic cultivation, or repair, sealing or resealing of a road, footpath, driveway or farm track; or b) do not exceed 300mm in depth within 12m of any National Grid support structure foundation; c) do not compromise the stability of a National Gird support structure; and d) do not result in a reduction in the ground to conductor clearance distances below what is required by Table 4 of NZECP34." b) Delete proposed restricted discretionary rule c) Insert as a new restricted discretionary rule (Any earthworks) 	Oppose. The provision made for buildings within the National Grid Yard as a restricted discretionary activity is inappropriate and does not give full effect to the NPSET. Other than very small buildings, buildings in the National Grid Yard should be non-complying activities. The rules do not control earthworks within the national grid yard. Earthworks can have significant adverse effects on the national grid, in particular on the stability of support structures. Earthworks within the national grid should therefore be specifically managed in the District Plan.	Reject		

within the National Grid Yard that exceed 300mm in depth and are within 12m of any National Grid support structure foundation."
d) Insert new matters of discretion for the new restricted discretionary rule: "a) Impacts on the operation, maintenance, upgrade and development of the National Grid. b) Compliance with NZECP34:2001. c) Technical advice provided by the National Grid owner (Transpower). d) The risk to the structural integrity of the National Grid. e) Any impact on the ability of the National Grid owner (Transpower) to access the National Grid. f) The risk of electrical hazards affecting public
affecting public or individual safety, and the risk of property damage."
e) Replace non- complying rule 13.4.3 with the following: " <u>In all activity areas,</u> within the National <u>Grid Yard:</u> a) The <u>establishment of</u> <u>sensitive land use</u> activity, including the change of the

<u>use of an existing</u>
<u>building or</u>
<u>structure.</u>
b) The construction
<u>of a new, or</u>
addition to an
existing, building
<u>or structure not</u>
<u>meeting</u>
permitted activity
<u>rule 13.4.1.</u>
<u>c) Earthworks not</u>
<u>meeting</u>
<u>permitted activity</u>
<u>rule 13.4.XX</u> ".

Amendment 75 – Other Provisions					
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point	
Transpower New Zealand	6.46	Amend clause (c) of all 'Other Provisions' as follows: "(c) Network Utilities, including the National Grid and all activities (such as earthworks, buildings and structures, residential activities, education facilities, health care facilities) within the National Grid Yard – See Chapter 13."	Oppose. The proposed change does not adequately direct parties to the National Grid provisions in proposed section 13.4 of the Plan.	Reject	

Amendment 76 – Planning Maps				
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Transpower New Zealand	6.47	Amend the Plan maps as included in amendment 76 of the Plan Change. Ensure the key included in amendment 76 is included on each relevant Plan map	Support in part. It is unclear from the plan change documentation whether it is proposed to include the key shown in the text of amendment 76 on each Plan map. The key labels the lines	Accept

	as 'National Grid Centre Line' and 'National Grid Corridor maximum extent (39 m from the	
	centre line)'. These labels should be included in the legend included on each Plan map.	

ADDITIONAL	AL PROVISIONS – NEW POLICY TO CHAPTER 13			
Submitter	Sub. Ref.	Decision/Relief Sought	Reasons	Commissioners' Recommendation on Submission Point
Greater Wellington Regional Council	2.1	Add new policy as follows: <u>"Seek to locate network</u> <u>utilities outside of flood</u> <u>hazard areas, and most</u> <u>importantly outside of</u> <u>stopbanks. Where this is</u> <u>not practicable, ensure</u> <u>that they are designed</u> <u>and installed in a</u> <u>manner to withstand a</u> <u>design flood event".</u>	Concerned about the interaction of activities undertaken by network utility providers with the infrastructure which protects communities from the adverse effects of flooding. Stopbanks are an integral aspect of this risk management to protect the development which has occurred in our communities. A particular issue is with any activity which may compromise the integrity of stopbanks. A common activity is the installation of pipes and other facilities through or on stopbanks which are carried out by various network utility providers. Excavation and backfill in stopbanks, even if completed to a high standard, can increase the chance of stopbanks failing during major floods. Because of the swift rising nature of the rivers and streams in the Wellington region, any excavation in the stopbanks, regardless of reason should only be undertaken exercising due caution.	Reject
Transpower	F31.14	nejett	TIATISPOWEI	ALLEPI

New Zealand			acknowledges the principle of the Policy; however it considers that the policy fails to recognise the need for linear infrastructure to commonly cross water bodies, including the flood hazard area. Transpower considers that linear infrastructure should be specifically excluded from the Policy to avoid confusion about its intent.	
Powerco	FS2.1	Accept in part	Powerco accepts the need to protect the integrity of flood protection assets and can support inclusion of the policy proposed by GWRC. The suggested wording is considered appropriate as it focuses on locating network utilities outside of flood hazard areas but recognises that this may not always be practicable in which case network utilities should be designed and installed to be resilient to the effects of flooding.	Reject



UPPER HUTT CITY DISTRICT PLAN

PROPOSED PLAN CHANGE 38 – NETWORK UTILITIES AND RENEWABLE ENERGY GENERATION

RECOMMENDATION ON DECISIONS REQUESTED

CONTENTS

CONTENTS
LIST OF SUBMITTERS
SUMMARY OF DECISIONS REQUESTED BY PROPOSED AMENDMENT NUMBER
General5
AMENDMENT 1: Amendments to Chapter 16 – Network Utilities Issues, Objectives and Policies
AMENDMENT 2: Amendments to Chapter 16A – Renewable Energy (Issues, Objectives and Policies)
AMENDMENT 3: Amendments to Chapter 18 – Residential Zone (Rules)
AMENDMENT 4: Amendments to Chapter 19 – Rural Zone (Rules) 32
AMENDMENT 5: Amendments to Chapter 20 – Business Zone (Rules)
AMENDMENT 6: Amendments to Chapter 21 – Open Space Zone Rules
AMENDMENT 7: Amendments to Chapter 22 – Special Activity Zone Rules
AMENDMENT 10: New Chapter 30A – Rules for Renewable Energy Generation 51
AMENDMENT 11: Amendments to Chapter 35 – Definitions (Antenna) 52
AMENDMENT 11: Amendments to Chapter 35 – Definitions (Building) 52
AMENDMENT 11: Amendments to Chapter 35 – Definitions (Cabinet)
AMENDMENT 11: Amendments to Chapter 35 – Definitions (Line) 53
AMENDMENT 11: Amendments to Chapter 35 – Definitions (Maintenance)
AMENDMENT 11: Amendments to Chapter 35 – Definitions (Minor above ground line) 54
AMENDMENT 11: Amendments to Chapter 35 – Definitions (Minor upgrading) 54
AMENDMENT 11: Amendments to Chapter 35 – Definitions (Network utility)
SUBMISSIONS RECEIVED THAT ARE NOT ON PLAN CHANGE 38, BY PROVISION
AMENDMENT 1: Amendments to Chapter 16 – Network Utilities Issues, Objectives and Policies
LIST OF SUBMITTERS

Submitter number	Submitter name	Address for service
1	Greater Wellington Regional Council	Greater Wellington Regional Council PO Box 11646 Manners Street Wellington 6142 Attention: Caroline Ammundsen
2	KiwiRail Holdings Limited	KiwiRail Holdings Limited Wellington Railway Station PO Box 593 Wellington 6140 Attention: Rebecca Beals
3	Spark New Zealand Trading Limited	Spark New Zealand Trading Limited Private Bag 92028 Auckland 1010 Attention: Mary Barton
4	Transpower New Zealand Limited	Transpower New Zealand Limited PO Box 1021 Wellington Attention: Sarah Shand
5	Chorus New Zealand Limited	Chorus New Zealand Limited PO Box 632 Wellington Attention: Mary Barton
6	Wellington Electricity Lines Limited	Wellington Electricity Lines Limited c/- Edison Consulting Group PO Box 4269 Mount Maunganui South 3149 Attention: Tim Lester
7	Powerco Limited	Burton Planning Consultants Limited Level 1, 2-8 Northcroft Street PO Box 33-817 Takapuna Auckland 0740 Attention: Georgina McPherson
8	Royal Forest and Bird Protection Society of New Zealand Incorporated – Upper Hutt Branch	Royal Forest and Bird Society of New Zealand Incorporated – Upper Hutt Branch 14 Cruickshank Road Upper Hutt 5018 Attention: Barry Wards
9	Roz Brown	Roz Brown PO Box 40718 Upper Hutt 5140

COMMISSIONERS' RECOMMENDATIONS ON DECISIONS REQUESTED BY PROPOSED AMENDMENT NUMBER

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
General					
83.1	Spart	General	Either (i) amend the proposed plan change as per the relief set out in the specific submissions by Spark Or (ii) such other relief to like effect to remedy the concerns outlined in the submissions And in relation to both (i) and (ii) above, (iii) any consequential amendments necessary as a result of the amendments to grant the relief sought above.	Spark's network is subject to constant maintenance, modification and upgrading as the number of customers and services increases, and changes in technology occur rapidly. Technological advances during the life to the District Plan, along with more instances of telecommunications providens seeking to co-locate on each other's assets, are also likely to occur requiring existing assets to be upgraded. Consider that any new District Plan provisions should be created with such provisions should be created with such provision for co-location of equipment being kept in mind.	Accept in part
2. 2.	Transpower	General	 That Proposed Plan Change 38 is approved, subject to appropriate amendments that ensure: Full effect is given to the National Policy Statement on Electricity Transmission 2008; Effect is given to the policies of the RPS; Recognition of the NESETA and ensure that there are no conflicts with provisions of the District Plan and the NESETA (s44A of the RMA); Provide an appropriate policy framework for the policytramework for the protection of National Grid infrastructure, including reverse sensitivity effects; and Provide for the ongoing operation, maintenance, upgrading and development of existing National Grid infrastructure. 	Support the general intent of the Plan Change subject to appropriate amendments.	Accept in part
S5.1	Chorus	General	Either (i) amend the proposed plan change as per the relief set out in the specific submissions by Chorus Or (ii) such other relief to like effect to remedy the concerns outlined in the submissions And in relation to both (i) and (ii) above, (i) (iii) any consequential amendments necessary as a result of the amendments to grant the relief sought above.	Chorus' general relief sought in this submission point (5.1) applies to submission points 5.2 to 5.21.	Accept in part

Commissioners' Recommendation on submission point	Accept in part		Accept	Accept in part
Reasons	Support the general intent and the move by Upper Hutt City, Porirua City and Hutt City Councils to provide a consistent approach to the management of network utilities across the three plan changes. Reliable and constant energy supply is critical to sustaining the regional economy, population and way of life, and demand for energy is constantly increasing number of constraints, in terms of providing a secure and reliable supply of gas to meet the increasing demand and population growth. Some amendments are sought to further clarify the specific regulatory approach for gas distribution infrastructure.	ies	Support. In particular the paragraph which recognises that utilities are vital for a community and the ability to construct, maintain, upgrade and operate these effectively is very important.	Support in part, subject to a minor amendment to the wording of a new sentence added to paragraph seven of the background section that was not included in the draft version of the Plan Change. The inclusion of 'new buildings and structures' in the sentence is in line with the rules in each of the zone chapters that control new buildings and structures within the transmission line corridor. Support in part. 'Environmental wellbeing' includes trying to live in harmony with the environment by recognising and understanding the impact(s) of our interactions with nature and taking action to protect that environment. And protecting ourselves from environmental hazards and minimising the negative impact(s) of our behaviours on the environment. Environmental wellness involves a number of different aspects of personal and societal responsibilities, including being aware of the earth's natural resources and their respective limits. Identify that the Local Government Act 2002 includes a focus on sustainability with the reference to the 'four well-beings' (social, economic, environmental and cultural) and that the purpose of the Act is cultural) and that the purpose of the Act is cultural) and that the purpose of the Act is
Decision Sought	Amend the relevant provisions of the District Plan to: • address the relevant provisions in sections 5-8 of the RMA, including with respect to the sustainable management of Powerco's assets as a physical resource; implement the statutory tests in section 32 and the requirements in the First Schedule of the RMA; • Recognise and provide for the ongoing operation, maintenance, upgrade and development of Powerco's gas and electricity networks.	Network Utilities Issues, Objectives and Polic	Retain as notified.	Network utilities can be vulnerable to reverse sensitivity effects when new buildings or the effects of the existing network utility) are established nearby. Jeading to constraints on the operation of the network utility" Include 'environmental wellbeing' as an intrinsic part of the provisions of people and communities affected and impacted by network utilities and infrastructure.
Provision	General	ts to Chapter 16 -	16.1 Background	16.1 Background Background
Submitter	Powerco	1: Amendmen	KiwiRail	Forest and Bird
Submission No.	S7.1	AMENDMENT	S2.1	88

Commissioners' Recommendation on submission point		Accept in part	Accept	Accept	Accept	Accept	Accept in part
Reasons	and action by, and on behalf of, communities; and (b) to promote the social, economic, environmental, and cultural well- being of communities, in the present and in the future.	Transpower considers that the adverse effects of network utilities are already appropriately addressed in the Plan Change.	Support.	Oppose in part. Support that the issue recognises the essential nature of network utilities, and the need to balance the benefits with the adverse effects. Also support the acknowledgement in the Explanation that it may not be possible to avoid, remedy or mitigate all adverse effects due to their constraints. Seek an amendment to the second to last sentence in the third paragraph of 6.2.1 to recognise that there will not necessarily always be an adverse effect from network utilities, for example from undergrounding.	Transpower agrees with the general support to the issue provided by the submitter, and also agrees that network utilities do not always have adverse environmental effects that cannot be either avoided, remedied or mitigated.	WELL support the submission point as it correctly notes that network utility benefits should be given consideration as part of the effects assessment.	Support in part. In particular, support the acknowledgement in paragraph three of the explanation that network utilities may have 'preceived' adverse effects on public health and safety. Also support the later part of paragraph three where it is acknowledged that in some cases, it may not be possible to avoid, remedy or mitigate all adverse effects associated with a utility and that in these circumstances, consideration will be given to both the benefits of the utility and the significance of the adverse effects on the surrounding environment. Seek a minor wording change so that this paragraph refers to 'network utility' as is defined in the Plan Change, not' utility'.
Decision Sought		Reject	Retain as notified.	In some cases, it might not be entirely possible to avoid, remedy or mitigate all adverse effects associated with a utility, meaning that there with may be some level of residual adverse effect on the environment.	Accept	Accept	The high voltage electricity transmission lines that form part of the <u>Maational Genid</u> plav a vital role in the well-being of the community the adverse effects of the high voltage electricity transmission lines are often local, while the benefits may be in a different locality and/or extent beyond the local to the regional and national Adverse effects may only occur at the time of construction or installation of the network utility, but in some instances may confinue throughout its operation or during maintenance and upgrade works. In some case, it might not mitigate all adverse effects and will be some elevel of adverse effect on the surrounding environment. In such circumstances, there is a need to carding conting the energits the endition of the benefits the
Provision			Issue 16.2.1	Issue 16.2.1			(Explanation)
Submitter		Transpower	KiwiRail	Spark	Transpower	WELL	Transpower
Submission No.		FS2.1	S2.2	S3.2	FS2.2	FS3.8	6. 400

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			of the adverse effects on the surrounding environment.		
			And any consequential amendments replacing 'national grid' with 'National Grid' and 'utility' with 'network utility'.		
FS3.12	WELL		Accept	WELL agree that not all effects of network utility development and operation can be avoided remedied or mitigated. This submission point reflects WELL's submission on the same matter.	Accept in part
S5:2	Chorus	Issue 16.2.1	In some cases, it might not be entirely possible to avoid, remedy or mitigate all adverse effects associated with a utility, meaning that there will may be some level of residual adverse effect on the environment.	Oppose in part. Support in that it recognises the essential nature of network utilities, and the need to balance the benefits with the adverse effects. Support the adverse effects. Explanation that it may not be possible to avoid, remedy or mitigate all adverse effects due to their constraints. Seek an amendment to the 2nd to last sentence in the 3rd paragraph of 16.2.1 to recognise that there will not necessarily always be an adverse effect from network utilities, for example from undergrounding.	Accept
FS3.29	WELL		Accept	WELL support the requested amendment as it reflects WELL's submission in that residual environmental effects may result from network utility development and/or operation.	Accept
S7.2	Powerco	lssue 16.2.1	Retain Issue 16.2.1 and the associated explanation without modification.	Support.	Accept
\$2.3	KiwiRail	Issue 16.2.2	Retain as notified.	Support. Not opposed to development on sites adjoining the rail network, however as trains can often be intermittent and cause effects an amenity, support these potential effects are mitigated. Where sensitive activities are proposed on land near the railway corridors, appropriate controls should therefore be imposed to ensure the long term amenity of those occupring those sites. Along with the need for mitigation is the need to manage the risk of objections and complaints leading to restraints on the operation, maintenance and enhancement of the rail corridor.	Accept
FS3.2	WELL		Accept	WELL support this submission point in principle only to the extent that it similarly reflects WELL's desire to protect critical electricity lines from reverse sensitivity.	Accept
S3.3	Spark	Issue 16.2.2	Retain Issue 16.2.2.	Support.	Accept
FS2.3	Transpower		Accept	Except for the amendment sought in its original submission relating to earthworks,	Accept

commissioners' Recommendation n submission point		ccept	ccept	ccept	ccept	ccept	ccept in part ccept ccept	eject
Reasons	adverse effects on natural and physical resources and on amenity values. See no reason to delete these specific reference highlights because they remind people of what could be affected.	Transpower is opposed to this submission A as this is no longer intended to be the scope of resource management issue 16.2.2. The issue as proposed addresses adverse effects on network utilities, rather than those arising from network utilities. Transpower supports the focus of this particular resource management issue and explanation. The matter of adverse effects arising from network utilities is addressed alsewhere in the plan change.	Issue 16.2.2 relates to effects <i>on</i> network A utilities / reverses sensitivities. The unsuision point relates to effects of network utilities. The effects of network utilities is adequately addressed throughout 16.3.4) and does not require the retention of the preceding issue.	Support. A	Support. A	Support. A	Protect regionally significant infrastructure as required under the RPS (e.g. Objective 10 and Policy 8 of the RPS). Seek inclusion of protect in the objective. The qualifier 'unreasonably' is not consistent with Policy 10 of the NPSET as the Policy does not use that word. Seek that the word 'unreasonably' be deleted from the objective. Seek an amendment to replace 'incompatible subdivision, use and development' with 'inappropriate subdivision, use and development' in order to achieve consistency with the explanation and other provisions of the Plan Change including the explanation to Issue 16.2.2 to which this objective relates. Support. A Support.	support in part, on the basis that the R reference to 'avoid, remedy or mitigate adverse effects on the environment'
Decision Sought		Reject	Reject	Retain Issue 16.2.5 without amendment.	Retain as notified.	Retain Objective 16.3.1.	Delective 16.3.1: To Decorpose the benefits of and protect regionally significant utilities and ensure their functions and operations are not unreasonably compromised by other activities. Explanation: The objective and supporting policies are focused on recognising the benefits that these regionally significant network utilities have locally. regionally and nationally and ensuring that these network utilities benefits are protected from incommentable inappropriate subdivision. use and development. And make any other consequential amendments to the Plan Change to ensure that the benefits of regionally significant network utilities are both recognised and protected. Retain Objective 16.3.1 and the associated explanation without modification.	Objective 16.3.2: The sustainable, secure and efficient use and development of the high voltage
Provision				Issue 16.2.5	Objective 16.3.1	Objective 16.3.1	Objective 16.3.1 (Explanation) (Explanation) Objective 16.3.1 Objective 16.3.1	Objective 16.3.2 Objective
Submitter		Transpower	WELL	Transpower	KiwiRail	Spark	Chorus Powerco Transpower	Indispower
Submission No.		FS2.4	FS3.44	S4.5	S2.2	S3.4	S5.4 S5.4 S37.4 S4.7	04.7

Provision Decision Sc 16.3.2 (110kV or g
(Explanation) (Ines which adverse effe extent prac technical and constraints o
Explanation: The efficient of the national (social, econol people. Techr requirements electricity trau extent to whiti avoid or mitig effects.
Accept in part
Objective Clarify the ap by removing t grid' in the ext
Objective 16.3.3 Retain as notifi

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				services for the region, and the consequences of an unmaintained network on public safety being significant.	
S3.5	Spark	Objective 16.3.3	Retain Objective 16.3.3.	Support.	Accept in part
FS2.5	Transpower		Accept	Except for the amendment sought in its original submission relating to maintenance and upgrading, Transpower agrees with Spark's submission and generally considers that the Objective is appropriate.	Accept in part
α.	Transpower	Objective 16.3.3	To recognise and provide for the sustainable, secure and efficient use, operation, maintenance, upgrading and development of network utilities within the City.	Support in part. Given that the definition of 'Network Utilities' specifically includes 'Regionally Significant Utilities' this objective covers the National Grid. Accordingly, seek the inclusion of the terms 'maintenance' and 'upgrading' to the objective to ensure consistency with Policy 2 of the NPSET. Policy 2 includes 'operation, maintenance, upgrading and development' and as such it can be implied that the term development does not capture or incorporate upgrading and maintenance activities.	Accept
S5.5	Chorus	Objective 16.3.3	Retain Objective 16.3.3	Support.	Accept in part
S7.5	Powerco	Objective 16.3.3	Retain Objective 16.3.3 and the associated explanation without modification.	Support.	Accept in part
\$2.4	KiwiRail	Objective 16.3.4	Retain as notified.	Support. While new rail networks are not common, maintenance and upgrading of existing networks to ensure safety and technology adjustments occur, are pivotal to the continued operation of the network. Support being able to undertake works with little impediment or restriction given the nature of the rail network, the purpose it services for the region, and the consequences of an unmaintained network on public safety being significant.	Accept in part
FS3.3	WELL	Objective 16.3.4	Council reject the submission in part and do not retain Objective 16.3.4 and Policy 16.4.8 unaltered – rather, the amendments sought by WELL are accepted by Council.	Oppose retention of Objective 16.3.4 as indicated in WELL's submission, there needs to be more explicit acknowledgement that not all adverse environmental effects of network utility development or operation can be avoided remedied or mitigated. WELL's opposition to Policy 16.4.8 being retained is based on the same rationale as the above.	Accept in part
o. S	Spark	Objective 16.3.4	Objective 16.3.4: To manage any adverse effects on the environment resulting from the design. location, upgrading and maintenance of network utilities to the extent practicable. Explanation to Objective 16.3.4: However, in some cases, it might not be	Oppose in part. Seek amendments to both Objective 16.3.4 and its associated Policy 16.4.12 to consistently recognise that adverse effects will be managed as far as practicable, and that not all adverse effects may be able to be avoided, remedied or mitigated. Seek an amendment to 16.6 'Anticipated environmental results' to reflect that all	Accept in part

ommissioners' Recommendation n submission point		ccept in part	ccept in part	ccept in part	ccept in part
Reasons C	adverse effects may not be able to be avoided, remedied or mitigated.	Powerco supports the requested A amendments, which recognise that it will not always be possible to avoid, remedy or mitigate all adverse effects due to the technical and operational constraints associated with network utilities.	Support for the submission points have been given because they reflect similar outcomes being sought by WELL in their submission. WELL consider that the new network utility VELL consider that the new network utility acknowledgement to the fact that not all adverse environmental effects can be avoided remedied or mitigated. WELL consider that the use of the words 'to the extent practicable' suitably acknowledges the that the diffects of network utility developments	Support in part. While the last part of the A explanation correctly recognises that all adverse effects of network utilines cannot always be mitigated, it then states that there will always be some level of adverse effect on the surrounding environment that requires mitigation. This statement is contradictory and as such, seek the requirement to mitigate 'some level of adverse effect be deleted.	WELL support this submission point as it A similarly reflects WELL's submissions on the same mater. WELL support the amended wording in the submission point as it recognises positive effects (i.e., community benefits) need to
Decision Sought	entirely possible to avoid, remedy or mitigate all adverse effects associated with a network utility due to their technical and operational constraints, meaning that there will may be some level of residual adverse effect on the surrounding environment—that—requires mitigation.	Accept	Accept	Amend the explanation to Objective 16.3.4 as follows: However in some cases, it might not be entirely possible to avoid, remedy or mitigate all adverse effects associated with a network utility, meaning there will be some level of adverse effect on the surrounding environment that requires mitigatien. Or amend the paragraph in line with the explanation of Issue 16.2.1 as follows: However in some cases, it might not be entirely possible to avoid, remedy or mitigate all adverse effect on the some level of adverse effect on the surrounding environment, associated with a network utility, meaning there will be some level of adverse effect on the some level of adverse offect on the surrounding environment.	Accept
Provision				Objective 16.3.4 (Explanation)	
Submitter		Powerco	WELL	Transpower	WELL
Submission No.		FS1.5	ດ ຕ ທ ມ	0.	FS3.14

Commissioners' Recommendation on submission point		Accept in part		Accept	Accept in part	Accept in part	Accept in part	Accept in part
Reasons	of environmental effects.	Oppose in part. Seek amendments to consistently recognise that adverse effects will be managed as far as practicable, and that not all adverse effects may be able to be avoided, remedied or mitigated.		Supported for the reasons stated in the submission. WELL also submitted that Objective 16.3.4 needed to better acknowledge that some residual environmental effects of network utility development and/or operation would not be able to be avoided remedied or mitigated. The recommended amendment to the Objective sexplanation is supported as it mirrors WELL's submission on the same matter. WELL's submission on the same matter. WELL's submission on the same matter. WELL's submission on the same matter in Proposed Policy 16.4.12 and the corresponding 16.6 Anticipated environmental result	Support in part. Current wording is confusing as it reads that any effects that cannot be avoided remedied or mitigated will be required to be mitigated - which is somewhat contradictory. Consider that the explanation should acknowledge that some residual effects may need to be accepted.	Transpower supports this submission in so far as it makes an important connection between the need to accept some degree of adverse environmental effects from network utilities and the benefits that arise from these physical resources. Transpower considers that what the submitter is seeking is relevant to providing for network utilities.	Support.	Support in part. Adverse effects on the environment resulting from network utilities also include those caused during
Decision Sought		Objective 16.3.4: <u>To manage any adverse effects on the</u> <u>environment resulting from the design</u> , location, <u>operation</u> , <u>upgrading</u> and <u>maintenance of network utilities to the extent</u> <u>practicable</u> .	Explanation to Objective 16.3.4: However, in some cases, it might not be entirely possible to avoid, remedy or mitigate all adverse effects associated with a network utility due to their technical and operational constraints, meaning that there will may be some level of residual adverse effect on the surrounding environment—that requires mitigation.	Accept	Amend the explanation to proposed Objective 16.3.4 to include that the development and operation of network utilities may result in residual effects that cannot be avoided, remedied or mitigated, and that such residual effects will be suitably offset by the public good that they provide.	Accept	Retain Objective 16.3.4 and the associated explanation without modification.	To effectively manage any adverse effects on the environment resulting from the design.
Provision		Objective 16.3.4			Objective 16.3.4		Objective 16.3.4	Objective 16.3.4
Submitter		Chorus		WELL	WELL	Transpower	Powerco	Forest and Bird
Submission No.		S5.6		Э. Э. Э. Э. Э. Э. Э. Э. Э. Э. Э. Э. Э. Э	S6.13	F82.6	S7.6	S8.3

Commissioners' Recommendation on submission point		Reject	Accept	Reject	Reject	Reject
Reasons	construction (as identified in the explanation), not just design, location, operation, upgrading and maintenance. Consider that the statement would be strengthened by including the word 'effectively' before manage.	Transpower opposes the inclusion of the term "effectively as it considers that this is an unnecessary addition to the Objective. It also raises questions about what is meant by the direction to 'manage' adverse effects in other provisions of the Plan, where this is not prefaced with the adverb 'effectively'. Transpower accepts that the effects of construction should be managed, and that this would be consistent with the drafting of Policy 16.4.12.	WELL consider it appropriate to refer to construction in Objective 16.3.4.	Oppose in part. Concerned about the interaction of activities undertaken by network utility providers with the infrastructure which providers of flooding. Stopbanks are an integral aspect of this risk management to protect the development which has occurred in our communities. A particular issue is with any activity which may compromise the integrity of stopbanks. A common activity is the installation of piese and other facilities through or on stopbanks which are carried out by various network utility providers. Excavation and backfill in stopbanks, even if completed to a high standard, can increase the chance of stopbanks, even if completed to a high standard, can increase the chance of stopbanks even if completed to a high standard, can increase the chance of stopbanks even if completed to a high standard, can increase the chance of stopbanks even if completed to a high standard, can increase the chance of stopbanks even if completed to a high standard, can increase the chance of stopbanks even if completed to a high standard, can increase the chance of stopbanks even if completed to a high standard, can increase the chance of stopbanks even if completed to a high standard, can increase the chance of stopbanks even if completed to a high standard, can increase the chance of stopbanks even in the stopbanks, regardless of reason should only be undertaken exercising due caution.	Powerco accepts the need to protect the integrity of flood protection assets and can support inclusion of the policy proposed by GWRC. The suggested wording is considered appropriate as it focuses on locating network utilities outside of flood hazard areas but recognises that this may not always be practicable in which case network utilities should be designed and installed to be resilient to the effects of flooding.	Support in part. Generally support breadth and scope of the policies in terms of
Decision Sought	location, operation, upgrading, construction and maintenance of network utilities.	Reject	Reject	Add a new policy as follows: Seek to locate network utilities outside of flood hazard areas, and most importantly outside of stopbanks. Where this is not practicable, ensure that they are designed and installed in a manner to withstand a design flood event,	Accept in part	Include a policy such as:
Provision				16.4 Policies		16.4 (all Policies)
Submitter		Transpower	WELL	GWRC	Powerco	Forest and Bird
Submission No.		FS2.7	FS3.45	2.7	FS1.1	S8.4

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			network utilities on ecological corridors and manage these as appropriate and practicable.	inclusion of a policy that focuses on recognising and providing for the impact(s) of network utilities on ecological corridors.	
FS1.8	Powerco		Accept	Powerco can accept the proposed policy. It is appropriate to manage the effects of network utilities on ecological corridors.	Reject
FS2.8	Transpower		Reject	Transpower considers that the policy is unnecessary and that issue of managing all adverse environmental effects is already appropriately addressed via proposed Policy 16.4.12.	Accept
FS3.46	WELL		Reject	Ecological corridors are not defined in the district plan. Ecological impacts of a development (whether on corridors or not) are already required to be addressed as a result of PC38.	Accept
S1.2	GWRC	Policy 16.4.1	Retain the following policy and its explanation:	Support.	Accept
			To ensure that the establishment, operation, maintenance and upgrading of essential utilities in the City avoids, remedies or mitigates any adverse environmental effects.		
FS2.9	Transpower		Reject	Transpower considers that the policy is unnecessary and that issues of managing all adverse environmental effects is already appropriately addressed via proposed Policy 16.4.12.	Accept
S2.2	KiwiRail	Policy 16.4.1	Retain as notified.	Support.	Accept
S3.7	Spark	Policy 16.4.1 (Explanation)	Due to the scale of the planning maps and the extensive nature of some regionally significant network utilities. It is however, not feasible to otherfity all regionally significant network utilities on Council planning maps, particularly the local gas distribution lines and the lecommunication networks.	Oppose in part. It is not appropriate to map telecommunications networks, due to the sheer volume of detail that would be required and because with technological advancements over the fife of the Plan, constant changes would be required to keep it up to date with little benefit.	Reject
S4.10	Transpower	Policy 16.4.1 Policy 16.4.1 (Explanation)	Retain Policy 16.4.1 as proposed. Amend the explanation to the policy as follows: in the case of the National Grid, which is not designated. this network will be is specifically recognised and mapped, as required by the National Policy Statement on Electricity Transmission.	Support in part, as the policy provides the policy framework for the inclusion of the National Gind network (or 'transmission line corridor' as per the District Plan) on the District Plan maps in accordance with the NPSET. Support the use of the term 'as practicable' in the policy, as, and in accordance with Transpower's comments on the draft Plan Change, this recognises that it is not practical or feasible to map local gas and electricity distribution networks and Transpower's underground fibre optic cables. Seek a minor wording amendment as proposed below as the transmission line corridor is already shown on the District Plan maps.	Accept

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
S5.7	Chorus	Policy 16.4.1 (Explanation)	Due to the scale of the planning maps and the extensive nature of some regionally significant network utilities, it is however not frassible to identify all regionally significant network utilities on Council planning maps, particularly the local gas distribution lines and telecommunication networks.	Oppose in part. It is not appropriate to map telecommunications networks, due to the sheer volume of detail that would be required and because with technological advancements over the life of the Plan, constant changes would be required to keep it up to date with little benefit.	Reject
<i>S7.7</i>	Powerco	Policy 16.4.1	Retain Policy 16.4.1 and the associated explanation without modification.	Support. Accept it would not be practicable to show gas distribution lines on the planning maps. As an altermative, it may be possible to make information about the location of Powerco's gas lines available on the Council's GIS system.	Accept in part
S2.2	KiwiRail	Policy 16.4.2	Retain as notified.	Support.	Accept
α. Ω. Ω.	Spart	Policy 16.4.2 (Explanation)	Regionally significant network utilities provide benefits within the City. regionally and nationally. There are benefits that are to be considered in respect of any matter relating to regionally significant network utilities. Some of the benefits are: That people and goods can travel to, and from and around the City and Region efficiently and safely. That community well-being and public health and safely is maintained through the provision of essential services including supply of potable water and the collection. transfer and appropriate treatment of sewage and stormwater. and People have access to electricity. telecommunications.	Oppose in part. Support the recognition of the importance and role of "regionally significant network utilities" and the benefits they provide. Seek amendments to Policy 16.4.2 (in conjunction with other changes elsewhere) to specifically recognise telecommunications and radiocommunications – both of which are similarly essential services to those listed.	Reject
S4.11	Transpower	Policy 16.4.2 Policy 16.4.2 (Explanation)	Amend Policy 16.4.2 as follows: Recognise the national, regional and local benefits of, and protect regionally significant network utilities. Amend the explanation of Policy 16.4.2 as follows: Regionally significant network utilities provide benefits within the City, regionally and nationally. There are benefits that are to be considered in respect of any matter relating to regionally significant network utilities. Some of the benefits are: That people and goods can travel to and from and around the City and Regionefity and safety. That community well-being and public health and safety is maintained through	Support in part, subject to amendments sought in accordance with the amendments also requested for Objective 16.3.1.	Reject

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			the provision of essential services including supply of potable water and the collection. transfer and appropriate treatment of sewage and stommater. and People have access to electricity and gas to meet their needs. and the security of supply of these services can be maintained or improved.		
			And make any other consequential amendments to the Plan Change to ensure that the benefits of regionally significant network utilities are both recognised and protected.		
FS3.15	WELL		Reject in part. Council to make it clear in Policy 16.4.2 (third bullet point) that access to electricity is provided by distribution rather than transmission.	WELL consider it important to distinguish between Transpower's function of the transmission of high voltage electricity, and the distribution of electricity to the majority of end users (or consumers). As indicated in WELL's submission – distribution in WELL's submission – distribution in WELL's aupmission – distribution inters people's 'access' to leactricity.	Reject
SS 5.8	Chorus	Policy 16.4.2 (Explanation)	Regionally significant network utilities provide benefits within the City. regionally and nationally. There are benefits that are to be considered in respect of any matter relating to regionally significant network utilities. Some of the benefits are: People have access to electricity relecommunications. In released.	Oppose in part. Support the recognition of the importance and role of "regionally significant network utilities" and the benefits that they provide. Seek amendments (in conjunction with other changes elsewhere) to specifically recognise telecommunications and radio communications - both of which are similarly essential services to those listed.	Reject
S6.14	WELL	Policy 16.4.2	Regionally significant network utilities provide benefits within the City, regionally and nationally. These are benefits that are to be considered in respect of any matter relating to regionally significant network utilities. Some of the benefits are: People have access to are provided with electricity and gas to meet their needs.	Support in part. The third bullet is incorrect as the electricity distribution network provides access to electricity, rather than the defined Regionally Significant Network Utilities. The National Grid transports electricity across the region, and apart from large electricity customers, does not provide 'people' access to this electricity.	Reject
FS2.10 S7.8	Transpower Powerco	Policy 16.4.2	Accept Retain Policy 16.4.2 and the associated	Transpower agrees with this point. Support.	Reject Accept in part
0.00		, .	explanation without modification.	Orimont Matamanad to donal amond on	
823.3	KwiXail	Policy 16.4.3	Retain as notified.	Support. Not opposed to development on sties adjoining the rail network, however as trains can often be intermittent and cause effects on amenity, support that these potential effects are mitigated. Where sensitive activities are proposed on land	Accept in part

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				near the railway corridors, appropriate controls should therefore be imposed to ensure the long term amenity of those occupying those sites. Along with the need for mitigation is the need to manage the risk of objections and complaints leading to restraints on the operation, maintenance and enhancement of the rail corridor.	
FS3.2	WELL		Accept	WELL support this submission point in principle only to the extent that it similarly reflects WELL's desire to protect critical electricity lines from reverse sensitivity.	Accept in part
S4.12	Transpower	Policy 16.4.3	Avoid, or as appropriate, remedy or mitigate, the potential for any adverse effects including reverse sensitivity effects on regionally significant network utilities from meomeatule inappropriate subdivision, use and development occurring under, over, or adlacent to regionally significant network utilities	Support in part. Seek a minor amendment to the wording of the policy to ensure consistency with the explanation to the policy, the explanation to Objective 16.2.2 and other provisions of the District Plan and plan change.	Reject
S6.15	WELL	Policy 16.4.3	Include a definition for Critical Electricity Lines (CEL) in PC38; and subsequently include such CEL protection from reverse sestivity as that provided to Regionally Significant Network Utilities by proposed Policy 16.4.3.	Support in part. CELs are at risk from reverse sensitivity effects and therefore, similar to the defined regionally significant network utilities, should also be afforded the same level of protection.	Reject
S7.9	Powerco	Policy 16.4.3	Retain Policy 16.4.3 and the associated explanation without modification.	Support.	Accept in part
S4.13	Transpower	Policy 16.4.7	To manage subdivision and development within close proximity to within close program within close proximity to within close program within close provident electricity electricity transmission times to provident to electricity (a) The safe secure and electricity tansmission numity values of the electricity the community and anonity values of thin to protect the safety and and and and transmission innes and the safety and the formunity of and anonity values of thin to protect the safety and and and and <td< td=""><td>Oppose in part. Seek the replacement of 'to manage' to 'to control' subdivision and development within close proximity to existing high voltage electricity transmission lines.</td><td>Reject</td></td<>	Oppose in part. Seek the replacement of 'to manage' to 'to control' subdivision and development within close proximity to existing high voltage electricity transmission lines.	Reject
FS3.16	MELL	Policy 16.4.7	Accept	WELL support the intent of the sought relief. In the that Transpower's WELL note that Transpower's divestment of 110kV lines mean that the new owners require the same level of protection, regardless of their inclusion	Reject

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				of the National Grid umbrella. This continued protection is required as the effects of the lines will remain same, regardless of ownership.	
S6.16	WELL	Policy 16.4.7	Retain proposed Policy 16.4.7 as it is currently worded.	Support.	Accept
\$2.4	KiwiRail	Policy 16.4.8	Retain as notified.	Support. While new rail networks are not common, maintenance and upgrading of existing networks to consure safety and technology adjustments occur, are pivotal to the continued operation of the network. Support being able to undertake works with little impediment or restriction given the nature of the rail network, the purpose it services of the region, and the consequences of an unmaintained network on public safety being significant.	Accept
FS3.3	WELL	Policy 16.4.8	Council reject the submission in part and do not retain Objective 16.3.4 and Policy 16.4.8 unaltered – rather, the amendments sought by WELL are accepted by Council.	Oppose retention of Objective 16.3.4 as indicated in WELL's submission, there needs to be more explicit acknowledgement that not all adverse environmental effects of network utility development or operation can be avoided remedied or mitigated. WELL's opposition to Policy 16.4.8 being retained is based on the same rationale as the above.	Reject
S3.9	Spark	Policy 16.4.8	Retain Policies 16.4.8, 16.4.9 and 16.4.14.	Support.	Accept
S4.14	Transpower	Policy 16.4.8 Policy 16.4.8	Retain Policy 16.4.8 as proposed.	Support in part. Seek a minor amendment to the explanation to ensure consistency	Accept
		(Explanation)	Amend explanation of Policy 16.4.8 as follows: In some cases, some level of adverse effects may need to be accepted to recognise the necessity for, and benefits derived from, some network utilities and meet their operational requirements. This policy also recognises the benefits that all network utilities have.		
FS3.17	WELL		Accept	WELL support the submission point to the extent that it seeks better acknowledgement in the policy explanation that beneficial effects will ensure as a result of network utility development in the district.	Accept
S5.9	Chorus	Policy 16.4.8	Retain policies 16.4.8, 16.4.9 and 16.4.14.	Support.	Accept
96.17	WELL	Policy 16.4.8	To recognise and provide for the: • need for new and the maintenance and upgrading of existing network utilities; and itechnical and operational requirements and constraints of network utilities in assessing their location, design, development, construction and appearance; • the possibility of residual effects that cannot be fully avoided, remedied or	Support in part. The explanation explicit enough in recognising that not all environmental effects can be avoided, remedied or mitigated. The policy does not state that some adverse effects may need to be accepted.	Reject

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			 mitigated: and benefits that network utilities provide to the economic. social and cultural functioning of the City, Region and Nation. 		
FS2.11	Transpower		Accept	Transpower considers that this clause is appropriate and is consistent with the explanation to the policy.	Reject
S7.10	Powerco	Policy 16.4.8	Retain Policy 16.4.8 and the associated explanation without modification.	Support.	Accept
S2.4	KiwiRail	Policy 16.4.9	Retain as notified.	Support. While new rail networks are not common, maintenance and upgrading of existing networks to ensure safely and technology adjustments occur, are pivotal to the continued operation of the network. Support being able to undertake works with little impediment or restriction given the nature of the rail network, the purpose it services for the region, and the consequences of an unmaintained network on public safety being significant.	Accept
S3.9	Spark	Policy 16.4.9	Retain Policies 16.4.8, 16.4.9 and 16.4.14.	Support.	Accept
S4.15	Transpower	Policy 16.4.9	Retain Policy 16.4.9 and Policy 16.4.10 as proposed without amendment.	Support.	Accept
S5.9	Chorus	Policy 16.4.9	Retain policies 16.4.8, 16.4.9 and 16.4.14.	Support.	Accept
S7.11	Powerco	Policy 16.4.9	Retain Policy 16.4.9 and the associated explanation without modification.	Support.	Accept
S4.15	Transpower	Policy 16.4.10	Retain Policy 16.4.9 and Policy 16.4.10 as proposed without amendment.	Support.	Accept
S7.12	Powerco	Policy 16.4.10	Retain Policy 16.4.10 and the associated explanation without modification.	Support.	Accept
S4.16	Transpower	Policy 16.4.11 (Explanation)	Network utility operators, particularly those who operate regionally significant network utilities, should use the notice of requirement for designation process where appropriate when they seek to develop new or extend existing network utilities.	Support in part. Seek minor amendments to the explanation of the policy to recognise that in all instances it may not be possible to utilise a notice of requirement process.	Accept
S7.13	Powerco	Policy 16.4.11	Retain Policy 16.4.11 and the associated explanation without modification.	Support.	Accept in part
S2.4	KiwiRail	Policy 16.4.12	Retain as notified.	Support. While new rail networks are not common, maintenance and upgrading of existing networks to ensure safety and technology adjustments occur, are pivotal to the continued operation of the network. Support being able to undertake works with little impediment or restriction given the nature of the rail network, the purpose it services for the region, and the consequences of an unmaintained network on public safety being significant.	Accept

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
FS1.5	Powerco		Accept	Powerco supports the requested amendments, which recognise that it will not always be possible to avoid, remedy or mitigate all adverse effects due to the technical and operational constraints associated with network utilities.	Accept
ο. 	WELL		Accept	Support for the submission points have been given because they reflect similar outcomes being sought by WELL in their submission. WELL consider that the new network utility chapter should provide more explicit acknowledgement to the fact that not all adverse environmental effects can be avoided remeded or mitigated. WELL consider that the use of the words "to the extent practicable" suitably acknowledges network utility developments.	Accept
S4.17	Transpower	Policy 16.4.12	Ensure that network utilities are designed, developed. constructed. located, upgraded, operated and maintained to avoid, remedy or mitigate any actual or potential adverse effects on the environment to the extent practicable.	Oppose, for the reasons outlined for Objective 16.3.2 and to ensure consistency with the NPSET.	Accept in part
FS3.18	WELL		Accept	WELL support this submission point as it reflects WELL's opinion the new network utility chapter needs greater emphasis that in some instances not all adverse environmental effects can be avoided remedied or mitigated.	Accept in part
ю Э	Chorus	Policy 16.4.12	Policy 16.4.12: Ensure that network utilities are designed, developed, constructed, located, upgraded, operated and maintained to avoid, remedy or mitigate any actual or potential adverse effects on the environment to the extent practicable.	Oppose in part. Seek amendments to consistently recognise that adverse effects will be managed as far as practicable, and that not adverse effects may be able to be avoided, remedied or mitigated. Seek an amendment to 16.6 'Anticipated environmental results' to reflect that all adverse effects may not be able to be avoided, remedied or mitigated.	Accept in part
лю 30 30	WELL		Accept	Supported for the reasons stated in the submission. WELL also submitted that Objective 16.3.4 needed to better acknowledge that some residual environmental effects of network utility development and/or operation would not be able to be avoided remedied or mitigated. The recommended amendment to the able to be avoided remedied or mitigated. The recommended amendment to the able to be avoided remedied or mitigated. The recommended amendment to the matter. WELL's submission on the same matter. WELL support the use of the words "to the extent practicable" as requested by the submitter in Proposed Policy 16.4.12 and the corresponding 16.6 Anticipated environmental result	Accept in part
S6.18	WELL	Policy 16.4.12	Ensure that network utilities are designed.	Support in part. Seek minor amendments	Accept in part

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			developed, constructed, located, upgraded, operated and maintained to the extent possible to avoid, remedy or mitigate any actual or potential adverse effects on the environment.	to acknowledge that not all effects derived from the development and operation of network utilities can be avoided, remedied or mitigated as acknowledged in the preceding Policy 16.4.8.	
S7.14	Powerco	Policy 16.4.12	Retain Policy 16.4.12 and the associated explanation without modification.	Support.	Accept
S4.18	Transpower	Policy 16.4.13 (Explanation)	Some network utilities may adversely affect health and safety. For example, telecommunication facilities generate radio detrimental effects on health. Any optential health feffects arising from radiofrequency emissions are adressed by Regulation 4 of the Resource Management (National Environmental Standards for Telecommunication Facilities) Regulations generate electromagnetic fields (EMF) which may be a risk (b health, and aleo generates the may be a risk (b health, and aleo generates the isk of electromagnetic fields (EMF) which Mational Environmental Standard for Electricity Statement on Electricity Transmission, and the National Environmental Standard for Electricity Irransmission on Non-lonising Radiation Protection (ICNIRP) to prevent the potential for health effects.	Oppose. Seek the deletion of reference to the generation of electrocution by electricity transmission / distribution. As currently worded the explanation regarding EMF is vague which is unnecessary given the clear direction provided by the NPSET and the NESETA. The amended wording requested is more consistent with the intention of the policy, which is that network utilities comply with relevant standards.	Accept in part
FS3.19	WELL		Accept	For the reasons stated in the submission point. WELL consider that reference /statements to adverse health and safety effects are inappropriate and should be termered as requested by the submitter. Reference to the International Commission on Non- ionising Radiation Protection ((CNIRP) is appropriate in relation to health effects commisance	Accept in part
S6.19	WELL	Policy 16.4.13	Ensure network utilities, in particular those emitting electric and magnetic fields, are designed, located, upgraded, operated and maintained to comply with relevant national environmental standards and to meet other nationally and internationally recognised standards.	Support in part. The policy refers twice to national standards - assume this is a drafting error. Seek amendments to recognise the International Commission on Non-Ionising Radiation Protection Guidelines for limiting exposure to time-varying electric, magnetic, and electromagnetic fields (up to 300 GHz) (Health Physics, 1998, 74(4); 494-522) (ICNIRP Guidelines).	Reject
FS2.12	Transpower		Accept	Transpower agrees that the reference to internationally is appropriate.	Reject
S7.15	Powerco	Policy 16.4.13	Policy: Manage effects on health and safety by ensuring Ensure network utilities, in particular	Support in part. While the explanation identifies that the focus is on health and safety matters, this is not reflected in the	Accept in part

Commissioners' Recommendation on submission point		Accept in part	Accept in part
Reasons	wording of the policy itself. Seek amendment of the policy to provide a clearer focus. Concern that the wording of the explanation relating electromagnetic fields (EMF) generated by electricity transmission / distribution activities may cause unnecessary concern about potential health effects of EMF. Despite numerous international studies, there is a still no persuasive evidence that EMF poses any health risks. Because of this, it is not appropriate to suggest that EMF from electricity transmission/distribution may be a six to health or for the policy to focus on utilities emitting electric and magnetic fields.	Transpower considers that this change is appropriate, and is consistent with the explanation.	For the reasons stated. WELL consider that reference /statements to adverse
Decision Sought	these emitting electric and magnetic fields are designed, located, upgraded, operated and maintailed to comply with relevant other nationality recognised standards and to methon: Expanation: Some network utilities may adversely affect health and safety. For example, relecommunication facilities generater radio frequency emissions which may have detimental effects anising from radiofrequency emissions are addressed by Regulation 4 of the Resource Management (National Environmental Environmental Environmental Environmental Environmental Environmental Environmental affects on bath Any botentian detimental effects anising from radiofrequency emissions are addressed by Regulation from addressed by Regulation 4 of the Resource Management (National EMH) which may be present a from activities can generate electromagnetic fields and asfety insks are actionation from gas or petroleum pipelines setationes from gas or petroleum pipelines and from address and and addressing health and safety matters and accidental spillage or leakage of hazardous substances from gas or petroleum pipelines addressing health and safety matters that are accidental spillage or leakage of hazardous addressing health and safety matters that are exclosed al spillage or leakage a risk if an accidental spillage or leakage a risk if an accidental spillage or leakage arisk if addressing health and safety matters that are exclosed al spillage or leakage arisk if addressing health and safety matters that are exclosed al spillage or leakage arisk if addressing health and safety matters are addressing health and safety matters are builtening and addressing health and safety matters are exclosed and or perioding and addressing health and safet	Accept	Accept
Provision			
Submitter		Transpower	WELL
Submission No.		FS2.13	FS3.40

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				health and safety effects are inappropriate and should be tempered as requested by the submitter. Reference to the International Commission on Non-ionising Radiation Protection (ICNIRP) is appropriate in relation to health effects compliance.	
S3.9	Spark	Policy 16.4.14	Retain Policies 16.4.8, 16.4.9 and 16.4.14.	Support.	Accept
0.	Spark	Policy 16.4.14	Policy 16.4.4 Enable the co-location or multiple use of network utilities where this is efficient and practicable and assists with avoiding, remedying or mitigating adverse effects on the environment to the extent practicable.	Oppose in part. Seek amendments to both Objective 16.3.4 and its associated Policy 16.4.12 to consistently recognise that adverse effects will be managed as far as practicable, and that not all adverse effects may be able to be avoided, remedied or mitigated. Seek an amendment to 16.6 'Anticipated environmental results' to reflect that all adverse effects may not be able to be avoided, remedied or mitigated.	Reject
S4.19	Transpower	Policy 16.4.14	Enable the co-location or multiple use of network utilities where this is efficient and practicable technically feasible and assists with avoiding. remedying or mitigating adverse effects on the environment.	Support in part, subject to a wording change to the policy that, like the explanation recognises that there are a number of technical constraints to the colocation of or multiple use of network utilities.	Accept in part
FS3.20	WELL		Accept	WELL support the submission to amend proposed Policy 16.4.14 as it will provide certainty in instances where co-location is not pursued by the network utility operator	Accept in part
S5.9	Chorus	Policy 16.4.14	Retain policies 16.4.8, 16.4.9 and 16.4.14.	Support.	Accept
S7.16	Powerco	Policy 16.4.14	Retain Policy 16.4.14 and the associated explanation without modification.	Support.	Accept
8.1.8	GWRC	Policy 16.4.14 (Explanation)	Amend the explanation of Policy 16.4.14 as follows: It is also recognised that co-location is not alwars possible due to operational issues such alwars possible due to operational issues and alwars possible due to operational is and alwars possible due to op	Support in part. Support the encouragement of co-location services, but seek recognition within the proposed policy or its explanation that co-location is not always appropriate, namely where its effects may be to constrict flood flows.	Reject
10.33 1.5 2.5 2.5 2.5 2.5 2.5 2.5 2.5 2.5 2.5 2	WELL		Reject Council rejects the submission points in their Council to retain undergrounding utilities as a Council to retain undergrounding utilities as a permitted activity in all areas regardless of flood mitigation assets.	WELL oppose both submission points. All undergrounding of lines (including removal, maintenance, operation and upgrading) should remain permitted activities as indicated in Table 30.1 regardless of flood protection devices. Restricted discretionary consent for underground line works within flood protection assets as sought by the submitter would be onerous and unnecessary as WELL construction standards specify correct backfill compacton requirements (as would most utility operators) and hence will avoid undergin requirem to the flood control	Accept

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				asset. A permitted activity standard relating to effects on flood control structures would be more appropriate.	
81.4	GWRC	Policy 16.4.15	Require the underground placement of new network utilities unless: • there are natural or physical features or structures or technological and operational constraints that makes underground placement impracted or unreasonable; they are of a temporary nature and required for emergency purposes or citical events; and they are of a nature that they can only operate aboverground; and the placement is through a flood protection stop-bank.	Support in part. Support the encouragement of co-location services, but seek recognition within the proposed policy or its explanation that co-location is not always appropriate, manely where its effects may be to constrict flood flows.	Reject
FS2.2	Powerco		Accept	Powerco accepts there is a need to protect the integrity of flood protection assets and can support the proposed amendment to Policy 16.4.15 to remove the policy requirement for the placement of new network utilities underground when they are through a flood protection stop-bank.	Reject
S4.20	Transpower	Policy 16.4.15	Except for transmission lines*. rRequire the underground placement of new network utilities * 'transmission lines' are already defined in the District Plan however as an alternative and consequential amendment Transpower requests rewording the corridor definition to 'National Grid Corridor', 'National Grid Yard' and including a definition of 'National Grid.'	Oppose. While it is accepted that the transmission and distribution of electricity can occur underground, the undergrounding of new high voltage transmission lines for the National Grid can be up to 10 – 20 times more expansive than above ground lines. This key constraint should be recognised in the policy or the National Grid should be specifically exempt from this policy.	Reject
FS3.21	WELL		Reject. Council does not specifically exclude transmission lines form the undergrounding policy as sought by the submitter.	The submission point requests an exemption to the policy based on non-RMA grounds (i.e., financial cost) and cannot be considered. Excessive undergrounding costs are not limited only to Transpower	Accept in part
S5.10	Chorus	Policy 16.4.15 (Explanation)	The adverse visual effects of certain network utilities can often be managed by putting the services underground. With some exceptions #this is the required generally the required approach for those network utilities, such as those with cables that can be located underground. For those network utility structures that need to be located aboveground, particular attention should be given to their design. location and minimising of any adverse visual effects as outlined in Policy 16.4.12. This can be achieved in a number of ways including, where practical through screening, careful placement, size and	Support in part. Seek amendment to signal that the addition of overhead fibre optic lines to existing support structures are provided for as a permitted activity in limited circumstances. The use of existing support structures to add additional overhead lines in the manner proposed is an efficient use of existing infrastructure with minimal adverse environmental effects. Seek further changes to remove inconsistencies in the current wording.	Accept in part

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			appearance and applying different activity status. With the exception of Protected Ridgelines and the Southern Hils. <u>new</u> overhead lines (where there is no existing overhead lines (where there is no existing overhead network) are provided as a permitted activity in the Rural and Open Space Zones provide for overhead lines. recognising the more visual absorptive capacity of those locations. and the practicatity of needing to provide for a cost effective means of needing to provide for a cost effective means of needing to provide for a cost effective means of a provide service development and maintenance in nemote less densely populated areas. <u>New above around lines and their associated</u> weating above or overhead lines. <u>new overhead</u> of a nexisting overhead lines. <u>an ensiting</u> provision is made for new outsomer upgrading. In recognition that this is an efficient use of an existing resource. <u>However: a ve</u> <u>eventions</u> and where provided for as minio upgrading. In recognition that this is an efficient use of an existing resource. <u>However: a se</u> <u>eventions</u> <u>upgrading</u> . In policy also recognises that there are <u>other</u> situations where placing lines <u>order for an order or sectorable or underground</u> is, or may be, impracticable or unreasonable.		
S7.17	Powerco	Policy 16.4.15	Retain Policy 16.4.15 and the associated explanation without modification.	Support.	Accept
S7.18	Powerco	Policy 16.4.16	Retain Policy 16.4.16 and the associated explanation without modification.	Support.	Accept
S7.19	Powerco	Policy 16.4.17	Retain Policy 16.4.17 and the associated explanation without modification.	Support.	Accept
с. S	Forest and Bird	Policy 16.4.17	While requiring consultation in all cases would be unreasonable, we suggest that the policy be reworded to (a) strengthen the word 'encourage' as a general terms, and (b) identify those situations where consultation is required.	Oppose in part. Consider that 'encourag[ing]' network utility providers to consult with local communities does not seem strong enough and that there will be occasions where such community consultation, especially with interest groups such as F&B, would be desirable, if not essential.	Reject
FS2.14	Transpower		Reject	Transpower opposes these requests as it considers that any requirement to consult would be inappropriate and unduly restrictive. Transpower considers that the policy and method, as proposed, are appropriate and should be retained without amendment.	Accept
FS3.47	WELL		Reject	Consultation should be at the discretion of the network utility owner. Such consultation will occur on the basis of effects and will form part of the network utility operators mitigation o environmental	Accept

	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				effects.	
Transp	ower	Method 16.5.1(5)	 The consideration of viable alternatives for new designations<u>Plan ehange(s) to intreduce</u> new provisions to manage reverse sensitivity effects 	Support in part, subject to minor wording changes requested below. Plan provisions for managing reverse sensitivity effects on regionally significant network utilities are already contained within the Operative District Plan.	Reject
WELL			Reject	WELL consider that the retention of the Plan Change method is suitable to deal with future instances of reverse sensitivity on infrastructure. Such plan changes could involve critical electrical lines being captured under any amended definition of regionally significant infrastructure.	Accept
Trans	power	Method 16.5.2 Method 16.5.6 Method 16.5.7 Method 16.5.9 Method 16.5.9	Retain methods 16.5.2, 16.5.6, 16.5.7, 16.5.9 and 16.5.10 as proposed.	Support. The methods as proposed are appropriate implementation mechanisms to address the resource management issues, objectives and policies outlined in amended Chapter 30.	Accept
Trans	power	Method 16.5.8	Education of and -bBuilding relationships with network utility providers.	Oppose. While the concept of 'building relationships with network utility providers' is supported, it is unclear as to what is intended by the 'education of' network utility providers.	Accept in part
WEL	_		Accept	For the reasons stated.	Accept in part
Fore Bird	st and	Method 16.5.9	The decision sought related to 16.4.17 (see S8.8) should be reflected in the methods.	See S8.8	Reject
Tran	spower		Reject	Transpower opposes these requests as it considers that any requirement to consult would be inappropriate and unduly restrictive. Transpower considers that the policy and method, as proposed, are appropriate and should be retained without amendment.	Accept
Spar	~	16.6 Anticipated environmental results and monitoring	16.6 Anticipated environmental result: The avoidance, remedying, or mitigation of the adverse effects of developing and maintaining the City's network utilities to the extent practicable.	Oppose in part. Seek an amendment to 16.6 'Anticipated environmental results' to reflect that all adverse effects may not be able to be avoided, remedied or mitigated.	Reject
Ром	erco		Accept	Powerco supports the requested amendments, which recognise that it will not always be possible to avoid, remedy or mitigate all adverse effects due to the technical and operational constraints associated with network utilities.	Accept
WEI	-		Accept	Support for the submission points have been given because they reflect similar outcomes being sought by WELL in their submission. WELL consider that the new network utility chapters hould provide more explicit acknowledgement to the fact that not all adverse environmental effects can be	Reject

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				avoided remedied or mitigated. WELL consider that the use of the words 'to the extent practicable' suitably acknowledges that there may be residual effects of network utility developments.	
S S S	Chorus	16.6 Anticipated environmental results and monitoring	16.6 Anticipated environmental result: The avoidance, remedying, or mitigation of the adverse effects of developing and maintaining the City's network utilities to the extent practicable.	Oppose in part. Seek amendments to consistently recognise that adverse effects will be managed as far as practicable, and that not all adverse effects may be able to be avoided, remedied on mitigated. Seek an amendment to 16.6 Anticipated environmental results' to reflect that all adverse effects may not be able to be avoided, remedied or mitigated.	Reject
о м м	WELL		Accept	Supported for the reasons stated in the submission. WELL also submitted that Objective 16.3.4 needed to better acknowledge that some residual environmental effects of network utility development and/or operation would not be able to be avoided remedied or mitgated. The recommended amendment to the Objective's explanation is supported as it mirrors WELL's submission on the same matter. WELL support the use of the words "to the extent practicable" as requested by the submitter in Proposed Policy 16.4.12 and the orresponding 16.6 Anticipated environmental result	Reject
AMENDMENT	2: Amendmen	its to Chapter 16A	- Renewable Energy (Issues, Objectives and P	olicies)	
S8.7	Forest and Bird	16A.3 (all Objectives)	 We would like to see the inclusion of Objectives addressing the following: Education and information about renewable energy generation and energy conservation; Subsidising renewable energy generation initiatives; and Identification of cost-effective renewable energy options that have reduced environmental impacts. 	Oppose in part. Consider that the Objectives in Chapter 16A are too limiting in terms of the potential scope that UHCC could lead in what it could do to advance renewable energy generation in Upper Hutt.	Accept in part
AMENDMENT	3: Amendmen	its to Chapter 18 -	Residential Zone (Rules)		
S2.5	KiwiRail	Rule 18.6	Retain as notified.	Support. KiwiRail encourage land use near the raiway corridor that does not compromise the short or long term ability to operate a safe and efficient rail network, both day and night. Support the consideration of the impact of the design and layout of the subdivision on the rail network as a regionally significant network utility, and the outcome of consultation with KiwRail, as matters Council seek to control.	Accept
S4.24	Transpower	Rule 18.6	Amend Rule 18.6 and 18.28A as follows:	Oppose. In particular, oppose the exclusion of the National Grid from the new matters	Reject

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			Council may impose conditions over the following matters: The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the owner or operator of regionally significant tables) Note: Ruth 18.20 eoverse subdivision within the Hallowing the owner of the site. Note: Ruth 18.20 eoverse subdivision within the Electricity Transmission Corrider. Subdivision which creates building platforms within 20m of high vottage (110kV or greater) electricity transmission lines as shown on the planning maps also requires resource consent under Rule 18.29.	of consideration bullet points in sections 18.6, 18.28A and 18.37. It is understood that the National Grid has been specifically excluded as specific matters of consideration are included in the District Plan for subdivision requiring consent under Rule 18.29 where the subdivision creates building platforms with 20m of electricity transmission lines. As currently drafted, there is no obligation for applicants to consult with Transpower if a subdivision were to take place on land where National Grid assets, or access to National Grid assets, or access to nore than 20m from the transmission line. While the building platforms were proposed more than 20m from the transmission line assets or access to National Grid assets or access to National Grid assets or access the opticate distance any subdivision of land containing National Grid assets to access the opticate distance any upgrade these assets through alterations to to potondaries and access arrangements.	
S6.20	WELL	Rule 18.6	Amend the permitted and controlled activity standards to include the outcome of consultation of the CEL's owner and that consequential amendments are made throughout the Residential Zone Rules.	Support in part. Consider that it is important that, as owner and operators of CELs, consultation is undertaken with them in relation to subdivision and other land use developments in close proximity to the sub- transmission network.	Reject
\$2.5 5	KiwiRail	Rule 18.28A	Retain as notified.	Support. KiwiRail encourage land use near the railway corridor that does not compromise the short or long term ability to operate a safe and efficient rail network, both day and night. Support the consideration of the impact of the design and layout of the subdivision on the rail network as a regionally significant network utility, and the outcome of consultation with KiwiRail, as matters Council seek to control.	Accept
S4.24	Transpower	Rule 18.28A	Amend Rule 18.6 and 18.28A as follows: Council may impose conditions over the following matters: The outcome of consultation with the owner or operator of regionally significant network utilities (axeluding the National Crid) located on or in proximity to the site.	Oppose. In particular, oppose the exclusion of the National Grid from the new matters of consideration bullet points in sections 18.6, 18.28A and 18.37. It is understood that the National Grid has been specifically excluded as specific matters of consideration are included in the District Plan for subdivision requiring consent under Rule 18.29 where the subdivision creates building platforms with 20m of electricity transmission lines. As currently drafted, there is no obligation for applicants to consult with Transpower if	Reject

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			which creates building platforms within 20m of high voltage (110kV or greater) electricity transmission lines as shown on the planning maps also requires resource consent under Rule 18.29.	a subdivision were to take place on land where National Grid assets, or access to National Grid assets are located where the proposed building platforms were proposed more than 20m from the transmission line. While the building platforms may be proposed at an appropriate distance any subdivision of land containing National Grid assets or access to National Grid assets has the potential to constrain Transpower's ability to access, inspect, maintain, or upgrade these assets through atterations to to fot boundaries and access arrangements.	
S2.5	KiwiRail	Rule 18.37	Retain as notified.	Support. KiwiRail encourage land use near the railway corridor that does not compromise the short or long term ability to operate a safe and efficient rail network, both day and night. Support the consideration of the impact of the design and layout of the subdivision on network utility, and the outcome of consultation with KiwiRail, as matters Council seek to control.	Accept
S2.6	KiwiRail	Rule 18.37	Insert the following provision as a matter for consideration: Account must be taken of the future development potential of adjoining or adjacent land and any potential reverse sensitivity effects on regionally significant network utilities (excluding the National Grid).	Support in part. Support that potential reverse sensitivity effects on regionally significant network utilities are considered through the sudvision process. Consideration of elements such as setbacks, design standards and mitgation requirements for development on a site at the time of subdivision, will ensure that future occupants are fully aware of the potential effects from residing adjoining a mitgated so that the railway is not subject to reverse sensitivity effects. Seek to insert a similar provision to ensure that considered in relation to regionally significant infrastructure when subdivision applications are being processed.	Accept in part
S4.24	Transpower	Rule 18.37	 Amend Rule 18.37 as follows: The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of and access to, regionally significant network utilities (excluding the National Grief) located on or in proximity to the site. The outcome of consultation with the owner or operator of regionally significant the twork utilities (excluding the National Grief) located on or in proximity to the site. 	Oppose. In particular, oppose the exclusion of the National Grid from the new matters of consideration bullet points in sections 18.6, 18.28A and 18.37. It is understood that the National Grid has been specifically excluded as specific matters of consideration are included in the District Plan for subdivision requiring consent under Rule 18.29 where the subdivision creates building platforms with 20m of electricity transmission lines. As currently drafted, there is no obligation	Reject

Commissioners' Recommendation on submission point	بان 6 ق ق ق ق ق ت ت ت ت ت ت ت ت ت ت ت ت ت ت ت		Reject	
Reasons	for applicants to consult with Transpowel a subdivision were to take place on la where National Grid assets, or access National Grid assets are located where t proposed building platforms were propos more than 20m from the transmission lin While the building platforms may proposed at an appropriate distance a subdivision of land containing National G assets or access to National Grid asse has the potential to constrain Transpowe ability to access, inspect, maintain, upgrade these assets through alterations lot boundaries and access arrangements		Oppose. See S4.24	
Decision Sought	to the site. Note: Rule 18.26 covers eublinision within the Electricity Transmission Corridor. Subdivision which creates building platforms within 20m of high vottage (110kV or greater) electricity transmission lines as shown on the planning maps also requires resource consent under Rule 18.29.	Rural Zone (Rules)	 Amend Rule 19.6 as follows: Amend Rule 19.6 as follows: Council may impose conditions over the following matters: The outcome of consultation with the owner or operator of regionally significant network utilities (exeluting the significant network utilities (exeluting the significant network utilities (exeluting the batter) electricity transmission form on the planning maps also requires resource consent under Rule 19.21. Amend Rule 19.22 as follows: 	transmission lines as sriuwri uri ure prariririy
Provision		ts to Chapter 19 -	Rule 19.6 Rule 19.28 Rules Rules	
Submitter		4: Amendmen	Transpower	
Submission No.		AMENDMENT	S4.25	

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
S6.20	WELL	Rule 19.6	Amend the permitted and controlled activity standards to include the outcome of consultation of the CEL's owner and that consequential amendments are made throughout the Residential Zone Rules.	Support in part. Consider that it is important that, as owner and operators of CELs, consultation is undertaken with them in relation to subdivision and other land use developments in close proximity to the sub- transmission network.	Reject
S2.5	KiwiRail	Rule 19.6	Retain as notified.	Support. KiwiRail encourage land use near the railway corridor that does not compromise the short or long term ability to operate a safe and efficient rail network, both day and night. Support the consideration of the impact of the design and layout of the subdivision on network utility, and layout of the subdivision on the rail network as a regionally significant network utility and he outcome of consultation with KiwRail, as matters Council seek to control.	Accept
S2.5	KiwiRail	Rule 19.28	 The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of and access to, regionally significant network utilities (exeluding the National Grid)—located on or in proximity to the site. The outcome of consultation with the owner or operator of regionally significant network utilities (exeluding the National National National National Construction with the subdivision which construct the site. Note: Rule 10.21 covers within the Electricity Hammission Lines as shown on the planning maps also requires resource consent under Rule 19.21. Retain as notified. 	Support. KiwiRail encourage land use near the railway corridor that does not compromise the short or long term ability to operate a safe and efficient rail network, both day and inght. Support the consideration of the impact of the design and layout of the subdivision on network utility, and the outcome of consultation with KiwRail as matters	Accept
AMENDMENT	5: Amendmen	ts to Chapter 20 -	Business Zone (Rules)	Council seek to control.	
S2.5	KiwiRail	Rule 20.6	Retain as notified.	Support. KiwiRail encourage land use near the railway corridor that does not compromise the short or long term ability to	Accept

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				operate a safe and efficient rail network, both day and night. Support the consideration of the impact of the design and layout of the subdivision on the rail network as a regionally significant network utility, and the outcome of consultation with KwiRail, as matters Council seek to control.	
S6.20	WELL	Rule 20.6	Amend the permitted and controlled activity standards to include the outcome of consultation of the CEL's owner and that consequential amendments are made throughout the Residential Zone Rules.	Support in part. Consider that it is important that, as owner and operators of CELs, consultation is undertaken with them in relation to subdivision and other land use developments in close proximity to the sub- transmission network.	Reject
S2.5	KiwiRail	Rule 20.32	Retain as notified.	Support. KiwiRail encourage land use near the railway corridor that does not compromise the short or long term ability to operate a safe and efficient rail network, both day and night. Support the consideration of the impact of the design and layout of the subdivision on network utility, and the outcome of consultation with KiwiRail, as matters Council seek to control.	Accept
AMENDMENT	6: Amendmen	nts to Chapter 21 -	Open Space Zone Rules		
\$4.27 \$2.5	Transpower KiwiRail	Rule 21.5A Rule 21.5A	Amend Rule 21.5A and Rule 21.28 as follows: Council may impose conditions over the following matters: The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the power or operator of regionally significant network utilities (excluding the power or operator of regionally Mathin the Electricity Transmission Corrider. Retain as notified.	Oppose, In particular, oppose the exclusion of the National Grid from the new matters of consideration bullet points in sections 21.5A, 21.2B and 21.41, it is understood that the National Grid has been excluded, as per the other zone chapters, as there are specific rules relating to subdivision that creates building platforms within the Electricity Transmission Corridor. However, no such subdivision rule is provided in the open space zone chapter and as such the covers buildings and structures within the Electricity Transmission Corridor) is incorrect and should be deleted. Support. KiwiRail encourage land use near the railway corridor that does not comperate a safe and efficient rail network, both day and layout of the subdivision on the trail on with KiwiRail, as matters Compont kutility, and the outcome of the design and layout of the subdivision on the trail.	Reject Accept
S6.20	WELL	Rule 21.5A	Amend the permitted and controlled activity	Support in part. Consider that it is	Reject

ubmission Io.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			standards to include the outcome of consultation of the CEL's owner and that consequential amendments are made throughout the Residential Zone Rules.	important that, as owner and operators of CELs, consultation is undertaken with them in relation to subdivision and other land use developments in close proximity to the sub- transmission network.	
20	KiwiRail	Rule 21.28	Retain as notified.	Support. KiwiRail encourage land use near the railway corridor that does not compromise the short or long term ability to operate a safe and efficient rail network, both day and night. Support the consideration of the impact of the design and layout of the subdivision on network utility, and the outcome of consultation with KiwiRail, as matters Council seek to control.	Accept
+.27	Transpower	Rule 21.28	Amend Rule 21.5A and Rule 21.28 as follows: Council may impose conditions over the following matters: The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the Matienal Grid)–located on or in close proximity to the site. <u>Note: Rule 21.16 covers subdivision</u> within the Electricity Transmission Corrider.	Oppose. In particular, oppose the exclusion of the National Grid from the new matters of consideration builter points in sections 21.54, 21.28 and 21.41. It is understood that the National Grid has been excluded, as per the other zone chapters, as there are specific rules relating to subdivision that creates building platforms within the Electricity Transmission Corridor. However, no such subdivision rule is provided in the open space zone chapter and as such the covers buildings and structures within the Electricity Transmission Corridor. However, no such subdivision rule is provided in the open space zone chapter and as such the covers buildings and structures within the Electricity Transmission.	Reject
بم د	KiwiRail	Rule 21.41	Retain as notified.	Support. KiwiRail encourage land use near the railway corridor that does not compromise the short or long term ability to operate a safe and efficient rail network, both day and night. Support the consideration of the impact of the design and layout of the subdivision on the rail network as a regionally significant network utility, and the outcome of consultation with KiwiRail, as matters Council seek to control.	Accept
MENDMENT	7: Amendmer	its to Chapter 22 -	Special Activity Zone Rules		
25	KwiRail	Rule 22.7	Retain as notified.	Support, KiwiRail encourage land use near the railway corridor that does not compromise the short or long term ability to operate a safe and efficient rail network, both day and night. Support the consideration of the impact of the design and layout of the subdivision on the rail network as a regionally significant network utility, and the outcome of consultation with KiwiRail, as matters	Accept

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				Council seek to control.	
S4.28	Transpower	Rule 22.7	Amend Rules 22.30 and 22.7 as follows: Council may impose conditions over the following matters: The outcome of consultation with the	Oppose. See 4.24	Reject
			The outcome of consumation with the owner or operator of regionally significant network utilities (excluding the Mational Grid) located on or in close proximity to the site. Note: Rule 22.18 covers subdivision within the Electricity Transmission Cerrider within 32m of high voltage platforms within 32m of high voltage (110kV or greater) electricity transmission lines as shown on the planning maps.		
S2.5	KiwiRail	Rule 22.30	Retain as notified.	Support. KiwiRail encourage land use near the railway corridor that does not compromise the short or long term ability to operate a safe and efficient rail network, both day and night. Support the consideration of the impact of the design and layout of the subdivision on the rail network as a regionally significant network utility, and the outcome of consultation with KiwRail, as matters Council seek to control.	Accept
S4.28	Transpower	Rule 22.30	Amend Rules 22.30 and 22.7 as follows: Council may impose conditions over the following matters: The outcome of consultation with the significant network utilities (excluding the significant network utilities (excluding the National Grid)-located on or in close proximity to the site. Note: Rule 22.18 covers subdivision within the Electricity Transmission Corridor within 32m of high voltage f110kV or greater) electricity transmission lines as shown on the planning maps.	Oppose. See 4.24	Reject
S6.20	MELL	Rule 22.30	Amend the permitted and controlled activity standards to include the outcome of consultation of the CEL's owner and that consequential amendments are made throughout the Residential Zone Rules.	Support in part. Consider that it is important that, as owner and operators of CELs, consultation is undertaken with them in relation to subdivision and other land use developments in close proximity to the sub- transmission network.	Reject
S2.5	KiwiRail	Rule 22.31	Retain as notified.	Support. KiwiRail encourage land use near the railway corridor that does not	Accept

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				compromise the short or long term ability to operate a safe and efficient rail network, both day and night. Support the consideration of the impact of the design and layout of the subdivision on the rail network as a regionally significant network utility, and the outcome of consultation with KiwiRail, as matters Council seek to control.	
S4.28	Transpower	Rule 22.31	Amend Rule 22.31 as follows: Matters that may be relevant in the consideration of any resource consent, other than for a restricted discretionary activity, may include the following: The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of and access to regionally significant network utilities (<i>exeluding the National eite.</i> The outcome of consultation with the owner or operator of regionally significant network utilities (<i>exeluding the National Citid</i>) located on or in proximity to the site. The outcome of consultation with the owner or operator of regionally bational Citid) located on or in proximity to the site.	Oppose. See 4.24	Reject
A AAT NICAAT NT	0. A second second	of actual of a	as shown on the planning maps.	antinan Bulan	
AMENDMENT	8: Amendmen	its to Chapter 28 -	 Southern Hills Overlay Area and Protected Rid 	gelines Rules	
S4.29	Transpower	Chapter 28 Southern Hills Overlay Area and Protected Ridgeline Rules	Delete all proposed amendments to Chapter 28.	Oppose. Support the activity status of network utilities are restricted discretionary activities under Rule 28.5, however Transpower's earlier comments on the draft Plan Change sought the exclusion of a requirement for network utilities to comply with the rules and standards in the zone chapter of the District Plan. This is requested to extend to overlay areas also. The inclusion of network utilities in this chapter does not make sense as the rule framework relies on a base rule in an underlying zone but network utilities are not covered by zone rules. Standards and matters of consideration consideration consideration consideration context to the amendments to this chapter activity to the mendments of the potential provide sufficient scope for the potential	Accept in part

ommissioners' Recommendation		cept in part	cept in part	cept	cept in part	
Reasons	adverse effects (namely landscape and visual effects) of network utilities within the Southern Hills Overlay Area and on protected ridgelines to be adequately assessed.	WELL support this submission point for the Adressons stated. Teasons stated. WELL supported PtC38 in general based on the Network Utility Chapter representing a complete code. The submission point by Transpower appropriately raises the matter of the complete code approach being diluted via the use of policy overlays, as well as the confusing matter of how zone rules apply to the concept of a complete code.	Support in part. It is unclear whether minor Ar above ground lines and additional fibre optic lines to existing overhead network utility structure (as requested in the relief sought in S5.18) are caught by Rules 28.1 and 28.2. Seek to make these activities permitted with no applicable standards as they have less than minor adverse effects, particularly in the context of the existing structures.	WELL support the submission points as At they will provide certainty to permitted activities within the Protected Ridgelines and Southern Hills Overlay areas.	Oppose in part. It is unclear and uncertain as to which area wide rules apply to network utilities. 16.1 (tenth paragraph) states that. The underiving zone objectives, policies and rules do not apply to network utilities. including roads, unless specifically referred to. City wide rules, such as those relating to earthworks, notable trees, historic heritage and hazardous substances will still apply. In relation to the City wide rules, there are no applicable earthworks rules, and eccordingly the reference to this should be deleted. Provisions have also been included for "New above ground network utilities" in the protected Ridgelines and Southern Hills Overlay area (through Amendment 8 of the Plan Change), but it is not specifically stated in the Utilities pontore a sonorific statement he invited	
Decision Sought		Reject	Amend Rules 28.1 and 28.2 to provide as a permitted activity in the Protected Ridgelines and Southern Hills Overlay for lines permitted under the minor above ground lines and minor upgrading provisions, with no applicable performance standards.	Accept Durlise for Motorised Helitetos	Insert a statement in Chapter 30 Rules for Utilities prior to the "Activities Table" that spells out that other than the rules in Chapter 30, the only other rules potentially relevant to network utilities are those relating to notable trees, historic heritage, the Southern Hills Overlay and Protected Ridgelines, and include a specific reference to the relevant rules in those chapters. Alternatively, include the relevant rules into Chapter 30. A consequential amendment is required to 16.1 (tent) paragraph) to remove the reference relevant rules in the Protected Ridgelines and Southern Hills Overlay.	
Provision			Rule 28.1 Rule 28.2	te to Chantar 20	Ch30 – Ch40 – Ch	
Submitter		WELL	Chorus	WELL	Spark	
Submission No.		FS3.24	S5.21	FS3.39	S3.10	
Commissioners' Recommendation on submission point		Accept in part				Accept in part
--	---	---	---	---	---	---
Reasons	Table") that clearly sets out which area wide rules apply to avoid any uncertainty.	S3.10 raises similar concerns to concerns to powerco in those raised by powerco in those submission S7.20, which the Council has potentially out of specifically, both submissions relating that further required around the relationship between the utilites rules in chiltes and which of those and the rules and which of those and tho network utilities.	In Powerco's opinion, there is currently significant uncertaintry around the approach to earthworks and vegetation clearance associated with the installation, maintenance and upgrade of network utilities. This is because the introduction to the chapter indicates that the City wide rules for earthworks will apply. However, the assessment citrate for controlled and restricted discretionary activities set out in Chapter 30 indicate the council may impose conditions on earthworks and vegetation clearance, suggesting that these matters will be considered in the context of Chapter 30 network utilities.	Powerco supports S3.10 to the extent that it seeks to clarify that only the City-wide rules for notable trees, historic heritage, the Southem Hills Overlay and Protected Ridgelines (not earthworks or vegetation clearance) will apply to network utilities.	Powerco would also support the inclusion of all rules relevant to network utilities within Chapter 30 in order to provide a consolidated approach to the management of network utilities.	WELL support the concept of the Network Utility Chapter being a complete code.
Decision Sought		Accept in part				Accept
Provision						
Submitter						WELL
Submission No.		т. 8. 1.0				FS3.10

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				Therefore, when other chapters of the plan require consideration, from a permitted activity perspective, the relevant plan sections should be made as clear as possible.	
S5.11	Chorus	Ch30: Application of city-wide rules	Insert a statement in Chapter 30 Rules for Utilities prior to the "Activities Table" that spells out that other than the rules in Chapter 30, the only other trules potentially relevant to network utilities are those relating to notable trees, historic heritage, hazardous substances, the Southern Hills Overlay, and Protected Ridgelines, and include a specific cross reference to the relevant rules in those chapters. Alternatively, include the relevant rules into Chapter 30. A consequential amendment is required to 16.1 (tenth paragraph) to remove the reference to "earthworks" and to signal that there are relevant rules in the Protected Ridgelines and Southern Hills Overlay.	Oppose in part. It is unclear and uncertain as to which area wide rules apply to network utilities. It is stated in 16.1 (tenth paragraph that: The underlying zone objectives, policies and rules do not apply to network utilities, including roads, unless specifically referred to. City wide rules, such as those relating beritage and hazardous substances will still apply. In relation to the City wide rules, there are no applicable earthworks rules, and accordingly the reference to this should be deleted. Provisions have also been included in the Protected Ridgelines and Southerm Hills Overlay area (through Amendment 8 of the Plan Change), but it is not specifically stated in the Utilities chapter that these provisions apply frequest a specific statement be included under Chapter 30 (prior to the "Activities Table") that clearly sets out which area wide rules apply to avoid any uncertainty.	Accept in part
FS3.31	WELL		Accept	WELL consider that it is important for the Network Utility Chapter to operate as much as possible as a complete code. Any provisions within the City Wide Rules must be clearly identified in Chapter 30 of the Plan Change.	Accept in part
FS2.15	Transpower		Reject	Transpower generally supports this request as it would help provide clarity for District Plan users. However, Transpower would not support inclusion of reference to the Protected Ridgeline and Southern Hills Overlay, as it does not consider that these provisions should apply to network utilities.	Accept in part
<i>S</i> 7.20	Powerco	Application of city-wide provisions	Amend Chapter 30 – Rules for Utilities to include all standards relevant to the operation, maintenance, upgrade and development of network utilities to provide a comprehensive suite of district wide provisions specific to network utilities. In addition to the existing provise for the following as a permitted activity in all zones and overlays: and clearance required to minitatin the safe and efficient operation of network	Oppose. Consider all standards relevant to the operation, maintenance, upgrade and development of network utilities should be provided for within Chapter 30 Rules for Utilities. Earthworks and vegetation clearance are typical maintenance activities required to ensure the continued efficient operation of existing network utilities and it is therefore appropriate to provide a set of standards that appropriately recognises the unique nature of these activities in relation to network utilities, and	Reject

Commissioners' Recommendation on submission point		Reject
Reasons	particularly in relation to lineal network utilities that traverse property and zone boundaries and which are typically located within the road reserve. Seek that earthworks and vegetation clearance standards relevant to network utilities to comply with the rules in Chapter 33 Flooding and Fault Band Hazards. Activity table 33.1 requires discretionary activity consent for buildings and structures to be erected within the 1% (1 in 100 year) flood within the 1% (1 in 100 year) flood within the 1% (1 in 100 year) flood extent of the Hutt River, as shown on the planning maps, which is considered to include network utilities. It will not always be practicable to completely avoid locating it within the 1% flood extent of the Hutt River, and that gas pipework can be appropriately locate the areas subject to inundation. Accept that it will be appropriate to incuside the areas at risk from flooding, and that it is appropriate to make some provision for underground network provided they do not affect flood protection structures.	Oppose. While accept the likely activity status of the rules expected to apply to National Grid assets (i.e. not non- complying as initially proposed in the draft Plan Change), the rules are unclear. As a result determining which rule applies to which activity is not easy. In addition, and in light of the new definition of 'line', it is unclear which rules specifically apply to transmission line support structures. Notwithstanding opposition, specifically support the noncomplying activity status for network utilities that do not comply with the permitted activity status for in standard 30.3. Transpower designs and operates its infrastructure to comply with these standards.
Decision Sought	 utilities; Earthworks and trenching undertaken by a network utility operator in the road reserve; Underground gas distribution and transmission pipelines and ancillary above ground equipment not affected by inundation within the 1% (1 in 100 year) flood extent of the Hurt River, as shown on the Planning Maps, provided such works are not located on or within 5m of the Council's flood control structures. 	 Amend the proposed activity status tables to achieve the following activity statuses: Permitted: The operation, maintenance, minor upgrading of transmission lines, upport structures and substations. Restricted discretionary: New transmission lines, transmission line, support structures and substations. Discretionary: New transmission lines, transmission line, support structures and substations. Non-complying: All network utilities which do not comply with the permitted activity standards for radiofrequency.
Provision		Rule 30.1 Activity table
Submitter		Transpower
Submission No.		S4.30

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			upgrading on transmission lines and transmission line support structures.		
			And, to avoid the potential for confusion and dispute, include a clear statement to the effect that the provisions of other chapters do not apply to network utilities (as stated in the last paragraph of Section 16.1).		
S7.21	Powerco	Rule 30.1	Number the activities listed in 30.1 Activities Table for ease of reference.	Support. Seek to number the various rules included in the table for ease of reference.	Reject
FS3.41	WELL		Accept	Numbering the various rules will facilitate District Plan usability	Reject
S7.22	Powerco	Rule 30.1	Retain the following rules in 30.1 Activities Table without modification:	Support.	Accept
			30.1 Activities		
			Removal, maintenance, operation ;		
			The removal of existing network utilities, ind any existing structures		
			The operation and maintenance of existing nutilities		
			The upgrading of all other network utilities, excl		
			electricity and telecommunication lines, an		
			 gas distribution and transmission pipeline pressure exceeding 2000 kilopascals 		
			The removal, operation and maintenance of n		
			utilities and the minor upgrading of electricit telecommunication lines that does not		
			permitted activity standards.		
			The upgrading of network utilities that do not the permitted activity standards. excluding:		
			electricity transmission lines above 110kV:		
			gas distribution and transmission pipeline pressure exceeding 2000 kilopascals		
			Subdivision		
			Subdivision for the purpose of accommodatin network utility		
			Gas distribution and transm		
			Underground gas distribution and transm		
			pipelines at a pressure not exceeding		
			structures or streams, and ancillary equit		
			including regulator stations but not comp stations.		
S7.23	Powerco	Rule 30.1	Retain the advice notes to the Rules in Activity	Support.	Accept
		(Advice Notes)	Table 30.1, which recognise regulations of relevance to undertaking works in close proximity to gas pipelines, as follows:		

Commissioners' Recommendation on submission point		Accept	Accept in part	Accept in part	Accept in part
Reasons		Support. Correctly references all relevant standards that control electricity transmission activities. Support the inclusion in the notes section of words inclusion in the NES applies to network utilities.	Oppose. Concerned about the implications of permitted activity status for utilities adverse effects on flood protection assets. With regard to the National Environmental Standard on Electricity Transmission Activities (NESETA), permitted activity clause 33(5) requires that earthworks related to an existing transmission line must not contribute to slope instability, drainage problems, or flooding of overland flow paths, otherwise earthworks are to be considered a controlled activity. Earthworks in stopbanks can contribute to all of these effects e.g. slope instability, therefore it is appropriate that where existing electricity transmission lines are existing new utilities, the activity status should be controlled. However, for all other network utilities including new utilities, the activity status should be a restricted discretionary activity, so that GWRC retains the ability to seek that an application be refused if the flood hazard risks/effects are inappropriate	Powerco accepts there is a need to protect the integrity of food protection assets and can support a rule requiring restricted discretionary consent for removal, maintenance or upgrades associated with undeground services where they are located in or on a stop-bank.	WELL oppose both submission points. All undergrounding of lines (including removal, maintenance, operation and upgrading) should remain permitted activities as indicated in Table 30.1 regardless of flood protection devices. Restricted discretionary consent for
Decision Sought	Other Relevant Regulations The following two standards are relevant for work in proximity to gas pipelines: • NZS/AS 2885 Pipelines – Gas and Lguid petroleum • NZS/AS 2993 Gas Distribution Network. Upber Hutt City Council is not responsible for enforcing these regulations or standards.	Retain the notes section of Chapter 30 as proposed.	Make any removal, maintenance or upgrading associated with underground services (excluding those associated with works on existing transmission lines) where they are located in or on a stopbank a restricted discretionary activity. Council's discretion should be restricted to the effects of the works on the integrity of the stop-bank. To accord with the NESETA, such works on existing electricity transmission lines in stopbanks should be a controlled activity. Council's control should be restricted to the effects of the works on the integrity of the stop- bank.	Accept	Reject Council rejects the submission points in their entirety. Council to retain undergrounding utilities as a permitted activity in all areas regardless of flood mitigation assets.
Provision		Rule 30.1 – activity table notes	Chapter 30 (Rules for Network Utilities)		
Submitter		Transpower	GWRC	Powerco	WELL
Submission No.		S4.31	ى. ئ	FS1.3	FS3.1

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				underground line works within flood protection assets as sought by the submitter would be onerous and unnecessary as WELL construction standards specify correct backfill compaction requirements (as would most utility operators) and hence will avoid undermining integrity of the flood control asset. A permitted activity standard relating to effects on flood control structures would be more appropriate.	
S1.6	GWRC	Chapter 30 (Rules for Network Utilities)	Make new underground utilities where they are located in or on a stopbank a restricted discretionary activity, not a permitted activity as currently proposed. Council's discretion should be restricted to the effects of the works on the integrity of the stop-bank.	Oppose. Concerned about the implications of permitted activity status for utilities operations where there is potential for adverse effects on flood protection assets.	Reject
FS1.4	Powerco		Accept	Powerco accepts there is a need to protect the integrity of flood protection assets and can support a rule requiring restricted discretionary consent for new underground utilities where they are located in or on a stop-bank.	Reject
S5.12	Chorus	Rule 30.1 (Minor above ground lines)	Retain the permitted provision for "Minor Above Ground Lines".	Support.	Accept
FS3.32	WELL		Accept	The submission point compliments the WELL submission point.	Accept
S2.7	KiwiRail	Rule 30.1 (Removal, maintenance, operation and upgrading)	Retain as notified.	Support.	Accept
80 17 10	KiwiRail	Rule 30.1 (Removal, maintenance, operation and upgrading)	The operation and maintenance of existing network utilities.	Support. Question the reference to 'existing' – new network utilities should also be able to be operated and maintained, with any consent or adverse effects being managed through the consent process to construct the network utility. At the time that consent is sought for construction, the operation and maintenance would not be a permitted activity as the network utility would not be existing. Centainity over whether that would also therefore require consent, or could at a future stage rely on a permitted activity rule would be removed with the removal of the word 'existing'.	Accept
FS2.16	Transpower		Accept	Transpower agrees that the term is unnecessary and may create confusion about when a network utility needed to exist to come within the scope of these rules. Transpower considers that once a network utility exists it should be subject to the permitted activity rules, regardless of	Accept

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				when it was constructed.	
S2.9	KiwiRail	Rule 30.1 (Removal, maintenance, operation and upgrading)	Retain as notified.	Support.	Accept in part
S2.10	KiwiRail	Rule 30.1 (Removal, maintenance, operation and upgrading)	Retain as notified.	Support.	Accept in part
\$2.11	KiwiRail	Rule 30.1 (Removal, maintenance, operation and upgrading)	Retain as notified.	Support but conditional on amendments to the definitions of 'upgrading' and 'maintenance' being clarified so that repair and replacement are clearly not upgrading. Without clarification, there is the potential that activities that would be permitted or controlled in the event the permitted standards are not complied with, become restricted discretionary.	Accept in part
S6.21	WELL	Rule 30.1 (Removal, maintenance, operation and upgrading)	Retain the Rules pertaining to Removal, maintenance, operation and upgrading in PC38.	Support, contingent upon the amendment sought to include additional circuits and conductors being included in the definition for Minor Upgrading.	Accept in part
S2.12	KiwiRail	Rule 30.1 (General)	Aerial crossings necessary for network utilities. located on or within existing bridges and structures or across watercourses streams. and including regulator stations but not compressor stations.	Support in part. The current wording of the rule could be clarified for certainty and consistency with the RMA.	Accept
FS2.17	Transpower		Accept	Transpower supports this change as avoids potential for confusion that the rule only applies to smaller watercourses.	Accept
FS3.4	WELL		Accept	For the reason given by the submitter.	Accept
S2.13	KiwiRail	Rule 30.1 (Underground utilities)	Retain as notified.	Support.	Accept
S6.22	WELL	Rule 30.1 (Radio communication, telecommunicati electricity distribution and transmission)	New and upgraded transformers, substations and switching stations (other than those encased within a cabinet and or those that are pole mounted and located within the Rural Zone or Open Space Zone) distributing electricity and ancillary buildings.	Oppose. Oppose the part of Rule 30.1 that identifies all new and upgraded transformers as a discretionary activity in all zones. The smaller transformers will be voltage. The smaller transformers will be able to meet the permitted activity standards for all zones as specified in 30.4 and 30.5 (note: subject to road reserve area standards being increased to 5m ² as sought below). The definition for Cabinet includes transformers and switchgear, consequently the permitted activity rule for cabinets should also apply to transformers and	Accept in part

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				switch gear that can be located within the cabinet. As currently worded resource consent (discretionary activity) would be required for overhead transformers in all zones. Consider that the definition of "line" in Section 2 of the Electricity Act 1992, which is inherited as the definition of "line" in this plan change, includes overhead transformers, and therefore placement of an overhead transformer (pole mounted) on an overhead line, within the a permitted activity to reflect that overhead lines in these zones are also permitted activities.	
S2.14	KiwiRail	Rule 30.1 (Roading, and traffic and transport structures)	Retain as notified.	Support.	Accept
S3.11	Sopark	Rule 30.1 (Masts, with or without antennas in the Residential and Open Space zones)	Provide as a permitted activity for masts, with or without associated antennas in the Residential and Open Space zones under Rule 30.1. Amend Rule 30.4 to provide for a maximum height of 12m for "Vlasts, antennas, lines and single pole support structures" (whether one or more providers) in the Residential and Open single pole support structures" (whether one or more providers) in the Residential and Open Space Zones. Amend Rule 30.5 to provide for a mast diameter of less than 600mm from 6m in height (whether one or two providers); and antennas attached to masts to be within a horizontal diameter circle of 750mm around Provide a default status of discretionary activity.	Oppose. There is no permitted provision for masts and associated antennas in the Residential and Open Space somes. Masts in these areas up to 12m high and complying with the criteria in Rule 30.14 are a restricted discretionary activity. This is considered unreasonable, particularly given the essential nature of telecommunications, the increasing technical requirement to locate telecommunication masts close to the areas that they serve, and the level of effects. The provisions are significantly more stringent than in the operative plan.	Reject
S5.13	Chorus	Rule 30.1 (Masts, with or without antennas in the Residential and Open Space Dones) Rule 30.4 Rule 30.5	Provide as a permitted activity for masts, with or without associated antennas in the Residential and Open Space zones under Rule 30.1. Amend Rule 30.4 to provide for a maximum height of 12m for "Masts, antennas, lines and single pole support structures" (whether one or more providers) in the Residential and Open Space zones. Amend Rule 30.5 to provide for a mast diameter of less than 600mm from 6m in height (whether one or two providers); and antennas attached to masts to be within a horizontal diameter circle of 750mm around the mast.	Oppose. There is no permitted provision for masts and associated antennas in the Residential and Open Space zones. Masts in these areas up to 12m high and complying with the criteria in Rule 30.14 are a restricted discretionary activity. This is considered unreasonable, particularly given the essential nature of telecommunications, the increasing technical requirement to locate telecommunication masts close to the areas that they serve, and the level of effects. The provisions are significantly more stringent than in the operative plan.	Reject

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			Provide a default status of discretionary activity.		
S3.12	Sopark	Rule 30.1 (Masts, with or without antennas in the and Special Activity zones) Rule 30.5 Rule 30.5	Amend 30.4 Maximum Height of Network Utilities to provide for a maximum height of 20m for one provider and 25m for two providers for masts in the Rural zone. Amend 30.4 Maximum Height of Network Utilities to provide a maximum height of 25m for masts with two or more providers in the Special Activity zone (other than the St Patricks Estate). Amend 30.5 Maximum Size and Diameter of Network Utilities to provide for a mast diameter of 1.5m in the Rural zone (whether one, two or more providers); and provide for a 1.5m mast diameter in the Business and Special Activity zones (except St Patricks Estate) for two or more providers. Amend 30.5 Maximum Size and Diameter of Network Utilities to provide for a horizontal diameter circle of 5m for antennas in the Rural	Oppose. The existing height and size provisions are too restrictive in terms of the level of potential adverse effects, and the necessity of telecommunications facilities. Further, while the policy framework purports to encourage co-location, restrictions whereby a mast with one provider is permitted up to 20m in the Special Activity zone, but the restriction on two or more providers is 12m will not achieve this.	Accept in part
FS3.11	WELL		Accept	WELL support the submissions on Rules 30.1 and 30.2 as they provide greater flexibility for support structure height in the rural zone, as well better enablement of collocating services on the one structure. Both amendments would facilitate WELL's operations in the district.	Accept in part
85.14 2	Chorus	Rule 30.1 (Masts, with or without antennas in the artennas in the Activity Zones) Rule 30.5 Rule 30.5	Amend 30.4 Maximum Height of Network Utilities to provide for a maximum height of 20m for one provider and 25m for two providers for masts in the Rural zone. Amend 30.4 Maximum Height of Network Utilities to provide a maximum height of 25m for masts with two or more providers in the Special Activity zone (other than the St Patricks Estate). Amend 30.5 Maximum Size and Diameter of Network utilities to provide for a mast diameter of 1.5m in the Rural zone (whether one, or two or more providers); and provide for a 1.5m mast diameter in the Business and Special Activity zones (except St Patricks Estate) for two or more providers. Amend 30.5 Maximum Size and Diameter of Network Utilities to provide for a horizontal Network Utilities to provide for a horizontal	Oppose. The existing height and size provisions are considered to be too restrictive in terms of the level of potential adverse effects, and the necessity of telecommunication facilities. Further, while the policy framework purports to encourage the policy framework purports to an ast with one provider is permitted up to 20m in the Special Activity zone, but the restriction on two or more providers is 12m will not achieve this.	Accept in part

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			diameter circle of 5m for antennas in the Rural zone.		
FS3.34	MELL	Rule 30.4	Accept	WELL support the submission point requesting greater height standards in the rural zone for support structures. WELL consider such increases will better enable network efficiency by allowing longer spans, as well as providing greater flexibility to co-locate services on single support structures (i.e., enable separation distances between services).	
83.11	Spark	Rule 30.4	Amend Rule 30.4 to provide for a maximum height of 12m for "Masts, antennas, lines and single pole support structures" (whether one or more providers) in the Residential and Open Space Zones.	Oppose. There is no permitted provision for masts and associated antennas in the Residential and Open Space zones. Masts in these areas up to 12m high and complying with the criteria in Rule 30.14 are a restricted discretionary activity. This is considered unreasonable, particularly given the essential nature of telecommunications, the increasing technical requirement to locate telecommunication masts close to the areas that they serve, and the level of effects. The provisions are significantly more strindent than in the operative plan.	Reject
S4 .32	Transpower	Rule 30.4 (Maximum height of network utilities)	Amend the maximum height standards to specify a specific and appropriate height standard which 'provides for' transmission line support structures.	Oppose. Given the proposed definitions of 'line' and of 'network utility structure' it is unclear what standard wind apply to transmission line support structures. The first height standard in the proposed table specifically refers to 'single pole support structures' and accordingly it could be perceived that any other support structure, that is not attached to a building, would fall under the following standards: under the following standards: tructures located within road reserve, that are not otherwise provided for. The height limits in the standard above are inappropriate for transmission line support structures.	Reject
S5.13	Chorus	Rule 30.4	Amend Rule 30.4 to provide for a maximum height of 12m for "Masts, antennas, lines and single pole support structures" (whether one or more providers) in the Residential and Open Space zones.	Oppose. There is no permitted provision for masts and associated antennas in the Residential and Open Space sones. Masts in these areas up to 12m high and complying with the criteria in Rule 30.14 are a restricted discretionary activity. This is considered unreasonable, particularly given the essential nature of telecommunications, the increasing technical requirement to locate	Reject

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				telecommunication masts close to the areas that they serve, and the level of effects. The provisions are significantly more stringent than in the operative plan.	
FS3.33	WELL	Rule 30.4	Accept	WELL support line height standards in the residential and open space zone, especially in regard to minor overhead lines.	Reject
S6.23	WELL	Rule 30.4 (Maximum height of network utilities)	Retain the standards contained in 30.4.	Support.	Accept
S7.24	Powerco	Rule 30.4 (Maximum height of network utilities)	Retain standard 30.4 as they apply to cabinets and other network utility structures (not otherwise provided for).	Support.	Accept
S3.11	Spark	Rule 30.5	Amend Rule 30.5 to provide for a mast diameter of less than 600mm from 6m in height (whether one or two providers); and antennas attached to masts to be within a horizontal diameter circle of 750mm around the mast. Provide a default status of discretionary activity. Residential and Open Space zones.	Oppose. There is no permitted provision for masts and associated antennas in the Residential and Open Space zones. Masts in these areas up to 12m high and complying with the criteria in Rule 30.14 are a restricted discretionary activity. This is considered unreasonable, particularly given the essential nature of telecommunications, the increasing technical requirement to locate telecommunication masts close to the areas that they serve, and the level of effects. The provisions are significantly more stringent than in the operative plan.	Reject
S5.13	Chorts	Rule 30.1 Rule 30.5 (Rule 30.5 (Rule 30.5 (Masts, with or without antennas in the antennas in the Residential and Open Space zones)	Provide as a permitted activity for masts, with or without associated antennas in the secidential and Open Space zones under Rule 30.1. Amend Rule 30.4 to provide for a maximum height of 12m for "Wasts, antennas, lines and single pole support structures" (whether one or more providers) in the Residential and Open Space zones. Amend Rule 30.5 to provide for a mast diameter of less than 600mm from 6m in height (whether one or two providers); and antennas attached to masts to be within a horizontal diameter circle of 750mm around the mast. Provide a default status of discretionary activity.	Oppose. There is no permitted provision for masts and associated anternas in the Residential and Open Space zones. Masts in these areas up to 12m high and complying with the criteria in Rule 30.14 are a restricted discretionary activity. This is considered unreasonable, particularly given the essential nature of telecommunications, the increasing technical requirement to locate technical requirement to locate telecommunication masts close to the areas that they serve, and the level of effects. The provisions are significantly more stringent than in the operative plan.	Reject
FS3.33	WELL	Rule 30.4	Accept	WELL support line height standards in the residential and open space zone, especially in regard to minor overhead lines.	Reject
S6.24	WELL	Rule 30.5	Amend Standard 30.5 to enable road reserve	Support in part. The size and diameter	Accept in part

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
		(Maximum size and diameter of network utilities)	Cabinets associated with electrical distribution networks to have an engineering envelope up to 5m2 as permitted activities adjacent to all land use zones	standards for Cabinets located within the road reserve are too restrictive. As shown in Attachment C to WELL's submission, electricity distribution cabinets are common elements within the City's road reserves and present the only above ground structures associates with otherwise permitted underground networks. The design and location of the distribution cabinets are thoroughly considered prior to their positioning, thereby ensuring that pedestrian access and safety will not be compromised by the cabinet. In order to be able to support the operation and function of underground distribution lines (permitted activities) a cabinet (containing a transformer and switch gear) requires an area foorprint of no less than $5m^2$ from a technical basis. Support Standard 30.5 <i>in that</i> a 15m ² area for cabinets not located within the road reserve is appropriate.	
S7.25	Powerco	Rule 30.5 (Maximum size and diameter of network utilities)	Retain standard 30.5 as they apply to cabinets and other network utility structures (not otherwise provided for).	Support.	Accept
FS3.42	WELL		Council reject in part the submission point to the extent is seeks retention of the current cabinet area within road reserves. A 5m2 area is being sought by WELL.	WELL seek that a 5m2 permitted activity standard for cabinets within the road reserve is provided due to technical requirements and limitations.	Reject
S2.15	KiwiRail	Rule 30.6 (Separation distance and setback from boundaries)	The following table applies to masts and antenna attached to masts and any cabinet or other network utility structure that is defined as a building that are not located in the road reserve or rail corridor.	Support in part. The standards as worded require that infrastructure in the rail corridor meet the setbacks. Seek that the rule be amended to exclude infrastructure in the rail corridor.	Accept
FS3.5	WELL		Accept the submission. Reference to infrastructure within the rail corridor to be applicable to Rule 30.6 as sought by the submitter.	WELL consider it appropriate for network utility structures not located within the rail corridor have different setback requirements to structures located within the rail corridor.	Accept
S6.25	WELL	Rule 30.6 (Separation distance and setback from boundaries)	Retain the separation and setback exemptions for cabinets and other network utility structures.	Support, in so far that Lines (not being defined as a building) are excluded from the setback provisions.	Accept
S7.26	Powerco	Rule 30.6 (Separation distance and setbacks from boundaries)	Retain Standard 30.6 but amend the introduction to address the anomaly created by the reference to the definition of 'building', which then explicitly excludes network utilities. This could be achieved by amendments along the following lines: <u>30.5 Separation distance and setbacks from</u>	Support in part. Use of the term 'building' in the introduction to Standard 30.6 creates an anomaly as the definition of 'building' explicitly excludes network utilities. Seek to amend the introduction to delete the cross-reference to the definition of 'building' and instead refer to the key parameters of that definition that should be	Accept

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			boundaries No network utilities shall be located within an esplanade or strip. The following table applies to masts and anterna attached to masts and any cabinet or other network utility structure that is <u>ower Em2</u> of a more than 1.2m defined as a building that are not located in the road reserve	used to trigger a need for compliance with the standard. For example, the definition of building applies to any structure over 5m2 in area with a height of more than 1.2m.	
FS3.43	WELL		Accept	The recommended amendment to Rule 30.6 will assist in making the district plan clearer and will avoid interpretation over what structures the rule applies to.	Accept
S4.33	Transpower	Rule 30.6 (Separation distances and setbacks from boundaries) Kule 30.7 (Specific standards for temporary above ground lines)	Exclude transmission lines and their associated support structure from the separation distance and setback standard. Delete section 30.7.	Oppose. The standards are entirely inappropriate to apply to transmission lines. Transmission lines, whether temporary or permanent, commonly cross rivers and therefore will encroach within the riparian setback, without adverse effect. It is unclear what effect applying this setback standard to lines is intended to manage.	Accept in part
S6.26	WELL	Rule 30.7 (Specific standards for temporary above ground lines)	Retain Standard 30.7 as currently worded.	Support.	Accept
S4.34	Transpower	Rule 30.12	Council will restrict its discretion to, and may impose conditions on: Local. national and/or regional benefits derived from the activity. <u>Any constraints arising from technical</u> and operational requirements of the network which may limit measures to avoid. remedy or mitigate environmental effects.	Oppose. As proposed, the matters of discretion fail to allow recognition of the benefits of regionally significant network utilities. Provision for the recognition of benefits is established through the inclusion of Objectu e16.3.1 and Policy 16.4.8 and this should be reflected as a matter of discretion when considering resource consent applications for upgracting. In addition, the matters of discretion must include recognition of the technical and operational effects of the network utility and how this may limit measures to avoid, remedy or mitigate environmental effects.	Accept
AMENDMENT	10: New Chap	ter 30A - Rules fo	r Renewable Energy Generation		
S9.9	Roz Brown	Rule 30A.10	Are we protected enough if the Network Utilities enter into the Biomass energy production?	Request clarification.	Reject
S9.10	Roz Brown	Rule 30A.10	Strengthen wording on ecological inputs.	None given.	Reject

Commissioners' Recommendation on submission point		dequately Accept ated with and providing ns will be	dequately Accept ated with radio providing sions for		Accept	efinition is Accept mission.	sfinition of Accept cludes all rea with a									-
Reasons		Support in part. The definition at provides for antennas associa telecommunications radiocommunications, as well as certainty as to how the provision applied.	Support in part. The definition at provides for antennas associa telecommunications and communications, as well as certainty as to how the provi antenna are to be applied.		Support.	Transpower agrees that the de appropriate and supports the subn	Support in part. Support the de 'building' to the extent that it ext structures of less than 5m2 in ar height of less than 1.2m.									
Decision Sought	- Definitions (Antenna)	Retain the exemption of " <u>(c)</u> any other device not otherwise defined above that is less than 1.5m2 in area" and the other two notes to the definition that make it clear how the area and diameter measurements are to be undertaken.	Retain the exemption of "(c) any other device not otherwise defined above that is less than 1.5m2 in area"; and the two "Notes" to the definition that make it clear how the area and diameter measurements are to be undertaken.	– Definitions (Building)	Retain the current definition for 'Building' in PC38.	Accept	Retain the definition of building to the extent that it excludes all structures of less than 1.2m in height, as follows:	Any structure whether temporary or permanent, movable or immovable, which, in addition to its ordinary and usual meaning, includes the following:	 Any structure of over 5m² in area with a height of more than 1.2m. Any fence or wall with a height of more than 2m. 	 Any retaining wall with a height of more than 1.5m above the finished ground level. 	(i) Which has a capacity of not less than 25,000 litres and is supported directly by the ground.	 (ii) Which has a capacity of 2.000 litres or more and is supported at a height of more than 2.0 metres 	from the base of its structure. (iii) <u>(Which has a capacity of 500</u> liftes or more and is supported at	<u>a height of more than 4.0 metres</u> from the base of its supporting structure.	This definition does not apply to network utilities as defined in this chapter.	
Provision	nts to Chapter 35	Ch35: Antenna	Ch35: Antenna	nts to Chapter 35	Ch35: Building		Ch35: Building									
Submitter	711: Amendme	Spark	Chorus	711: Amendme	WELL	Transpower	Powerco									
Submission No.	AMENDMENT	S3.13	S5.15	AMENDMENT	S6.1	FS2.18	S7.27									

Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
		necessary to give effect to this submission.		
ndme	ents to Chapter 35	- Definitions (Cabinet)		
	Ch35: Cabinet	Means a box-shaped structure which houses radio and telecommunication equipment, electrical equipment, equipment associated with the continued operation of network utilities. which includes single transformers and associated switching qear distributing electricity at a voltage up to, and including, 400444 (10kV).	Support in part. 110kV, not 100kV, is the industry standard.	Accept
endme	ants to Chapter 35	 Definitions (Critical Electricity Line) 		
	Ch35: Critical electricity line	Include a definition for Critical Electricity Line as follows: Critical Electricity Lines means electricity infrastructure which directly convey. Jarge quantities of electricity from point to point. Typically such electricity conveyance is across cities, districts or regions between Grid Exit Points and Zone Substations. For the avoidance of doubt, Critical Electricity Lines includes assets which were part of the national grid but are no longer owned by Transpower, or existing and new assets which berform the function of transmission or sub-transmission but are not owned by Transpower.	Not stated. Consider PC38 is limited in the recognition of electricity infrastructure that is not a component of the Transpower National Grid, or infrastructure that is used to distribute electricity to customers (homes or businesses). Seek a new section (and definition) to schowledge Critical Electricity Lines that are associated with the subtransmission network within and across the City.	Reject
ndme	ents to Chapter 35	- Definitions (Line)		
	Ch35: Line	Retain the definition of "line".	Support.	Accept
wer	Ch35: Line	Means line' as defined in section 5 of the Telecommunications Act 2001 and <u>or</u> in section 2 of the Electricity Act 1992.	Support in part. Seek to clarify that the activity does not need to meet the definition of line under both Acts referenced.	Accept
		Accept	For the reasons stated.	Accept
	Ch35: Line	Retain the definition of "Line".	Support.	Accept
	Ch35: Line	Retain definition for Line.	Support.	Accept
endme	ints to Chapter 35	- Definitions (Maintenance)		
_	Ch35: Maintenance	Retain definition as notified.	Support.	Accept
		Accept	WELL supports these submission points as they reflect the relief sought by WELL in their retention.	Accept
wer	Ch35: Maintenance	Replace the definition of maintenance with one that is clear and which provides for standard National Grid maintenance activities; Or alternatively, delete the definition of maintenance.	Oppose. Too restrictive and does not appropriately 'provide for' standard maintennatic activities that Transpower regularly undertakes on the National Grid. The definition as drafted provides no greater right than what is provided for	Reject

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				through existing use rights. The definition also lacks certainty and it is unclear which of Transpower's maintenance activities are covered by it.	
FS3.26	WELL		Reject	WELL consider that the definition for maintenance is appropriate for defining permitted activities fairly and evenly across all network utility operators. It is unclear from submission S4.36 what specific maintenance that require an altered definition.	Accept
S6.5	WELL	Ch35: Maintenance	Retain definition for Maintenance.	Support.	Accept
S7.28	Powerco	Ch35: Maintenance	Retain the definition of 'maintenance' without modification.	Support.	Accept
AMENDMENT	11: Amendme	ents to Chapter 35 -	- Definitions (Minor above ground line)		
S5.17	Chorus	Ch35: Minor above ground line	Retain the definition of "Minor Above Ground Line".	Support.	Accept
FS3.35	WELL		Accept	Seek retention of the definition.	Accept
S6.6	WELL	Ch35: Minor above ground line	Retain definition for minor above ground line.	Support.	Accept
AMENDMENT	11: Amendme	ents to Chapter 35 -	- Definitions (Minor upgrading)		
S4.37	Transpower	Ch35: Minor upgrading	Means an increase in the carrying capacity, efficiency or security of electricity and telecommunication lines, which utilise the existing or replacement support structures and includes: (7) <u>Hower</u> Support structure and includes: (7) <u>Hower</u> Support structure and includes: (7) <u>Hower</u> Support structures and includes: (7) <u>Hower</u> Support structures and includes: (7) <u>Hower</u> Support structures and or the transmission line corridor. and of the transmission line corridor. and intility structures. Note: The Resource Management (National Environmental Standards for Electricity Transmission Activities Regulations 2009 applies to <u>all</u> transmission lines that were operational, or able to <u>be</u> operated, on <u>14</u> <u>January 2010</u> .	Oppose in part. Sub-section (iv) of the definition requires clarification to ensure that in stating the addition of it means an increase in the number of, rather than the replacement of existing circuits, lines and utility structures, which should be considered a minor upgrade. In addition, note associated with the definition is not clearly worded.	Accept in part
FS3.27	WELL		Accept	As indicated in the corresponding WELL submission point, it is considered that a new collocated cricuit that does not constitute a new line or require additional support structures should be considered a	Accept

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
				minor upgrade based on the minor effects. WELL support submission 4.37 to the extent that only circuits requiring the construction of a new line or support structure are exempt from the minor upgrade definition	
S5.18	Chorus	Ch35: Minor upgrading	Means an increase in the carrying capacity, efficiency or security of electricity and or replacement support structures and includes: 	Oppose in part. Seek amendment to allow for: (a) the provision of up to two fibre optic ines utilising existing or replacement support structures subject to specific controls; and (b) the replacement of an existing support poles in road reserve within 2 metres of the existing pole which it is replacing. These changes are sought to allow for the efficient and effective use of existing infrastructure for activities that have less than minor adverse effects.	Accept in part
FS3.36	WELL		Council rejects in part the submission point to the extent it relates to a 2m relocation area for replacement support structures.	The submission seeks to limit replacement support structures in road reserve to within 2m of the existing pole. The definition already allows for a 5m replacement area which is appropriate. The sought replacement envelope is unnecessary and confusing	Accept in part
S6.7	WELL	Ch35: Minor upgrading	Amend the definition for Minor Upgrading by including a new provision for the addition of circuits and conductors on existing support structures. Amend the definition for Minor Upgrading to allow replacement support structures to be located within 5 meters of the structure being replaced.	Support in part. Minor activities such as the addition of circuits and conductors should be added to the definition of Minor Upgrading as it will enable WELL to provide for growth and future demand for electricity across the city. This would also improve adherence with proposed Policy 16.4.14 as it would provide for co-location, mitigating visual impacts.	Accept in part

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
AMENDMENT	11: Amendme	nts to Chapter 35	 Definitions (Network utility) 		
S2.17	KiwiRail	Ch35: Network utility	Retain definition as notified.	Support.	Accept
FS3.6	WELL		Accept	WELL supports these submission points as they reflect the relief sought by WELL in their retention.	Accept
S4.38	Transpower	utility Network	Means any activity undertaken by a network utility operator as defined in s166 of the RMA, relating to: diffue prevision, operation and maintenance of works-for the conveyancing of the defined in section 2 of the Electricity Act 1992 works' (as defined in section 2 of the Electricity Act 1992) for conveyancing of	Oppose in part. Seek to clarify the reference to the Electricity Act 1992.	Accept
S6.8	WELL	Ch35: Network utilitv	Retain Definition for Network Utility.	Support.	Accept
AMENDMENT	11: Amendme	nts to Chapter 35 -	 Definitions (Network utility structure) 		
S2.18	KiwiRail	Ch35: Network utility structure	Retain definition as notified.	Support.	Accept
FS3.6	WELL		Accept	WELL supports these submission points as they reflect the relief sought by WELL in their retention.	Accept
85 . 33	Transpower	utility structure	Means any structure associated with a network utility and includes, but it not limited to, detecticity line support poles and towers, pipes, valves, meters, requilator stations, support poles and towers, transformers (other than pole mounted transformers), substations (other than overhead substations), compressor stations, pumping stations, anvigational aids, meteorological installations, containers, cabinets and similar structures. It does not include lines, antennas and masts,	Oppose. Does not specifically include transmission line poles or towers / support structures. It is appropriate that these structures are explicitly included in the definition.	Accept in part
S6.9	WELL	Ch35: Network utility structure	Retain Definition for Network Utility Structure.	Support.	Accept
AMENDMENT	11: Amendme	nts to Chapter 35	- Definitions (Noise sensitive activity)		
\$2.19	KimRail	Ch35: Noise sensitive activity	Noise sensitive activity means any. (a) residential activity. (a) visitor accommodation boarding house or other premises where residential accommodation for five or more travellens is offered at a daily tariff or other specified time. or (c) childcare facility.	Oppose. There is no definition proposed, or included within the existing District Plan, in relation to "noise sensitive activities". This is important in an assessment of the impacts of development on network utilities, particularly regionally significant network utilities, as provided for through the subdivision provisions, and the definition will assist in identifying where mitigation may or may not be required. Support the indusion of a definition consistent with that already existing in the Hutt City District Plan.	Reject

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
AMENDMENT	11: Amendme	ints to Chapter 35 -	- Definitions (Regionally significant network ut	ilities)	
S2.20	KiwiRail	Ch35: Regionally significant network utilities	Regionally significant network utilities includes: (f) the Strategic Transport Network, as detailed in Appendix 1.1e the Wellington Regional land Transport Strategy-2010-2040.	Support in part. References to the RLTS may not remain relevant when it is updated.	Reject
S3.15	Spark	Ch35: Regionally significant network utilities	Regionally significant network utilities incudes. Facilities for provision of telecommunications Act 2011 and facilities for the provision of radiocommunications' as defined in section 2(1) of the provision of the actions' as defined in section the provision of radiocommunications Act 1989.	Oppose in part. Seek inclusion of telecommunications and radiocommunications in the definition of "regionally significant network utilities" in recognition of the essential nature of these services, including in times of emergency and natural disaster.	Reject
S4.40	Transpower	Ch35: Regionally significant network utilities	Retain the definition of Regionally Significant Network utilities without amendment.	Support, in particular bullet point two which refers to the National Grid, as defined by the NPSET. It is important that reference to the NPSET remains as the National Grid has not been or is not proposed to be defined separately in the District Plan.	Accept
S5.19	Chorus	Ch35: Regionally significant network utilities	Regionally significant network utilities incudes. - Eaclifies for provision of telecommunications' as defined in section 2 of and facilities for the provision of radiocommunications. and facilities for the provision of radiocommunications. as defined in section 2 of the Radiocommunications. Act 1989. Act 1989.	Oppose in part. Seek amendment to include telecommunications and radiocommunications in the definition in recognition of the essential nature of these services, including in times of emergency and natural disaster.	Reject
S7.29	Powerco	Ch35: Regionally significant network utilities	Retain the definition of regionally significant network utilities to the extent that it includes pipelines for the distribution of natural or manufactured gas, as follows: Regionally significant network utilities includes: • pipelines for the distribution or transmission of natural or manufactured gas or petroleum Adopt any other such relief, including additions, deteitons or consequential amendments necessary as a result of the matters raised in these submission, as necessary to give effect to this submission.	Support in part. Support the proposed definition of 'regionally significant network utilities' to the extent that it applies to its gas distribution network.	Accept
AMENDMENT	11: Amendme	ints to Chapter 35 -	- Definitions (Upgrading)		
S2.21	KiwiRail	Ch35: Upgrading	Upgrading: as it applies to network utilities. upgrading means the replacement, repair, renewal ar improvement or physical works that result in an increase in carrying capacity, operational efficiency, security or safety of	Support in part. The definition of 'upgrading' as proposed includes 'replacement, repair, renewal.' It is unclear whether replacement, repair or renewal activities would be maintenance of upgrading as both definitions cover the	Accept

mmissioners' Recommendation submission point		.ept in part	.ept in part	.ept in part	ept in part
Reasons Co	same activity. Clarity is required. The improvement in carrying capacity and operational efficiency is not something that necessarily results in physical changes, or one that would necessarily be an upgrade. For example, changing engine type or putting an extra carriage on a train are improvements in operational efficiency and carrying capacity respectively. In some instances, some activities will be frect operational activities will be instances, some activities will be etechnically an 'upgrade' where they are in effect operational activities. Needing to get of the provision or the RMA in relation to designations. Would support wording that restricted utility itself, rather than the operation of the utility itself, rather than the operation of the utility.	WELL consider that 'upgrading' can involve Acc renewal of plant and equipment that is above and beyond maintenance (i.e., is not like for like works). Consequently, WELL consider that the proposed definition for upgrading is retained.	Oppose in part. The currently drafted Acc definition as is unclear in application. Bullet point three excludes all activities under Rule 30.1, while bullet point four allows for upgrading of activities listed under that rule that can comply with standards 30.4, 30.5 and 30.6. It is understood that the intention was to provide for upgrading of network utilities under the "Upgrading" provisions, other than telecommunication and electricity lines that are provided for under the "Minor Upgrading" provisions. As currently drafted the rule does not reflect this approach.	Powerco agrees there would be benefit in Acc amending the definition of 'upgrading' to clarify the distinction between network utilities covered by the definition of 'minor upgrading' and those covered by the definition of 'upgrading' it should be clear from any such amendment that the definition of 'upgrading' will apply to Powerco's gas distribution network.	Oppose. Unclear what the difference is Acc between minor upgrading, upgrading and new facilities and unable to determine
Decision Sought	 existing network utilities but excludes: 'maintenance' (as it relates to network utilities.' 'minor upgrading': and Any activity specifically otherwise provided for under Rule 30.1; and Any increase in height or size or change in location. unless such an increase or change is specifically provided for and would comply with the applicable permitted activity standard specified in Rules 30.4, 30.5 and 30.6. 	Reject	Amend the wording of the definition of "upgrading" to make it clear that it applies to all telecommunications and radiocommunications, other than lines which are provided for separately.	Accept in part	Amend the definition of upgrading so that it is clear that all activities which are not operation, maintenance or minor upgrades of
Provision			Ch35: Upgrading		Ch35: Upgrading
Submitter		WELL	раң Х	Powerco	Transpower
Submission No.		FS3.7	S3.16	FS1.7	S4.41

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			substations, are defined as 'upgrading'.	Replacement', 'repair', 'renewal' or an improvement in the operating efficiency of upgrades. In relation to transmission lines, including their support structures, the definition should provide for all upgrading that is not 'minor upgrading' and which is not the establishment of new transmission lines. The definition also needs to allow for upgrades to substations. Concerned with dause (c) which references permitted activity standards. Standards relevant to 'upgrading' activities rules should be listed in the rule itself, and not in both the definition adds to the outping adds to the outping adds to the complexity of the plan change.	
FS3.28	WELL		Accept	WELL support the submission point to the extent that clarification is required that the definition covers substations as well as lines.	Accept in part
S5.20	Chorus	Ch35: Upgrading	Amend the wording of the definition of "Upgrading" to make it clear that it applies to all telecommunications and radiocommunications, other than lines which are provided for separately.	Oppose in part. The definition as currently drafted is unclear and does not reflect the intended approach of providing for all network utilities under the 'Upgrading' provisions rather than telecommunication and electricity lines which are provided for in the 'Minor Upgrading' provisions.	Accept in part
S6.10 Section 32 Ev	WELL aluation Repor	Ch35: Upgrading t	Retain Definition for Upgrading.	Support.	Accept in part
S9.1	Roz Brown	Section 32 – 4.2.5	I feel it would be helpful for the councillors and officers who did not attend the open forum to become aware that the 'community' included interested persons from Upper Hutt Forest and Bird, Upper Hutt Town and Country, Upper Hutt Rural Women, Friends of the Hutt River.	The in-house (library, economic development) knowledge of this PPC 38 does not appear to be a happening thing, what are the communication channels and opportunities in-house?	Accept in part
S9.2	Roz Brown	Section 32 – 6.4	The opportunity for biomass energy generation should include the forestry sector. This comment relates to other sections of the Plan as well. Wondering where DHB and Regional Health input is reflected.		Accept in part
S9.4	Roz Brown	Section 32 - 7.1	Light pollution.		Accept in part
S9.5	Roz Brown	Section 32 – 9.3	Agree with option three [outlined in section 9.3]. With alterations to include light pollution, micro hydro, biomass (forestry), economic advantages.		Accept in part
S9.5	Roz Brown	Section 32 – 11.0	Vast economic and employment effects as we acknowledge that the economy is a wholly owned subsidiary of the environment. Biomass and micro hydro could greatly turn Upper Hutt		Accept in part

Submission No.	Submitter	Provision	Decision Sought	Reasons	Commissioners' Recommendation on submission point
			into an energy secure city.		
S9.6	Roz Brown	Section 32 -	Any influence over RPS to include bird		Accept in part
		Appendix 4 – Objective 9	corridors would be appreciated by karearea et al.		
S9.7	Roz Brown	Section 32 -	A bit light on significant infrastructure - railway,		Accept in part
		Appendix 4 –	SH2.		
		Relevant			
		Definitions			

SUBMISSIONS RECEIVED THAT ARE NOT ON PLAN CHANGE 38, BY PROVISION

Upper Hutt City Council considers that the decisions sought in the following submission points are not on or within the scope of what is proposed through Plan Change 38.

Submission No.	Submitter	Provision	Decision Sought	Reasons
AMENDMENT	1: Amendmer	its to Chapter 16 - Network	c Utilities Issues, Objectives and Policies	
S6.13	MELL	Objective 16.3.2	Clarify the application of Objective 16.2.3 by removing the reference to the 'national grid' in the explanation.	Oppose in part. Consider that the explanation to Objective 16.3.2 is confusing in that it refers to both high voltage electricity transmission ines, as well as the transmission of electricity on the National Grid. WELL own high voltage transmission assets that were previously owned by Transpower. However, as ownership of the asset has changed, acknowledgement in PC38 of high voltage transmission lines is applied interchangeably between High voltage (110kV or greater) transmission lines (which would include WELL assets). Seek amendments to (Transpower Assets). Seek amendments to
				clarify this.

APPENDIX TWO – AMENDED TEXT OF PROPOSED PLAN CHANGE 34

Proposed District Plan Change 34

NETWORK UTILITIES AND RENEWABLE ENERGY GENERATION

Amendments to Chapter 3 - Definitions

AMENDMENT 1 [Chapter 3 Definitions]

Amend definition of Antenna:

Antenna:

any broadcasting and/or telecommunication or radiocommunication apparatus for transmission or reception including the antenna mounting but not any supporting mast or similar structure. This definition includes any satellite dish.

means antenna as defined in the Resource Management (National Environmental Standard for Telecommunications Facilities) Regulations 2008.

An antenna does not include:

- (a) devices used in amateur radio configurations;
- (b) devices used only for television reception; and
- (c) any other device not otherwise defined above that is less than 1.5m² in area.

Notes:

- The mountings of any antenna and any radiofrequency equipment or similar device shall not be included in the measurement of area of diameter of each antenna, provided that the radiofrequency unit or similar device is smaller in area or diameter than the antenna itself.
- Any antenna only need meet the area or diameter measurement, as appropriate to the type of antenna and the measurement is of each individual antenna and is not a cumulative measurement.

AMENDMENT 2 [Chapter 3 Definitions]

Add new definition for Anemometer

<u>Anemometer:</u> means a mast and supporting sensors for the purpose of wind resource measurement. This includes guy wires and various meteorological instruments to be erected at varying heights, including:

- (a) anemometers to measure the average wind speed, wind gust speeds, turbulence intensity and wind shear;
- (b) wind vanes to measure wind direction; and
- (c) other meteorological instruments to measure temperature, air pressure, humidity and rainfall.

AMENDMENT 3 [Chapter 3 Definitions]

Amend definition for Building to reflect proposed network utility amendments

Building:

means any structure or part of a structure, whether temporary or permanent, movable or immovable, but for the purposes of this Plan excludes:

- (a) any fence not exceeding 2 metres in height;
- (b) any retaining wall not exceeding 1.2 metres in height;
- (c) satellite dishes with a diameter not exceeding 0.6m; and antennas 2.5m above the maximum height permitted in the activity area or the rules in Chapter 13 - Utilities.
- (d) decks less than 500mm in height;
- (e) all structures less than 1.2 metres in height;
- (f) all signs, as defined in this Plan.

AMENDMENT 4 [Chapter 3 Definitions]

Add new definition for Cabinet

Cabinet:

means a box-shaped structure which houses radio and telecommunication equipment, electrical equipment, equipment associated with the continued operation of network utilities, which includes single transformers and associated switching gear distributing electricity at a voltage up to, and including, 1010KV.

It also has the same meaning as in the Resource Management (National Environmental Standards for Telecommunication Facilities) Regulations 2008 and means a casing around equipment that is necessary to operate a telecommunication network.

AMENDMENT 5 [Chapter 3 Definitions]

Add new definition for Commercial Scale Renewable Energy Generation Activities:

Commercial Scale Renewable Energy Generation Activities:

means the land, buildings, substations, turbines, structures, underground cabling earthworks, access tracks and roads associated with the generation of electricity from a renewable energy source and the operation of the renewable energy generation activity. It does not include:

- (a) small scale wind turbines of less than 5kW
- (b) community scale renewable energy generation activities
- (c) any cabling required to link the wind energy facility to the point of entry into the electricity network, whether transmission or distribution in nature.

AMENDMENT 6 [Chapter 3 Definitions]

Add new definition for Community Scale Renewable Energy Generation

Community Scale Renewable Energy Generation:

means renewable energy generation for the purpose of supplying electricity to a whole community which is not connected to the distribution network ('off grid'); or to supplying an immediate neighbourhood in an urban area with some export back into the distribution network. Comment [GS1]: S9.3 WELL

AMENDMENT 7 [Chapter 3 Definitions]

Add new definition for Construction and Commissioning Activities

Construction and Commissioning Activities:

in respect of renewable electricity generation activities includes those activities directly involved with the building and operation of a new renewable electricity generation activity. This includes site preparation, earthworks, quarrying, concrete batching, plant construction, road construction and widening, traffic generation, reservoir formation, clearance or inundation of vegetation, but specifically excludes investigative activities such as geological sampling, surveys and geotechnical investigations.

Activities associated with "construction and commissioning" includes rapid and temporary population increases and the associated effects on infrastructure and community facilities; the need to reroute or relocate network utilities and community facilities; the need to construct new infrastructure including the system of electricity conveyance transmission (including substations) required to convey electricity to the distribution network and/or the national grid as provided for in the definition of 'renewable electricity generation activity.

AMENDMENT 8 [Chapter 3 Definitions]

Add new definition for Distribution Network:

Distribution Network: for the purpose of Chapter 14L, has the same meaning as in the National Policy Statement for Renewable Electricity Generation and means a distributor's lines and associated equipment used for the conveyance of electricity on lines other than lines that are part of the national grid.

AMENDMENT 9 [Chapter 3 Definitions]

Add new definition for Distributor

 Distributor:
 for the purpose of Chapter 14L, has the same meaning as in the National Policy Statement for Renewable Electricity Generation and means a business engaged in distribution of electricity.

AMENDMENT 10 [Chapter 3 Definitions]

Amend the definition of Height to reflect proposed network utility amendments

Height:

Maximum Height shall be the perpendicular distance between the lowest ground level at any point and the highest part of the building immediately above that point.

- (a) For the purposes of calculating maximum height, ground level shall be deemed to be:
 - (i) the natural level of the ground or the finished level of the ground as a result of an approved subdivision, and shall not include earthworks which have resulted or will result from work undertaken as part of the construction of any building or development of the site.

- (ii) where the natural ground level has fluctuated over time, as a direct result of nature, the natural ground level will be the level that exists at the time the level has to be ascertained.
- (b) When calculating maximum height the following shall be excluded:
 - antennas, satellite dishes with a diameter not exceeding 0.6m, flagpoles, finials or other similar decorative features where the maximum height is not more than 3.0m above the maximum height permitted for the activity area and this shall also apply to utilities listed in Chapter 13 Utilities.
 - (ii) chimneys, flues and ventilation shafts.
 - (iii) conductors relating lightning rods attached to network utilities.

Maximum Overall Height shall be the vertical distance between the lowest ground level and the highest part of the building immediately above that point.

- (a) For the purposes of calculating maximum overall height, ground level shall be deemed to be the lowest of the following levels:
 - (i) the finished level of the ground as a result of an excavation for building construction works.
 - (ii) the finished level of the ground as a result of any other works.
- (b) When calculating maximum overall height the following shall be excluded:
 - antennas, satellite dishes with a diameter not exceeding 0.6m, flagpoles, finials or other similar decorative features where the maximum height is not more than 3.0m above the maximum height permitted for the activity area and this shall also apply to utilities listed in Chapter 13 - Utilities.
 - (ii) chimneys, flues and ventilation shafts.

(iii) conductors relating lightning rods attached to network utilities.

"Note: Compliance with the New Zealand Electrical Code of Practice for Electrical Safe Distances (NZECP34:2001) is mandatory under the Electricity Act 1992. All activities regulated by NZECP34, including buildings, structures, earthworks and the operation of mobile plant, must comply with that regulation. Activities should be checked for compliance even if they are permitted by the District Plan."

AMENDMENT 11 [Chapter 3 Definitions]

Add new definition for Line

means 'line' as defined in section 5 of the Telecommunications Act 2001 and includes the definition of 'line' or in section 2 of the Electricity Act 1992

AMENDMENT 12 [Chapter 3 Definitions]

New definition of Maintenance

Maintenance:

Line:

<u>ce:</u> as it applies to network utilities, means the replacement, repair or renewal of existing network utilities and where the effects of that utility remain the Comment [GS2]: S6.3 Transpower

Comment [GS3]: S6.4 Transpower

same or similar in character, intensity and scale, and excludes 'minor upgrading' and 'upgrading'.

AMENDMENT 13 [Chapter 3 Definitions]

Amend definition of Mast

Mast:

any mast, pole, tower or similar structure which is fixed to the ground and specifically designed to carry an antenna to facilitate broadcasting, telecommunications and radiocommunications the transmission of telecommunication and radiocommunication signals. This definition excludes any arm supports.

AMENDMENT 14 [Chapter 3 Definitions]

Add new definition for Minor Above Ground Line

Minor Above Ground Line:

means a line that provides an above ground connection to a site, including any connection to a building within that site, from an existing or permitted new above ground line provided that no more than one new support structure is required for that connection.

AMENDMENT 15 [Chapter 3 Definitions]

Add new definition for Minor Upgrading

Minor Upgrading:

means an increase in the carrying capacity, efficiency or security of electricity and telecommunication lines, which utilise the existing or replacement support structures and includes:

- (a) the reconductoring of the line with higher capacity conductors; and
- (b) the resagging of conductors; and
- (c) the addition of longer and more efficient insulators; and
- (d) a support structure replacement within 5m of the support structure that is to be replaced; and
- (e) the addition of earthwires, which may contain telecommunication lines, earthpeaks and lightning rods; and
- (f) the addition of electrical or telecommunication fittings, excluding antenna; and
- (g) support structure replacement in the same location or within the existing alignment of the transmission line corridor; and
- (h) the replacement of existing cross arms, including with cross arms of an alternative design; and
- (i) an increase in support structure height to achieve compliance with the clearance distances specified in NZECP34:2001;
- (i) an increase in the height of replacement poles in the road reserve by a maximum of 1m, for the purpose of achieving road controlling

authority clearance requirements, provided the permitted height in Rule 13.4.2.1 is not exceeded; and

- (k) an increase in voltage of electricity lines from 11kV to no more than <u>33Kv; and</u>
- (I) the addition of a new overhead telecommunication fibre optic line provided that:

(i) the maximum number of fibre optic lines on existing support structures does not exceed two lines;

(ii) the diameter of new fibre optic lines does not exceed 25mm; and (iii) the location of the new fibre optic line is consistent with the

following figure.

Comment [GS4]: S8.6 Chorus

Figure X: Location of new fibre optic line





Except where provided for above, minor upgrading shall not include:

- (a) any increase in the voltage of the line unless the line was originally constructed to operate at the higher voltage but has been operating at a reduced voltage; or
- (b) any increase in any individual wire, cable, or other similar conductor to a diameter that exceeds 35mm; or
- (c) the bundling together of any wire, cable, or other similar conductor so that the bundle exceeds 43mm in diameter; or
- (d) the addition of any new circuits, lines or utility structures, where this results in an increase in the number of new circuits, lines or utility structures except as provided for in (I) above.
- Note: The Resource Management (National Environmental Standards for Electricity Transmission Activities) Regulations 2009 applies to all the existing <u>aNational gGrid</u>, and applies to all Transmission Lines that were operational, or able to be operated, on 14 January 2010.

Comment [GS6]: S6.6 Transpower

Comment [GS8]: S6.6 Transpower

AMENDMENT 16 [Chapter 3 Definitions]

Add new definition for National Grid

National Grid: means 'a National Grid' as defined in the National Policy Statement on Comment [GS9]: S6.7 Transpower Electricity Transmission.

AMENDMENT 17 [Chapter 3 Definitions]

Add new definition for National Grid Corridor

National Grid Corridor:

means the area located within:

- (a) 32m of a 110kV National Grid transmission line;
- (b) 37m of a 220kV National Grid transmission line;
- (c) 39m of a 350kV National Grid transmission line

measured either side of the centreline of the transmission line, as depicted in Diagram 1, below of the definition of National Grid Yard. The measurement of setback distances from National Grid transmission lines shall be taken from the centre line of the National Grid transmission line and the outer edge of any support structure. The centre line at any point is a straight line between the centre points of the two support structures at each end of the span.

Note: The National Grid Corridor does not apply to underground cables or any transmission lines (or sections of line) that are designated.

AMENDMENT 18 [Chapter 3 Definitions]

Add new definition for National Grid Yard

National Grid Yard: means

- (a) the area located 12 metres either side of the centreline of an above ground a National Grid transmission line which is 110kV or greater;
- (b) the area located 12 metres in any direction from the edge of a <u>National Grid support structure which supports a National Grid</u> <u>transmission line which is 110kV or greater</u>

as depicted in Diagram 1 below.

The measurement of setback distances from National Grid transmission lines shall be taken from the centre line of the National Grid transmission line and the outer edge of any support structure. The centre line at any point is a straight line between the centre points of the two support structures at each end of the span.

Note: The National Grid Yard does not apply to underground cables or any transmission lines (or sections of line) that are designated. Comment [GS10]: S6.8 Transpower

Comment [GS11]: S6.9 Transpower





AMENDMENT 19 [Chapter 3 Definitions]

Add new definition for Network Utility

Network Utility:

- means any activity undertaken by a network utility operator as defined in section 166 of the RMA, relating to:
- (a) distribution or transmission by pipeline of natural or manufactured gas, petroleum, biofuel or geothermal energy; or
- (b) telecommunication as defined in section 5 of the Telecommunications Act 2001; or radiocommunications as defined in section (2)(1) of the Radiocommunications Act 1989; or
- (c) the provision, operation and maintenance of works as defined in section 2 of the Electricity Act 1992 for the conveyancing of electricity, as defined in section 2 of the Electricity Act 1992; or

Comment [GS12]: S6.10 Transpwer

- (d) the distribution of water for supply including irrigation; or
- (e) sewerage or drainage reticulation; or
- (f) construction, and operation of roads and railway lines; or
- (g) the operation of an airport as defined by the Airport Authorities Act 1966; or
- (h) the provision of any approach control service within the meaning of the <u>Civil Aviation Act 1990; or</u>
- (i) undertaking a project or work described as a 'network utility operation' by regulations made under the Resource Management Act 1991

and includes:

- (a) lighthouses, navigation aids, beacons, signal and trig stations and natural hazard emergency warning devices;
- (b) meteorological services;
- (c) all associated structures; and
- (d) regionally significant network utilities.

AMENDMENT 20 [Chapter 3 Definitions]

Add new definition for Network Utility Structure

Network Utility Structure:

means any structure associated with a network utility and includes, but is not limited to, pipes, valves, meters, regulator stations, support poles and towers, transformers (other than pole mounted transformers), substations (other than overhead substations), compressor stations, pumping stations, navigational aids, meteorological installations, containers, cabinets and similar built structures. It does not include lines, antennas and masts.

AMENDMENT 21 [Chapter 3 Definitions]

Amend definition of Recession Plane

Recession Plane: a control relating to the receipt of natural light by adjoining sites; this control does not apply to chimneys, antennas, satellite dishes with a diameter not exceeding 0.6m, and flagpoles, and any part of a mast that above a height of 2.5m from ground level has a diameter not exceeding 0.6m.

AMENDMENT 22 [Chapter 3 Definitions]

Add definition for Regionally Significant Network Utilities

Regionally Significant Network Utilities:

means:

- (a) pipelines for the distribution or transmission of natural or manufactured gas or petroleum;
- (b) the National Grid
- (c) facilities for the generation and transmission of electricity where it is supplied to the network, as defined by the Electricity Industry Act 2010;
- (d) the local authority water supply network and water treatment plants;
- (e) the local authority wastewater and stormwater network, systems and wastewater treatment plants; and
- (f) the Strategic Transport Network, as detailed in Appendix 1 to the Wellington Regional Land Transport Strategy 2010-2040

AMENDMENT 23 [Chapter 3 Definitions]

Add definition for Renewable Electricity Generation

Renewable Electricity Generation:

means generation of electricity from solar, wind, hydro, geothermal, biomass, tidal, wave, or ocean current sources.

AMENDMENT 24 [Chapter 3 Definitions]

Add definition for Renewable Electricity Generation Activities

Renewable Electricity Generation Activities:

has the same meaning as under the National Policy Statement for Renewable Electricity Generation and means the construction, operation and maintenance of structures associated with renewable electricity generation. This includes small and community-scale distributed renewable generation activities and the system of electricity conveyance required to convey electricity to the distribution network and/or the national grid and electricity storage technologies associated with renewable electricity.

AMENDMENT 25 [Chapter 3 Definitions]

Add definition for Renewable Energy

Renewable Energy:

as defined in section 2 of the Resource Management Act 1991

AMENDMENT 26 [Chapter 3 Definitions]

Add definition for Sensitive Activity

Sensitive Activity:	as it applies to the National Grid Yard, means the following activities:	Comment [GS13]: S3.5 KiwiRail
	(a) residential <u>buildings</u>	
	(b) the accommodation or care of people, including hospitals	Comment [GS14]: S6.13 Transpower

(c) childcare facilities, kohanga reo, primary, intermediate and secondary schools.

AMENDMENT 27 [Chapter 3 Definitions]

Add definition for Small Scale Renewable Energy Generation

Small Scale Renewable Energy Generation:

means small scale renewable energy generation development for the purpose of using or generating electricity on a particular site (single household or business premise) with or without exporting back into the distribution network.

AMENDMENT 28 [Chapter 3 Definitions]

Add definition for Small Scale Wind Turbines

Small Scale Wind Turbines:

means wind turbines that are capable of generating up to 10kW of electricity.

AMENDMENT 29 [Chapter 3 Definitions]

Add definition for Solar Panel

Solar Panel: means a panel exposed to radiation from the sun, used to heat water or, when mounted with solar cells, to produce electricity directly.
AMENDMENT 30 [Chapter 3 Definitions]

Add definition for Temporary Renewable Energy Assessment and Research Structures:

Temporary Renewable Energy Assessment and Research Structures:

means structures for the investigation, identification and assessment of potential sites and energy sources for renewable electricity generation by existing and prospective generators and includes the following activities:

- (a) erecting an anemometer mast;
- (b) digging test pits, drilling boreholes, constructing investigation drives and removing samples to investigate geological conditions;
- (c) installing instruments into drill holes for monitoring groundwater levels and land movement;
- (d) erecting survey monuments and installing instruments to monitor land movement;
- (e) erecting telemetry stations for the transmission of instrument data;
- (f) installing microseismic stations to measure microseismic activity and ground noise; and
- (g) erection of signs or notices giving warning of danger.

AMENDMENT 31 [Chapter 3 Definitions]

Add definition for Transmission Line

Transmission Line: means the facilities and structures used for, or associated with, the overhead or underground transmission of electricity in the National Grid;

and
(a) includes transmission line support structures, telecommunication
cables, and telecommunication devices to which paragraph (a) applies;
but
(b) does not include an electricity substation.

AMENDMENT 32 [Chapter 3 Definitions]

Add definition for Upgrading

Upgrading:

As it applies to network utilities, upgrading means the replacement, repair, renewal er improvement or physical works that result in an increase in carrying capacity, operational efficiency, security or safety of existing network utilities but excludes:

- (a) 'maintenance' (as it relates to network utilities);
- (b) 'minor upgrading'; and
- (c) any activity specifically provided for under Rules 13.3.1.9 to 13.3.1.41; and any increase in height or size or change in location, unless such increase or change is specifically provided for and would comply with the applicable permitted activity standards specified in 13.3.2.2, 13.3.2.3 or 13.3.2.4.

Comment [GS17]: S6.14

Comment [GS16]: S3.6 KiwiRail

Comment [GS15]: S6.14 Transpower

AMENDMENT 33 [Chapter 3 Definitions]

Delete definition of Utilities

		C 11 ·
I Itilitiac'	comprises tr	$\Delta tollow/lnd$
oundes.		ic ronowing.

- transformation, transmission or distribution of electricity provided by network utility operators or requiring authorities, and private connections to such utilities;
- (b) drainage or wastewater reticulation provided by network utility operators or requiring authorities and private connections to such utilities;
- (c) the distribution of water for supply, including irrigation;
- (d) broadcasting, telecommunication and radio communication facilities including transmitting/receiving devices such as antennas, dishes, wires, insulators, casings, tunnels and associated equipment as well as support structures such as towers, masts and poles and ancillary buildings;
- (e) pipes for the distribution or transmission of petroleum or natural or manufactured gas, and necessary incidental equipment provided by network utility operators or requiring authorities, and private connections to such utilities;
- (f) pipes for the conveyance of irrigation water, or drainage of water or wastewater, and necessary incidental equipment including pumping stations provided by network utility operators or requiring authorities and private connections to such utilities;
- (g) lighthouses, meteorological facilities, navigational aids and beacons including approach control services within the meaning of the Civil Aviation Act 1990;
- (h) roads, footways, cycleways and service lanes;
- (i) street lighting poles, traffic signals and equipment (including surveillance cameras);
- (j) street furniture and traffic signs, parking meters, parking control equipment, including Pay and Display Booths;
- (k) culverts;
- (I) recycling depots; and
- (m) wastewater treatment plants and booster pumping stations.

Amendments to Chapter 11 - Subdivision

AMENDMENT 34 [Chapter 11 Subdivision (11.2.2.2 Matters in which Council Seeks to Control)]

Add new matters (h) and (i) for controlled subdivisions in relation to network utilities

11.2.2.2 Matters in which Council Seeks to Control

The matters over which control is reserved are:

- (a) The design and layout of the subdivision, including the size, shape and position of any lot, any roads or the diversion or alteration to any existing roads, access, passing bays, parking and manoeuvring standards, and any necessary easements;
- (b) The provision of servicing, including water supply, waste water systems, stormwater control and disposal, roads, access, street lighting, telephone and electricity;
- (c) Management of construction effects, including traffic movements, hours of operation and sediment control;
- (d) Provision of esplanade reserves, esplanade strips and access strips;
- (e) Site contamination remediation measures and works;
- (f) Protection of significant sites, including natural, cultural and archaeological sites;
- (g) Avoidance or mitigation of natural hazards;
- (h) The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of and access to regionally significant network utilities (excluding the National Grid) located on or in proximity to the site;
- (i) The outcome of consultation with the owner and operator of regionally significant network utilities (excluding the National Grid) located on or in proximity to the site; and
- (hj) Those matters described in Section 108 and 220 of the Resource Management Act 1991.

Note: Rule 11.2.3 (b) covers subdivision within the National Grid Corridor.

AMENDMENT 35 [Chapter 11 Subdivision (11.2.2.3 Assessment Criteria)]

Amend assessment criteria (a) to include reverse sensitivity consideration effects on regionally significant network utilities

11.2.2.3 Assessment Criteria

The following assessment criteria will be used:

(a) Allotment Design:

- Allotments to have the appropriate net site area and dimensions to enable activities, buildings or structures to be sited to comply with the specified activity area requirements.

- Subdivisions should be designed so as to give areas a strong and positive identity by taking into account characteristics of the area and ensuring that roading patterns, public open space/reserves and community facilities are well integrated.
- Account must be taken of the future development potential of adjoining or adjacent land <u>and any potential reverse sensitivity effects on</u> regionally significant network utilities (excluding the National Grid).
- The roading pattern must take into account the future development pattern of adjacent land.
- Subdivisions should be designed in a manner which recognises and gives due regard to the natural and physical characteristics of the land and adverse effects are avoided, remedied or mitigated.

AMENDMENT 36 [Chapter 11 Subdivision (11.2.3 Restricted Discretionary Activities)] Add new restricted discretionary activities (b) and (c)

11.2.3 Restricted Discretionary Activities

- (a) Any subdivision that does not comply with the standards and terms for controlled activity under Rule 11.2.2.1 in respect of (b) Engineering Design, (c) Contamination and (e) Earthworks.
- (b) Any subdivision located within a National Grid Corridor that complies with the standards and terms under Rule 11.2.3.42.
 - (i) Non-notification

In respect of Rule 11.2.3 (b), public notification of applications for resource consent is precluded. Limited notification will be served on the National Grid Operator as the only affected party under section 95B of the Act.

Note: Rule 11.2.3 (b) (i) prevails over Rule 17.2.2.

(c) Any subdivision located within close proximity to consented and existing renewable energy generation activities.

(i) Non-notification

In respect of Rule 11.2.3 (c), public notification of applications for resource consent is precluded. Limited notification will be served on the renewable energy generation activities' operator as the only affected party under section 95B of the Act.

Note: Rule 11.2.3 (c) (i) prevails over Rule 17.2.2.

AMENDMENT 37 [Chapter 11 Subdivision (11.2.3.1 Matters in which Council has restricted its discretion)]

Renumber and clarify the matters in which Council has restricted its discretion relating to 11.2.3 (a)

11.2.3.1 Matters in which Council has restricted its discretion

(a) Any subdivision that does not comply with the standards and terms for controlled activity under Rule 11.2.2.1 in respect of (b) Engineering Design, (c) Contamination and (e) Earthworks.

Comment [GS18]: S6.16 Transpower

- Any actual or potential adverse effects arising from the proposed noncompliance, and measures to avoid, remedy or mitigate such effects.
- (ii) Amenity Values:

The extent to which any earthworks proposal will affect adversely the visual amenity values of the area, and the extent to which the earthworks will result in unnecessary scarring and be visually prominent.

The effects on the amenity values of neighbouring properties including dust and noise.

The extent to which replanting, rehabilitation works or retaining structures are included as part of the proposal to mitigate adverse effects. Earthworks should not result in the permanent exposure of excavated areas or visually dominant retaining structures when viewed from adjoining properties or public areas, including roads.

(iii) Existing Natural Features and Topography:

The extent to which the proposed earthworks reflect natural landforms, and are sympathetic to the natural topography.

(iv) Historical or Cultural Significance:

The extent to which the proposed earthworks will affect adversely land and features which have historical and cultural significance.

(v) Natural Hazards:

Consideration should be given to those areas prone to erosion, landslip and flooding. Earthworks should not increase the vulnerability of people or their property to such natural hazards. In the Primary and Secondary River Corridors of the Hutt River, consideration should be given to the effects on the flood protection structures.

(vi) Construction Effects:

The extent to which the proposed earthworks have adverse short term and temporary effects on the local environment.

(vii) Engineering Requirements:

The extent of compliance with NZS 4431 1989 (Code of Practice for Earth Fill for Residential Development).

The extent of compliance with Part 2 NZS 4404:2004 (Land Development and Subdivision Engineering).

(viii) Erosion and Sediment Management:

The extent of compliance with the "Erosion and Sediment Control Guidelines for the Wellington Regional 2003" and "Small Earthworks – Erosion and Sediment Control for small sites" by Greater Wellington Regional Council.

(ix) Contaminated Land:

The extent to which works are consistent with the Ministry for the Environment, Contaminated Land Management Guidelines 1 - 5.

(x) Vegetation protection and presence:

The extent to which protection is given and how the safe, continuous presence of vegetation is provided for in the area as shown in Appendix Subdivision 7 by using an appropriate legal mechanism.

(<u>xi</u>) Visual effects of built development on the wider area (Appendix Subdivision 8):

Consideration shall be given to any actual and potential adverse effects of built development in the area identified on Appendix Subdivision 8 on visual amenity of the wider area (i.e. the valley floor and upper Holborn Drive). To assist, an expert assessment shall be undertaken, and the extent to which development controls are placed on identified individual lots as a result of the assessment's findings shall be taken into account.

For the purposes of this rule, built development includes but is not limited to structures of any height such as dwellings and ancillary buildings, decks, fences, walls and retaining walls.

AMENDMENT 38 [Chapter 11 Subdivision (11.2.3.1 Matters in which Council has restricted its discretion)]

Add new matters in which Council has restricted its discretion for Rule 11.2.3(b)

- (b) Any subdivision located within a National Grid Corridor that complies with the standards and terms under Rule 11.2.3.1.
 - (i) the extent to which the design, construction and layout of the subdivision (including landscaping) demonstrates that a suitable building platform(s) for a dwelling can be located outside of the National Grid Yard for each new lot to ensure adverse effects on and from the National Grid and on public health and safety are appropriately avoided, remedied or mitigated;
 - (ii) The provision for the on-going operation, maintenance (including access) and planned upgrade of Transmission Lines;
 - (iii) The risk to the structural integrity of the National Grid;
 - (iv) The extent to which the subdivision design and consequential development will minimise the risk of injury and/or property damage from such lines;
 - (v) The extent to which the subdivision design and consequential development will minimise the potential reverse sensitivity on and amenity and nuisance effects of the transmission asset: and
 - (vi) The extent to which landscaping will impact on the operation, maintenance, upgrade and development (including access) of the National Grid.

Advice Note: Compliance with the New Zealand Electrical Code of Practice for Electrical Safe Distances (NZECP34:2001) is mandatory under the Electricity Act 1992. All activities regulated by NZECP34, including buildings, structures, earthworks and the operation of mobile plant, must comply with that regulation. Activities should be checked for compliance even if they are permitted by the District Plan.

Vegetation to be planted within proximity to Transmission Lines the National Grid Yard as shown on the planning maps should be selected and/or managed to ensure that it will not result in that vegetation breaching the Electricity (Hazards from Trees) Regulations 2003 or prevent access to support structures. To discuss works, including tree planting **near** any

Comment [GS19]: S6.18 Transpower

Comment [GS20]: S6.18 Transpower

Comment [GS21]: S6.18 Transpower



Amendments to Chapter 13 – Network Utilities, including the National Grid

AMENDMENT 42 [Chapter 13 Utilities]

Delete current Chapter 13 Utilities in its entirety.

AMENDMENT 43 [new Chapter 13 Network Utilities, including the National Grid (Title and Introduction)]

Add a new Title and Introduction

13 Network Utilities, including the National Grid

Introduction

This chapter outlines the provisions of the District Plan that relate to network utilities within Lower Hutt City, including the National Grid. Network utilities provide the infrastructure which enables a community to undertake its everyday activities and functions and allows people to provide for their social and economic wellbeing, and their health and safety. Network utilities which are managed through this Chapter include those defined by way of section 166 of the Resource Management Act 1991.

The City has a range of important network utilities that serve an important function locally, regionally and nationally, some of which are critical and life-supporting. The City is traversed by State Highway 2, linked to Porirua by State Highway 58, contains Transpower's Hayward's Substation which is the northern end of the DC link with the Benmore substation in the South Island, the National Grid, the Regional Wellington – Wairarapa railway line and the Hutt Valley and Melling railway lines.

The Regional Policy Statement for the Wellington Region recognises the importance of regionally significant infrastructure within the Region, as forming part of national or regional networks that enable communities to provide for their social, economic and cultural wellbeing and their health and safety. There are a number of network utilities within Lower Hutt City that are identified as being regionally significant infrastructure in the Regional Policy Statement. The Regional Policy Statement requires that the benefits of such regionally significant infrastructure be recognised and protected in the District Plan.

The Council is required to give effect to any National Policy Statement. The National Policy Statement on Electricity Transmission came into force in 2008 and applies to effects on and effects of the transmission network National Grid. The National Policy Statement on Electricity Transmission's objective is to recognise the national significance of the electricity transmission network by facilitating the operation, maintenance and upgrade of the National Grid and the establishment of new transmission resources to meet the needs of present and future generations while: managing the adverse environmental effects of the network; and managing the adverse effects of other activities on the network.

There are many providers of network utilities within Lower Hutt City including the Council, Crown agencies, the Greater Wellington Regional Council, State Owned Enterprises, trading enterprises and private companies. The Council is in itself a major provider of network utilities and services, supplying water, sewage and stormwater reticulation, waste disposal and roads.

Comment [GS25]: S6.21 Transpower

Other utilities that are managed through this Chapter because of their nature and function are lighthouses, navigation aids, beacons, signal stations and natural hazard emergency warning devices and meteorological services¹. These other utilities are owned and / or operated by Maritime New Zealand, local authorities, or the Meteorological Service in order to provide for the health, safety and wellbeing of the local community, region and nation.

The successful functioning of the City depends on network utilities. It is therefore very important that construction, maintenance, upgrade and operation of these services be effectively provided for, technical and geographical constraints on the operation of network utilities are acknowledged, and that the benefits that derive from them are adequately recognised. Network utilities can be vulnerable to reverse sensitivity effects when new activities (that are sensitive to the effects of the existing network utility) are established nearby, leading to constraints on the operation of the network utility. However, network utilities can also have adverse effects resulting from their construction, operation or associated maintenance activities.

For example, network utilities may typically include buildings, poles, overhead wires, pylons, pipes or antennas, which may have an adverse visual impact depending on their location and proximity to other land use activities. The installation and upgrading of network utilities will also typically involve earthworks. However, network utilities may also involve few structures and have limited visual impact, such as underground powerelectricity and telecommunication lines. Network utilities are also often seen as a necessary and normal part of the environment, such as a road.

Comment [GS26]: S6.21 Transpower

The network utility rules apply where network utility operators do not hold a designation for their activities under the designation procedures of the Act. They may, however, also be used by Council to help assess any notices of new requirement for new designations.

The provisions in this Chapter apply to network utilities throughout all zones of the City. The underlying zone objectives, policies and rules do not apply to network utilities, including roads, unless specifically referred to. City wide rules, such as those relating to historic heritage, notable trees, earthworks and hazardous substances will still apply. Under Rule 14A (a), network utilities that are located in the road reserve are subject to the provisions of the activity area where the road reserve is located. Where the road reserve is between two different activity areas, the centre line of the road reserve will become the boundary between such activity areas.

AMENDMENT 44 [new Chapter 13 Network Utilities, including the National Grid (13.1 Issues, Objectives and Policies)]

Add new Issue, Objective, Policies and Explanation and Reasons relating to "Regionally Significant Network Utilities" as Section 13.1.1

13.1 Issues, Objectives and Policies

13.1.1 **Regionally Significant Network Utilities**

¹ The Meteorological Service is a requiring authority for its network operation of a system comprising telecommunication links to permit telecommunication and radiocommunication. Therefore, these aspects of meteorological service activities and facilities are network utilities.

The benefits of regionally significant network utilities to the City, region and nation need to be recognised and protected. **Objective** To recognise and protect the benefits of regionally significant network utilities. Comment [GS27]: S6.22Transpower Policies To identify regionally significant network utilities within the City on Council (a) planning maps, as practicable. To recognise the national, regional and local benefits of regionally significant (b) network utilities **Explanation and Reasons** The importance of and benefits arising from regionally significant network utilities within the City needs to be identified and recognised and the Regional Policy Statement needs to be given effect to. The objective and supporting policies are focused on recognising the benefits that these regionally significant network utilities have locally, regionally and nationally. Policy (a) requires the Council to identify regionally significant network utilities within the City on its planning maps, as practicable. The majority of any new and Comment [GS28]: S7.6 Spark, S8.9 Chorus extensions to existing regionally significant network utilities are expected to be identified on Council planning maps by network utility operators through a notice of requirement for designation process. In the case of the National Grid, which is not designated, this network will be specifically recognised and mapped, as required by the National Policy Statement on Electricity Transmission. Due to the scale of the planning maps and the extensive nature of some regionally significant network utilities, it is however not feasible to identify all regionally significant network utilities on Council planning maps, particularly the local gas distribution lines. Policy (b) recognises that regionally significant network utilities provide benefits within the City, as well as regionally and nationally. These benefits need to be protected and considered in respect of any matter relating to regionally significant network utilities. Some of these benefits are: (i) That people and goods can travel to, and from and around the City and Region efficiently and safely; That community well-being and public health and safety is maintained through (ii) the provision of essential services including supply of potable water, the

(iii) People have access to electricity and gas to meet their needs.

emergency services;

collection and transfer of sewage and stormwater, and the provision of

lssue

AMENDMENT 45 [new Chapter 13 Network Utilities, including the National Grid (13.1 Issues, Objectives and Policies)]

Add new Issue, Objective, Policies and Explanation and Reasons relating to "Managing Adverse Effects, including Reverse Sensitivity Effects, on Regionally Significant Network Utilities" as Section 13.1.2

13.1.2 Managing Adverse Effects, including Reverse Sensitivity Effects, on Regionally Significant Network Utilities

<u>Issue</u>

Inappropriate subdivision, use and development in the vicinity of regionally significant network utilities may lead to adverse effects including reverse sensitivity effects that have the potential to impact upon the effective and efficient operation, maintenance, upgrading and development of such utilities.

Objective

To ensure the operation, maintenance, upgrading and development of regionally significant network utilities is not unreasonably compromised by other activities.

Policies

- (a) To avoid, or as appropriate, remedy or mitigate, the potential for any adverse effects, including reverse sensitivity effects on regionally significant network utilities from incompatible new subdivision, use and development occurring under, over, or adjacent to regionally significant network utilities.
- (b) To ensure the safe and efficient maintenance, operation, upgrade and development of the National Grid by avoiding the incompatible establishment of or changes to sensitive activities and incompatible buildings and structures within a defined National Grid Yard.

Explanation and Reasons

Inappropriate subdivision, use and development may result in adverse effects on regionally significant network utilities and / or restrict access to such network utilities including the ability to undertake maintenance or upgrade work. Reverse sensitivity can occur when sensitive activities locate near to or intensify by existing regionally significant network utilities and seek to or constrain the operation or expansion of these utilities. This may mean that the local, regional and national benefits of those regionally significant network utilities may be compromised. The City has a lot of well-established regionally significant network utilities located in close proximity to existing land use activities. The Council is concerned with new more intensive land use activities may lead to adverse effects, including reverse sensitivity effects on those utilities.

Policy (a) requires that any potential adverse effects, including reverse sensitivity effects on regionally significant network utilities are appropriately managed, with priority given to avoiding adverse effects, where practicable, on those utilities. The location of inappropriate new subdivision, use or development in proximity to existing regionally significant network utilities has the potential to compromise the Comment [GS29]: HS6.23 Transpower efficient operation and use of the network utility including by restricting access and result in the benefits of that network utility being reduced. In addition, the safety and amenity values of the community may be adversely affected by locating in too close proximity to regionally significant network utilities. The potential for reverse sensitivity effects may arise when the pattern and density of land use activities changes through the subdivision or rezoning of land. At the time of rezoning, the Council will seek to introduce new provisions to manage those potential reverse sensitivity effects on existing or designated regionally significant network utilities. Any applications for subdivision that involve potential intensification located in proximity to regionally significant network utilities will require assessment in terms of the potential effects on those utilities as well as consultation with the relevant network utility operator.

Policy (b) recognises the importance of the National Grid and seeks to protect the continued operation, <u>maintenance</u>, and <u>upgrade</u> and <u>functioning</u> of that network. The policy provides for the establishment of a National Grid Yard within which sensitive activities and incompatible buildings and structures will be avoided. The management of buildings and structures within a National Grid Yard is aimed at:

- 1. <u>Maintaining access to the National Grid for its on-going operation</u>, <u>maintenance</u>, <u>upgrading and development</u>;
- 2. Mitigating safety risks for occupants and users of properties:
- 3. Protecting the structural integrity of transmission lines; and
- 4. <u>Maintaining the opportunity to further optimise existing National Grid lines in</u> <u>the future.</u>

ensuring that these do not hinder required access to the network for its on-going operation and maintenance, which is a This is a matter of national significance under the National Policy Statement for on Electricity Transmission. This Chapter contains specific rules that apply to the use and development of land within the National Grid Yard throughout the District Plan. Chapter 11 Subdivision, contains specific rules that apply to subdivision within the National Grid Corridor.

The on-going operation, upgrade and maintenance of the existing National Grid is provided for by the Resource Management (National Environmental Standards for Electricity Transmission Activities) Regulations 2009. These regulations specify that existing electricity transmission activities are permitted, subject to terms and conditions to ensure that these activities do not have significant adverse effects. The standards also specify resource consent requirements for electricity transmission activities that do not meet the terms and conditions for permitted activities.

AMENDMENT 46 [new Chapter 13 Network Utilities, including the National Grid (13.1 Issues, Objectives and Policies)]

Add new Issue, Objective, Policies and Explanation and Reasons relating to "Recognising and Providing for Network Utilities" as Section 13.1.3

13.1.3 Recognising and Providing for Network Utilities

Issue

The key role that network utilities play and the benefits they have needs to be recognised and the technical and operational requirements of the network utility concerned should not be unreasonably restricted. Failing to adequately provide for network utilities may result in the desired level of well-being and quality of life not being achieved within the City.

Comment [GS30]: S6.23 Transpower

Objective

To recognise and provide for the sustainable, secure and efficient use, operation and development of network utilities within the City

Policies

(a) To recognise and provide for the:

- (i) need for new and the maintenance and upgrading of existing network utilities;
- (ii) technical and operational requirements and constraints of network utilities in assessing their location, design, development, construction and appearance; and
- (iii) benefits that network utilities provide to the economic, social and cultural functioning of the City.
- (b) To enable the efficient construction, installation, operation, upgrading and maintenance of network utilities.
- (c) To ensure that the provision and operation of utilities that cross jurisdictional boundaries is managed in an integrated manner.
- (d) To encourage the appropriate use of designations for new network utilities and extensions to existing network utilities that are not designated.

Explanation and Reasons

It is important that the benefits of all network utilities, including those that are not identified as regionally significant, are recognised and provided for. Network utilities provide essential services to people's homes and businesses, such as water, transport means, electricity, gas, radiocommunications and telecommunications, and are critical for the effective functioning and liveability of the City. Failing to adequately provide for network utilities may result in the desired level of well-being and quality of life not being achieved within the City.

Policy (a) recognises that the provision of new and the upgrading of existing network utilities is necessary to meet the needs of City, both now and into the future. In considering any proposals for new or upgrades to existing network utilities, the technical and operational requirements that may constrain where and how they can locate and be designed need to be recognised. In some cases, some level of adverse effects may need to be accepted to recognise the necessity for some network utilities and meet their operational requirements. This policy also recognises the benefits that all network utilities have.

Policy (b) acknowledges the important role that network utilities have in providing for the wellbeing of the City's community. Network utilities form an essential part of the efficient functioning of the City and their maintenance and development allows their benefits to be realised. There are a range of network utilities that enable communities to undertake everyday activities and functions and provide essential services to people's homes and businesses. It is therefore important that the District Plan provides for network utilities to be constructed, installed, operated, upgraded and maintained. Policy (c) reflects that by their nature, many network utilities cross jurisdictional boundaries between councils. Cross boundary issues can result for network utility providers and for the community, particularly where different councils have different rules or processes for how they recognise and provide for network utilities and manage their effects. It is important that councils work together in an integrated manner both when developing plan provisions and when dealing with proposals for new or upgrades to existing network utilities.

Policy (d) is focussed on encouraging network utility operators, particularly those who operate regionally significant network utilities, to use the notice of requirement for designation process when they seek to develop new or extend existing network utilities. This is particularly encouraged for operators where such new or extended network utilities involve restrictions on the use of privately owned land and may require land acquisition. It is recognised that not all network utility operators use designations, particularly those that do not operate linear infrastructure.

AMENDMENT 47 [new Chapter 13 Network Utilities, including the National Grid (13.1 Issues,

Objectives and Policies)]

Add new Issue, Objective, Policies and Explanation and Reasons relating to "Managing Environmental Effects" as Section 13.1.4

13.1.4 Managing Environmental Effects

lssue

The actual and potential adverse environmental effects arising from network utilities need to be managed.

Objective

To manage any adverse effects on the environment resulting from the design, location, operation, upgrading and maintenance of network utilities.

Policies

- (a) To ensure that network utilities are designed, located, developed, constructed, upgraded, operated and maintained to avoid, remedy or mitigate any actual or potential adverse effects on the environment.
- (b) To manage effects on health and safety by ensureing network utilities, in particular those utilities emitting electric and magnetic fields, are designed, located, upgraded, operated and maintained to comply with relevant national environmental standards and to meet other nationally recognised standards and guidelines.
- (c) To enable the co-location or multiple use of network utilities where this is efficient, technically feasible and practicable and assists with avoiding, remedying or mitigating adverse effects on the environment.
- (d) To require the underground placement of new network utilities unless
 - (i) there are natural or physical features or structures, or technological and operational constraints that makes underground placement impractical or unreasonable;

Comment [GS31]: S10.9 Powerco

Comment [GS32]: S6.26 Transpower

- (ii) they are of a temporary nature and required for emergency purposes or critical events; and
- (iii) they are of a nature that they can only operate aboveground.
- (e) To encourage the use of roads as network utility corridors in accordance with the National Code of Practice for Utility Operators' Access to Transport Corridors.
- (f) To encourage network utility providers to consult with local communities, <u>landowners and the Regional Council (where relevant)</u> on the appropriate placement, location and design of new network utilities.

Explanation and Reasons

The issue and supporting objective recognise that the construction, operation, upgrading and/or maintenance of network utilities can have adverse effects and adversely affect the amenity of areas of the City, as a result of noise, emissions, and visual dominance, for example. Some network utilities are relatively large, visually prominent and capable of generating significant adverse effects on the surrounding environment. Such network utilities may also have actual or perceived adverse effects on public health and safety. Adverse effects may only occur at the time of construction or installation of the utility, but in some instances may continue throughout its operation or during maintenance and/or upgrade works. For new linear network utilities, adverse effects are often best able to be mitigated through the route selection process. However, in some cases, it might not be entirely possible to avoid, remedy or mitigate all adverse effects associated with a network utility due to their technical and operational constraints, meaning there will may be some level of residual adverse effect on the surrounding environment that requires mitigation. In such circumstances, there is a need to carefully consider both the benefits the utility will provide and the significance of the adverse effects on the surrounding environment.

Policy (a) recognises the importance of managing the design, location, operation, upgrading, construction, operation and maintenance of network utilities and requires that any potential adverse effects arising from network utilities are avoided, remedied or mitigated. It is acknowledged that it is not always possible to do so. and that there may be some level of residual effect, due to the technical and operational requirements of network utilities, as reflected through Policy 13.1.3(a)(ii). This policy is sufficiently broad to recognise that there are a range of different network utilities with different potential adverse effects on the environment. For instance, above ground network utilities can have adverse effects including visual, noise, traffic, odour and amenity, depending on their size, location, frequency and their scale in comparison with the character of a particular environment. For instance, a different activity status and different performance standards apply to some network utilities in the Historic Residential, Landscape Protection Residential, Recreation and Rural Residential Zones and the Coastal Environment identified as Significant Natural Resource 9, to reflect that these zones have special environments that are more vulnerable to adverse effects and associated loss of amenity.

Policy (b) recognises that some network utilities may adversely affect health and safety. For example, telecommunication facilities generate radio frequency emissions which may have detrimental effects on health. Any potential health effects

Comment [GS33]: S2.3 GWRC

Comment [GS34]: S6.28 Transpower

Comment [GS36]: S6.28 Transpower

Comment [GS35]: S8.12 Chorus

Comment [GS37]: Clause 20 – duplicated word

Comment [GS38]: S7.9, S8.12, S9.14, HFS2.10 Spark, Chorus, WELL, Powerco arising from radiofrequency emissions are addressed by Regulation 4 of the Resource Management (National Environmental Standards for Telecommunication Facilities) Regulations 2008. Electricity transmission/distribution can generate electromagnetic fields (EMF) which may be a risk to health and also generates the risk of electrocution. Electricity transmission/distribution can present a risk to public health and safety, primarily through the risk of electrocution from direct contact with conductors or as a result of a flashover. The National Policy Statement on Electricity Transmission requires that the exposures be limited to the guidelines of the International Commission on Non-Ionising Radiation Protection (ICNIRP) to prevent the potential for public health effects. Other possible health and safety risks are accidental spillage or leakage of hazardous substances from gas or petroleum pipelines, explosions from gas or petroleum pipelines; accidental overflow from sewage pump stations, and flooding from damaged/inoperative stormwater systems. Chemicals used in conjunction with some network utilities, such as water treatment plants for example, also pose a risk if an accidental spill occurs. There are also a number of national and international standards and guidelines addressing health and safety matters that are external to the District Plan but that must be complied with, including the Resource Management (National Environmental Standards for Telecommunication Facilities) Regulations 2008, and the New Zealand Electrical Code of Practice. and tThe International Commission on Non-Ionising Radiation Protection (ICNIRP) Guidelines provides best practice guidance.

Policy (c) recognises that the co-location and co-siting of network utilities may provide environmental benefits in terms of reduced visual impacts and consolidation of network utilities in existing areas thereby reducing adverse effects on amenity by reducing the need for more network utility structures. While co-location is encouraged it needs to be understood that technical requirements will generally mean that associated structures may need to be taller or bulkier to avoid interference between the two or more providers, such as radio-frequency bands. It is also recognised that co-location is not always possible due to operational issues such as radiofrequency interference, electrical interference, lease arrangements and structural capacity.

Policy (d) requires the underground placement of new network utilities unless particular circumstances apply. The adverse visual effects of certain network utilities can often be managed by putting the services underground. With some exceptions. **This** is the required approach for those network utilities, such as those with cables that can be located underground. For those network utility structures that need to be located aboveground, particular attention should be given to their design, location and minimising of any adverse visual effects as outlined in Policy (a). This can be achieved in a number of ways including, where practical, through screening, careful placement, size and appearance and applying different activity status. In particular, the underground placement of electricity and telecommunications lines is required in most circumstances by only providing for aboveground lines in particular defined situations, such as for customer connections, and through different activity status. New above ground lines and their associated supporting structures in areas that do not have existing above ground lines are generally considered to be unacceptable within the City, except in those areas that can visually absorb new aboveground lines, such as the Rural Zone where they are permitted. However it is recognised that particular consideration needs to be given to the efficient use of resources and that there are situations where placing lines underground is, or may be, impracticable or unreasonable.



Comment [GS41]: S10.14 Powerco

Comment [GS42]: S8.12 Chorus

Comment [GS43]: S8.12 Chorus

Policy (e) promotes the use of the road corridor for the location of network utilities, in line with the National Code of Practice for Utility Operators' Access to Transport Corridors. Locating network utilities in the road can assist to minimise the adverse effects of network utilities on amenity and other values as these locations generally have a range of existing network utilities and are less sensitive to new network utilities. However, the effects of these activities require some management to ensure conflicts with the primary function of the road corridor and with each other are avoided.

Policy (f) encourages network utility operators to engage with the local community when they are considering the location, placement and design of new network utilities. In some cases, engaging early with the community about a proposed new network utility may result in an alternative more appropriate location to be identified that both meets the needs of the network utility operator and addresses any concerns that the community may have. In encouraging consultation, the Council recognises that it cannot require network utility operators to consult on permitted activities.

AMENDMENT 48 [new Chapter 13 Network Utilities, including the National Grid (13.2 How to Use the Network Utilities Rules)]

Add a new Section 13.2 How to Use the Network Utilities Rules

<u>13.2</u> How to Use the Network Utility Rules

The following is an advice note on how to navigate and use the suite of Rules, Standards and Terms and Conditions for this Chapter in the District Plan. This Chapter applies to all network utilities as defined in Chapter 3 (Definitions) and with the exception of rules in General Chapters 14A – 14L, these rules override all zone rules.

- (1) Before using the Rules, check which Activity Area (zone) the site(s) which is intended to be used for network utility activities is located in. The District Plan Activity Areas are available to view on the Council's website (via the District Plan Chapters and Maps or through the interactive GIS viewer) as well as hard copies at all libraries and the administration building.
- (2) Once the Activity Area has been identified, check the rules in table 13.3.1 to find a description of which activity you want to carry out. Make sure the Activity Area of your site(s) is listed next to the rule otherwise it will not apply.
- (3) Each rule can have associated standards and matters of control or discretion associated with it.
- (4) Section 13.3.2 contains the standards. Where a rule has a standard(s) associated with it, that standard(s) must be complied with for the activity status to apply. If the activity cannot comply with the associated standard(s), it will have a different status and may need a land use consent.
- (5) Sections 13.3.3, 13.3.4 and 13.3.5 contain matters of control and discretion. If the rule has a status of Controlled, Restricted Discretionary or Discretionary (i.e.: a land use consent is required), Matters of Control or Discretion may apply. Where a rule has matters associated with it, the Council will use them to assess the consent application. Chapter 17 of the District Plan outlines what information needs to be submitted with any land use consent application.

(6) Section 13.4 contains provisions for activities within the National Grid and has its own set of rules, standards and terms and conditions. Use this section if your activity is going to be within the National Grid Corridor or Yard (defined in Chapter 3 of the District Plan). AMENDMENT 49 [new Chapter 13 Network Utilities, including the National Grid (13.3 Rules – Network Utilities and 13.3.1 Activity Status)] Add new Sections 13.3 Rules – Network Utilities and 13.3.1 Activity Status **Rules – Network Utilities**

<u>13.3</u>

Activity Status 13.3.1

<u>Rule</u> Number	Rule	Activity Area	<u>Status</u>	<u>Standards</u>	<u>Matters of Control</u> or Discretion	
<u>Removal, N</u>	Aaintenance and Upgrading					
13.3.1.1	The removal of existing network utilities. including any existing associated structures.	AI	Permitted	Earthworks: 13.3.2.5.1 Vegetation: 13.3.2.6 Noise: 13.3.2.7		
<u>13.3.1.2</u>	The operation and maintenance of existing network utilities.	AI	Permitted	Earthworks: 13.3.2.5.1 Vegetation: 13.3.2.6 Noise: 13.3.2.7		Comment [GS44]: S3.14 KiwiRail
<u>13.3.1.3</u>	The minor upgrading of existing electricity and telecommunication lines and support structures.	AI	Permitted	Earthworks: 13.3.2.5 Vegetation: 13.3.2.6 Noise: 13.3.2.7		Comment [GS45]: S1.10 NZTA, FS1.8 Transpower Comment [GS46]: S6.30 Transpower
13.3.1.4	 The upgrading of existing network utilities. excluding: Electricity and telecommunication lines; Gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals. 	AII	Permitted	Health and Safety: 13.3.2.1 Earthworks: 13.3.2.5 Vegetation: 13.3.2.6 Noise: 13.3.2.7		
<u>13.3.1.5</u>	The removal, operation and maintenance of network utilities and the minor upgrading of electricity and telecommunication lines that does not meet permitted activity standards.	AII	Controlled		<u>13.3.3 (e), 13.3.3 (h)</u>	

ber	Rule	Activity Area	<u>Status</u>	<u>Standards</u>	<u>Matters of Control</u> or Discretion	
1.6	 The upgrading of network utilities. excluding: Electricity transmission lines above 110KV: and Gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals that does not meet permitted activity standards. 	<u>AI</u>	Restricted Discretionary	Health and Safety: 13.3.2.1	13.3.4 (a), 13.3.4 (r), 13.3.4 (s), <u>13.3.4(u),</u> 13.3.4(v)	Comment [GS47]: S6.44 Transpower
<u>1.7</u>	The upgrading of:• Electricity transmission lines above 110 kVand above; and• Gas distribution and transmission pipelinesat a pressure exceeding 2000 kilopascals.	<u>AI</u>	<u>Restricted</u> <u>Discretionary</u>	Health and Safety: 13.3.2.1	13.3.4 (b), 13.3.4 (p), 13.3.4 (r), 13.3.4 (s); 13.3.4(u), 13.3.4(v)	Comment [GS49]: S6.44 Transpower Comment [GS48]: S6.31 Transpower
divisio						
1.8	Subdivision for the purpose of accommodating any network utility.	<u>AII</u>	Controlled		13.3.3 (a), 13.3.3 (b), 13.3.3 (c), 13.3.3 (e), 13.3.3 (f), 13.3.3 (g), 13.3.3 (l)	
eral						
1.9	Cabinets and other network utility structures not otherwise listed in this table.	All, excluding <u>Historic</u> <u>Residential</u> <u>and</u> <u>Landscape</u> <u>Protection</u> <u>Residential</u>	Permitted	Health and Safety: 13.3.2.1 Height: 13.3.2.2.4 Size and Diameter: 13.3.2.2.5 Size and Diameter: 13.3.2.3.6 Separation/Setback: 13.3.2.4.1 Earthworks: 13.3.2.5		

		Comment [GS50]: S6.44 Transpower	Comment [GS52]: S6.44 Transpower Comment [GS51]: S7.11 Spark and HS8.14 Chorus, FS1.11 Transpower	
<u>Matters of Control</u> or Discretion		13.3.4 (a), 13.3.4 (b) 13.3.4 (e), 13.3.4 (f) 13.3.4 (g), 13.3.4 (f) 13.3.4 (l), 13.3.4 (k) 13.3.4 (l), 13.3.4 (m) 13.3.4 (r); 13.3.4 (m) 13.3.4 (v)	13.3.4 (a), 13.3.4 (b) 13.3.4 (e), 13.3.4 (f) 13.3.4 (g), 13.3.4 (f) 13.3.4 (l), 13.3.4 (h) 13.3.4 (l), 13.3.4 (m) 13.3.4 (r), 13.3.4 (m) 13.3.4 (r)	
Standards	<u>Vegetation: 13.3.2.6</u> Noise: 13.3.2.7	Health and Safety: 13.3.2.1 Height: 13.3.2.2.4 Height: 13.3.2.2.5 Size and Diameter: 13.3.2.3.6 Separation/Setback: 13.3.2.4.1 Separation/Setback: 13.3.2.4.1 Separation/Setback: 13.3.2.4.1 Vegetation: 13.3.2.6 Noise: 13.3.2.6	Health and Safety: 13.3.2.1 Height: 13.3.2.4 Height: 13.3.2.4 Size and Diameter: 13.3.2.3.6 Separation/Setback:13.3.2.4.1 13.3.2.4 Earthworks: 13.3.2.6 Vegetation: 13.3.2.6 Noise: 13.3.2.6	Health and Safety: 13.3.2.1
<u>Status</u>		Restricted Discretionary	Restricted Discretionary	Discretionary
Activity Area		<u>Historic</u> <u>Residential,</u> <u>Landscape</u> <u>Protection</u> <u>Residential</u>	All, excluding <u>Historic</u> <u>Residential</u> <u>and</u> <u>Landscape</u> <u>Protection</u> <u>Residential</u>	<u>Historic</u> <u>Residential</u> <u>and</u> <u>Landscape</u> <u>Protection</u> <u>Residential</u>
Rule		Cabinets and other network utility structures not otherwise listed in this table.	Cabinets and other network utility structures not otherwise listed in this table that do not meet the permitted activity standards in Rule 13.3.9.	Cabinets and other network utility structures not otherwise listed in this table that do not meet the restricted discretionary activity standards.
<u>Rule</u> Number		13.3.1.10	13.3.1.11	13.3.1.12

<u>Rule</u> Number	Rule	Activity Area	<u>Status</u>	Standards	<u>Matters of Control</u> or Discretion
13.3.1.13	Network utilities located within existing buildings.	AII	Permitted	Health and Safety: 13.3.2.1 Noise: 13.3.2.7	
13.3.1.14	Aerial crossings necessary for network utilities, located on or within existing bridges and structures or across streamswatercourses . and including regulator stations but not compressor stations.	AI	Permitted	Health and Safety: 13.3.2.1 Earthworks: 13.3.2.5	
13.3.1.15	All network utilities that are not otherwise listed as a permitted, controlled, restricted discretionary or non-complying activity.	AII	Discretionary	Health and Safety: 13.3.2.1	
<u>13.3.1.16</u>	All network utilities which do not comply with the permitted activity standards for radiofrequency and electro-magnetic fields in standard 13.4.1 Health and Safety	AII	<u>Non-</u> Complying		
Undergrou	ind Utilities				
13.3.1.17	The construction, installation and development, of new underground network utilities, except for:• Electricity transmission lines above 110kV; and• Gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals.	AI	Permitted	Health and Safety: 13.3.2.1 Earthworks: 13.3.2.5 Vegetation: 13.3.2.6	
Radiocom	munication, Telecommunications and Electricity	/ Distribution an	d Transmissio	Ę	
13.3.1.18	Masts with or without associated antennas.	Commercial	Permitted	Health and Safety: 13.3.2.1	

Matters of Control or Discretion		13.3.4 (b), 13.3.4 (e) 13.3.4 (f), 13.3.4 (g) 13.3.4 (f), 13.3.4 (j) 13.3.4 (j), 13.3.4 (j) 13.3.4 (n), 13.3.4 (n) 13.3.4 (o), 13.3.4 (n) 13.3.4 (u), 13.3.4 (n)	
<u>Standards</u>	Height: 13.3.2.2.1, Bize and Diameter: 13.3.2.22 Size and Diameter: 13.3.2.3.1, 13.3.2.3.2, 13.3.2.3.2, Earthworks: 13.3.2.4.2	Health and Safety: 13.3.2.1 Height: 13.3.2.2.1 Height: 13.3.2.2.1 Size and Diameter: 13.3.2.3.1 13.3.2.3.1 13.3.2.3.1 Size and Diameter: 13.3.2.3.1 13.3.2.3.1 13.3.2.3.1 Separation/Setback: 13.3.2.4.1 Earthworks: 13.3.2.4.2	Health and Safety: 13.3.2.1
Status		Restricted Discretionary	Discretionary
Activity Area	(All), Business (All), General Rural, Community Health, Community Iwi	Residential (excluding <u>Historic and</u> <u>Landscape</u> <u>Protection</u> <u>Residential)</u> , <u>Rerral</u> <u>Rural</u> <u>Residential</u>	The Coastal Environment identified as SNR 9, shown in Map Appendices 2A, 2B and 2C, Historic Residential, Landscape
Rule		Masts with or without associated antennas.	Masts with or without associated antennas.
<u>Rule</u> Number		13.3.1.19	13.3.1.20

Comment [GS53]: S6.44 Transpower

<u>Rule</u> Number	Rule	Activity Area	<u>Status</u>	<u>Standards</u>	<u>Matters of Control</u> or Discretion	
		<u>Protection</u> Residential				
13.3.1.21	Antenna and support structure attached to buildings.	All, except for <u>Historic</u> Residential	Permitted	Health and Safety: 13.3.2.1 Height: 13.3.2.2.3 Size and Diameter: 13.3.2.3.5		
13.3.1.22	Antenna and support structure attached to buildings.	<u>Historic</u> Residential	<u>Restricted</u> Discretionary	Health and Safety: 13.3.2.1 Height: 13.3.2.2.3 Size and Diameter: 13.3.2.3.5	<u>13.3.4 (d), 13.3.4 (e),</u> 13.3.4 (f), 13.3.4 (g)	
13.3.1.23	Masts, with or without associated antennas that do not meet permitted activity standards.	Commercial (All), Business (All), General Rural, Community Health, Community Iwi	Restricted Discretionary	Health and Safety: 13.3.2.1	13.3.4 (a), 13.3.4 (b), 13.3.4 (c), 13.3.4 (d), 13.3.4 (f), 13.3.4 (g), 13.3.4 (h), 13.3.4 (g), 13.3.4 (l), 13.3.4 (g), 13.3.4 (l), 13.3.4 (m), 13.3.4 (l), 13.3.4 (m), 13.3.4 (r); 13.3.4 (u), 13.3.4 (r); 13.3.4 (u),	Comment [GS54]: S6.44 Transpower
13.3.1.24	Antenna attached to buildings that do not meet permitted activity standards.	All, except for <u>Historic</u> Residential	<u>Restricted</u> Discretionary	Health and Safety: 13.3.2.1	13.3.4 (a), 13.3.4 (b), 13.3.4 (d), 13.3.4 (e), 13.3.4 (f), 13.3.4 (g), 13.3.4 (l), 13.3.4 (g), 13.3.4 (l), 13.3.4 (k), 13.3.4 (l), 13.3.4 (k),	Comment [GS55]: S6.44 Transpower
13.3.1.25	Antenna attached to buildings that do not	Historic	Discretionary	Health and Safety: 13.3.2.1		

<u>Rule</u> Number	Rule	Activity Area	Status	Standards	<u>Matters of Control</u> or Discretion	
	meet restricted discretionary activity standards.	Residential				
13.3.1.26	New and additional above ground lines, including support structures, excluding electricity transmission lines above 110kV	Rural (All)	Permitted	Health and Safety: 13.3.2.1 Height: 13.3.2.2.1 Separation/Setback:13.3.2.5 Earthworks: 13.3.2.5		Comment [GS56]: S6.29 and S6.32 Transpower
13.3.1.26A	New and additional above ground lines, including support structures. excluding electricity transmission lines above 110kV that do not meet permitted activity standards	Rural (All)	<u>Restricted</u> <u>Discretionary</u>	Health and Safety: 13.3.2.1	<u>13.3.4 (a), 13.3.4 (b),</u> <u>13.3.4 (e),</u> <u>13.3.4 (f), 13.3.4 (g),</u> <u>13.3.4 (h), 13.3.4 (h),</u> <u>13.3.4 (h), 13.3.4 (h),</u>	Comment [GS57]: S6.32, S6.37 Transpower
13.3.1.27	New and additional above ground lines,	All, except Bural	Discretionary	Health and Safety: 13.3.2.1 Height	13.3.4 (l). 13.3.4 (m). 13.3.4 (n). 13.3.4 (r). 13.3.4 (s). 13.3.4 (t). 13.3.4(u). 13.3.4(t).	Comment [GS58]: S6.44 Transpower
	provided for	5		Separation/Sotback: 13.3.2.4 Earthworks: 13.3.2.6 Vegetation: 13.3.2.6		Comment [GS59]: S8.18 Chorus Comment [GS60]: S6.32, S6.38
13.3.1.27A	New and additional above ground electricity transmission lines above 110kV, including support structures	All	Discretionary	Health and Safety: 13.3.2.1		Comment [GS61]: S6.29 Transpower
13.3.1.28	Minor above ground lines	AI	Permitted	Health and Safety: 13.3.2.1 Height: 13.3.2.2.1		

			Comment [GS62]: S6.31 Transpower	Comment [GS63]: S9.20 WELL			Comment [GS64]: S6.44 Transpower
<u>Matters of Control</u> or Discretion							13.3.4 (b), 13.3.4 (f), 13.3.4 (h), 13.3.4 (i), 13.3.4 (j), 13.3.4 (n), 13.3.4 (s), 13.3.4 (t); 13.3.4 (u), 13.3.4 (t);
Standards	Separation/Setback:13.3.2.4 Earthworks: 13.3.2.5	Health and Safety: 13.3.2.1 Height: 13.3.2.2.1 Separation/Setback:13.3.2.4 Earthworks: 13.3.2.5 Temporary above ground lines: 13.3.2.8	Health and Safety: 13.3.2.1			Health and Safety: 13.3.2.1 Earthworks: 13.3.2.5 Vegetation: 13.3.2.6 Noise: 13.3.2.7	Health and Safety: 13.3.2.1
<u>Status</u>		Permitted	Discretionary			Permitted	Restricted Discretionary
Activity Area		AI	AII			AII	<u>AII</u>
Rule		Temporary above ground lines	New and upgraded transformers,	substations and switching stations distributing electricity and ancillary buildings, except for those encased within a cabinet or located on a line that is otherwise a permitted activity.	oution and Transmission	Underground gas distribution and transmission pipelines at a pressure not exceeding 2000 kilopascals, including aerial crossings of bridges, structures or streams, and ancillary equipment, including regulator stations but not compressor stations.	Underground gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals, including aerial crossings of bridges, structures or streams, and ancillary equipment, including compressor
<u>Rule</u> Number		13.3.1.29	13.3.1.30		<u>Gas Distrib</u>	13.3.1.31	13.3.1.32

Rule		Activity Area	Status	Standards	<u>Matters of Control</u> or Discretion	
compounds with compressor hous	ses.					
tewater and Stormwater						
Water reservoirs.		AI	Restricted Discretionary		13.3.4 (a), 13.3.4 (b), 13.3.4 (e), 13.3.4 (f), 13.3.4 (g), 13.3.4 (f), 13.3.4 (j), 13.3.4 (j), 13.3.4 (g), 13.3.4 (j),	
Water and wastewater treatment pla	nts.	AII	Discretionary			
ical Activities						
Meteorological enclosures and buildin automatic weather stations and anemo masts, voluntary observer sites and asso microwave links.	<u>gs:</u> meter iciated	AI	Permitted	Health and Safety: 13.3.2.1 Height: 13.3.2.2, Height: 13.3.2.2, Size & Diameter: 13.3.2.3, Separation/Setback:13.3.2.4 13.3.2.6 Vegetation: 13.3.2.6 Noise: 13.3.2.6		
Meteorological enclosures and building automatic weather stations and anemol masts, voluntary observer sites and asso microwave links that are not permitted act	<u>is:</u> inter inties.	AI	Restricted Discretionary	Health and Safety: 13.3.2.1	13.3.4 (a), 13.3.4 (b), 13.3.4 (c), 13.3.4 (e), 13.3.4 (f), 13.3.4 (g), 13.3.4 (h), 13.3.4 (g), 13.3.4 (j), 13.3.4 (j),	

Comment [GS65]: S6.44 Transpower

	Comment [GS66]: S6.44 Transpower		Comment [GS67]: S3.20 KiwiRial		Comment [GS68]: S6.44 Transpower	
Matters of Control or Discretion	<u>13.3.4 (l), 13.3.4 (r);</u> <u>13.3.4(u), 13.3.4(v)</u>				13.3.3 (c), 13.3.3 (d), 13.3.3(g), 13.3.3 (j), 13.3.4(u), 13.3.4(v)	
Standards			Earthworks: 13.3.2.5			
Status			Permitted	Discretionary	Controlled	<u>Non-</u> complying
Activity Area			AI	AII	All, except for the Coastal Environment identified as SNR 9, shown in Map Appendices 2A, 2B and 2C	<u>The Coastal</u> <u>Environment</u> identified as <u>SNR 9, shown</u> in Map
Rule		of Traffic and Transport Structures	Traffic control signals and devices, light and decorative poles and associated structures and fittings, post boxes, landscaped gardens, artworks and sculptures, bus stops and shelters, phone boxes, public toilets and road furniture located within the road reserve and the rail corridor.	The construction, alteration or diversions of roads, excluding any such construction works which is part of a subdivision.	 <u>Any:</u> grade separated facility where a structure is used to separate roadways, railways, footways, cycleways or bodies of water <u>viaduct or tunnel</u> <u>bridges for roads, tramways, railways and underpasses</u> 	 <u>Any:</u> grade separated facility where a structure is used to separate roadways, railways, footways, cycleways or bodies of water viaduct or tunnel
<u>Rule</u> Number		Roading ar	13.3.1.37	13.3.1.38	13.3.1.39	13.3.1.40

<u>Rule</u> Number	Rule	Activity Area	<u>Status</u>	<u>Standards</u>	<u>Matters of Control</u> or Discretion
	 bridges for roads, tramways, railways and underpasses 	Appendices 2A, 2B and 2C			
Extreme A	dverse Weather and Tsunami Warning Devices				
13.3.1.41	Extreme adverse weather and tsunami warning devices.	AII	Permitted	Height: 13.3.2.2.7 Size and Diameter: 13.3.2.3.9 Earthworks: 13.3.2.5 Vegetation: 13.3.2.6	
Notes:					
Resource N	<u> 1anagement Regulations – National Environmental</u>	<u>Standards</u>			
<u>The operati</u> January 20 Regulations	ion, maintenance, upgrading, relocation or remov 010 is largely controlled by the Resource Man 2009, separate to this District Plan.	al of an electric agement (Nation	ity transmissio lal Environme	n line and ancillary structures tha ntal Standards for Electricity Tra	at existed prior to 14 insmission Activities)
<u>The installa</u> <u>controlled b</u> <u>District Plar</u>	ttion and operation of telecommunications facilities by the Resource Management (National Environm on the talso controls all radio-frequency emissions from	(antennas attac ental Standards 1 telecommunicat	hed to existing for Telecommu tion facilities th	I structures and cabinets in the rounications Eacilities) Regulations 2 rough specific exposure standards.	ad reserve) is largely 008, separate to this
<u>Hutt City Cc</u> <u>this Plan an</u>	ouncil is responsible for enforcing these standards. Id the requirements of the NES's identified above, t	For clarification, he provisions of t	where there is the NES shall a	s conflict or perceived conflict betw Ipply.	een the provisions of

Other Relevant Regulations

The National Environmental Standards are available for viewing at www.mfe.govt.nz and at Hutt City Council offices.

Compliance with the New Zealand Electrical Code of Practice for Electrical Safe Distances 34:2001 is mandatory for buildings, earthworks and mobile plants within close proximity to all electricity lines. Compliance with the Electricity (Hazards from Trees) Regulations 2003 is mandatory for tree trimming and planting in proximity to electricity transmission and distribution lines.

To discuss works, including tree planting, near electrical lines especially within 20m of those lines, contact the line operator.

Hutt City Council is not responsible for enforcing these standards.

AMENDMENT 50 [new Chapter 13 Network Utilities, including the National Grid (new 13.3.2 Standards and Terms)] Add a new Title 13.3.2 Standards and Terms and Section 13.3.2.1 Health and Safety

13.3.2 Standards and Terms

13.3.2.1 Health and Safety

Where specified as relevant, network utilities shall comply with the following standards:

- The maximum exposure levels shall not exceed the levels specified in NZS 2772:1999 'Radiofrequency Fields- Maximum exposure levels – 3kHz to 300 GHz'. (a)
- Protection Guidelines for limiting exposure to time-varying electric and magnetic fields (1 Hz 100 Hz), Health Physics 99(6):818-836; 2010, and the recommendations from the World Health Organisation monograph Environmental Health Criteria Network utilities that emit electric and magnetic fields shall comply with the International Commission on Non-ionising Radiation (No 238, 2007) (q)
- Note: The Resource Management (National Environmental Standards for Telecommunications Facilities) Regulations 2008, separate to this District Plan controls all radio-frequency emissions from telecommunication facilities through specific exposure standards.

AMENDMENT 51 [new Chapter 13 Network Utilities, including the National Grid (new 13.3.2 Standards and Terms)] Add a new Section 13.3.2.2 Height

13.3.2.2 Height

The maximum height of any utility structure listed in the table below shall include any antenna and support structures and exclude any lightning rod.

Standard	Utility	Comm	ercial	BI	<u>usiness</u>	Community	Reside	<u>ential</u>	Ru	ral	<u>Recreation</u>
		<u>Central,</u> Petone	Suburban and Special	General, Special, Extraction	nolevA	,Aifealth, Iwi	<u>General,</u> <u>Special,</u> General,	<u>Historic,</u> Landscape Protection	General	<u>Residential</u>	
13.3.2.2.1	<u>Masts, antennas, lines</u> and single-pole support structures.	<u>20m</u>	<u>15m</u>	<u>25m</u>	<u>Area 1 = 20m</u> <u>Area 2 = 15m</u>	20m	<u>12m</u>	<u>10m</u>	<u>15m</u>	<u>12m</u>	<u>12m</u>
13.3.2.2.2	<u>Masts and antennas</u> (involving two or more providers).	<u>25m</u>	<u>18m</u>	<u>30m</u>	<u>Area 1 = 20m</u> <u>Area 2 = 15m</u>	20m	<u>12m</u>	<u>10m</u>	<u>20m</u>	<u>12m</u>	<u>12m</u>
<u>13.3.2.3.3</u>	Maximum height of an antenna and support structure, measured from the highest point of the building to which it is attached.						<u>3.5m</u>	<mark></mark>		<u>3.5m</u>	
13.3.2.2.4	<u>Cabinets and other</u> <u>network utility</u> <u>structures within the</u> <u>road reserve (not</u> otherwise provided for)				<u>mo</u>		<u>1.8</u>	E		<u>2.0m</u>	

Recreation			<u>12m</u>		
ral	Residential		E		
Ru	General		15		
ential	<u>Historic,</u> Landscape Protection		E		
Resid	<u>General,</u> Special, <u>Hill</u>	SI	12		
Community	<u>Health,</u> <u>iwi</u>	<u>3.5r</u>	<u>15m</u>	4 1	
usiness	nolsvA		<u>30m</u>		
B	<u>General,</u> Special, Extraction				
nercial	Suburban		<u>n</u>		
Comn	<u>Central,</u> Petone		15		
<u>Utility</u>		Cabinets and network utility structures that are not otherwise provided for and that are not located within the road reserve.	Anemometer masts.	Maximum height of an extreme adverse weather and tsunami warning device, measured from the point of attachment.	
Standard		13.3.2.5	13.3.2.2.6	13.3.2.2.7	

AMENDMENT 52 [new Chapter 13 Network Utilities, including the National Grid (new 13.3.2 Standards and Terms)] Add a new Section 13.3.2.3 Size and Diameter

13.3.2.3 Size and Diameter

	nunity	<u>iwl</u>	height	<u>neight</u>	tinna tited ontal e of mm
	Comn	<u> НівэН</u>	l mi n	6m in h	Ante loca Anti Anti Anti Anti Anti Ante Ante Ante Ante Ante Ante Ante Ante
	ral	General	nm fron	1m from	Ante Inna ed Nuithi ontal ontal of Circle
	Ru	Residential	st <600r	it <600n	de of
	۲	<u>evisse</u>	mas	mas	n circ
	atio	River	er of	er of	a loc
	ecre	Special	mete	mete	
	Ř	General	Dia	Dia	a l
		Extraction	5		
	less	nolevA	ter o 1.5m	ter o	
	Busir	Special	ame	ame ast `	Ante ated of 5
	ш	General		E	
		<u>Special</u>	<u>t</u>	ht	E C C C C C C C C C C C C C C C C C C C
	nercial	Suburban	<u>ôm in heig</u>	3m in heig	sircle of 75
	Comr	<u>9not99</u>	mm from (mm from (orizontal o
		Central	st <600	st <600	h niti h
		Landsc. Prot.	f mas	ma	N NI
	ntial	II!H	ter of	ter of	ocate
	side	Historic	ame	ame	la
	Re	<u>Isioaq2</u>	Ö	D	
-		General			
	<u>Utility</u>		Masts	<u>Masts</u> (involving two or more providers)	<u>Antenna</u> attached to masts
	Standard		13.3.2.3.1	13.3.2.3.2	13.3.2.3.3

Comment [GS69]: HS7.17 Spark; HS8.22 Chorus

Comment [GS70]: S7.19 Spark; S8.24 Chorus

		ment [GS71]: S7.19 Spark;	24 Chorus mment [GS72]: S7.20 Spark and 25 Chorus
		38	
unity	iwl	nna ied inna inn inn	nna ter of area 3m ²
Comm	<u>dtlsəH</u>	Anter locat borizc circle 750n	Anter diamet of 1.5
ral	General	Ante nna locat ed withi n a horiz ontal of of Sm	3m or
Ru	Residential	Anten na locat ed within a horiz ontal circle of 1.2m	er of 1.5 1.2m ²
	Passive		amet a of '
reatic	River	tenna d with 50mr	are
Rec	Special	An orizol of 7	Anter
	Extraction		
less	nolsvA		1.8m ²
Busin	<u>Special</u>	Anter zonta of 5	a of ,
	General	loc hori	nna c
	Special	20 mm	Ante 2m 0
mercial	Suburban	circle of 75	Antenna diameter of 1.3m or area of 0.81.2m
Com	<u>9not99</u>	orizontal	tenna neter of .8m ²
	Central	thin a h	
	Landsc. Prot.	be with	er of 8m ²
ential	II!H	locati	amet 1 of 0
eside	Historic	anna	area
Ω	General	Ante	Anter 1m oi
I			
<u>Utility</u>		Antenna attached to masts (involving to or more providers)	Antenna attached to other buildings
Standard		13.3.2.3.4	13.3.2.3.5

				Comment [GS73]: S9.23	
nunity	iwl				
Comn	Health				
ural	General				
<u>N</u>	<u>Residential</u>				
۲	<u>evisse</u>				
atio	River				
ecre	<u>Special</u>				
Ř	General				
	Extraction				
ess	nolevA				
usin	Special		٩_		
١	General		<u>15m</u>	<u>5m²</u>	
	Special				
mercial	<u>Suburban</u>				46
Com	<u>Petone</u>				
	Central				
	Landsc. Prot.				
itial	IIIH	NI			
iden	Historic				
Res	<u>Special</u>				
	General				
Utility		Cabinets and other network utility structures located within the road reserve (not otherwise provided for)	Cabinets and other network <u>utility</u> structures not othenwise provided for that are not located within the road reserve	Cabinets located within the road reserve containing an electricity distribution	
Standard		13.3.2.3.6	13.3.2.3.7	<u>13.3.2.3.X</u>	

unity	iwl					
Comm	<u>Health</u>					
<u>ıral</u>	General					
<u>Rı</u>	Residential					
п	Passive					
atio	River					
ecre	<u>Isioaq2</u>					
R	General			than 2.5m x 1.		
-	Extraction					
ness	nolevA					
Busi	<u>Isicad</u>		m ²	ion th		
	General		30	ter in dimens		
<u>Commercial</u>	Special					
	<u>Suburban</u>				No grea	No great
	<u>Petone</u>					
	Central					
	Landsc. Prot.					
Intial	II!H					
side	Historic					
Re	Special					
	General					
К		ation	orologic closures uildings	es es es es es es es es es es es es es e		
Utilit		substa	<u>Mete</u> <u>al en</u>	Extre advei weatt tsuna warni devic		
Standard			13.3.2.3.8	13.3.2.3.9		
	Comment [GS74]: S9.26 WELL Comment [GS75]: S9.2 WELL	Comment [GS76]: S3.21 KiwiRa				
--------------	---	------------------------------	--------------------	---------------------------------------	---	--
	g table that <u>is</u>	unity	<u>iwi</u>		s than om a ary in Areas.	
	e followin a building	Comm	<u>Health</u>		No less 10m fr bound Activity	
	l lines, the sfined as s dor <u>:</u>	Iral	General		than 10m / property idary. 15m in I 15m in in less mall of elling ulding ies and ies a	
	including that is d∉ rail corri	Ru	<u>Residential</u>		No less t from any boun Under than 20 than 20 eccli leccli dec	
	tures, cture ve or		Passive	naintained	<u>0m</u> titial itty	
	<u>y stru</u> <u>eser</u>	ation	River		han 1 bound Activ as.	
	utility cutility oad I	Recre	Special	be m	Are	
	vork u		General	shall		
	l netw ner ne ted in		Extraction	tback	e e	
	to al or oth	iess	nolsvA	an se		
	pplies Ibinet re nof	Busir	Special	ı ripari	Areas	
	any ce and a		General	n 20m	ctivity	
	and and a		Special	inimun	0m fro	
	<u>3.3.2.4</u> masts 1.2 m	ercial	<u>Suburban</u>	A A A A A A A A A A A A A A A A A A A	eside 1	
S	<u>441:</u> ed to ethan	umo	Petone			
acks	13.4 Ittach	0	Central			
Setb	of standard nd antenna a th a height of		Landsc. Prot.			
e and		ntial	IIIH		ss than 10m i boundary ii sidential an Activity Area	
tanc	s and a with	side	<u>Historic</u>			
n Dis	excep mast n are:	Re	<u>Special</u>	-	No le he Re tural /	
aratic	the es to 5m ² j		General		(N) (N) (
13.3.2.4 Sep	<u>With</u> appli	Standard		<u>13.3.2.4.1</u> Riparian setback	<u>13.3.2.4.2</u> Separation distanc. or setback for masl and antenna attached to masts	

AMENDMENT 53 [new Chapter 13 Network Utilities, including the National Grid (new 13.3.2 Standards and Terms)] Add a new Section 13.3.2.4 Separation Distance and Setbacks

4100 4 ä 1 1000

ent [GS75]: S9.2 WELL ent [GS76]: S3.21 KiwiRail

community	<u>Htalth</u> <u>Iwi</u>		o less than 2 etres to any oundary in a <u>Rural.</u> <u>Residential</u> d Recreation <u>cetivity Area</u> nd to a road service lane boundary.	
0				
ral	General	<u>m from</u> ist wall of elling uding ies and ks).	all	
Ru	<u>Residential</u>	than 50 the close a dw (excli balcon	metres to aries.	
	Pasise		an 2 und	
eation	River		bo bo	
Recr	Special		0 Z	
	General			
	Extraction		in a vity dary.	
iness	nolsvA		undary on Acti e boun	
Busi	Special		iny bou	
	General		<u>es to a</u> and <u>Re</u> r servi	
_	Special		metro pad o	
nercia	<u>Suburban</u>		than 2 Resider to a rc	
Com	Petone		Mo less than 2 metres No less than 2 metres to any boundary in a to all boundaries. No less than 2 metres to any boundary in a to all boundaries. for to all boundaries. Rural, Residential and Recreation Activity. No less than 2 metres to any boundary. for to all boundaries. Area and to a road or service lane boundary. No less than 2 metres to any boundary.	
	Central		N N N N	
	Landsc. Prot.		No less than 2 metres to all boundaries.	
ntia	II!H			
side	Historic		bound	
Res	Special		o all br	
	General		No	
Standard			13.3.2.4.3 Separation distance or setback for cabinets and other network utility structures	

AMENDMENT 54 [new Chapter 13 Network Utilities, including the National Grid (new 13.3.2 Standards and Terms)] Add a new Section 13.3.2.5 Earthworks

13.3.2.5 Earthworks

13.3.2.5.1 Sediment and Erosion Control

Erosion and sediment control measures shall be installed and maintained for all network utility activities, in accordance with the "Erosion and Sediment Control Guidelines for the Wellington Region – September 2002" – reprinted 2006.

13.3.2.5.2 Slope, Height, Depth and Area of Earthworks

The following shall apply to all network utility activities, except to earthworks within 2.0 metres of the exterior walls of any network utility building or structure or the outer edge of a network utility structure without walls that is defined as a building measured in plain view, trenching in the road reserve or rail corridor, and to piling undertaken for associated with the installation of a network utility mast, where that piling is contained within 2.0 metres of the edge of the mast.

- (i) Slope No earthworks shall be carried out on a slope greater than 45 degrees.
- (ii) Height, Depth Earthworks shall not exceed 1.5 metres in height or depth.
- (iii) Recession Plane Any earthworks that involve the raising of the height of land above existing ground level shall not exceed a height recession plane measured at an angle of 45 degrees from any neighbouring boundary.
- (iv) Area: Riparian Areas 25m²

All Recreation and Residential Activity Areas – 100m²

All Rural Activity Areas – 1000m²

All Other Activity Areas - 500m²

Rail corridor and state highway – 1,000m²

AMENDMENT 55 [new Chapter 13 Network Utilities, including the National Grid (new 13.3.2 Standards and Terms)]

Add a new Section 13.3.2.6 Native Vegetation Clearance – Rural Residential and General Rural Activity Areas

13.3.2.6 Native Vegetation Clearance – Rural Residential and General Rural Activity Areas

Within the **Rural Residential and General Rural Activity Areas** there shall be no destruction of any native vegetation where:

- (a) the area of native vegetation in one site exceeds 1 hectare with an average height of 3 metres or more, or
- (b) the area of native vegetation is part of an area in one or more sites, which exceeds 1 hectare with an average height of 3 metres or more.

The word "area" in (a) and (b) above refers to the existing area covered by native vegetation, (i.e. it is the sum of the area of native vegetation which is proposed to be disturbed or removed, plus the balance area of native vegetation).

Comment [GS77]: HS10.18, HS10.20 Powerco, HS7.21 and HS8.26 Spark and Chorus, HS9.27 WELL, HS 6.42 Transpower

Comment [GS78]: S1.11 NZTA, S3.22 KiwiRail Tree removal and trimming undertaken solely for maintenance activities under the Electricity (Hazard and Trees) Regulations 2003 are exempt from this standard.

AMENDMENT 56 [new Chapter 13 Network Utilities, including the National Grid (new 13.3.2 Standards and Terms)] Add a new Section 13.3.2.7 Noise

13.3.2.7 Noise

Noise associated with the activity shall not exceed the permitted activity noise standard(s) within the zone in which the activity is located.

AMENDMENT 57 [new Chapter 13 Network Utilities, including the National Grid (new 13.3.2 Standards and Terms)]

Add a new Section 13.3.2.8 Temporary Above Ground Lines

13.3.2.8 **Temporary Above Ground Lines**

The line(s) shall be in place for no longer than six calendar months from the date of erection until its removal.

AMENDMENT 58 [new Chapter 13 Network Utilities, including the National Grid (new 13.3.3 Matters in which Council Seeks to Control)] Add a new Section 13.3.3 Matters in which Council Seeks to Control

13.3.3 Matters in which Council Seeks to Control

Matters over which control is reserved are:

- (a) Site design, frontage and area.
- (b) Legal and physical access to the lots.
- (c) Risks to public health and safety.
- (d) Design and external appearance.
- (e) Earthworks and sediment and erosion control.
- (f) Landscaping and screening.
- (g) Traffic and parking management.
- (h) Noise.
- (i) The route of the road.
- (i) The imposition of financial contributions in accordance with Chapter 12 of this Plan.
- The design and construction of the road, including safety, traffic engineering, (k) landscaping and noise mitigation measures.

AMENDMENT 59 [new Chapter 13 Network Utilities, including the National Grid (new 13.3.4 Matters in which Council has Restricted its Discretion)] Add a new Section 13.3.4 Matters in which Council has Restricted its Discretion

Comment [GS79]: S6.43 Transpower; S9.28 WELL

13.3.4 Matters in which Council has Restricted its Discretion

Matters that the Council has restricted the exercise of discretion over are as follows:

- (a) The degree, extent and effects of the non-compliance with the Permitted Activity Conditions.
- (b) Risks to public health and safety.
- (c) The maximum height of the mast and area or diameter of any antenna.
- (d) The maximum height, area or diameter of any antenna.
- (e) Design and external appearance.
- (f) Any effect on heritage and cultural values.
- (g) Visual effects including impacts on:
 - (i) The residential and recreational use of land in the vicinity of the proposed utility;
 - (ii) The existing character, landscape, streetscape and amenity values of the locality;
 - (iii) Key public places, public viewing points, the coast, and significant recreational areas.
- (h) Amenity effects, including noise, vibration, odour, dust, earthworks and lighting.
- (i) Cumulative effects.
- (i) Any potential interference with public use and enjoyment of the land and the operation of land uses in the vicinity.
- (k) Measures to mitigate the bulk and scale of the utility, including screening, colour and finish treatment, earth mounding and / or planting, viewing distances, the location of support structures.
- (I) Whether the size and scale of the proposal is generally compatible with other development in the area.
- (m) Any adverse effects on traffic and pedestrian safety including sight lines and the visibility of traffic signage.
- (n) The extent to which alternative locations, routes or other options have been appropriately considered.
- (o) The extent to which it is technically, economically and practically reasonable for the masts or antennas can be co-sited with similar structures or other buildings.
- (p) Where antennas are proposed to be sited on the top of a building, the extent to which they can be designed or screened so that they form an integral part of the total building design.
- (q) Rehabilitation of the site following any construction or future maintenance period.
- (r) The extent to which the affected persons / community has been consulted with.
- (s) Earthworks and erosion and sediment control.

- (t) Any adverse effects on an area of native vegetation.
- (u) Any positive effects to be derived from the activity.
- (v) Any constraints arising from technical and operational requirements of the network which may limit measures to avoid, remedy or mitigate environmental effects.

AMENDMENT 60 [new Chapter 13 Network Utilities, including the National Grid (new 13.3.5 Assessment Criteria for Discretionary Activities)]

Add a new Section 13.3.5 Assessment Criteria for Discretionary Activities

13.3.5 Assessment Criteria for Discretionary Activities

In considering an application for a discretionary activity, the Council's discretion is unrestricted. The Council shall consider any relevant matter with particular regard to the objectives and policies of the Plan. In addition, the Council shall have particular regard to the relevant matters outlined in 13.3.4 – Matters of Discretion for Restricted Discretionary Activities.

AMENDMENT 61 [new Chapter 13 Network Utilities, including the National Grid (new 13.4 Rules – National Grid)]

Add a new Section 13.4.1 Permitted Activities

<u>13.4</u> <u>Rules – National Grid</u>

13.4.1 Permitted Activities

In all activity areas, buildings and structures less than 2.5m in height and less than 10m² in area located within the National Grid Yard, that meet all the permitted activity conditions of that activity area, provided that they are not being used for a Sensitive Activity.

Note: Compliance with the New Zealand Electrical Code of Practice for Electrical Safe Distances (NZECP34:2001) is mandatory under the Electricity Act 1992. All activities regulated by NZECP34, including buildings, structures, earthworks and the operation of mobile plant, must comply with that regulation. Activities should be checked for compliance even if they are permitted by the District Plan.

AMENDMENT 62 [new Chapter 13 Network Utilities, including the National Grid (new 13.4 Rules – National Grid)]

Add a new Section 13.1.2 Restricted Discretionary Activities

13.4.2 Restricted Discretionary Activities

(a) In all activity areas, new buildings and structures and additions and alterations to existing buildings and structures that are over 2.5m in height and / or more than 10m² in area located within the National Grid Yard.

(i) Non-notification

Comment [GS80]: S6.44 Transpower; S69.14, S7.9, S8.12, FS2.10, FS1.4 WELL, Spark, Chorus, Transpower, Powerco In respect of Rule 13.4.2 (a), public notification of applications for resource concent is procluded. Limited notification will be served on the National Grid Operator as the only affected party under section 95B of the Act.

Note: Rule 13.4.2 (a) (i) prevails over Rule 17.2.2.

Comment [GS81]: S6.45

 AMENDMENT 63 [new Chapter 13 Network Utilities, including the National Grid (new 13.4 Rules - National Grid)]
 Add a new Section 13.4.2.1 Matters in which Council has restricted its discretion
 13.4.2.1 Matters in which Council has restricted its discretion

 (a) Any risk to the structural integrity of the transmission line;
 (b) Any effects on the ability of the transmission line owner to operate, maintain and/or upgrade the National Grid;
 (c) The proximity of buildings and structures to electrical hazards;
 (d) Operational risks relating to health or public safety, and the risk of property damage;
 (e) Amenity effects; and
 (f) Any actual or potential reverse sensitivity effects.

> Advice Note: Compliance with the New Zealand Electrical Code of Practice for Electrical Safe Distances (NZECP34:2001) is mandatory under the Electricity Act 1002. All activities regulated by NZECP34, including buildings, structures, earthworks and the operation of mobile plant, must comply with that regulation. Activities should be checked for compliance even if they are permitted by the District Plan.

> Vegetation to be planted within the National Grid Yard as shown on the planning maps should be selected and/or managed to onsure that it will not result in that vegetation breaching the Electricity (Hazards from Trees) Regulations 2003 or prevent access to support structures. To discuss works, including tree planting near any electricity line especially works within the transmission corridor; contact the relevant network utility operator.

Comment [GS82]: S6.45

AMENDMENT 64 [new Chapter 13 Network Utilities, including the National Grid (new 13.4 Rules – National Grid)]

Add a new Section 13.4.3 Non-Complying Activities

13.4.3 Non-Complying Activities

- a) The establishment of sensitive land use activity, including the change of the use of an existing building or structure.
- b) The construction of a new, or addition to an existing, building or structure that does not meet permitted activity rule 13.4.1.

In all activity areas, the construction and use, establishment, or change of use, of any building and structure located within the National Grid Yard for a Sensitive Activity.

Comment [GS83]: S6.45

AMENDMENT 65 [new Chapter 13 Network Utilities, including the National Grid (new 13.5 Anticipated Environmental Results)]

Add a new Section 13.5 Anticipated Environmental Results

<u>13.5</u> <u>Anticipated Environmental Results</u>

The following environmental outcomes are anticipated in respect of network utilities:

- (a) Network utilities are able to operate, upgrade and expand as required to provide safe, effective and efficient services to the City.
- (b) The Lower Hutt City community is able to provide for its social, economic, cultural and environmental wellbeing and for its health and safety.
- (c) Potential conflicts between regionally significant network utilities and incompatible development, use and subdivision are avoided.
- (d) Adverse effects from network utilities on amenity and the environment are appropriately avoided, remedied or mitigated.

Amendment to Chapter 14C - Noise

AMENDMENT 66 [Chapter 14C Noise (14C 2.1 Permitted Activity Conditions)] Delete Permitted Activity Condition 14C 2.1 (h)

14C 2.1 Permitted Activity - Conditions

In all Activity Areas

(h) Noise from electricity distribution transformers not exceeding 1.5 MVA capacity, which are designed, constructed, installed and maintained to good New Zealand industry practice, providing the maximum noise level, measured at a distance of two metres in front of any window of any living area or bedroom of any residential building, shall be less than 50dBA.

Amendments to Chapter 14I - Earthworks

AMENDMENT 67 [Chapter 14I Earthworks (14I 1.3 Essential Works)] Amend the Issue, Policies and Explanation and Reasons for 14I 1.3 Essential Works

14I 1.3 Essential Works

Issue

Some earthworks are necessary for the provision of services essential to the health and safety of the community; in particular. These may include earthworks necessary for the establishment and maintenance of utilities, or those associated with the management of a river or stream to avoid, remedy or mitigate a flood event. It is necessary to ensure that opportunity for such works is provided for in the Plan.

Objective

To ensure that provision is made for earthworks to be carried out for services which are essential to the health and safety of the community.

Policyies

- (a) To provide for essential earthworks to be carried out which are necessary for the management of any stream or river.
- (b) To provide for earthworks associated with the establishment and maintenance of a utility in accordance with the Plan.

Explanation and Reasons

Some earthworks are essential to ensure property and lives are not at risk, or to mitigate adverse effects of a naturally occurring event such as a flood. <u>Opportunity</u> is made in the Plan for earthworks associated with flooding events to be carried out without restriction. This is to recognise their importance to the community.

Earthworks are often necessary in the establishment and maintenance of network utilities services which are essential to the well being of the community.

Opportunity is made in the Plan for such earthworks to be carried out without restriction. This is to recognise their importance to the community.

AMENDMENT 68 [Chapter 14I Earthworks (14I 2 Rules)]

Amend Rule 14I 2(i) to reflect proposed new Chapter 13 Network Utilities

14I 2 Rules

These provisions shall not apply to the following:

- Earthworks associated with the establishment of <u>network</u> utilities in accordance with Chapter 13 – <u>Network</u> Utilities, including the National Grid.
- (ii) Earthworks carried out as part of a subdivision consent under Chapter 11.

- (iii) Earthworks in the River Recreation Activity Area for the purposes of the management of any river or stream in accordance with Chapter 7C River Recreation Activity Area
- (iv) Earthworks associated with extraction activities in Chapter 6D Extraction Activity Area.

New Chapter 14L - Renewable Energy Generation

AMENDMENT 69 [New Chapter 14L Renewable Energy Generation (Introduction)] Add a new Introduction to Chapter 14L Renewable Energy Generation

14L Renewable Energy Generation

Introduction

Energy is vital to the efficient functioning of our country. As both a natural and physical resource, the generation and use of energy is a relevant resource management matter. In particular, section 7(j) of the Resource Management Act requires decision-makers to give particular regard to the benefits derived from the use and development of renewable energy.

This chapter is focused on renewable energy generation; and in particular, the conversion of natural resources into electricity. Energy generation from non-renewable sources, energy efficiency and energy conservation are addressed in other Plan Chapters.

The National Policy Statement for Renewable Electricity Generation came into force in May 2011 and forms part of central government's strategic target to achieve 90% of electricity generated is from renewable energy sources by 2025. The Council is required to give effect to any national policy statement. The objective of the National Policy Statement is to recognise the national significance of renewable electricity generation activities by providing for the development, operation, maintenance and upgrading of new and existing renewable electricity generated from renewable energy sources increases to a level that meets or exceeds the New Zealand Government's national target for renewable electricity generation. Policy D of the National Policy Statement requires that decision makers shall, to the extent reasonably possible, manage activities to avoid reverse sensitivity effects on consented and on existing renewable electricity generation activities.

The Regional Policy Statement for the Wellington Region recognises the importance of energy within the Region, and in particular that the benefits from renewable energy, from small to large scale generation, be recognised. Hutt City Council is required to give effect to the Regional Policy Statement and include policies and/or methods to ensure these benefits are recognised. These benefits include security of supply and diversification of energy sources, reducing dependency on imported energy sources and reducing greenhouse gas emissions.

Although energy is essential to our communities, its generation has the potential to have both adverse and positive environmental effects. The positive effects of renewable energy generation are often felt nationally (and/or globally) as well as locally, however adverse effects are generally more localised. Given the nature of renewable energy generation activities, many developments are unlikely to be able to internalise all potential adverse effects that they may generate within the site, and may include effects on amenity, landscape, ecology, cultural and heritage values. The benefits of any development therefore need to be weighed up against potential adverse effects.

The application of renewable energy can be in a number of different forms. At the domestic or small scale, there are various passive approaches including orientating buildings to assist passive heating and cooling, and natural lighting and more active approaches such as solar water heating and panels, and wind turbines. The passive approaches are addressed through other chapters in the District Plan. This chapter addresses the active approaches to renewable energy generation, primarily in the form of electricity generation. At present, renewable energy generation facilities in Lower Hutt City are limited to small scale wind turbines and solar panels. There are no existing larger scale renewable energy generation activities within the City.

A study of the Wellington Region's renewable energy sources undertaken by the Energy Efficiency and Conservation Authority in August 2006 identifies wind and solar as the primary renewable energy sources within Lower Hutt City that fall within the functions of the Hutt City Council. Other options such as large scale solar generation or biomass energy may become more viable in the future, and this chapter should be amended to address the various resource management issues that may arise as more information about these options becomes available. This chapter therefore focusses on wind and small-scale solar resources.

The provisions in this Chapter apply to renewable energy generation activities throughout all zones of the City. The underlying zone objectives, policies and rules do not apply to renewable energy generation activities unless specifically referred to. City wide rules, such as those relating to historic heritage, notable trees, earthworks and hazardous substances will still apply.

AMENDMENT 70 [New Chapter 14L Renewable Energy Generation (14L 1 Issues Objectives and Policies)]

Add new Issues, Objectives and Policies relating to Balancing the Effects of Renewable Energy Generation as Section 14L 1.1

14L 1 Issue, Objectives and Policies

14L 1.1 Balancing the Effects of Renewable Energy Generation

<u>Issue</u>

Balancing conflicts created by the effects of renewable energy generation with its local, regional and national benefits.

Explanation:

Lower Hutt City contains some renewable energy resources, the use of which would provide benefits locally, regionally and nationally. However, the use of such resources can also have adverse environmental effects, which are generally felt at a local level. These effects can create conflicts between renewable energy generation and a wide range of biophysical and community held values, and raise issues of scale and location.

Objectives

(a) Provide for the development of renewable energy generation that is designed, located, constructed, operated, maintained and upgraded so as to:

(i) Avoid, remedy or mitigate adverse effects on the environment; and

- (ii) Promote the local, regional, and national benefits of the use and development of renewable energy resources.
- (b) Enable small-scale renewable energy generation and the identification and assessment of potential renewable energy sources and sites in appropriate locations within the City.

Policies

- (a) Recognise the local, regional and national benefits of renewable energy generation, including:
 - (i) Maintaining and increasing security of electricity supply;
 - (ii) Using renewable rather than finite energy sources;
 - (iii) Reducing dependency on imported energy sources;
 - (iv) Reducing greenhouse gas emissions; and
 - (v) The reversibility of the adverse effects of some renewable energy generation technologies.
- (b) Enable small-scale renewable energy generation to be developed and operated in a manner that avoids, remedies or mitigates adverse environmental effects.
- (c) Enable the identification and assessment of potential renewable energy sources and sites in a manner that avoids, remedies or mitigates adverse environmental effects.
- (d) Provide for the operation, maintenance and development of community scale and commercial scale renewable energy generation activities.
- (e) Manage the adverse environmental effects of community scale and commercial scale renewable energy generation activities by recognising that these activities have the potential to cause significant adverse effects on the environment. In particular, activities that use wind as a source of energy have the potential for significant adverse effects on landscape, ecology and amenity values, and noise (including low frequency noise) and may be inappropriate in some locations.
- (f) Recognise the technical and operational constraints of renewable energy generation, including the location of the resource, development and maintenance of facilities and the location of the electricity distribution network.
- (g) Encourage community and commercial scale renewable energy generation providers to consult early with the local community, including Māori, on the appropriate placement, location and design of renewable energy generation activities.
- (h) Protect consented and existing renewable energy generation activities from incompatible subdivision, land use and development.
- (i) Ensure that the provision and operation of renewable energy generation activities that cross jurisdictional boundaries is managed in an integrated manner.

Explanation and Reasons

There are significant benefits available at the local, regional and national level from renewable energy generation activities. The benefits from any renewable energy generation proposal can range from large significant contributions, to small incremental gains, but they are all cumulative. New renewable generation capacity can contribute to the New Zealand Energy Strategy target, and to increasing the diversity of supply. The City offers opportunities in particular for wind and solar generation.

Some parts of the City, particularly those that are elevated, are potentially suited to the development of renewable energy sources, particularly commercial scale renewable energy generation activities where wind is the energy source. There are other parts of the City where smaller community scale renewable energy generation activities may be viable to serve the electricity needs of local areas. There is little potential in the City for the community or commercial scale use of solar energy for electricity generation.

However, small-scale renewable energy generation particularly from wind and the sun provides an opportunity for Lower Hutt City to become partially self-reliant for energy supply. Technologies that currently exist and enable individuals to harness energy sources are likely to improve and become more cost-effective into the future. As demand for energy increases, self-reliance will have economic benefits for both individuals and the City.

The nature and scale of effects from renewable energy generation will vary depending on the scale and location of the activity and the characteristics of the surrounding area. Potential effects include adverse visual impacts, impacts on indigenous flora and fauna, culturally and historically significant areas and noise effects. While the Plan permits some renewable energy generation activities, those that are of a significant scale or do not meet the standards to be a permitted activity, will require a full assessment of their environmental effects through the resource consent process. This will allow the Council to weigh the benefits of any new generation activities on sites containing an item listed in the Appendix Heritage 1 and 2 and where turbines do not comply with NZS6808:2010, as these are likely to have significant adverse effects on the environment.

The locational, functional and technical constraints on the siting of renewable energy generation facilities also need to be considered when development proposals are assessed and conditions of consent are imposed. This recognition includes the need for renewable energy generation facilities to be located where such resources are available and the location of existing structures and infrastructure.

The Plan also recognises that new subdivision, land use and development activities can result in reverse sensitivity effects on existing and consented renewable energy generation facilities and may result in the benefits of facilities being reduced. In addition, community amenity values may be adversely affected by locating in too close proximity to renewable energy generation facilities. At present, the City only contains small scale renewable energy generation facilities with no established community scale or commercial facilities. It is likely any such larger scale facilities would be established in the rural areas and that any reverse sensitivity effects would arise from subsequent new subdivision, land use and development which would provide an opportunity for assessment of any such potential effects.

Some renewable energy generation activities and the effects arising from them may cross jurisdictional boundaries between councils. Cross boundary issues can result for renewable energy generation providers and for the community, particularly where different councils have different rules or processes for how they recognise and provide for renewable energy generation activities and manage their effects. It is important that councils work together in an integrated manner both when developing plan provisions and when dealing with proposals for new or upgrades to existing renewable energy generation activities.

AMENDMENT 71 [New Chapter 14L Renewable Energy Generation (new Section 14L 2 Rules)] Add new Section 14L 2.1 Activity Status and Standards

14L 2 Rules

14L 2.1 Activity Status and Standards

<u>Rule</u> number	Activity	Activity Area	<u>Activity</u> <u>Status</u>	<u>Standards</u> <u>Ma</u>	atters of scretion
14L2.1.1	The operation and maintenance of existing renewable energy generation facilities.	<u>All</u>	Permitted	Must comply with the permitted activity standards for the Activity Area.	
14L 2.1.2	Any solar panel mounted to any building, including buildings listed in Appendix Heritage 1 and 2	All	Permitted	 (a) May exceed the permitted height for the Activity Area by no more than 1m. (b) May exceed the recession plane standard for the Activity Area by no more than 1m (measured vertically). (c) Where located on a heritage building listed in Appendix Heritage 1 and 2, any solar panel shall: (i) be located on a roof plane which is not visible from any adjacent public areas; (ii) be aligned with the plane of the roof. 	
14L 2.1.3	Roof-mounted small scale wind turbines	<u>All</u>	Permitted	 (a) Shall comply with recession plane, yard and noise standards for the Activity Area in which the site is located. (b) May exceed the permitted height for the Activity Area by no more than 2m. (c) May not exceed more than one turbine per site. 	

<u>Rule</u> number	Activity	Activity Area	<u>Activity</u> <u>Status</u>	Standards	<u>Matters of</u> discretion
14L 2.1.4	Freestanding small scale wind turbines	AII	Permitted	 (a) Must comply with recession plane, yard and noise standards for the Activity Area that the site is located in. (b) May exceed the permitted height for the zone by up to, but no more than, 2m. (c) May not exceed more than one turbine per site. 	
<u>14L 2.1.5</u>	Temporary renewable energy assessment and research structures	Rural (All) Recreation (All) Business(All)	Permitted	 (a) Shall comply with the New Zealand Standards in relation to noise, radio frequency emissions and any other emissions. (b) Must comply with the noise standards for the Activity Area that the site is located in. (c) Any anemometer must not exceed 80 metres in height. (d) All structures shall comply with the height in relation to boundary and yard requirements for the Activity Area in which they are located. (e) Any public road damaged in such investigation or assessment activity shall be reinstated to a condition of a similar or improved standard to that which existed prior to commencement of the work. (f) No such investigation or assessment period shall exceed five years. (g) All equipment and structures shall be reinstated to a condition of the investigation and the site shall be restored and rehabilitated to a condition no less than that which existed prior to the works commencing. 	

<u>Rule</u> number	Activity	Activity Area	<u>Activity</u> <u>Status</u>	Standards	<u>Matters of</u> discretion
<u>14L 2.1.6</u>	Solar panels that do not comply with one or more of the permitted activity standards	AII	Restricted Discretionary		14L 2.2 (a), (b), (c), (e), (h), (i), (m)
14L 2.1.7	Small scale roof-mounted and freestanding wind turbines that do not comply with one or more of the permitted activity standards	AII	<u>Restricted</u> <u>Discretionary</u>		<u>14L 2.2 (a), (b), (c),</u> (d), (e), (g), (i), (j), (k), (m)
<u>14L 2.1.8</u>	Temporary renewable energy assessment and research structures that do not comply with one or more of the permitted activity standards and are not located on a site	<u>All</u>	Restricted Discretionary		<u>14L 2.2 (a) to (m)</u> inclusive
<u>14L 2.1.9</u>	Land based structures that support in-stream hydro or marine energy generation	Rural (All) Recreation (All) Business (All)	Restricted Discretionary	 (a) Shall not be located: (i) within an esplanade reserve or strip; (ii) within any riparian setback; (iii) on any legal road (formed or unformed). 	<u>14L 2.2 (a) to (m)</u> inclusive
14L 2.1.10	Community scale renewable energy generation activities, including associated construction and commissioning activities	Rural (All)	Restricted Discretionary	 (b) Shall not be located: (i) within an esplanade reserve or strip; (ii) within any riparian setback (iii) on any legal road (formed or unformed). (c) Any structures shall be set back at distance of not less than three times the height of 	14L 2.2 (a) to (m) inclusive

<u>Rule</u> number	Activity	Activity Area	<u>Activity</u> <u>Status</u>	Standards	<u>Matters of</u> discretion
				generating device or support structure from any site boundary.	
14L 2.1.11	The installation or upgrading of any commercial scale renewable energy generation activities, including associated construction and commissioning activities	Rural (All)	Discretionary	(a) NZS6808:2010 Acoustics - Wind Farm Noise	
141.2.1.12	Any renewable energy generation activity that is not otherwise provided for as a permitted, restricted discretionary or non- complying activity.	All	Discretionary		
14L 2.1.13	Except where otherwise specifically provided for. any renewable energy generation activity located on a site containing an item listed in the Appendix Heritage 1 and 2	All	<u>Non-</u> complying		
1412.1.14	<u>The installation of or</u> upgrading of any community or commercial scale renewable energy	<u>All</u>	<u>Non-</u> complying		

<u>Rule</u> number	Activity	Activity Area	<u>Activity</u> <u>Status</u>	Standards	<u>Matters of</u> discretion
	generation activities, including associated construction and commissioning activities that does not comply with NZS6808:2010, where wind is the energy source for the activity.				

AMENDMENT 72 [New Chapter 14L Renewable Energy Generation (new Section 14L 2 Rules)]

Add new Section 14L 2.2 Matters of Discretion for Restricted Discretionary Activities

14L 2.2 Matters of Discretion for Restricted Discretionary Activities

- (a) Any positive effects to be derived from the activity.
- (b) The contribution to achieving national, regional and local energy policy objectives and renewable energy targets.
- (c) Health and safety.
- (d) Suitability of the site for the proposed activity, including consideration of geotechnical and natural hazard constraints.
- (e) Layout, design and location of proposed structure.
- (f) Traffic effects.
- (g) Extent of any earthworks.
- (h) Effects on historic heritage.
- (i) Visual, character and amenity effects.
- (j) Noise and lighting effects.
- (k) Effects on public access.
- (I) Effects on natural character.
- (m) Adequacy of the methods of mitigation/remediation or ongoing management.

AMENDMENT 73 [New Chapter 14L Renewable Energy Generation (new Section 14L 2 Rules)]

Add new Section 14L 2.3 Assessment Criteria for Discretionary Activities

14L 2.3 Assessment Criteria for Discretionary Activities

The following assessment criteria shall guide the assessment of discretionary activity applications and shall be read in conjunction with the District Plan's objectives and policies.

- (a) Any positive effects to be derived from the activity.
- (b) The contribution to achieving national, regional and local energy policy objectives and renewable energy targets.
- (c) Health and safety.
- (d) Noise effects, including compliance with NZS 6808:2010 (Acoustics wind farm noise).
- (e) The impact on reserves and other protected public and private land, and recreation areas, community facilities, infrastructure and services.
- (f) The impact on public access to, and along, the margins of the coast, lakes and rivers or to natural and physical features.
- (g) The impact on landscape values and natural features.
- (h) The potential impact of natural hazard events and the effect the activity itself may have on exacerbating or relieving natural hazards.

- (i) Any geotechnical constraints of the affected area.
- (j) Impacts of earthworks and the modification of natural landforms, including proposed remedial and mitigation measures.
- (k) Impacts on the amenity values of the surrounding environment, including a consideration of electromagnetic interference, vibration, aviation navigation lighting and turbine blade shadow or glare flicker.
- (I) Ecological impacts, particularly impacts on the coastal environment, water bodies, impacts associated with native vegetation removal and impacts on indigenous avifauna, indigenous fauna and their habitats.
- (m) Impacts on archaeological and historic features and items, and sites of significance to tangata whenua.
- (n) Traffic impacts (including construction and post-construction traffic) and impacts on the roading network, including the nature and extent of vehicle movements, access, management and mitigation measures; road safety and levels of service; and impacts associated with traffic distraction.
- (o) Impacts on aviation and navigation.
- (p) Construction effects, including construction noise (and the ability to meet NZS <u>6803:1999 Acoustics – Construction Noise).</u>
- (q) Any cumulative effects.

AMENDMENT 74 [New Chapter 14L Renewable Energy Generation (new Section 14L 3 Anticipated Environmental Results)]

Add new Section 14L 3 Anticipated Environmental Results

<u>14L 3</u> <u>Anticipated Environmental Results</u>

The following environmental outcomes are anticipated in respect of renewable energy generation:

- (a) The benefits to be derived to Lower Hutt City from renewable energy generation activities are realised.
- (b) The Lower Hutt City community is able to provide for its social, economic, cultural and environmental wellbeing.
- (c) The health and safety of Lower Hutt City's community is not adversely affected by the construction, operation, upgrading and maintenance of renewable energy generation activities.
- (d) Adverse effects from renewable energy generation activities on amenity and the environment are appropriately avoided, remedied or mitigated.

Amendments to "Other Provisions" in all Activity Areas

AMENDMENT 75 [All Activity Areas (Chapters 4A, 4B, 4C, 4D, 4E, 5A, 5B, 5C, 5D, 6A, 6B, 6C, 7A, 7B, 7C, 7D, 8A, 8B, 9A, 10A)]

Amend wording of "Other Provisions" as follows:

4A 2.6 Other Provisions

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network</u> Utilities, including the National Grid See Chapter 13.
- (d) General Rules See Chapter 14.

4B 2.5 Other Provisions

- (a) Subdivision See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network</u> Utilities, including the National Grid See Chapter 13.
- (d) General Rules See Chapter 14.

4C 2.5 Other Provisions

- (a) Subdivision See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network</u> Utilities, including the National Grid See Chapter 13.
- (d) General Rules See Chapter 14.

4D 2.5 Other Provisions

- (a) Subdivision See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network Utilities including the National Grid</u> See Chapter 13.
- (d) General Rules See Chapter 14.

4E 2.5 Other Provisions

- (a) Subdivision See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network Utilities</u>, including the National Grid See Chapter 13.
- (d) General Rules See Chapter 14.

5A 2.4 Other Provisions

- (a) Subdivision See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network Utilities</u>, including the National Grid See Chapter 13.
- (d) General Rules See Chapter 14.

5B 2.1.5 Other Provisions

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network</u> Utilities, including the National Grid See Chapter 13.
- (d) General Rules See Chapter 14.

5B 2.2.5 **Other Provisions**

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- Network Utilities, including the National Grid See Chapter 13. (c)
- (d) General Rules See Chapter 14.

5C 2.5 **Other Provisions**

- (a) Subdivision See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) Network Utilities including the National Grid - See Chapter 13.
- (d) General Rules See Chapter 14.

5D(i) 2.5 **Other Provisions**

- (a) Subdivision See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) Network Utilities, including the National Grid - See Chapter 13.
- (d) General Rules See Chapter 14.

5D(ii) 2.5 **Other Provisions**

- (a) Subdivision See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- Network Utilities, including the National Grid See Chapter 13. (c)
- (d) General Rules See Chapter 14.

6A 2.6 **Other Provisions**

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) Network Utilities, including the National Grid - See Chapter 13.
- (d) General Rules See Chapter 14.

6B 2.5 **Other Provisions**

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- Network Utilities, including the National Grid See Chapter 13. (c)
- (d) General Rules See Chapter 14.

6C 2.5 **Other Provisions**

- (a) Subdivision See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) Network Utilities including the National Grid - See Chapter 13.
- (d) General Rules See Chapter 14.

6D 2.4 **Other Provisions**

- (a) Subdivision See Chapter 11.
- Financial Contributions See Chapter 12. (b)
- Network Utilities, including the National Grid See Chapter 13.
- (c) <u>Network Utilities</u>, <u>including the Na</u>
 (d) General Rules See Chapter 14.

7A 2.4 Other Provisions

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network Utilities</u>, including the National Grid See Chapter 13.
- (d) General Rules See Chapter 14.

7B(i) 2.5 Other Provisions

- (a) Subdivision See Chapter 11
- (b) Financial Contributions See Chapter 12
- (c) <u>Network</u> Utilities, including the National Grid See Chapter 13
- (d) General Rules See Chapter 14.

7B(ii) 2.5 Other Provisions

- (a) Subdivision See Chapter 11
- (b) Financial Contributions See Chapter 12
- (c) <u>Network Utilities</u>, including the National Grid See Chapter 13
- (d) General Rules See Chapter 14.

7B(iii) 2.5 Other Provisions

- (a) Subdivision See Chapter 11
- (b) Financial Contributions See Chapter 12
- (c) <u>Network Utilities, including the National Grid</u> See Chapter 13
- (d) General Rules See Chapter 14.

7C 2.4 Other Provisions

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network Utilities</u>, including the National Grid See Chapter 13.
- (d) General Rules See Chapter 14

7D 2.4 Other Provisions

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network Utilities</u>, including the National Grid See Chapter 13.
- (d) General Rules See Chapter 14.

8A 2.5 Other Provisions

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network Utilities</u>, <u>including the National Grid</u> See Chapter 13.
- (d) General Rules See Chapter 14.

8B 2.5 Other Provisions

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network Utilities</u>, including the National Grid See Chapter 13.
- (d) General Rules See Chapter 14.

9A 2.4 **Other Provisions**

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
- (c) <u>Network Utilities, including the National Grid</u> See Chapter 13.
- (d) General Rules See Chapter 14.

10A 2.4 **Other Provisions**

- (a) Subdivisions See Chapter 11.
- (b) Financial Contributions See Chapter 12.
 (c) <u>Network Utilities, including the National Grid</u> See Chapter 13.
 (d) General Rules See Chapter 14.

Amendments to Planning Maps

AMENDMENT 76 [Planning Maps A3, C3, D2, D3, D5, D6, E1, E2, E5, F1, F4, F5, G1, H1, H2, H3, R1, R2, R3, R4]

Amend the above planning maps to display the location of the National Grid and the maximum extent of the National Grid Corridor (39m from centre line):

KEY	
	National Grid Centre Line
	National Grid Corridor maximum extent (39m from centre line)

APPENDIX THREE – AMENDED TEXT OF PROPOSED PLAN CHANGE 38

APPENDIX 2

Proposed changes to District Plan text for Proposed District Plan Change 38: Network Utilities and Renewable Energy Generation

Table of Amendments:

Amendment	Title	Page reference
1	Amendments to Chapter 16; Utilities (Issues Objectives and Policies)	2 - 17
2	Insert new Chapter 16A: Renewable Energy Issues Objectives and Policies	18 - 23
3	Amendments to Chapter 18: Residential Zone Rules	24 - 26
4	Amendments to Chapter 19: Rural Zone Rules	27 – 28
5	Amendments to Chapter 20: Business Zone Rules	29 – 30
6	Amendments to Chapter 21: Open Space Zone Rules	31 – 32
7	Amendments to Chapter 22: Special Activity Zone Rules	33 – 34
8	Amendments to Chapter 28: Southern Hills Overlay Area and Protected Ridgeline Rules	35 – 39
9	Amendments to Chapter 30: Rules for Utilities	40 – 58
10	New Chapter 30A: Rules for Renewable Energy Generation	59 – 64
11	Amendments to Chapter 35: Definitions	65 - 72

Guide to amendments:

Proposed new text is <u>underlined</u>. Text proposed to be deleted is struckthrough. For the sake of brevity, only proposed amendments to the content of relevant chapters, parts and sections are shown, rather than replicating the entire chapter, part or section.

Amendment 1: Amendments to Chapter 16: Utilities Issues Objectives and Policies

NETWORK UTILITIES

16.1 Background

16

The term <u>'network</u>_utilities' refers to the facilities and services that are necessary for the functioning of the City, and includes:

- Transportation, such as roading, rail and pedestrian transport.
- Telecommunications.

Utilities, such as energy supply, electricity transmission, sewerage, waste disposal and water supply systems.

This chapter outlines the provisions of the District Plan that relate to network utilities within Upper Hutt City. Network utilities provide the infrastructure which enables a community to undertake its everyday activities and functions and allows people to provide for their social and economic wellbeing, and their health and safety. Network utilities which are managed through this Chapter include those defined through reference in section 166 of the Resource Management Act 1991. A definition of network utilities is included in Chapter 35, Definitions.

The City has a range of network utilities that serve an important function locally, regionally and nationally, some of which are critical and life-supporting. In particular, State Highway 2 and the Wairarapa Railway Line provide the link from Wellington and Hutt City to Upper Hutt and the Wairarapa. There are also a number of transmission assets associated with the National Grid located within the City.

The Regional Policy Statement for the Wellington Region recognises the importance of regionally significant infrastructure within the Region, as forming part of national or regional networks that enable communities to provide for their social, economic and cultural wellbeing and their health and safety. There are a number of network utilities within Upper Hutt City that are identified as being regionally significant infrastructure in the Regional Policy Statement. The Regional Policy Statement requires that the benefits of such regionally significant infrastructure be recognised and protected in the District Plan.

The Council is required to give effect to any National Policy Statement. The National Policy Statement on Electricity Transmission came into force in 2008 and applies to effects on and effects of the transmission network". The National Policy Statement on Electricity Transmission's objective is to recognise the national significance of the National Grid by facilitating the operation, maintenance and upgrade of the existing transmission network and the establishment of new transmission resources to meet the needs of present and future generations while: managing the adverse environmental effects of the network; and managing the adverse effects of other activities on the network.

There are many providers of network utilities within Upper Hutt City including the Council, Crown agencies, the Greater Wellington Regional Council, State Owned Enterprises, trading enterprises and private companies. The Council is in itself a major provider of network utilities and services, supplying water, sewage and stormwater reticulation, waste disposal and roads.

Other utilities that are managed through this Chapter because of their nature and function are navigation aids, beacons, signal stations and natural hazard warning devices and meteorological services¹. These other utilities are owned and / or operated by Maritime New Zealand, local authorities or the Meteorological Service in order to provide for the health, safety and wellbeing of the local community, region and nation.

The successful functioning of the City depends on network utilities. It is therefore very important that construction, maintenance, upgrade and operation of these services be effectively provided for, technical and geographical constraints on the operation of network utilities are acknowledged and that the benefits that derive from them are adequately recognised. Network utilities can be vulnerable to reverse sensitivity effects when new buildings or structures and activities (that are sensitive to the effects of the existing network utility) are established nearby, leading to constraints on the operation of the network utility. However, network utilities can also have adverse environmental effects resulting from their construction, operation or associated maintenance activities.

For example, network utilities may typically include buildings, poles, overhead wires, pylons, pipes or antennas, which may have an adverse visual impact depending on their location and proximity to other land use activities. The installation and upgrading of network utilities will also typically involve earthworks. However, network utilities may also involve few structures and have limited visual impact, such as underground power and telecommunication lines. Network utilities are also often seen as a necessary and normal part of the environment, such as a road.

The network utility rules apply where network utility operators do not hold a designation for their activities under the designation procedures of the Act. They may, however, also be used by Council to help assess any notices of requirement for new designations.

The provisions in this Chapter apply to network utilities throughout all zones of the City. The underlying zone objectives, policies and rules do not apply to network utilities, including roads, unless specifically referred to. City wide rules, such as those relating to earthworks, notable trees, flooding and fault band hazards, the Southern Hills Overlay and Protected Ridgelines, historic heritage and hazardous substances will still apply. As identified in Council's Planning Maps, all activities that take place within roads are subject to the relevant zone rules which pertain to the area in which the road is situated, as well as the City-wide rules. Where a road separates different zones on opposite sides of the road, the centre line of the road defines the boundary of the two zones.

¹ <u>The Meteorological Service is a requiring authority for its network operation of a system comprising</u> telecommunication links to permit telecommunication and radiocommunication. Therefore, these aspects of meteorological service activities and facilities are network utilities. Comment [GS1]: S4.2 Transpower

Comment [GS2]: S8.1 Forest and Bird

Comment [GS3]: S1.5 GWRC
Comment [GS4]: S3.10 and S5.11

16.2 Resource Management Issues

16.2.1

The sustainable management of the City's utilities is important as they are necessary to the functioning of the City and provide the framework for future development and use of resources, including water resources.

<u>Balancing the national, regional and local benefits of network</u> <u>utilities with effects on the local environment</u>

An important issue is managing the actual and potential adverse environmental effects arising from network utilities while also recognising the key role they play and benefits they have locally, regionally and nationally, and to ensure the technical and operational requirements of the network utility concerned is not unreasonably restricted. Failing to adequately provide for network utilities may result in the desired level of well-being and quality of life not being achieved within the City. This is because network utilities provide essential services to people's homes and businesses, such as water, transport means, electricity, gas and telecommunications, and are critical for the effective functioning and liveability of the City.

The high voltage electricity transmission lines that form part of the national grid play a vital role in the well-being of the community. The adverse effects of the high voltage electricity transmission lines are often local, while the benefits may be in a different locality and/or extend beyond the local to the regional and national.

However, the construction, operation and/or maintenance of network utilities can have adverse effects and adversely affect the amenity of areas of the City, as a result of, for example, noise, emissions, and visual dominance. Some network utilities are relatively large, visually prominent and capable of generating significant adverse effects on the surrounding environment. They may also have potential or perceived adverse effects on public health and safety. Adverse effects may only occur at the time of construction or installation of the network utility, but in some instances may continue throughout its operation or during maintenance and upgrade works. In some cases, it might not be entirely possible to avoid, remedy or mitigate all adverse effects associated with a network utility, meaning there will may be some level of residual adverse effect on the surrounding environment. In such circumstances, there is a need to carefully consider both the benefits the network utility will provide and the significance of the adverse effects on the surrounding environment.

The City would not be able to function effectively or efficiently without essential facilities and services. Utilities are important physical resources that enable residents to provide for their economic, cultural and social well being and their health and safety.

The future well being of the City's population will continue to rely on major investment and sustainable management of utilities. This investment should be soundly managed and developed, taking into account the long term environmental values and needs of the Comment [GS6]: S4.3 Transpower Comment [GS7]: S3.2 Spark, FS2.2 Transpower; FS3.8 WELL Comment [GS8]: S4.3 Transpower

Comment [GS5]: S4.3 Transpower

community and the environment.

An urban area with an integrated public transportation system is more resource efficient than a City completely dependent on private vehicle use. The current scale and pattern of development in Upper Hutt is based on both private vehicle and public transport use. Private motor vehicles are used by more than half of Upper Hutt's workforce as a means to get to work. More than half of the work force travels outside of the City to work, using either public transport or private motor vehicles. Access to public transport is an important resource management issue.

Stormwater drainage systems are necessary, particularly in urban areas, for health and safety reasons. In low lying areas drainage is needed for healthy and acceptable living conditions. In urban areas, a stormwater drainage network with safe inlets, secures the community from flooding. River and stream control systems work to minimise human and economic losses from flooding. Utilities can be affected by flooding and it is also possible for them to adversely impact on flooding patterns.

The Council's sewerage system provides a safe, cost effective and socially acceptable system for removing human waste. An efficient sewerage system with properly designed and maintained pumping and drainage systems, sufficient treatment, and suitably located final effluent discharge points, is the best protection against pollution.

The high voltage electricity transmission lines that form part of the national grid play a vital role in the well-being of the community. The adverse effects of the high voltage electricity transmission lines are often local, while the benefits may be in a different locality and/or extend beyond the local to the regional and national.

16.2.2 Potential adverse effects on the environment caused by the development and use of utilities. Managing adverse effects including reverse sensitivity effects on regionally significant network utilities

Providing, locating and operating a network of facilities and services can impact upon both the natural and physical environment. As an example, adverse effects from the development of utilities may occur during their establishment when the visual impact of the facilities is not in harmony with environmental amenity. On the other hand, utilities often have special technical and operational requirements which place constraints on where they can be located.

Changes to the infrastructure of the City will need to be managed to avoid, remedy, or mitigate adverse effects on natural and physical resources and on amenity values, taking into account the practical requirements for the establishment and operation of utilities.

Inappropriate subdivision, use and development in the vicinity of regionally significant network utilities may lead to adverse effects including reverse sensitivity effects that have the potential to impact upon the effective and efficient operation of such utilities. Inappropriate subdivision, use and development may result in adverse effects on regionally significant network utilities and / or restrict access to such network utilities including the ability to undertake maintenance or upgrade work. Reverse sensitivity can occur when sensitive or inappropriate activities locate near to or intensify by existing network utilities. This may mean that the local, regional and national benefits of those regionally significant network utilities may be compromised. The City has a lot of well-established regionally significant network utilities located in close proximity to existing land use activities. The Council is predominantly concerned with new more intensive land use activities establishing in proximity to existing regionally significant network utilities that may lead to reverse sensitivity effects on those utilities.

The efficient, convenient and safe movement of people, vehicles and goods in the City.

16.2.3

The location, design and characteristics of activities, subdivision and development can adversely affect the safety, accessibility and efficiency of the roading network and the quality of the environment. Appropriately located activities, and well-designed subdivision and development, can contribute to the convenience and viability for access by walking, cycling and public transport. Roads themselves (including the State Highway network) contribute to the convenience, viability, and access to activities enjoyed by City residents.

16.2.4 The limits that rural roading places on subdivision and development

Mangaroa Hill Road, Blue Mountains Road, Akatarawa Road, and parts of Moonshine Hill Road and Mount Cecil Road require major upgrading to be able to accommodate further significant development. Such upgrading may have significant adverse environmental effects.

The limits that the rural roading system places on further development not only apply to formed and sealed roads, but also to the large number of 'paper roads' within the City. The pressure to subdivide with access to these paper roads can pose a public interest issue. Council could be placed in a position where it may have to spend public money on road upgrading where the community would receive little benefit in return. The limited access provisions applying to State Highway 2 and the nature of other roads in the Kaitoke area impose limits on further development in this part of the City.

16.2.5 The potential adverse effects generated by subdivision and development in close proximity to high voltage (110kV or greater) electricity transmission lines.

There can be a risk to the health and safety of nearby people and property when development occurs within close proximity to high voltage electricity transmission lines. Equally, development located under or in close proximity to high voltage electricity transmission lines can pose a risk to the efficient operation of the national grid.

Additionally, development in close proximity to high voltage electricity transmission lines generally does not provide a good level of amenity, particularly in the case of residential development or other sensitive activities.

16.3 Objectives

16.3.1

The sustainable management of physical infrastructure so that it can meet both the needs of today's community and the reasonably foreseeable needs of future generations.

Infrastructure is an essential part of the City's physical resources, and its maintenance is essential to the social, economic, and cultural well-being of people, as well as to their health and safety. Efficient resource use relates not only to the shorter term economic use of resources, but also to the longer term environmental effects of resource use.

An efficient city needs effectively operating utilities. The location and operation of these utilities can adversely affect the environment.

Many sites for utility operations will be provided by statutory designations.

<u>16.3.1</u> <u>To recognise and protect the benefits of regionally significant network</u> <u>utilities and ensure their functions and operations are not unreasonably</u> <u>compromised by other activities.</u>

This objective seeks to identify the importance of regionally significant network utilities within the City and to give effect to the Regional Policy Statement. The objective and supporting policies are focused on recognising the benefits that these regionally significant network utilities have locally, regionally and nationally and ensuring that these benefits they are protected from incompatible subdivision, use and development.

16.3.2 The sustainable, secure and efficient use and development of the high voltage (110kV or greater) electricity transmission lines which avoids, remedies or mitigates adverse effects on the environment and recognises the technical and operational requirements and constraints of the network.

The efficient transmission of electricity on the national grid plays a vital role in the social, economic and cultural well-being of people. Technical, operational and security requirements associated with high voltage electricity transmission lines can limit the extent to which it is feasible to avoid or mitigate all adverse environmental effects. Comment [GS9]: S4.6 Transpower

Comment [GS10]: S4.6 Transpower

Comment [GS11]: S4.6 Transpower
16.3.3 To recognise and provide for the sustainable, secure and efficient use, operation, maintenance and upgrading and development of network utilities within the City.

This objective requires that the benefits of network utilities including those that are not identified as regionally significant be recognised and provided for. Network utility services form an essential part of the City's physical resource and provide for the community's social, cultural and economic well-being. They provide essential services to people's homes and businesses, such as water, transport means, electricity, gas, radiocommunications and telecommunications and are critical for the functioning and liveability of the City. Failing to adequately provide for network utilities may result in the desired level of well-being and quality of life not being achieved within the City.

16.3.4 To manage any adverse effects on the environment resulting from the design, location, construction, operation, upgrading and maintenance of network utilities.

This Objective recognises that the construction, operation, upgrade and maintenance of network utilities can adversely affect the environment and amenity, and seeks to manage potential adverse effects, particularly through design and location. This recognises that some network utilities are relatively large, visually prominent and capable of generating significant effects on the environment. They may also have adverse effects on public health and safety. Adverse effects may only occur at the time of construction or installation of the utility, but in some instances may continue throughout its operation or during maintenance and / or upgrade works. For new lineal infrastructure, adverse effects are often best able to be mitigated through the route selection process. However, in some cases, it might not be entirely possible to avoid, remedy or mitigate all adverse effects associated with a network utility, meaning there will may be some level of residual adverse effect on the surrounding environment that requires mitigation. <u>In such circumstances, there is a need to consider</u> both the benefits the network utility will provide and the significance of the adverse effects on the surrounding environment.

16.4 Policies

16.4.1 To ensure that the establishment, operation, maintenance and upgrading of essential utilities in the City avoids, remedies or mitigates any adverse environmental effects.

The City is dependent upon the efficient provision of facilities and services to maintain people's health and safety and to support economic, social and cultural activities. There should, therefore, be opportunity to provide for the development, maintenance and use of essential facilities and services, as long as they can meet, or do not significantly compromise, environmental standards. This includes standards to address the potential adverse effects that are specific to utilities and services, such as the impacts of transmission lines and services will be promoted where appropriate, in order to help avoid, remedy or mitigate adverse effects, especially on visual amenity. Where utilities are required to be sited above ground, the site or route selection process should seek to minimise adverse effects.

Comment [GS12]: S4.8 Transpower

Comment [GS13]: S8.3 Forest and Bird

Comment [GS14]: S3.6, S4.9, S5.6 and S6.13, Spark, Chorus, Transpower and WELL and FS1.5, FS2.6, FS3.9, FS3.14 and FS3.30 Transpower, Powerco, WELL Subdivision may be required to provide for special facilities or activities (such as substations, transformer sites, pumping station sites, and roads). There are therefore subdivision standards in each zone which recognise the special subdivision requirements for utilities.

Utilities can both be affected by flooding, and affect flood patterns. Therefore, it is appropriate that the location of utilities in flood prone areas is regulated (Chapter 33).

<u>Identify regionally significant network utilities within the City on Council</u> <u>planning maps, as practicable.</u>

The Council has identified regionally significant network utilities within the City on its planning maps. The majority of any new and extensions to existing regionally significant network utilities are expected to be identified on Council planning maps by network utility operators through a notice of requirement for designation process. In the case of the National Grid, which is not designated, this network will be is specifically recognised and mapped, as required by the National Policy Statement on Electricity Transmission. Due to the scale of the planning maps and the extensive nature of some regionally significant network utilities, it is however not feasible to identify all regionally significant network utilities on Council planning maps, particularly the local gas distribution lines.

16.4.2 To require an assessment of alternative methods and sites for proposed utility developments which have significant implications for resource use, or which may cause significant adverse environmental effects.

The development of the City's infrastructure can have major implications for resource use. Large new proposals, such as a new energy distribution facility, highway, or sewage treatment plant, can be significant users of resources, and can cause significant adverse environmental effects. They can also have implications for other aspects of the City's economic activity and social life. For example, a major improvement to the roading network will generate changes in vehicle use, and open up opportunities for land development in some areas and may foreclose opportunities in other areas. Major developments in the City's infrastructure should therefore be carefully managed to promote sustainable management.

Consideration of alternative methods and sites is an important means of promoting sustainable resource use, and should be undertaken for those activities which have significant resource use implications. The broader considerations of the use of these resources should be integrated into the District Plan process.

<u>Recognise the national, regional and local benefits of regionally</u> <u>significant network utilities.</u>

Regionally significant network utilities provide benefits within the City, regionally and nationally. These are benefits that are to be considered in respect of any matter relating to regionally significant network utilities. Some of the benefits are:

- <u>That people and goods can travel to, and from and around the City and</u> <u>Region efficiently and safely;</u>
- That community well-being and public health and safety is maintained through the provision of essential services including supply of potable water and the collection, transfer and appropriate treatment of sewage and stormwater; and
- <u>People have access to electricity and gas to meet their needs.</u>

Comment [GS15]: S4.10 Transpower

16.4.3 To promote the efficient and effective use of existing utilities and to encourage the co-siting of compatible facilities.

Utilising the spare capacity of existing systems can be a more efficient use of resources than the development of new systems. Where such use can be demonstrated to be practicable and reasonable, it should be promoted. This policy is not intended to prevent competition between resource users or to prohibit growth; it simply seeks to promote the efficient use of natural and physical resources. For example, urban infrastructure will generally be used more efficiently if new residential developments are contained within existing urban areas, rather than dispersed or ribbon forms of development. Consolidation of settlements promotes the efficient use of existing infrastructure capacity and facilitates the viable introduction of new facilities. Council intendes to support this policy by promoting costing and charging regimes for Council's services and utilities which take into account the longer term environmental implications of resource use and development.

Questions of long term efficiency should be addressed in the planning of new infrastructure systems, and this can be promoted through liaison with infrastructure providers and by requiring the assessment of such matters in designation, subdivision and land use consent processes.

Co-siting of facilities can reduce adverse effects on the environment by reducing the proliferation of similar facilities. Co-siting may be promoted by consultation between relevant organisations, and by ensuring that such matters are addressed in designation and resource consent processes.

Avoid, or as appropriate, remedy or mitigate, the potential for any adverse effects including reverse sensitivity effects on regionally significant network utilities from incompatible inappropriate new subdivision, use and development occurring under, over, or adjacent to regionally significant network utilities.

Any potential adverse effects including reverse sensitivity effects, on regionally significant network utilities are to be appropriately managed, with priority given to avoiding adverse effects, where practicable, on those utilities. The location of inappropriate new subdivision, use or development in proximity to existing regionally significant network utilities has the potential to compromise the efficient operation and use of the network utility including restricting access and result in the benefits of that network utility being reduced. In addition, the safety and amenity values of the community may be adversely affected by locating in too close proximity to regionally significant network utilities. The potential for adverse effects including reverse sensitivity effects may arise when the pattern and density of land use activities changes through the subdivision or rezoning of land. At the time of rezoning, the Council will seek to introduce new provisions to manage those potential adverse effects on existing or designated regionally significant network utilities. Any applications for subdivision that involve potential intensification located in proximity to regionally significant network utilities will require assessment in terms of the potential effects on those utilities as well as consultation with the relevant network utility operator.

16.4.4 To promote the safe and efficient use and development of the transportation network.

The transportation network is a major physical resource in the City. The land and other resources used for transportation need to be sustainably managed.

There are a number of reasons for promoting a safe and efficient transportation network, including:

Comment [GS16]: S4.12 Transpower

- The land and associated resources required by the existing transport system represent a significant level of investment and commitment. To promote the purpose of the Act, it is desirable that existing systems are used and developed efficiently.
- The efficient use of energy and resources in the design, management and use of transportation systems should be promoted.
- Efficiency would be promoted through the integration of different modes and types of transport and by improving the network. The beneficial effects of any development such as increases in safety or reduction in travel times must also be taken into account.

The transport system also needs to be maintained and developed without creating significant adverse effects on the environment.

Minor changes to the transport system can be undertaken without the need for stringent controls, although they may need monitoring to identify and manage their cumulative effects. However, large-scale transportation projects, or developments within areas of environmental sensitivity, require careful assessment to identify potential effects and possible mitigation measures.

Rural roads place limits on further development in certain areas of the City because of their condition or potential capacity. Closer subdivision in these areas may be restricted because of the demand that it would place on these roads and the likely costs incurred by the Council. Other rural activities can cause damage to roads or create dangerous situations where roads are not designed to accommodate such traffic. The upgrading of such roads can place a heavy financial burden on the community and have significant adverse environmental effects. Therefore, a requirement for financial contributions and/or limitations on development is an appropriate response.

16.4.5 To promote accessibility within the City and between the City and neighbouring areas.

Access into and around the Central Business District, suburban shopping areas and industrial areas is important for both businesses and the community. This is facilitated by the availability of adequate car parking facilities and the close proximity of railway stations and the bus interchange.

In certain circumstances, car parking can have an adverse effect on the environment of an area. The proliferation of on-street car parking can adversely affect the visual and amenity values of an area, generate noise and make manoeuvring of vehicles difficult and unsafe. It may be necessary to require that sufficient on-site car parking is provided for any proposed activity, or that a financial contribution is made so that it can be developed by Council to avoid problems. The car parking requirements of the Plan have been developed on the basis of anticipated car parking demand and availability of car parking facilities.

The ability of people to have access to a variety of transport modes enables greater choice and means that transportation services can be used in a more efficient manner.

Most of Upper Hutt is well suited to cycling because of its topography. The bicycle is a useful, efficient and environmentally friendly form of transport. Although cycling is catered for within the present roading system, conflict can arise between cyclists, pedestrians and vehicular traffic. These conflicts need to be minimised or avoided to promote safety and encourage people to use dedicated cycling facilities. Convenient cycling and walking routes to community focal points need to be provided by linking streets, reserves, car parking areas and shopping centres.

16.4.6 To ensure that the subdivision, use and development of land is served by safe and adequate access from the roading network

The roading network provides access to a wide range of activities. It is important to ensure that connections to the network are located, designed and maintained so as to provide for the safety of all road users.

16.4.7 To manage subdivision and development within close proximity to existing high voltage (110kV or greater) electricity transmission lines to protect both:

- (a) the safe, secure and efficient use and development of the electricity transmission network; and
- (b) the safety and amenity values of the community.

A corridor management approach involves setting minimum buffer distances from high voltage electricity transmission lines to manage development both in the immediate proximity of and adjacent to the lines.

<u>16.4.8</u> <u>To recognise and provide for the:</u>

- <u>need for new and the maintenance and upgrading of existing</u> <u>network utilities; and</u>
- <u>technical and operational requirements and constraints of network</u> <u>utilities in assessing their location, design, development,</u> <u>construction and appearance; and</u>
- <u>benefits that network utilities provide to the economic, social and</u> <u>cultural functioning of the City, Region and Nation.</u>

The provision of new and the upgrading of existing network utilities is necessary to meet the needs of City, both now and into the future. In considering any proposals for new or upgrades to existing network utilities, the technical and operational requirements that may constrain where and how they can locate and be designed need to be recognised. In some cases, some level of adverse effects may need to be accepted to recognise the necessity for and benefits derived from, some network utilities and meet their operational requirements. This policy also recognises the benefits that all network utilities have.

<u>16.4.9</u> Enable the efficient construction, installation, operation, upgrading and maintenance of network utilities.

Network utilities have an important role in providing for the wellbeing of the City's community and beyond. Network utilities form an essential part of the efficient functioning of the City and their maintenance and development allows their benefits to be realised. There are a range of network utilities that enable communities to undertake everyday activities and functions and provide essential services to people's homes and businesses. It is therefore important that the District Plan provides for network utilities to be constructed, installed, operated, upgraded and maintained.

<u>16.4.10</u> Ensure that the provision and operation of utilities that cross jurisdictional boundaries is managed in an integrated manner.

Most network utilities cross jurisdictional boundaries between councils. Cross boundary issues can result for network utility providers and for the community, particularly where different councils have different rules or processes for how they recognise and provide for network utilities and manage their effects. It is important that councils work together in an integrated manner both when Comment [GS17]: S4.14 Transpower developing plan provisions and when dealing with proposals for new or upgrades to existing network utilities.

<u>16.4.11</u> Encourage the appropriate use of designations for new network utilities and extensions to existing network utilities that are not designated.

Network utility operators, particularly those who operate regionally significant network utilities, should use the notice of requirement for designation process, where appropriate when they seek to develop new or extend existing network utilities. This is particularly encouraged for operators where such new or extended network utilities involve restrictions on the use of privately owned land and may require land acquisition. It is recognised that not all network utility operators use designations, particularly those that do not operate linear infrastructure.

<u>16.4.12</u> Ensure that network utilities are designed, developed, constructed, located, upgraded, operated and maintained to avoid, remedy or mitigate any actual or potential adverse effects on the environment,

There are a range of different network utilities with different potential adverse effects on the environment. For instance, above ground network utilities can have adverse effects including visual, noise, traffic, odour and amenity, depending on their size, location, frequency and their scale in comparison with the character of a particular environment. A different activity status applies to some network utilities in the Southern Hills Overlay Area, Open Space and Residential Zones, to reflect that these zones have special environments that are more vulnerable to adverse effects and associated loss of amenity.

16.4.13 <u>Ensure Manage effects on health and safety by ensuring network utilities,</u> in particular those emitting electric and magnetic fields, are designed, located, upgraded, operated and maintained to comply with relevant national environmental standards and to meet other nationally recognised standards and guidelines.

Some network utilities may adversely affect health and safety. For example, telecommunication facilities generate radio frequency emissions which may have detrimental effects on health. Any potential health effects arising from radiofrequency emissions are addressed by Regulation 4 of the Resource Management (National Environmental Standards for Telecommunication Facilities) Regulations 2008. Electricity transmission/distribution activities can generate electromagnetic fields (EMF) which may be present a risk to health and safety, primarily through and also generates the risk of electrocution from direct contact with conductors or as a result of a flashover. The National Policy Statement on Electricity Transmission, and the National Environmental Standard for Electricity Transmission require that the exposures be limited to the guidelines of the International Commission on Non-Ionising Radiation Protection (ICNIRP) to prevent the potential for public health effects. Other possible health and safety risks are accidental spillage or leakage of hazardous substances from gas or petroleum pipelines, explosions from gas or petroleum pipelines, accidental overflow from sewage pump stations, and flooding from damaged/inoperative stormwater systems. Chemicals used in conjunction with some network utilities, such as water treatment plants for example, also pose a risk if an accidental spill occurs. There are a number of relevant national and international standards and guidelines addressing health and safety matters that are external to the District Plan but that must be complied with, including Resource Management (National Environmental Standards for the Telecommunication Facilities) Regulations 2008, and the New Zealand Electrical Code of Practice. and The International Commission on Non-Ionising Radiation Protection (ICNIRP) Guidelines provide best practice guidance.

Comment [GS18]: S4.16 Transpower

Comment [GS19]: S7.15 Powerco

Comment [GS20]: S7.15 Powerco

Comment [GS21]: S7.15 Powerco

Comment [GS22]: S7.15 Powerco

Comment [GS23]: S4.18 Transpower

Comment [GS24]: S7.15 Powerco

Comment [GS25]: S7.15 Powerco

16.4.14 Enable the co-location or multiple use of network utilities where this is efficient, technically feasible and practicable and assists with avoiding, remedying or mitigating adverse effects on the environment.

The co-location and co-siting of network utilities may provide environmental benefits in terms of reduced visual impacts and consolidation of network utilities in existing areas thereby reducing adverse effects on amenity by reducing the need for more network utility structures. While co-location is encouraged it needs to be understood that technical requirements will generally mean that associated structures may need to be taller or bulkier to avoid interference between the two or more providers, such as radio-frequency bands. It is also recognised that co-location is not always possible due to operational issues such as radiofrequency interference, electrical interference, lease arrangements, safety and structural capacity.

Require the underground placement of new network utilities unless

16.4.15

- there are natural or physical features or structures, or technological and operational constraints that makes underground placement impractical or unreasonable;
- <u>they are of a temporary nature and required for emergency purposes</u> <u>or critical events; and</u>
- <u>they are of a nature that they can only operate aboveground.</u>

The adverse visual effects of certain network utilities can often be managed by putting the services underground. **With some exceptions, this** is the required approach for those network utilities, such as those with cables that can be located underground. For those network utility structures that need to be located aboveground, particular attention should be given to their design, location and minimising of any adverse visual effects as outlined in Policy 16.4.12. This can be achieved in a number of ways including, where practical, through screening, careful placement, size and appearance and applying different activity status. With the exception of Protected Ridgelines and the Southern Hills, new overhead lines, including electricity lines below 110kV, are provided for as a permitted activity in the Rural and Open Space Zones provide for overhead lines, recognising the more visual absorptive capacity of those locations, and the practicality of needing to provide for a cost effective means of enabling service development and maintenance in remote less densely populated areas.

<u>New customer connections to existing lines and minor upgrading of existing lines</u> are provided for within the City in recognition that this is an efficient use of an <u>existing resource. However, new</u> above ground lines and their associated supporting structures in areas that do not have existing above ground lines are generally considered to be unacceptable within the City. However it is recognised The policy recognises that particular consideration needs to be given to the efficient use of resources and that there are situations where placing lines underground is, or may be, impracticable or unreasonable.

<u>16.4.16</u> Encourage the use of roads as network utility corridors in accordance with the National Code of Practice for Utility Operators' Access to Transport Corridors.

Locating network utilities in the road corridor can assist to minimise the adverse effects of network utilities on amenity and other values as these locations generally have a range of existing network utilities and are less sensitive to new network utilities. However, the effects of these activities require some management to ensure conflicts with the primary function of the road corridor and with each other are avoided.

<u>16.4.17</u> <u>Encourage network utility providers to consult with local communities on</u> <u>the appropriate placement, location and design of new network utilities.</u>

Comment [GS26]: S4.19 Transpower

Comment [GS28]: S5.10 Chorus

Comment [GS27]: S5.10 Chorus

Comment [GS29]: S5.10 Chorus

In some cases, engaging early with the community about a proposed new network utility may result in an alternative more appropriate location to be identified that both meets the needs of the network utility operator and addresses any concerns that the community may have. In encouraging consultation, the Council recognises that it cannot require network utility operators to consult on permitted activities.

16.5 Methods

16.5.1

District Plan provisions consisting of the following:

- 1. Provision for the operation of_utilities by designations and rules. Planning Maps that identify the location of both designated and undesignated regionally significant network utilities within the District to the extent practicable.
- 2. Designations for State Highways and railways. Encourage designations for new network utilities and extensions to existing network utilities that are not currently designated
- 3. Management of the location of traffic generating uses through zoning rules and the resource consents process to avoid, remedy or mitigate adverse effects on the safety and efficiency of the transport system.
- 4. Performance standards to avoid, remedy or mitigate any adverse effects on the environment associated with the provision and operation of the City's infrastructure. Regulatory Assessment Framework that includes rules and matters of control and discretion to guide assessment of the construction, operation, upgrading and maintenance of network utilities,=and inappropriate subdivision, use and development within Electricity Transmission Corridors and subdivision that occurs in proximity to regionally significant network utilities. The framework utilises permitted, controlled, restricted discretionary, discretionary and non-complying activity status and specific matters of control and discretion to assess and manage the actual and potential adverse effects.
- 5. The consideration of viable alternatives for new designations. Plan change(s) to introduce new provisions to manage reverse sensitivity effects on regionally significant network utilities where there are pressures for new or intensification of existing development in proximity to regionally significant infrastructure.
- 6. Provision of appropriate infrastructure at the time of subdivision.
- 7. Financial contributions for the upgrading or extension of public infrastructure, or the avoidance, remedying, or mitigation of any adverse effects on public infrastructure.
- 8. Identification of designations on the Planning Maps and the inclusion of their details in Chapter 36.
- 9. Identification of high voltage (110kV or greater) electricity transmission lines on the Planning Maps.
- 10. Management of buildings, structures, earthworks and vegetation within a determined transmission corridor either side of the centreline of high voltage (110kV or greater) electricity transmission lines.
- 11. Administer, monitor and enforce compliance with the Resource Management (National Environmental Standards for Telecommunication Facilities) Regulations 2008 and the Resource Management (National Environmental Standards for Electricity Transmission Activities) Regulations 2009.

Application of any applicable National Environmental Standards (NES) such as the NES for Electricity Transmission Activities.

- 12. Information requirements for certificates of compliance and resource consents for network utilities.
- Advice notes identifying the relevant national and international codes and standards that also apply to network utilities in addition to the District Plan's Regulatory Assessment Framework.
 Monitoring and review of the District Plan network utility
- 14. Monitoring and review of the District Plan network utility provisions to assist in assessing the effectiveness of the network utilities provisions in the Plan.
- **16.5.2** Planning for the efficient development of infrastructure by liaison and consultation with requiring authorities.
- **16.5.3** Council provision and maintenance of transportation infrastructure through its Annual and Strategic Plans and the strategies of roading infrastructure providers.
- **16.5.4** The Code of Practice for Civil Engineering Works.
- **16.5.5** Efficient management of Council's works and utilities. This would include Council demonstrating a leadership role in the sustainable management and use of its infrastructure.
- **16.5.6** Consultation with Transpower when applying policies relating to the transmission network within the City with the potential to adversely affect transmission assets.
- **16.5.7** Compliance with relevant national and international codes and standards that also apply to network utilities.
- **16.5.8** Education of and building relationships with network utility providers.
- 16.5.9
 Encouraging network utility providers to engage with the local community when considering new network utilities within the City.
- 16.5.10
 Where appropriate, hold joint hearings with adjacent territorial authorities in instances where network utilities cross territorial boundaries and undertake joint plan changes to achieve consistency.

Anticipated environmental results and monitoring

The following results are expected to be achieved by the objective, policies and methods in this Chapter. The means of monitoring whether this Plan achieves the anticipated environmental results are also set out below.

Anticipated environmental results	Monitoring indicators	Data source
The sustainable management of <u>network</u> utilities throughout the City	Complaints and enforcement procedures	Council complaints register
	Types of <u>network</u> utility development	Council records
	System failures	
	Consultation with regional and national organisations to ensure network utilities management is co-ordinated and consistent	
The avoidance, remedying, or mitigation of the adverse effects of developing and	Complaints and enforcement procedures	Council complaints register
maintaining the City's <u>network</u> utilities	Assessment of the effectiveness of selected methods in implementing policies relating to <u>network</u> utilities	Council resource consent records
The avoidance of potential conflicts between regionally significant network utilities	Complaints and enforcement procedures	<u>Council complaints</u> <u>register</u>
and incompatible development, use and subdivision	Assessment of the effectiveness of selected methods in implementing policies relating to regionally significant network utilities	<u>Council resource</u> <u>consent records</u>

16.6

Amendment 2:

Insert new Chapter 16A: Renewable Energy Issues Objectives and Policies

16A RENEWABLE ENERGY GENERATION

<u>16A.1</u> <u>Background</u>

Energy is vital to the efficient functioning of our country. As both a natural and physical resource, the generation and use of energy is a relevant resource management matter. In particular, section 7(j) of the Resource Management Act requires decision-makers to give particular regard to the benefits derived from the use and development of renewable energy.

This chapter is focused on renewable energy generation; and in particular, the conversion of natural resources into electricity. Energy generation from non-renewable sources, energy efficiency and energy conservation are addressed in other Plan chapters.

The National Policy Statement for Renewable Electricity Generation came into force in May 2011 and forms part of central government's strategic target to achieve 90% of electricity generated is from renewable energy sources by 2025. The Council is required to give effect to any national policy statement. The objective of the National Policy Statement is to recognise the national significance of renewable electricity generation activities by providing for the development, operation, maintenance and upgrading of new and existing renewable electricity generation activities, such that the proportion of New Zealand's electricity generated from renewable energy sources increases to a level that meets or exceeds the New Zealand Government's national target for renewable electricity generation. Policy D of the National Policy Statement requires that decision makers shall, to the extent reasonably possible, manage activities to avoid reverse sensitivity effects on consented and on existing renewable electricity generation activities.

The Regional Policy Statement for the Wellington Region recognises the importance of energy within the Region, and in particular that the benefits from renewable energy, from small to large scale generation, be recognised. Upper Hutt City Council is required to give effect to the Regional Policy Statement and include policies and/or methods to ensure these benefits are recognised. These benefits include security of supply and diversification of energy sources, reducing dependency on imported energy sources and reducing greenhouse gas emissions. Although energy is essential to our communities, its generation has the potential to have both adverse and positive environmental effects. The positive effects of renewable energy generation are often felt nationally (and/or globally) as well as locally, however adverse effects are generally more localised. Given the nature of renewable energy generation activities, many developments are unlikely to be able to internalise all potential adverse effects that they may generate within the site, and may include effects on amenity, landscape, ecology, cultural and heritage values. The benefits of any development therefore need to be weighed up against potential adverse effects.

The application of renewable energy can be in a number of different forms. At the domestic or small scale, there are various passive approaches including orientating buildings to assist passive heating and cooling, and natural lighting and more active approaches such as solar water heating and panels, and wind turbines. The passive approaches are addressed through other chapters in the District Plan. This chapter addresses the active approaches to renewable energy generation, primarily in the form of electricity generation. At present, renewable energy generation facilities in Upper Hutt City are limited to small scale wind turbines and solar panels. There are no existing larger scale renewable energy generation activities within the <u>City.</u>

A study of the Wellington Region's renewable energy sources undertaken by the Energy Efficiency and Conservation Authority in August 2006 identifies wind and solar as the primary renewable energy sources within Upper Hutt City that fall within the functions of the Upper Hutt City Council. Other options such as large scale solar generation or biomass energy may become more viable in the future, and this chapter should be amended to address the various resource management issues that may arise as more information about these options becomes available. This chapter therefore focusses on wind and small-scale solar resources.

The provisions in this Chapter apply to renewable energy generation activities throughout all zones of the City. The underlying zone objectives, policies and rules do not apply to renewable energy generation activities unless specifically referred to. City wide rules, such as those relating to historic heritage, notable trees, earthworks and hazardous substances will still apply.

16A.2 Resource Management Issue

<u>16A.2.1</u> <u>Balancing conflicts created by the effects of renewable energy</u> generation with its local, regional and national benefits.

Upper Hutt City contains some renewable energy resources, the use of which would provide benefits locally, regionally and nationally. However, the use of such resources can also have adverse environmental effects, which are generally felt at a local level. These effects can create conflicts between renewable energy generation and a wide range of biophysical and community held values, and raise issues of scale and location.

16A.3 Objectives

- <u>16A.3.1</u> Provide for the development of renewable energy generation that is designed, located, constructed, operated, maintained and upgraded so as to:
 - a) Avoid, remedy or mitigate adverse effects on the <u>environment; and</u>
 - b) Promote the local, regional, and national benefits of the use and development of renewable energy resources.

There are significant benefits available at the local, regional and national level from renewable energy generation activities. The benefits from any renewable energy generation proposal can range from large significant contributions, to small incremental gains, but they are all cumulative. New renewable generation capacity can contribute to the New Zealand Energy Strategy target, and to increasing the diversity of supply. The City offers opportunities in particular for wind and solar generation.

The nature and scale of effects from renewable energy generation will vary depending on the scale and location of the activity and the characteristics of the surrounding area. Potential effects include adverse visual impacts, impacts on indigenous flora and fauna, culturally and historically significant areas and noise effects.

The District Plan provides for renewable energy generation activities at different scales, ranging from commercial-scale activities where the purpose is to generate electricity for financial gain to small scale activities which provide for domestic energy needs. Different activity statuses are used to reflect the nature, scale and associated levels of effects of the different scales of renewable energy generation activities.

16.A.3.2 Enable small-scale renewable energy generation and the identification and assessment of potential renewable energy sources and sites in appropriate locations within the City.

This objective recognises that there are parts of the City where smallscale renewable energy generation activities may be viable to serve the electricity needs of individuals and local areas. The objective also recognises the importance of enabling the identification and assessment of potential renewable energy sources and sites for the Upper Hutt community.

16A.4 Policies

<u>16A.4.1</u> <u>Recognise the local, regional and national benefits of renewable</u> <u>energy generation activities.</u>

Renewable energy generation activities have the potential to provide a range of benefits within the City, regionally and nationally. These benefits need to be considered by decision makers when determining any application for a new renewable energy generation activity within

the City. Some of the benefits of renewable energy generation activities include:

- <u>Maintaining and increasing security of electricity supply:</u>
- Using renewable rather than finite energy sources;
- Reducing dependency on imported energy sources;
- <u>Reducing greenhouse gas emissions; and</u>
- <u>The reversibility of the adverse effects of some renewable energy</u> <u>generation technologies.</u>

16A.4.2 Enable small-scale renewable energy generation to be developed and operated in a manner that avoids, remedies or mitigates adverse environmental effects.

Small-scale renewable energy generation particularly from wind and the sun provides an opportunity for Upper Hutt City to become partially self-reliant for energy supply. Technologies that currently exist and enable individuals to harness energy sources are likely to improve and become more cost-effective into the future. As demand for energy increases, self-reliance will have economic benefits for both individuals and the City.

Small-scale renewable energy generation activities may still have adverse environmental effects that should be avoided, remedied or mitigated. The District Plan includes such controls as are necessary to manage the potential effects of small-scale renewable energy generation activities.

<u>16A.4.3</u> Enable the identification and assessment of potential renewable energy sources and sites in a manner that avoids, remedies or mitigates adverse environmental effects.

A precursor to developing renewable energy generation activities is identifying and subsequently assessing potential renewable energy sources. This may require testing over a number of years, for instance, to determine whether wind speeds are of an appropriate velocity and are consistent enough to efficiently and effectively generate electricity.

While the equipment needed for identification and assessment is generally of a lesser overall scale than a renewable energy generation facility itself, the equipment has the potential to generate adverse environmental effects that need to be appropriately managed. Accordingly, the District Plan includes controls as are necessary to avoid, remedy or mitigate adverse effects.

<u>16A.4.4</u> <u>Provide for the operation, maintenance and development of community scale and commercial scale renewable energy generation activities.</u>

Some parts of the City, particularly those that are elevated, are potentially suited to the development of renewable energy sources, particularly commercial scale renewable energy generation activities where wind is the energy source. There is little potential in the City for the community or commercial scale use of solar energy for electricity generation. 16A.4.5 <u>Manage the adverse environmental effects of community scale and</u> <u>commercial scale renewable energy generation activities by</u> <u>recognising that these activities have the potential to cause</u> <u>significant adverse effects on the environment. In particular,</u> <u>activities that use wind as a source of energy have the potential</u> <u>for significant adverse effects on landscape, ecology and amenity</u> <u>values, and noise (including any low frequency noise) and may be</u> <u>inappropriate in some locations.</u>

> While the Plan permits some renewable energy generation activities, those that are of a significant scale or do not meet the standards to be a permitted activity, will require a full assessment of their environmental effects through the resource consent process. This will allow the Council to weigh the benefits of any new generation activity with its environmental effects. A non-complying activity status applies to activities located within the Southern Hills Overlay Area and on sites containing listed Heritage Features and where turbines do not comply with NZS6808:2010, as these are likely to have significant adverse effects on the environment.

16A.4.6 <u>Recognise the technical and operational constraints of renewable</u> <u>energy generation, including the location of the resource,</u> <u>development and maintenance of facilities and the location of the</u> <u>electricity distribution network.</u>

The locational, functional and technical constraints on the siting of renewable energy generation facilities also need to be considered when development proposals are assessed and conditions of consent are imposed. This recognition includes considering the need for renewable energy generation facilities to be located where such resources are available and the location of existing structures and infrastructure. The District Plan includes relevant matters of discretion which ensures the Council recognises these technical and operational constraints associated with any application.

16A.4.7 Encourage community and commercial scale renewable energy generation providers to consult early with the local community, including Māori, on the appropriate placement, location and design of renewable energy generation activities.

In some cases, engaging early with the community about a proposed new renewable energy generation activity may result in an alternative more appropriate location to be identified that both meets the needs of the developer and addresses any concerns that the community may have. In encouraging consultation, the Council recognises that it cannot require providers or individuals to consult on permitted activities.

16A.4.8Protect consented and existing renewable energy generation
activities from incompatible subdivision, land use and
development.

The Plan recognises that new subdivision, land use and development activities can result in reverse sensitivity effects on existing and consented renewable energy generation facilities and may result in the benefits of facilities being reduced. In addition, community amenity values may be adversely affected by locating in too close proximity to renewable energy generation facilities.

At present, the City only contains small scale renewable energy generation facilities with no established community scale or commercial facilities. It is likely any such larger scale facilities would be established in the rural areas and that any reverse sensitivity effects would arise from subsequent new subdivision, land use and development which would provide an opportunity for assessment of any such potential effects.

16A.4.9 Ensure that the provision and operation of renewable energy generation activities that cross jurisdictional boundaries is managed in an integrated manner.

Some renewable energy generation activities and the effects arising from them may cross jurisdictional boundaries between councils. Cross boundary issues can result for renewable energy generation providers and for the community, particularly where different councils have different rules or processes for how they recognise and provide for renewable energy generation activities and manage their effects. It is important that councils work together in an integrated manner both when developing plan provisions and when dealing with proposals for new or upgrades to existing renewable energy generation activities.

16A.5 Methods

- **16A.5.1** District Plan provisions which include a Regulatory Assessment Framework that includes rules and matters of control and discretion to guide assessment of renewable energy generation activities. The framework utilises permitted, controlled, restricted discretionary, discretionary and non-complying activity status and specific matters of control and discretion to assess and manage the actual and potential adverse effects.
- **16A.5.2** Plan change(s) to introduce new provisions to manage reverse sensitivity effects on renewable energy generation activities where there are pressures for new or intensification of existing development in proximity to consented or existing renewable energy generation activities.
- **16A.5.3** Monitoring and review of the District Plan renewable energy generation provisions to assist in assessing the effectiveness of the renewable energy generation provisions in the Plan.
- **16A5.4** Education of, and building relationships with, renewable energy generation providers.
- **16A5.5** Encouraging renewable energy generation providers to engage with the local community when considering new renewable energy generation activities within the City.
- **16A5.6** Education of the local community about renewable energy.

<u>16A.6</u>

Anticipated environmental results and monitoring

Anticipated environmental results	Monitoring indicators	Data source
The benefits to be derived to Upper Hutt City from renewable energy generation activities are realised.	Types of renewable energy generation activities Consultation with regional and national organisations to ensure renewable energy generation activities are co-ordinated and consistent	<u>Council records</u> <u>Customer surveys</u> <u>Council resource</u> <u>consent records and</u> <u>monitoring compliance</u>
<u>The Upper Hutt City</u> <u>community is able to</u> provide for its social, <u>economic, cultural and</u> <u>environmental wellbeing.</u>	<u>Types of renewable energy</u> <u>generation activities located within</u> <u>Upper Hutt City</u> <u>Level of update of renewable energy</u> <u>generation activities.</u>	<u>Council records</u> <u>Council resource</u> <u>consent records and</u> <u>monitoring compliance</u> <u>Community surveys</u>
<u>The health and safety of</u> <u>Upper Hutt City's</u> <u>community is not</u> <u>adversely affected by the</u> <u>construction, operation,</u> <u>upgrading and</u> <u>maintenance of</u> <u>renewable energy</u> <u>generation activities.</u>	<u>Complaints and enforcement</u> <u>procedures</u> <u>System failures</u>	<u>Council complaints</u> <u>register</u> <u>Council resource</u> <u>consent records and</u> <u>monitoring compliance</u>

Amendment 3: Amendments to Chapter 18: Residential Zone Rules

18

RESIDENTIAL ZONE RULES

Standards for Permitted and Controlled Activities

18.6 Subdivision which complies with the standards in rule 18.5 and 18.9, and subdivision of land for utilities, reserves or conservation purposes

Policies 4.4.2, 4.4.7, 4.4.12, 9.4.3, 10.4.1<u>,</u> <u>16.4.3</u> Council may impose conditions over the following matters:

- Design, appearance and layout of the subdivision.
- Landscaping.
- Provision of and effects on <u>network</u> utilities and/or services.
- Standard, construction and layout of vehicular access.
- Earthworks.
- Provision of esplanade reserves and strips.
- Protection of any special amenity feature.
- Financial contributions.
- The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the National Grid located on or in proximity to the site.

Note: Rule 18.29 covers subdivision within the Electricity Transmission Corridor.

• The outcome of consultation with the owner or operator of consented or existing renewable energy generation activities located on or in proximity to the site.

In relation to the land identified in Appendix Residential 1 Council may impose conditions over the following matters:

• Landscape character and visual amenity.

Restricted Discretionary Activities

18.28A Comprehensive Residential Development

Policies 4.4.1, 4.4.2, 4.4.4, 4.4.12, 4.4.13<u>,</u> <u>16.4.3</u> **Note:** Comprehensive Residential Development within Residential (Centres Overlay) Areas is not subject to the minimum net site area requirements of rule 18.10

Standards and terms for Comprehensive Residential Development:

- Compliance with the access standards of rule 18.9.
- Compliance with the site coverage standard of rule 18.11.
- Compliance with the yard setback standards of rule 18.12 and 18.17 for external boundaries only.
- Compliance with the outdoor living court standard of rule 18.13
- Compliance with the maximum building height standard of rule 18.15.
- Compliance with the sunlight access planes of rule 18.16 for external boundaries only.
- Compliance with the on-site soakage standard of rule 18.18A.

For subdivision of a Comprehensive Residential Development under Rule 18.1, Council's discretion is also limited to the matters specified in this rule, but not the standards and terms.

Council will restrict its discretion to, and may impose conditions on:

- Site layout, design and external appearance.
- Provision of and effects on <u>network</u> utilities and/or services.
- Landscaping.
- Standard, construction and layout of vehicular access.
- Protection of any special amenity feature.
- Financial contributions.
- The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the National Grid) located on or in proximity to the site.

Note: Rule 18.29 covers subdivision within the Electricity Transmission Corridor.

In considering a resource consent application, Council's discretion is also restricted to an assessment against the Design Guide for Residential (Centres Overlay) Areas (refer to Appendix Residential 2).

Matters for Consideration

18.37

Matters that may be relevant in the consideration of any resource consent, other than for a restricted discretionary activity, may include the following:

Su	bdivision
•	The requirements of section 106 of the Act.
•	Whether the proposed allotments are capable of accommodating a range of activities in compliance with zone standards.
•	Whether the subdivision compromises future subdivision potential of the land.
•	The cumulative effect on existing infrastructure as a result of the proposed subdivision.
•	The extent of compliance with the Council's Code of Practice for Civil Engineering Works.
•	For subdivisions with a net site area less than 400m ² located in a Residential (Centres Overlay) Area, the extent to which the proposal meets the requirements for 'small site design and development' of the Design Guide for Residential (Centres Overlay) Areas;
•	For subdivisions of a Comprehensive Residential Development, the extent to which the proposal meets the requirements of the Design Guide for Residential (Centres Overlay) Areas.
•	The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of, and access to, regionally significant network utilities (excluding the National Grid) located on or in proximity to the site.
•	The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the National Grid) located on or in proximity to the site.
<u>No</u>	te: Rule 18.29 covers subdivision within the Electricity Transmission Corridor.
•	The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of, and access to, consented or existing renewable energy generation activities located on or in provimity to the site
	The systems of consultation with the surger or engenter of concented or existing
•	renewable energy generation activities located on or in proximity to the site
•	Account must be taken of the future development potential of adjoining or adjacent
	land.

• Account must be taken of any potential reverse sensitivity effects on regionally significant network utilities (excluding the National Grid).

Comment [GS30]: S2.6 KiwiRail

Amendment 4: Amendments to Chapter 19: Rural Zone Rules

RURAL ZONE RULES

Standards for Permitted and Controlled Activities

19.6

19

Subdivision which complies with the standards in rule 19.5 and 19.8, and subdivision of land for utilities, reserves or conservation purposes

Policies 5.4.12, 9.4.3, 10.4.1<u>, 16.4.3</u> Council may impose conditions over the following matters:

- Design, appearance and layout of the subdivision.
- Landscaping.
- Provision of and effects on <u>network utilities and/or services</u>.
- Standard, construction and layout of vehicular access.
- Earthworks.
- Provision of esplanade reserves and strips.
- Protection of any special amenity feature.
- Financial contributions.
- The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the National Grid) located on or in proximity to the site.

Note: Rule 19.21 covers subdivision within the Electricity Transmission Corridor.

• <u>The outcome of consultation with the owner or operator of consented or existing renewable energy generation activities located on or in proximity to the site.</u>

Matters for Consideration

19.28

Matters that may be relevant in the consideration of any resource consent, other than for a restricted discretionary activity, may include the following:

Subdivision

- The requirements of section 106 of the Act.
- Whether the proposed allotments are capable of accommodating a range of activities in compliance with zone standards.
- Whether the subdivision compromises future subdivision potential of the land.
- The cumulative effect on existing infrastructure as a result of the proposed subdivision.
- The extent of compliance with Council's Code of Practice for Civil Engineering Works.
- The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of, and access to, regionally significant network utilities (excluding the National Grid) located on or in proximity to the site.
- The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the National Grid) located on or in proximity to the site.
- The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of, and access to, consented or existing renewable energy generation activities located on or in proximity to the site.
- The outcome of consultation with the owner or operator of consented or existing renewable energy generation activities located on or in proximity to the site.
- Account must be taken of the future development potential of adjoining or adjacent land.
 Account must be taken of any potential reverse sensitivity effects on regionally significant network utilities (excluding the National Grid).

Comment [GS31]: S2.6 KiwiRail

Amendment 5: Amendments to Chapter 20: Business Zone Rules

BUSINESS ZONE RULES

Standards for Permitted and Controlled Activities

20.6

20

Subdivision which complies with the standards in rule 20.5 and 20.8, and subdivision of land for utilities, reserves or conservation purposes

Policies 6.4.3, 6.4.4, 9.4.3<u>,</u> <u>16.4.3</u> Council may impose conditions over the following matters:

- Design, appearance and layout of the subdivision.
- Landscaping.
- Provision of and effects on <u>network</u> utilities and/or services.
- Standard, construction and layout of vehicular access.
- Earthworks.
- Provision of esplanade reserves and strips.
- Protection of any special amenity feature.
- Financial contributions.
- The outcome of consultation with the owner or operator of consented or existing renewable energy generation activities located on or in proximity to the site.
- The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the National Grid) located on or in proximity to the site.

<u>Note: Rule 20.28 covers subdivision within the Electricity</u> <u>Transmission Corridor.</u>

Matters for Consideration

20.32

Matters that may be relevant in the consideration of any resource consent, other than for a restricted discretionary activity, may include the following:

Subdivision The requirements of section 106 of the Act. • Whether the proposed allotments are capable of accommodating a range of activities in compliance with zone standards. Whether the subdivision compromises future subdivision potential of the land. The cumulative effect on existing infrastructure as a result of the proposed subdivision. . The extent of compliance with Council's Code of Practice for Civil Engineering Works. The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of, and access to, regionally significant network utilities (excluding the National Grid) located on or in proximity to the site. The outcome of consultation with the owner or operator of regionally significant . network utilities (excluding the National Grid) located on or in proximity to the site. Note: Rule 20.28 covers subdivision within the Electricity Transmission Corridor. The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of, and access to, consented or existing renewable energy generation activities located on or in proximity to the site. The outcome of consultation with the owner or operator of consented or existing renewable energy generation activities located on or in proximity to the site. Account must be taken of the future development potential of adjoining or adjacent land.

 Account must be taken of any potential reverse sensitivity effects on regionally significant network utilities (excluding the National Grid).

Comment [GS32]: S2.6 KiwiRail

Amendment 6: Amendments to Chapter 21: Open Space Zone Rules

21

OPEN SPACE ZONE RULES

Zone-wide provisions (including Speedway Area)

Permitted and Controlled Activities

Unless otherwise specified, all permitted and controlled activities within the Open Space Zone shall comply with the standards specified below.

21.5A All subdivision provided for as a controlled activity in Table 21.1

Policy 9.4.3<u>,</u> <u>16.4.3</u> Council may impose conditions over the following matters:

- Design, appearance and layout of the subdivision.
- Landscaping.
- Provision of and effects on <u>network</u> utilities and/or services.
- Earthworks.
- Provision of esplanade reserves and strips.
- Protection of any special amenity feature.
- Financial contributions.
- The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the National Grid) located on or in proximity to the site.

Note: Rule 21.16 covers subdivision within the Electricity Transmission Corridor.

• <u>The outcome of consultation with the owner or operator of consented or existing renewable energy generation activities located on or in proximity to the site.</u>

Provisions for the Speedway Area only

Permitted and Controlled Activities

Unless otherwise specified, all permitted and controlled activities within the Speedway Area of the Open Space Zone shall comply with the standards specified below.

21.28 Subdivision which complies with the standards in rule 21.6 and 21.27, and subdivision of land for utilities, reserves or conservation purposes

Policies 7.4.3, Council may impose conditions over the following matters: 7.4.5, 9.4.3, 16.4.3

- Design, appearance and layout of the subdivision.
- Landscaping.
- Provision of and effects on <u>network</u> utilities and/or services.
- Standard, construction and layout of vehicular access.
- Earthworks.
- Provision of esplanade reserves and strips.
- Protection of any special amenity feature.
- Financial contributions.
- The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the National Grid) located on or in proximity to the site.

Note: Rule 21.16 covers subdivision within the Electricity Transmission Corridor.

• <u>The outcome of consultation with the owner or operator of consented or existing renewable energy generation activities located on or in proximity to the site.</u>

Matters for Consideration

21.41 Matters that may be relevant in the consideration of any resource consent, other than for a restricted discretionary activity, may include the following:

•	The requirements of section 106 of the Act	
•	Whether the proposed allotments are capable of accommodating a range of activities in compliance with zone standards.	
•	Whether the subdivision compromises future subdivision potential of the land.	
•	The cumulative effect on existing infrastructure as a result of the proposed subdivision.	
•	The extent of compliance with the Council's Code of Practice for Civil Engineering Works.	
•	The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of, and access to, regionally significant network utilities located on or in proximity to the site.	
•	The outcome of consultation with the owner or operator of regionally significant network utilities located on or in proximity to the site.	
•	The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of, and access to, consented or existing renewable energy generation activities located on or in proximity to the site.	
•	The outcome of consultation with the owner or operator of consented or existing renewable energy generation activities located on or in proximity to the site.	
•	Account must be taken of the future development potential of adjoining or adjacent land.	
	Account must be taken of any potential reverse sensitivity effects on regionally significant network utilities (excluding the National Grid).	Comment [GS33]: S2.6 Ki

Amendment 7: Amendments to Chapter 22: Special Activity Zone Rules

SPECIAL ACTIVITY ZONE RULES

Zone-wide provisions

(including St. Patrick's Estate Area)

Provisions for the St Patrick's Estate Area only

Controlled Activities

22.30 Council may impose conditions over the following matters:

Policies 8.4.2, 8.4.3<u>, 16.4.3</u>

22

- Bulk, location and design of the buildings.
 - Design and layout of car parks, loading, manoeuvring, pedestrian links and access.
 - Provision of and effects on <u>network</u> utilities and/or services.
 - Landscaping.
 - Flood mitigation measures.
 - Screening.
 - Financial contributions.
 - The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the National Grid) located on or in proximity to the site.

<u>Note: Rule 22.18 covers subdivision within the Electricity</u> <u>Transmission Corridor.</u>

• The outcome of consultation with the owner or operator of consented or existing renewable energy generation activities located on or in proximity to the site.

Matters for Consideration

22.31

Subdivision

Matters that may be relevant in the consideration of any resource consent other than for a restricted discretionary activity may include the following:

The requirements of section 106 of the Act. Whether the proposed allotments are capable of accommodating a range of activities in compliance with zone standards. Whether the subdivision compromises future subdivision potential of the land. The cumulative effect on existing infrastructure as a result of the proposed subdivision. The extent of compliance with the Council's Code of Practice for Civil Engineering Works. The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of, and access to, regionally significant network utilities (excluding the National Grid) located on or in proximity to the site. The outcome of consultation with the owner or operator of regionally significant network utilities (excluding the National Grid) located on or in proximity to the site. Note: Rule 22.18 covers subdivision within the Electricity Transmission Corridor. The design and layout of the subdivision where any lot may affect the safe and effective operation and maintenance of, and access to, consented or existing renewable energy generation activities located on or in proximity to the site. The outcome of consultation with the owner or operator of consented or existing renewable energy generation activities located on or in proximity to the site. Account must be taken of the future development potential of adjoining or adjacent land.

• Account must be taken of any potential reverse sensitivity effects on regionally significant network utilities (excluding the National Grid).

Comment [GS34]: S2.6 KiwiRail

Standards for Permitted and Controlled Activities

22.7 Subdivision which complies with the standards in rule 22.6 and 22.9, and subdivision of land for utilities, reserves or conservation purposes

Policies 8.4.3, 9.4.3<u>, 16.4.3</u> Council may impose conditions over the following matters:

- Design, appearance and layout of the subdivision.
- Landscaping.
- Provision of and effects on <u>network utilities and/or services</u>.
- Standard, construction and layout of vehicular access.
- Earthworks.
- Provision of esplanade reserves and strips.
- Protection of any special amenity feature.
- Financial contributions.

• <u>The outcome of consultation with the owner or operator of</u> <u>regionally significant network utilities (excluding the National</u> <u>Grid) located on or in proximity to the site.</u>

<u>Note: Rule 22.18 covers subdivision within the Electricity</u> <u>Transmission Corridor.</u>

• <u>The outcome of consultation with the owner or operator of consented or existing renewable energy generation activities located on or in proximity to the site.</u>

Amendment 8:

Amendments to Chapter 28: Southern Hills Overlay Area and Protected Ridgelines Rules

28 RULES FOR SOUTHERN HILLS OVERLAY AREA AND PROTECTED RIDGELINES

Activity Table

Policies 12.4.6 - 12.4.8, 15.4.1, <u>16.4.12</u>

Protected Ridgelines		
28.1 Activity	All Zones	
Any building or structure <u>or new aboveground network utility</u> , otherwise permitted, located on a site subject to a protected ridgeline (identified on the Planning Maps) which complies with the standard specified in rule 28.4	Р	
Any building or structure or new aboveground network utility (excluding	RD	

<u>minor above ground lines</u>), otherwise permitted which does not comply with the standard specified in rule 28.4

Southern Hills Overlay Area		
28.2 Activity	All Zones	
Subdivision, otherwise controlled under the underlying zone, within an are identified as Southern Hills Overlay Area	ea RD	
Subdivision, otherwise restricted discretionary or discretionary under the underlying zone, within an area identified as Southern Hills Overlay Area	he D	
Subdivision, otherwise non-complying under the underlying zone within the Southern Hills Overlay Area	he NC	
Accessory buildings, otherwise permitted under the underlying zone, locate within an area identified as Southern Hills Overlay Area (see definition "accessory building" for limitations on size and floor area)	ed P of	
Any building or structure <u>or new aboveground network utility</u> (excludinaccessory buildings <u>and minor above ground lines</u>), otherwise permitte	ng RD ed	
under the underlying zone, located within an area identified as Southern Hil Overlay Area	lls	

Comment [GS36]: S5.21 Chorus, FS3.39 WELL

Comment [GS35]: S5.21 Chorus, FS3.39 WELL

38

Any building or structure <u>or new aboveground network utility</u> (including accessory buildings), otherwise controlled or restricted discretionary under the underlying zone, located within an area identified as Southern Hills Overlay Area		
Any building or structure <u>or new aboveground network utility</u> , otherwise non- complying under the underlying zone, located within an area identified as Southern Hills Overlay Area	NC	

Key

- Permitted activity which complies with standards for permitted activities specified in the Plan
- **RD** Restricted discretionary activity
- **D** Discretionary activity
- **NC** Non complying activity

28.3 City-wide provisions

Ρ

Each building, structure, <u>new aboveground network utility</u> or subdivision located within an area identified as Southern Hills Overlay Area shall comply with the relevant permitted activity standards in the City-wide provisions of the Plan.

Standard for Permitted Activities

- 28.4 Buildings or structures <u>or new aboveground network utilities</u> located on a site subject to a protected ridgeline
- Policy 12.4.6.Any building or structure or new aboveground network utility is to be16.4.12located and designed so as not to project through the protected
ridgelines identified on the Planning Maps, as viewed from any point on
State Highway 2.

Restricted Discretionary Activities

28.5	Any activity located on a site subject to a protected ridgeline (identified on the Planning Maps) which does not comply with the standard specified in rule 28.4
Policy 12.4.6, <u>16.4.12</u>	Council will restrict its discretion to, and may impose conditions on:
	• The extent to which any building <u>.</u> or structure <u>or new aboveground</u> <u>network utility</u> is visually obtrusive.
	• Design, appearance, scale and siting of the building or structure.
	• Materials used (including their colour, texture and reflectivity).
	Landscaping, planting and screening.
	Access arrangements.
28.7	Any building or structure <u>or new aboveground network utility</u> (excluding accessory buildings), otherwise permitted under the underlying zone, located within an area identified as Southern Hills Overlay Area
Policies 12.4.8, 15.4.1 <u>, 16.4.12</u>	Council will restrict its discretion to, and may impose conditions on:
	• Design, appearance, scale and siting of the building-or structure, or new aboveground network utility.
	• Effects on visual values.
	• Effects on landscape values.
	• Effects on ecological values.

• Measures to avoid, remedy or mitigate potential adverse effects.

Matters for Consideration

28.8	Matters that may be relevant in the consideration of any resource
Sout	thern Hills Overlay Area
•	Matters for consideration within 18.35 of the Residential Zone, 19.28 of the Rural Zone, 21.41 of the Open Space Zone, and 22.31 of the Special Activity Zone.
•	Compliance with the relevant standards of the underlying zoning of the site.
•	The extent to which any natural feature will be retained or enhanced by the development of anticipated consequential development(s).
•	The ability of affected natural features to absorb the overall development impact, including the extent to which any natural feature, and/or intrinsic value will be disturbed or modified as a result of the development or anticipated consequential development(s).
•	The suitability of any proposed mitigation mechanisms in avoiding, remedying or mitigating adverse visual or landscape effects on the development or anticipated consequential development(s) on any natural or amenity values, including screening through plantings or the reinstatement of any previous plantings.
•	The suitability of any proposed mitigation mechanisms in avoiding, remedying or mitigating adverse effects of the development or anticipated consequential development(s) on any ecological values.
•	The suitability of the site for the proposed development and anticipated consequential development(s), including the extent to which alternative sites or locations have been considered.
•	In respect of the proposed location of any new residential building <u>or network utility</u> , the extent to which other viable alternatives are available within the application site, and the effectiveness of the proposed location compared to any identified alternatives.
•	The suitability of the site in relation to natural science factors including geological, topographical, ecological and dynamic components.
•	The effects of the activity on the ecological, intrinsic, cultural or amenity values of the area.
•	The effectiveness of any existing or proposed protection or enhancement mechanisms.
•	The significance of the affected indigenous vegetation or habitat of indigenous fauna, in terms of the following generic criteria:
	 Representativeness: i.e. contains or supports an ecosystem that is unrepresented, uncommon or unique. Rarity: i.e. contains or supports threatened ecosystems, threatened species, or endemic species. Diversity: i.e. contains or supports diverse ecosystems, species, vegetation. Distinctiveness: i.e. its natural state, significance as a habitat. Continuity: i.e. role as an ecological buffer area or corridor.
Prot	ected Ridgelines

• The extent to which any building or structure <u>or network utility</u> is visually obtrusive.

- Design, appearance, scale and siting of the building or structure.
- Materials used (including their colour, texture and reflectivity).

•	Land	lscaping,	planting	and	screening.	

Access arrangements.

Amendment 9: Amendments to Chapter 30: Rules for Utilities

30

RULES FOR <u>NETWORK</u>UTILITIES

Activities Table

30.1 Activities	All Zones	
The construction, operation, maintenance, minor upgrading, replacement and refurbishment of:		
- Masts		
Transformers and lines for conveying electricity voltage	B	
Connections from allotments to gas, water, drainage and sewer pipes		
Pipes for distribution of natural or manufactured gas and necessary incidental equipment and compressor stations		
Wastewater, drainage and water supply networks]	
Service reservoirs	1	
Lines as defined by section 5 of the Telecommunications Act 2001		
 Works, facilities, services, transmission and equipment associated with the operation of a telecommunication and radiocommunication network, including microwave stations, radio stations, telephone exchanges, antennas, cables, lines, radio links and satellite dishes 		
 Footways, accessways and service lanes including footbridges over roads and railways 		
Bridges for roads and railways, and underpasses		
Street furniture and lighting		
Public toilets		
Automatic weather stations, voluntary observer sites and anemometer mast sites		
• Trig stations, benchmarks and other survey marks required as part of any engineering, land transfer or other land surveying activity		
The operation, maintenance, minor upgrading, replacement and refurbishment of existing formed roads within the meaning of section 43 Transit New Zealand Act 1989 and section 315 Local Government Act 1974	₽	
Construction of new roads, except where part of a subdivision for which consent is required	Ð	
Weather radar	Ð	
Any activity listed above that does not meet the relevant standards in rules 30.2 to 30.12	Ð	
<u>30.1</u> <u>Activities</u>	<u>Status</u>	Zone
--	---------------	-------------
Removal, maintenance, operation and upgrading	,	1
The removal of existing network utilities, including any existing structures	<u>P</u>	<u>A11</u>
The operation and maintenance of existing network utilities	<u>P</u>	<u>A11</u>
The minor upgrading of existing electricity and telecommunication lines	<u>P</u>	<u>A11</u>
 <u>The upgrading of all other network utilities, excluding:</u> <u>electricity and telecommunication lines and</u> <u>Gas distribution and transmission pipelines at a pressure</u> <u>exceeding 2000 kilopascals</u> 	<u>P</u>	<u>A11</u>
The removal, operation and maintenance of network utilities and the minor upgrading of electricity and telecommunication lines that does not meet permitted activity standards	<u>C</u>	<u>A11</u>
 The upgrading of network utilities that do not meet the permitted activity standards, excluding: Electricity transmission lines above 110kV; and Gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals 	RD	<u>A11</u>
 <u>The upgrading of:</u> <u>Electricity transmission lines above 110kV; and</u> <u>Gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals</u> 	<u>RD</u>	<u>A11</u>
Subdivision		
Subdivision for the purpose of accommodating any network utility	<u>C</u>	<u>A11</u>
General		
<u>Cabinets and other network utility structures not otherwise listed in</u> <u>this table that meet the relevant standards</u>	<u>P</u>	<u>A11</u>
Network utilities located within existing buildings	<u>P</u>	<u>A11</u>
Aerial crossings necessary for network utilities, located on or within existing bridges and structures or across streams watercourses, and including regulator stations, but not compressor stations.	<u>P</u>	
Cabinets and other network utility structures not otherwise listed in this table that do not meet all of the relevant standards.	RD	<u>A11</u>
<u>All network utilities that are not otherwise listed as a permitted,</u> <u>controlled, restricted discretionary or non-complying activity</u>	D	<u>A11</u>
All network utilities which do not comply with the permitted activity standards for radiofrequency and electro-magnetic fields in standard 30.3.	<u>NC</u>	<u>A11</u>
Underground utilities		
 The construction, installation and development, of new underground network utilities, except for: Electricity transmission lines above 110kV; and Gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals 	<u>P</u>	<u>A11</u>
Radiocommunication, Telecommunication and Electricity Distrib	ution and	Transmissio

Masts with or without associated antennas	<u>P</u>	<u>Rural</u> <u>Business</u> <u>Special</u> <u>Activity</u>
Masts with or without associated antennas which meet the criteria in Rule 30.14	<u>RD</u>	<u>Residential</u> <u>Open</u> <u>Space</u>
Antenna and support structure(s) attached to buildings	<u>P</u>	<u>A11</u>
<u>Masts, with or without associated antennas that do not comply with</u> <u>the standards to be a permitted activity.</u>	<u>RD</u>	<u>Rural</u> <u>Business</u> <u>Special</u> <u>Activity</u>
Antenna and support structure(s) attached to buildings that do not comply with the standards to be a permitted activity.	<u>RD</u>	<u>A11</u>
New above ground lines, excluding electricity transmission lines above 110kV.	<u>P</u>	<u>Rural</u> <u>Open</u> Space
Minor above ground lines	<u>P</u>	<u>A11</u>
Temporary above ground lines	<u>P</u>	<u>A11</u>
New or additional above ground lines that are not otherwise permitted, or restricted discretionary activities	D	<u>A11</u>
New and additional lines (above ground and underground) and support structures for conveying electricity at a voltage above 110kV	D	<u>A11</u>
New and upgraded transformers, substations and switching stations distributing electricity and ancillary buildings, except for those encased within a cabinet or located on a line that is otherwise a permitted activity.	D	<u>A11</u>
Gas Distribution and Transmission		*
<u>Underground gas distribution and transmission pipelines at a pressure not exceeding 2000 kilopascals, including aerial crossings of bridges, structures or streams, and ancillary equipment, including regulator stations but not compressor stations.</u>	<u>P</u>	<u>A11</u>
Underground gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals, including aerial crossings of bridges, structures or streams, and ancillary equipment, including compressor compounds with compressor houses	<u>RD</u>	<u>A11</u>
Water, Wastewater and Stormwater		
Water reservoirs	RD	<u>A11</u>
Water and wastewater treatment plants	D	<u>A11</u>
Meteorological Activities		
Meteorological enclosures and buildings; automatic weather stations and anemometer masts, voluntary observer sites and associated microwave links	P	<u>A11</u>
Meteorological enclosures and buildings, automatic weather stations and anemometer masts voluntary observer sites and associated microwave links that are not permitted activities	RD	<u>A11</u>
Roading, and Traffic and Transport Structures		

<u>Traffic control signals and devices, light and decorative poles and associated structures and fittings, post boxes, landscaped gardens, artworks and sculptures, bus stops and shelters, phone boxes, public toilets, road furniture located within the road reserve or rail corridors</u>	P	<u>A11</u>
The construction, alteration or diversion of roads, but excluding any such construction works which are part of a subdivision	<u>D</u>	<u>Business</u> <u>Rural</u> Residential
The construction, alteration or diversion of roads, but excluding any such construction works which are part of a subdivision	<u>NC</u>	<u>Open</u> Space
Extreme Adverse Weather Warning Devices		
Extreme adverse weather warning devices	<u>P</u>	<u>A11</u>

Notes:

Resource Management Regulations - National Environmental Standards

The operation, maintenance, upgrading, relocation or removal of an electricity transmission line and ancillary structures that existed prior to 14 January 2010 is largely controlled by the Resource Management (National Environmental Standards for Electricity Transmission Activities) Regulations 2009, separate to this District Plan.

The installation and operation of telecommunications facilities (antennas attached to existing structures and cabinets in the road reserve) is largely controlled by the Resource Management (National Environmental Standards for Telecommunications Facilities) Regulations 2008, separate to this District Plan. It also controls all radio-frequency emissions from telecommunication facilities through specific exposure standards.

<u>Upper Hutt City Council is responsible for enforcing these standards</u>. For clarification, where there is conflict or perceived conflict between the provisions of this Plan and the requirements of the NESs identified above, the provisions of the NES shall apply.

<u>The National Environmental Standards are available for viewing at www.mfe.govt.nz and at Upper Hutt City Council offices.</u>

Other Relevant Regulations

<u>Compliance with the New Zealand Electrical Code of Practice for Electrical Safe Distances</u> 34:2001 is mandatory for buildings, earthworks and mobile plants within close proximity to all electricity lines.

Compliance with the Electricity (Hazards from Trees) Regulations 2003 is mandatory for tree trimming and planting in proximity to electricity transmission and distribution lines.

To discuss works, including tree planting, near electrical lines especially within 20m of those lines, contact the line operator.

The following two standards are relevant for work in proximity to gas pipelines:

- <u>NZS/AS 2885 Pipelines Gas and Liquid petroleum</u>
- <u>NZS 5258:2993 Gas Distribution Network</u>

Upper Hutt City Council is not responsible for enforcing these regulations or standards.

Key

P Permitted activity which complies with standards for permitted activities specified in the Plan

- RD Restricted discretionary activity
- **D** Discretionary activity
- NC Non-complying activity

30.1A City-Wide Provisions

Each activity shall comply with the relevant permitted activity standards in the City-wide provisions of the Plan as listed below.

<u>Chapter</u>	<u>City-wide provisions</u>
<u>23</u>	Earthworks and Indigenous Vegetation Clearance
<u>26</u>	Heritage Features
27	Notable Trees
28	Southern Hills Overlay and Protected Ridgelines
<u>32</u>	Noise and Vibration
<u>33</u>	Flooding and Fault Band Hazards
<u>34</u>	Hazardous Substances and Contaminated Land

Comment [GS40]: S1.5, S3.10, S5.11, FS1.6, FS3.31, FS2.1, GWRC, Spark, Chorus, WELL, Powerco, Transpower

Standards for Permitted Activities

Exemptions:

- a) The operation, maintenance, minor upgrading, replacement and refurbishment of existing utilities are exempt from compliance with the following standards.
- b) All electricity transmission activities affecting national grid assets existing as at 14 January 2010 must comply with the National Environmental Standards for Electricity Transmission Activities, and no rule in the Plan shall apply unless required to by virtue of a specific regulation in the National Environmental Standards for Electricity Transmission Activities.

Policy 16.4.1

<u> 30.2</u>

Height

Utilities shall not exceed the maximum height for the zone in which they are located, except as follows:

 Masts* (freestanding or attached to a building) including any attached antennas, and/or overhead cables and lines shall not exceed the following heights:

	ZONE	MAXIMUM HEIGHT ABOVE GROUND**					
	Residential	15m					
	Rural	15m					
	Business	30m (50m in the CBD)					
	Open Space	15m (25m in the Speedway Zone)					
	Special Activity	20m (15m within the St Patrick's Estate Area)					
	For the purposes pole, tower or sim operation of a utility	- of the Plan, these are defined as 'any mast, ilar structure which is used to facilitate the r.²					
	<u>** Maximum height</u> mast may be attach	above ground includes any building to which a ed.					
	 Antennas directly attached to existing buildings may project the finished height of a building up to a maximum of 5m, pro that where the existing building already exceeds the max building height for the zone, antennas shall not exceed the of the building by more than 2m. 						
30.2A	Mast diameter						
Policy 16.4.1							
	The diameter of any ma height of 3 metres or mo in diameter from ground	st shall not exceed 0.5m when measured at a re above ground level, and shall not exceed 1m level up to 3m.					
30.3	Setbacks						
Folicy 10. 11	 Utilities may encro of half the setback- provided that no se Any service reserve boundary of any dwelling and its top 	ach into boundary setbacks up to a maximum specified for the zone in which they are located, tback shall be less than 1.5m. roir shall be located at least 15m from the Residential Zone or any site containing a shall be below the apex of any ridge or hill.					
	Exemption:						
	Utilities within roads, all lines (excluding their su setback requirements.	underground utilities and overhead cables and apport structures) are exempt from the above					
30.4	Screening						
Policy 16.4.1	Any structure associated road, an adjoining resi screening shall be in t	with utilities shall be screened from an arterial idential site or an Open Space Zone. The the form of trees and/or shrubs which will					

	effectively screen the structure. This requirement shall not apply to:
	Utilities within roads and all underground utilities
	Masts and overhead cables and lines
	 Antennas attached to buildings or other utility structures
30. 5 2	Development associated with <u>network</u> utilities shall comply with the Council's Code of Practice for Civil Engineering Works where that development involves assets which are managed by or to be vested in
Polic <u>yies</u> <u>16.4.4, 16.4.6,</u> <u>16.4.12</u>	Council.
30.6	Cables and lines for the distribution, transmission or supply of telecommunications or electricity shall be placed underground in all zones except the Rural Zone.
Policy 16.4.1	
30.7	Lines for conveying electricity must not exceed a voltage of 110kV.
Policy 16.4.1	
30.8	The pressure in pipes for the distribution of natural or manufactured gas shall not exceed 2000 kPa.
Policy 16.4.1	
30.9	Any above ground structure associated with a water or drainage supply facility shall not exceed 50m ² in area
Policy 16.1.1	
30.10	Any equipment shelter or plant associated with a telecommunication or radiocommunication activity shall not exceed 9m ² in floor area.
Policy 16.4.1	
30.11	The diameter of an antenna (not including its bracket or attachment) shall not exceed 5m.
Policy 16.4.1	
<u>30.3</u>	<u>Health and Safety</u>
<u>Policy 16.4.12,</u> <u>16.4.13</u>	Where relevant, network utilities shall comply with the following standards:
	in NZS 2772:1999
	(b) Network utilities that emit electric and magnetic fields shall comply with the International Commission on Non-ionising Radiation Protection Guidelines for limiting exposure to time-
	varying electric and magnetic fields (1 Hz – 100 Hz), Health Physics 99(6):818-836; 2010, and the recommendations from the World Health Organisation monograph Environmental Health Criteria (No 238, 2007).
	Note: The Resource Management (National Environmental Standards for

District Plan controls all radio-frequency emissions from

telecommunication facilities through specific exposure standards.

<u>30.4</u> <u>Maximum Height of Network Utilities</u>

 Policy 16.4.12
 The maximum height of any utility structure listed in the table below shall include any antenna and support structures and exclude any lightning rod.

Network Utility	Business	Business	<u>Residential</u>	<u>Rural</u>	<u>Open</u> Space	<u>Special</u> Activity		
					Spuce	<u></u>		
Masts, antennas, lines and single- pole support structures	<u>20m</u>	<u>25m</u>	<u>n/a</u>	<u>15m</u>	<u>n/a</u>	20m 15m in the St Patrick's Estate Area	<u>.</u>	
<u>Masts and</u> <u>antennas (where</u> <u>there are two or</u> <u>more providers)</u>	<u>25m</u>	<u>30m</u>	<u>n/a</u>	<u>20m</u>	<u>n/a</u>	12m 25m (other than in the St Patrick's Estate Area	Ú.	Comment [GS41]: S3.12 Spark; S5.14 Chorus
Maximum height of an antenna and support structure measured from the highest point of the building to which it is attached.	51	<u>m</u>		<u>3.</u>	<u>5m</u>			
<u>Cabinets, and</u> <u>network utility</u> <u>structures located</u> <u>within road</u> <u>reserve, that are</u> <u>not otherwise</u> <u>provided for</u>	2.0	<u>)m</u>	<u>1.8m</u>		<u>2.0m</u>			
<u>Cabinets, and</u> <u>network utility</u> <u>structures, that</u> <u>are not otherwise</u> <u>provided for</u>			3.	<u>5m</u>				
Anemometer masts	<u>15m</u>	<u>30m</u>	<u>12m</u>	<u>15m</u>	<u>1</u>	<u>2m</u>		

<u>Network Utility</u>	<u>Business</u> - CBD	<u>Business</u>	<u>Residential</u>	<u>Rural</u>	<u>Open</u> Space	<u>Special</u> <u>Activity</u>
<u>Maximum height</u> of an extreme			4	<u>Im</u>		
<u>measured from the</u> point of						
attachment.						

<u>30.5</u>

Maximum Size and Diameter of Network Utilities

Policy 16.4.12

The maximum size and diameter of network utilities for each zone is outlined in the table below.

Network Utility	CBD	Business	<u>Residential</u>	<u>Rural</u>	<u>Open</u>	Special	
					Space	Activity	
<u>Masts</u>	Diameter of mast <600mm from 6m in height	Diameter of mast 1.5m	<u>n/a</u>	Diameter of mast <600mm from 6m in height	<u>n/a</u>	Diameter of mast 1.5m Except in the St Patrick's Estate Area: Diameter of mast <600mm from 6m in height	
<u>Masts (where there</u> <u>are two or more</u> <u>providers)</u>	Diameter of mast <600mm from 6m in height	Diameter of mast 1.5m	<u>n/a</u>	Diameter of mast <600mm from 6m in height	<u>n/a</u>	Diameter of mast <600mm from 6m in height Diameter of mast 1.5m	Comment [GS42]: S3.12 Spark; S5.14 Chorus Comment [GS43]: S3.12 Spark; S5.14 Chorus
Antenna attached to masts	Antenna located within a horizonta l diameter circle of 750mm around the mast	<u>Antenna</u> <u>located</u> <u>within a</u> <u>horizont</u> <u>al circle</u> <u>of 5m</u> <u>around</u> <u>the mast</u>	<u>n/a</u>	Antenna located within a horizont al circle of 1.2m 5m around the mast	<u>n/a</u>	Antenna located within a horizontal circle of 5m around the mast <u>Except</u> In the St Patrick's Estate Area	Comment [GS44]: S3.12 Spark; S5.14 Chorus

<u>Network Utility</u>	CBD	Business	<u>Residential</u>	<u>Rural</u>	<u>Open</u> Space	<u>Special</u> <u>Activity</u>	
						<u>- Antenna</u> <u>located</u> <u>within a</u> <u>horizontal</u> <u>circle of</u> <u>750mm</u> <u>around the</u> <u>mast</u>	
<u>Antenna attached</u> <u>to buildings</u>	Antenna d 2m or are	liameter of a of 1.8m ²	<u>Antenna</u> <u>diameter of</u> <u>1m or area</u> <u>of 0.8m²</u>	Antenna diameter of 1.3m or area of 1.2m ²	Antenna diameter of 1m or area of 0.8m ²	<u>Antenna</u> <u>diameter of</u> <u>2m or area</u> <u>of 1.8m²</u>	
Cabinets and other network utility structures located within road reserve (not otherwise provided for)	<u>2r</u>	<u>n²</u>	<u>1.4m²</u>		<u>2m²</u>		
Cabinets and other network utility structures (not otherwise provided for) that are not located within road reserve			<u>1</u>	<u>5m²</u>			
<u>Cabinets located</u> <u>within the road</u> <u>reserve containing</u> <u>an electricity</u> <u>distribution</u> <u>substation</u>			<u>5</u>	<u>m²</u>			Comment [GS45]: S6.24 WELL
<u>Meteorological</u> <u>enclosures and</u> <u>buildings</u>			<u>3(</u>	<u>0m²</u>			
Extreme adverse weather warning devices		<u>No g</u>	reater dimensi	on than 2.5	<u>m x 1.5m</u>		

<u> 30.6</u>

Separation distance and setbacks from boundaries

Policy 16.4.12

No network utilities shall be located within an esplanade or strip.

The following table applies to masts and antenna attached to masts and any cabinet or other network utility structure that is over 5m² in area with a height of more than 1.2m defined as a building that are **not** located in the road reserve or rail corridor:

Comment [GS46]: S7.26 Powerco Comment [GS47]: S2.15 KiwiRail

-		
Zone	Setback distance or setback for	Setback distance or setback for
	<u>masts and antenna attached to</u>	cabinets and other network utility
	masts	structures
<u>A11</u>	Not located within an esplanade reserve	<u>e or strip</u>
Business	No less than 10m from a Residential	No less than 2 metres to any boundary
	Zone boundary	in a Rural, Residential, Open Space
		and Special Activity Zone and to a
		road or service lane boundary.
Rural	No less than 10m from any property	No less than 2 metres to all
	boundary	boundaries
	<u>Under 15m in height - no less than</u>	
	20m from the closest wall of a	
	dwelling (excluding balconies and	
	decks)	
	<u>Over 15m in height – no less than</u>	
	<u>50m from the closest wall of a</u>	
	dwelling (excluding balconies and	
	decks)	
D		
Residential, Open	No less than 10m from a Residential	No less than 2 metres to all
Space and	or Rural Zone boundary.	<u>boundaries</u>
Special Activity		

30.7 Specific standards for temporary above ground lines

Policy 16.4.12.
16.4.15The line(s) shall be in place for no longer than six calendar months from
the date of erection until its removal.

30.128Where any work is undertaken on a road or service lane the persons
responsible for the work shall notify Council at least 10 working days
before work begins.

Controlled Activities



The removal, operation and maintenance of network utilities and the minor upgrading of electricity and telecommunication lines

Policy 16.4.8, 16.4.9. that does not meet permitted activity standards

Council may impose conditions over the following matters:

- <u>Earthworks and sediment and erosion control;</u>
- Noise;
- Any adverse effects on a heritage site or area of native vegetation

<u>30.10</u> Subdivision for the purpose of accommodating any network utility

<u>Policy 16.4.8, 16.4.9, 16.4.9</u>

16.4.12,

<u>Council may impose conditions over the following matters:</u>

- <u>Site design, frontage and area;</u>
- <u>Legal and physical access to the lots;</u>
- <u>Risks to public health and safety;</u>
- <u>Earthworks and sediment and erosion control;</u>
- Landscaping and screening;
- <u>Traffic and parking management;</u>
- <u>The imposition of financial contributions in accordance with</u> Part E of this Plan
- Any adverse effects on a heritage site or area of native vegetation

Restricted Discretionary Activities 30.11 The upgrading of network utilities that do not meet the permitted activity standards, excluding: Policy 16.4.8, 16.4.9, Electricity transmission lines above 110kV; and 16.4.12 Gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals Council will restrict its discretion to, and may impose conditions on: The degree, extent and effects of the non-compliance with the Permitted Activity Standards The extent to which there are difficult ground conditions, technical or financial constraints that make compliance impracticable/ unreasonable Earthworks and erosion and sediment control Any adverse effects on an identified heritage site or an area of native vegetation. 30.12 The upgrading of: Electricity transmission lines above 110kV; and Policy 16.4.8, 16.4.9, Gas distribution and transmission pipelines at a pressure 16.4.12 exceeding 2000 kilopascals Council will restrict its discretion to, and may impose conditions on: Risks to public health and safety Rehabilitation of the site following any construction or future maintenance period. The extent to which the affected persons / community has been consulted with. Earthworks and erosion and sediment control.

- <u>Any potential interference with public use and enjoyment of</u> the land and the operation of land uses in the vicinity
- <u>Any adverse effects on an identified heritage site or an area of native vegetation</u>
- <u>Local, national and / or regional benefits derived from the</u> <u>activity</u>
- <u>Any constraints arising from technical and operational</u> requirements of the network which may limit measures to avoid, remedy or mitigate environmental effects.

Cabinets and other network utility structures not otherwise listed
in this table that do not meet all of the relevant standards.6.4.9.Council will restrict its discretion to, and may impose conditions on:

<u>Policy 16.4.8, 16.4.9, 16.4.9</u>

30.13

- <u>The degree, extent and effects of the non-compliance with the</u> Permitted Activity Standards
 - Risks to public health and safety
 - <u>Any effect on heritage and cultural values</u>
 - <u>Any enect on nentage and cultural values</u>
 <u>Design and external appearance</u>
 - Visual effects including impacts on:
 - <u>The residential and recreational use of land in the vicinity</u>
 <u>of the proposed utility;</u>
 - <u>The existing character</u>, <u>landscape</u>, <u>streetscape</u> and <u>amenity values of the locality</u>;
 - <u>Key public places, public viewing points and significant</u> recreational areas
 - Amenity effects, including noise, vibration, odour, dust, earthworks and lighting
 - Any potential interference with public use and enjoyment of the land and the operation of land uses in the near vicinity
 - <u>Measures to mitigate the bulk and scale of the utility,</u> including screening, colour and finish treatment, earth mounding and / or planting, viewing distances, the location of support structures
 - Whether the size and scale of the proposal is generally compatible with other development in the area
 - Any adverse effects on traffic and pedestrian safety including sight lines and the visibility of traffic signage
 - <u>The extent to which the affected persons / community has</u> been consulted with.

<u>30.14</u>

<u>Policy 16.4.8, 16.4.9,</u> <u>16.4.12</u>

<u>Masts with or without associated antennas which meet the</u> <u>following criteria:</u>

- Have a maximum height of 12m
- <u>The diameter of the mast is <600mm, from 6m in height</u> above ground level
- <u>The antenna are located within a horizontal circle of 750mm</u>
- Are located no less than 10m from a Residential or Rural Zone boundary

Council will restrict its discretion to, and may impose conditions on:

- <u>Risks to public health and safety</u>
- <u>The maximum height, area or diameter of any antenna</u>
- Design and external appearance
 - Any effect on heritage and cultural values
- <u>Visual effects including impacts on:</u>

Comment [GS48]: S4.34 Transpower; S4.17, S5.6 and US6.18, and FS3.9, FS1.5, FS3.18 and FS3.30; Spark, Chorus, Transpower, WELL

- <u>The residential and recreational use of land in the vicinity</u> of the proposed utility;
- <u>The existing character, landscape, streetscape and</u> <u>amenity values of the locality;</u>
- <u>key public places, public viewing points, and significant</u> recreational areas
- Amenity effects, including noise, vibration, odour, dust, earthworks and lighting
- <u>Cumulative effects</u>
- <u>Whether the size and scale of the proposal is generally</u> <u>compatible with other development in the area.</u>
- <u>Any adverse effects on traffic and pedestrian safety including</u> sight lines and the visibility of traffic signage.
- <u>The extent to which alternative locations, routes or other</u> options have been appropriately considered.
- Any potential interference with public use and enjoyment of the land and the operation of land uses in the near vicinity
- The extent to which it is technically, economically and practically reasonable for the masts or antennas can be cosited with similar structures or other buildings
- <u>The extent to which the affected persons / community has</u> been consulted with.

Masts, with or without associated antennas that do not comply with the standards to be a permitted activity.

<u>Policy 16.4.8, 16.4.9,</u> <u>16.4.12</u>

30.15

Council will restrict its discretion to, and may impose conditions on:

- The degree, extent and effects of the non-compliance with the Permitted Activity Standards
- Risks to public health and safety
- The maximum height of the mast and area or diameter of any antenna
- The maximum height, area or diameter of any antenna
- <u>Any effect on heritage and cultural values</u>
- Visual effects including impacts on:
 - <u>The residential and recreational use of land in the</u> vicinity of the proposed utility;
 - The existing character, landscape, streetscape and amenity values of the locality;
 - key public places, public viewing points, and significant recreational areas
- <u>Amenity effects, including noise, vibration, odour, dust,</u> <u>earthworks and lighting</u>
- <u>Cumulative effects</u>
- Any potential interference with public use and enjoyment of the land and the operation of land uses in the near vicinity
- <u>Measures to mitigate the bulk and scale of the utility, including</u> <u>screening, colour and finish treatment, earth mounding and /</u> <u>or planting, viewing distances, the location of support</u> <u>structures.</u>
- Whether the size and scale of the proposal is generally compatible with other development in the area.
- <u>Any adverse effects on traffic and pedestrian safety including</u> <u>sight lines and the visibility of traffic signage.</u>
- <u>The extent to which alternative locations, routes or other</u> <u>options have been appropriately considered.</u>
- The extent to which it is technically, economically and practically reasonable for the masts or antennas can be cosited with similar structures or other buildings

• <u>The extent to which the affected persons / community has</u> been consulted with.

<u>Antenna attached to buildings that do not comply with the</u> standards to be a permitted activity.

<u>Policy 16.4.8, 16.4.9, 16.4.9</u>

30.16

Council will restrict its discretion to, and may impose conditions on:

- <u>The degree, extent and effects of the non-compliance with the</u> <u>Permitted Activity Standards</u>
- Risks to public health and safety
 - The maximum height, area or diameter of any antenna
- <u>Design and external appearance</u>
- Any effect on heritage and cultural values
- <u>Visual effects including impacts on:</u>
 - <u>The residential and recreational use of land in the vicinity</u> of the proposed utility;
 - <u>The existing character, landscape, streetscape and</u> <u>amenity values of the locality;</u>
 - <u>key public places, public viewing points, and significant</u> recreational areas
- <u>Amenity</u> effects, including noise, vibration, odour, dust, earthworks and lighting
- <u>Cumulative effects</u>
- Any potential interference with public use and enjoyment of the land and the operation of land uses in the vicinity
- <u>Measures to mitigate the bulk and scale of the utility,</u> <u>including screening, colour and finish treatment, earth</u> <u>mounding and / or planting, viewing distances, the location of</u> <u>support structures.</u>
- <u>Whether the size and scale of the proposal is generally</u> <u>compatible with other development in the area.</u>
- <u>Any adverse effects on traffic and pedestrian safety including</u> <u>sight lines and the visibility of traffic signage.</u>
- Where antennas are proposed to be sited on the top of a building, the extent to which they can be designed or screened so that they form an integral part of the total building design.
- The extent to which the affected persons / community has been consulted with.

30.17

Policy 16.4.8, 16.4.9, 16.4.12

Underground gas distribution and transmission pipelines at a pressure exceeding 2000 kilopascals, including aerial crossings of bridges, structures or streams, and ancillary equipment, including compressor compounds with compressor houses

Council will restrict its discretion to, and may impose conditions on:

- <u>Risks to public health and safety</u>
- <u>Any effect on heritage and cultural values</u>
- <u>Amenity effects, including noise, vibration, odour, dust, earthworks and lighting</u>
- Cumulative effects
- Any potential interference with public use and enjoyment of the land and the operation of land uses in the vicinity
- <u>The extent to which alternative locations, routes or other</u> options have been appropriately considered.
- Earthworks and erosion and sediment control.
- <u>The extent to which the work is able to be conveniently</u> <u>accommodated underground without adversely affecting</u> <u>existing underground utility services.</u>

• <u>Any adverse effects on an identified heritage site or an area of native vegetation.</u>

30.18 Water reservoirs

Policy 16.4.8, 16.4.9, Council will restrict its discretion to, and may impose conditions on:

- Risks to public health and safety
- Design and external appearance
- Any effect on heritage and cultural values
- Visual effects including impacts on:
 - The residential and recreational use of land in the vicinity of the proposed utility:
 - The existing character, landscape, streetscape and amenity values of the locality;
 - <u>key public places, public viewing points, and significant</u> recreational areas
- <u>Amenity effects, including noise, vibration, odour, dust,</u> <u>earthworks and lighting</u>
- <u>Cumulative effects</u>
- <u>Any potential interference with public use and enjoyment of the land and the operation of land uses in the vicinity.</u>
- <u>Measures to mitigate the bulk and scale of the utility,</u> including screening, colour and finish treatment, earth mounding and / or planting, viewing distances, the location of support structures.
- <u>The extent to which alternative locations, routes or other</u> options have been appropriately considered.
- <u>Rehabilitation of the site following any construction or future</u> <u>maintenance period.</u>
- <u>The extent to which the affected persons / community has</u> been consulted with.
- Earthworks and erosion and sediment control.
- <u>Any adverse effects on an identified heritage site or an area of native vegetation.</u>

<u> 30.19</u>

16.4.12

<u>Policy 16.4.8, 16.4.9, 16.4.9</u>

<u>Meteorological enclosures and buildings, automatic weather</u> <u>stations and anemometer masts voluntary observer sites and</u> <u>associated microwave links that are not permitted activities</u>

Council will restrict its discretion to, and may impose conditions on:

- The degree, extent and effects of the non-compliance with the Permitted Activity Standards
- <u>Risks to public health and safety</u>
- <u>The maximum height of the mast and area or diameter of any</u> <u>antenna</u>
- Design and external appearance
- <u>Any effect on heritage and cultural values</u>
- Visual effects including impacts on:
 - The residential and recreational use of land in the vicinity of the proposed utility;
 - <u>The existing character, landscape, streetscape and</u> <u>amenity values of the locality;</u>
 - key public places, public viewing points, and significant recreational areas
- <u>Amenity effects, including noise, vibration, odour, dust,</u> <u>earthworks and lighting</u>
- <u>Cumulative effects</u>
- Any potential interference with public use and enjoyment of

the land and the operation of land uses in the vicinity

- <u>Measures to mitigate the bulk and scale of the utility,</u> including screening, colour and finish treatment, earth mounding and / or planting, viewing distances, the location of support structures.
- Whether the size and scale of the proposal is generally compatible with other development in the area.
- <u>The extent to which the affected persons / community has</u> been consulted with.

Matters for Consideration

30.1320

Matters that may be relevant in the consideration of any <u>discretionary</u> <u>activity</u> resource consent may include the following:

Network Utilities

- Compliance with the Code of Practice for Civil Engineering Works.
- Whether alternative locations or other options are physically or technically possible.
- Whether the size and scale of the development is generally compatible with other development in the area.
- Whether the local topography, existing vegetation or screening will diminish the impact of the new structure.
- The extent to which the development can be designed to reflect the form of development in the immediate locality and/or harmonise with the natural or built features of the area.
- The degree to which the proposed activity is appropriately located in terms of visibility. Where ridges or hilltop locations are necessary, Council encourages the co-siting of utilities to reduce the effect on visual amenities.
- Whether the utility structure damages habitats or ecosystems or causes the loss of significant vegetation that contributes to the amenity of an area.
- <u>All those matters listed for restricted discretionary activities.</u>

Traffic generation and access

- Accessibility for public transport, cyclists and pedestrians.
- Any vehicle/pedestrian/cyclist conflict.
- Accessibility to the site.
- Current traffic problems in the area.
- Ability of the existing roading network to cater for increased traffic generation.
- Ingress and egress to and from the site.
- Neighbourhood amenity.

Roads

- Any need to create a public road.
- Traffic Safety.
- Alternative routes or alignments.
- Visual Impacts.
- Environmental Effects.

Amendment 10: New Chapter 30A: Rules for Renewable Energy Generation

30A RULES FOR RENEWABLE ENERGY GENERATION

Activities Table

<u>30A.1</u> <u>Activities</u>	Activity Status	Zone
<u>The operation and maintenance of existing</u> renewable energy generation facilities.	<u>P</u>	<u>A11</u>
Any solar panel mounted to any building	<u>P</u>	<u>All</u>
Roof-mounted small scale wind turbines	<u>P</u>	<u>All</u>
Freestanding small scale wind turbines	<u>P</u>	<u>A11</u>
<u>Temporary renewable energy assessment and</u> <u>research structures</u>	<u>P</u>	<u>Rural</u> <u>Open Space</u> <u>Business</u> Industrial
<u>Solar panels that do not comply with one or more of</u> <u>the permitted activity standards</u>	<u>RD</u>	<u>A11</u>
<u>Small scale roof-mounted and freestanding wind</u> <u>turbines that do not comply with one or more of the</u> <u>permitted activity standards</u>	RD	<u>A11</u>
Temporary renewable energy assessment and research structures that do not comply with one or more of the permitted activity standards and are not located on a site	RD	<u>A11</u>
 Land based structures that support in-stream hydro or marine energy generation provided that they are not located: on a site containing an item listed in the Schedule of Significant Heritage Features 26.8 within an esplanade reserve or strip within any riparian setback on any legal road (formed or unformed) 	<u>RD</u>	<u>Rural</u> <u>Open Space</u> <u>Business</u> <u>Industrial</u>
Community scale renewable energy generation activities, including associated construction and commissioning activities provided that they are not located: • on a site containing an item listed in the Schedule of Significant Heritage Features 26 • within an esplanade reserve or strip • on any legal road (formed or unformed) and any structures are set back at a distance of not less than three times the height of any generating device or support structure from any site boundary. The installation or upgrading of any commercial	<u>RD</u>	Rural
scale renewable energy generation activities, including associated construction and commissioning activities, whereby for any activity where wind is the energy source, the activity	_	

complies with NZS6808:2010.		
Any renewable energy generation activity that is not	D	<u>A11</u>
otherwise provided for as a permitted, restricted		
discretionary or non-complying activity.		
The installation of or upgrading of any community	NC	<u>A11</u>
and commercial scale renewable energy generation		
activities, including associated construction and		
commissioning activities, renewable energy		
generation activity:		
 located on a site containing an item listed in 		
the Schedule of Significant Heritage		
Features 26.8; or		
 located within the Southern Hills Overlay; or 		
• that does not comply with NZS6808:2010,		
where wind is the energy source.		
Any renewable energy generation activity located on	NC	<u>A11</u>
a site containing an item listed in the Schedule of		
Significant Heritage Features 26.8		

Key

Permitted activity which complies with standards for permitted activities specified in the Plan

С Controlled activity

Р

- RD Restricted Discretionary activity
- D Discretionary activity
- **NC** Non Complying activity

Standards for Permitted Activities

<u>30A.2</u> The operation and maintenance of existing renewable energy generation activities

<u>Policy</u> 16A.4.4 The permitted activity standards for the zone must be complied with.

<u>30A.3</u> Any solar panel mounted to any building

<u>Policy</u> 16A.4.2

•

- The permitted building height may be exceeded by no more than 1m
- The sunlight access standard may be exceeded by no more than 1m • (measured vertically)
- *Where located on a heritage building listed in the Schedule of ٠ Significant Heritage Features 26.8, any solar panel shall:
 - be located on a roof plane which is not visible from any adjacent

<u>public areas</u>

• <u>aligned with the plane of the roof.</u>

<u>30A.4</u> <u>Roof-mounted small scale wind turbines</u>

- <u>Policy</u> 16A.4.2
- The sunlight access, setback and noise standards for the zone in which the site is located shall be complied with.
- The permitted building height may be exceeded by no more than 2m
- <u>*Shall not be located on a site containing an item listed in the Schedule</u> of Significant Heritage Features 26.8

30A.5 Freestanding small scale wind turbines

- <u>Policy</u> 16A.4.2
- <u>The sunlight access, setback and noise standards for the zone that the</u> site is located in must be complied with.
- <u>The permitted building height for the zone may be exceeded by up to,</u> <u>but no more than, 2m.</u>
- There shall be no more than one turbine per site
- <u>*Shall not be located on a site containing an item listed in the Schedule of Significant Heritage Features 26.8.</u>

<u>30A.6</u> <u>Temporary renewable energy assessment and research structures</u>

<u>Policy</u> 16A.4.3

- <u>Shall comply with the following New Zealand Standards as well as any others relevant to emissions:</u>
 - NZS6808:2010 Acoustics Wind Farm Noise
 - NZS 2772:1999 Radiofrequency Fields
- <u>Must comply with the noise standards for the zone in which they are located.</u>
- Shall not be located on a site containing an item listed in the Schedule of Significant Heritage Features 26.8.
- Any anemometer must not exceed 80 metres in height.
- <u>All structures shall comply with the sunlight access height control</u> plane and setback requirements for the zone in which they are located.
- Any public road damaged in such investigation or assessment activity shall be reinstated to a condition of a similar or improved standard to that which existed prior to commencement of the work;
- <u>No such investigation or assessment period shall exceed five years</u>
- All equipment and structures shall be removed at the end of the investigation and the site shall be restored and rehabilitated to a condition no less than that which existed prior to the works commencing.

Restricted Discretionary Activities

30A.7	Solar Panels that do not comply with one or more of the permitted
<u></u>	activity standards
<u>Policies</u> <u>1644.1.</u> <u>16A.4.2</u> <u>16A.4.5</u>	 <u>Any positive effects to be derived from the activity;</u> <u>The contribution to achieving national, regional and local energy policy objectives and renewable energy targets;</u> <u>Health and safety;</u> <u>Layout, design and location of proposed structure;</u> <u>Effects on historic heritage;</u> <u>Visual, character and amenity effects;</u> <u>Adequacy of the methods of mitigation/remediation or ongoing management.</u>
30A.8	Small scale roof-mounted and freestanding wind turbines that do
	not comply with one or more of the permitted activity standards
<u>Policies</u> <u>16A4.1,</u> <u>16A.4.2,</u> <u>16A.4.5</u>	 <u>All those matters under Rule 30A.11, and:</u> <u>Extent of any earthworks</u> <u>Noise and lighting effects</u>
30A.9	Temporary renewable energy assessment and research structures.
	Land based structures that support in-stream hydro or marine
<u>Policies</u> 16A4.1.	energy generation and Community scale renewable energy
<u>16A.4.3,,</u>	commissioning activities
<u>16A.4.5</u>	<u>Council will restrict its discretion to, and may impose conditions on:</u>
	• All those matters under Rule 30A.10, and:
	<u>Traffic effects</u>
	Effects on public access
	Effects on natural character

• <u>Suitability of the site for the proposed activity, including</u> <u>consideration of geotechnical and natural hazard constraints</u>

Matters for Consideration

<u>30A.10</u>

<u>Matters that may be relevant in the consideration of any discretionary</u> <u>activity resource consent may include the following:</u>



•	Any positive effects to be derived from the activity;
•	The contribution to achieving national, regional and local energy
	policy objectives and renewable energy targets;
•	Health and safety;
•	Noise effects, including compliance with NZS 6808:2010
	(Acoustics – wind farm noise);
•	The impact on reserves and other protected public and private
	land, and recreation areas, community facilities, infrastructure
	and services;
•	The impact on public access to, and along, the margins of the
	coast, lakes and rivers or to natural and physical features;
•	The impact on landscape values and natural features;
•	The potential impact of natural hazard events and the effect the
	activity itself may have on exacerbating or relieving natural
	hazards;
•	Any geotechnical constraints of the affected area:
•	Impacts of earthworks and the modification of natural landforms.
	including proposed remedial and mitigation measures:
•	Impacts on the amenity values of the surrounding environment.
	including a consideration of electromagnetic interference.
	vibration, aviation navigation lighting and turbine blade shadow
	or glare flicker;
•	Ecological impacts, particularly impacts on water bodies, impacts
	associated with native vegetation removal and impacts on
	indigenous avifauna, indigenous fauna and their habitats;
•	Impacts on archaeological and historic features and items, and
	sites of significance to tangata whenua;
•	Traffic impacts (including construction and post-construction
	traffic) and impacts on the roading network, including the nature
	and extent of vehicle movements, access, management and
	mitigation measures; road safety and levels of service; and
	impacts associated with traffic distraction;
•	Impacts on aviation and navigation;
•	Construction effects, including construction noise (and the ability
	to meet NZS 6803:1999 Acoustics - Construction Noise);
	Any cumulative effects

Amendment 11: Amendments to Chapter 35: Definitions

35	DEFINITIONS	
<u>Anemometer</u>	 means a mast and supporting sensors for the purpose of wind resource measurement. This includes guy wires and various meteorological instruments to be erected at varying heights, including: Anemometers to measure the average wind speed, wind gust speeds, turbulence intensity and wind shear; wind vanes to measure wind direction; and other meteorological instruments to measure temperature, air pressure, humidity and rainfall. 	
Antenna	any telecommunication or radio communication apparatus, including satellite dishes and aerials, used for transmission or reception, including the antenna mounting but not any supporting mast. means antenna as defined in the Resource Management (National Environmental Standard for Telecommunications Facilities) Regulations 2008.	
	 <u>An antenna does not include:</u> <u>Devices used in amateur radio configurations</u> <u>Devices used only for television reception; and</u> <u>Any other device not otherwise defined above that is less</u> <u>than 1.5m² in area</u> 	
	Advice note: The mountings of any antenna and any radiofrequency equipment or similar device shall not be included in the measurement of area of diameter of each antenna, provided that the radiofrequency unit or similar device is smaller in area or diameter than the antenna itself. Any antenna only need meet the area or diameter measurement, as appropriate to the type of antenna and the measurement is of each individual antenna and is not a cumulative measurement.	
Building	 any structure whether temporary or permanent, movable or immovable, which, in addition to its ordinary and usual meaning, includes the following: Any structure of over 5m² in area with a height of more than 1.0m 	
	 Any fence or wall with a height of more than 2m. Any retaining wall with a height of more than 1.5m above the finished ground level. Any tank or pool, and any structural support: 	

- (i) Which has a capacity of not less than 25,000 litres and is supported directly by the ground.
- (ii) Which has a capacity of 2,000 litres or more and is supported at a height of more than 2.0 metres from the base of its structure.
- (iii) Which has a capacity of 500 litres or more and is supported at a height of more than 4.0 metres from the base of its supporting structure.

This definition does not apply to <u>network</u> utilities as defined in this chapter.

Cabinet

<u>Commercial scale</u> renewable energy

Community scale

renewable energy

Construction and

Commissioning

activities:

generation

generation

activities

means a box-shaped structure which houses radio and telecommunication equipment, electrical equipment, equipment associated with the continued operation of network utilities and includes single transformers and associated switching gear distributing electricity at a voltage up to, and including, 100KV.

means the land, buildings, substations, turbines, structures, underground cabling earthworks, access tracks and roads associated with the generation of electricity from a renewable energy source and the operation of the renewable energy generation activity. It does not include:

- <u>Small scale wind turbines turbines of less than 5kW</u>
- <u>Community scale renewable energy generation activities</u>
 <u>Any cabling required to link the wind energy facility to</u> the point of entry into the electricity network, whether transmission or distribution in nature.

means renewable energy generation for the purpose of supplying electricity to a whole community which is not connected to the distribution network ('off grid'); or to supplying an immediate neighbourhood in an urban area with some export back into the distribution network.

in respect of renewable electricity generation activities includes those activities directly involved with the building and operation of a new renewable electricity generation activity. This includes site preparation, earthworks, quarrying, concrete batching, plant construction, road construction and widening, traffic generation, reservoir formation, clearance or inundation of vegetation, but specifically excludes investigative activities such as geological sampling, surveys and geotechnical investigations.

Activities associated with "construction and commissioning" includes rapid and temporary population increases and the associated effects on infrastructure and community facilities; the need to reroute or relocate network utilities and community facilities;; the need to construct new infrastructure including the system of electricity conveyance transmission (including substations) required to convey electricity to the distribution network and/or the national grid as provided for in the definition of 'renewable electricity generation activity.

Comment [GS49]: S6.2 WELL

<u>Distribution</u> <u>network</u>	For the purpose of Chapter 30A, has the same meaning as in the National Policy Statement for Renewable Electricity Generation and means a distributor's lines and associated equipment used for the conveyance of electricity on lines other than lines that are part of the national grid.	
<u>Distributor</u>	for the purpose of Chapter 30A, has the same meaning as in the National Policy Statement for Renewable Electricity Generation and means a business engaged in distribution of electricity.	
Line	means 'line' as defined in Section 5 of the Telecommunications Act 2001 and or Section 2 of the Electricity Act 1992.	Comment [GS50]: S6.4 Transpower
<u>Maintenance</u>	as it applies to network utilities, means the replacement, repair or renewal of existing network utilities and where the effects of that utility remain the same or similar in character, intensity and scale, and excludes 'minor upgrading' and 'upgrading'.	
Mast	any mast, pole, tower or similar structure which is used to facilitate the operation of a utility. any pole, tower or similar structure which is fixed to the ground specifically designed to carry an antenna to facilitate the transmission of—telecommunication and radiocommunication signals.	
<u>Minor above</u> ground line	means a line that provides an above ground connection to a site, including any connection to a building within that site, from an existing or permitted new above ground line provided that no more than one new support structure is required for that connection.	
Minor upgrading	any work necessary to continue or improve the operation and/or functioning of an existing activity where, as a result of the work, the activity concerned would have effects that are the same or similar in character, intensity, and scale as those that existed before the work was carried out. In respect of above ground cables and lines, minor upgrading includes the provision of one new or additional connection per site within existing developed areas already serviced by existing above ground lines.	
	 (i) the addition of circuits and conductors; (ii) the reconductoring of the line with higher capacity conductors; (iii) the resagging of conductors; 68 	

(iv) the addition of longer or more efficient insulators;

- (v) the addition of earthwires which may contain telecommunication lines, earthpeaks and lightning rods;
- (vi) an increase in the voltage of the line where the line was originally constructed to operate at the higher voltage but has been operating at a reduced voltage.

means an increase in the carrying capacity, efficiency or security of electricity and telecommunication lines, which utilise the existing or replacement support structures and includes:

(1) the reconductoring of the line with higher capacity conductors; and

(2) the resagging of conductors; and

- (3) the addition of longer and more efficient insulators; and
- (4) A support structure replacement within a similar location as 5 metres of the support structure that is to be replaced; and
- (5) The addition of earthwires, which may contain telecommunication lines, earthpeaks and lightning rods; and
- (6) The addition of electrical or telecommunication fittings; and
- [7] <u>Tower Support structure replacement in the same</u> location or within the existing alignment of the transmission line corridor; and
- (8) The replacement of existing cross arms, including with cross arms of an alternative design; and
- (9) An increase in tower height to achieve compliance with the clearance distances specified in NZECP34:2001; and
- (10) an increase in the height of replacement poles in the road reserve by a maximum of 1m, for the purpose of achieving road controlling authority clearance requirements, provided the permitted height in Rule 30.4 is not exceeded; and
- (11) an increase in voltage of electricity lines from 11kV to no more than 33kV.and
- (12) the addition of a new overhead telecommunication fibre optic line provided that:
 - (i) the maximum number of fibre optic lines on existing support structures does not exceed two lines;
 - (ii) the diameter of new fibre optic lines does not exceed 25mm; and
 - (iii) the location of the new fibre optic line is consistent with the following figure

Figure X: Location of new fibre optic line

Comment [GS51]: S5.18 Chorus; S6.7 and UFS3.36 WELL

Comment [GS52]: S4.37 Transpower

Comment [GS53]: Clause 20



LEGEND



Minor upgrading shall not include:

- <u>Any increase in the voltage of the line unless the line</u> was originally constructed to operate at the higher voltage but has been operating at a reduced voltage, or
 <u>Any increase in any individual wire, cable, or other</u>
- (ii) Any increase in any individual wire, cable, or other similar conductor to a diameter that exceeds 35mm, or
 (iii) The bundling together of any wire, cable, or other similar
- ii) The bundling together of any wire, cable, or other similar conductor so that the bundle exceeds 43mm in diameter, or
- (iv) The addition of any new circuits, lines or utility structures, where this results in an increase in the number of circuits, lines or utility structures except as provided for in (12) above...
- Note: The Resource Management (National Environmental Standards for Electricity Transmission Activities) Regulations 2009 applies to the existing National Grid, and applies to all transmission lines that were operational, or able to be operated, on 14 January 2010. (vii)

Comment [GS54]: S5.18 Chorus

Comment [GS55]: S5.18 Chorus
Comment [GS56]: S4.37
Transpower

Comment [GS57]: S4.37 Transpower

Network Utility	means any activity undertaken by a network utility operator as
	defined in s166 of the RMA, relating to:
	[1] Distribution or transmission by pipeline of natural or
	manulactured gas, perforedini, biolider of geomerinar
	(ii) Telecommunication as defined in section 5 of the
	Telecommunications Act 2001: or radiocommunications
	as defined in section (2)(1) of the Radiocommunications
	Act 1989, or
	(iii) the provision, operation and maintenance of works <u>as</u>
	defined in section 2 of the Electricity Act 1992 for the
	conveyancing of electricity , as defined in section 2 of the
	(iv) <u>Electricity Act 1992, or</u> Comment [GS58]: S4.38
	<u>or</u>
	(v) Sewerage or drainage reticulation, or
	(vi) Construction-and operation of roads and railway lines, or
	(vii) The operation of an airport as defined by the Airport
	Authorities Act 1966, or
	(viii) the provision of any approach control service within the
	meaning of the Civil Aviation Act 1990, or
	(ix) Undertaking a project of work described as a network
	Management Act 1991
	and includes:
	Lighthouses, navigation aids, beacons, signal and trig
	stations and natural hazard emergency warning devices.
	Meteorological services,
	all associated structures; and
	regionally significant network utilities
Network utility	means any structure associated with a network utility and
structure	includes, but is not limited to, pipes, valves, meters, regulator
	stations, support poles and towers for lines, transformers (other Comment IGS591: \$4.39
	than pole mounted transformers), substations (other than Transpower
	overhead substations), compressor stations, pumping stations,
	navigational aids, meteorological installations, containers,
	cabinets, and similar structures. It does not include lines,
	antennas and masts.
Regionally	includes:
significant	• pipelines for the distribution or transmission of natural
network utilities	or manufactured gas or petroleum
	• the National Grid, as defined by the National Policy
	Statement on Electricity Transmission
	• facilities for the generation and transmission of
	electricity where-it is supplied to the network, as
	defined by the Electricity Governance Rules 2003
	• the local authority water supply network and water
	treatment plants
	• the local authority wastewater and stormwater
	network s , systems and wastewater treatment plants
	• <u>the Strategic Transport Network, detailed in Appendix 1</u>
	to the Wellington Regional Land Transport Strategy
	<u>2010-2040</u>
Demonstelle	has the same manifest as an day the National D 11 Of the st
MANAMANIA	nas the same meaning as under the National Policy Statement

<u>electricity</u> <u>generation</u> <u>activities</u>	for Renewable Electricity Generation and means the construction, operation and maintenance of structures associated with renewable electricity generation. This includes small and community-scale distributed renewable generation activities and the system of electricity conveyance required to convey electricity to the distribution network and/or the national grid and electricity storage technologies associated with renewable electricity.
<u>Renewable</u> <u>electricity</u> generation	means generation of electricity from solar, wind, hydro, geothermal, biomass, tidal, wave, or ocean current sources.
<u>Renewable energy</u>	is defined in section 2 of the Resource Management Act 1991
<u>Small Scale</u> <u>Renewable Energy</u> <u>Generation</u>	means small scale renewable energy generation development for the purpose of using or generating electricity on a particular site (single household or business premise) with or without exporting back into the distribution network.
<u>Small scale wind</u> <u>turbines</u>	means wind turbines that are capable of generating up to 10kW of electricity.
<u>Solar panel</u>	means a panel exposed to radiation from the sun, used to heat water or, when mounted with solar cells, to produce electricity direct
<u>Temporary</u> <u>renewable energy</u> <u>assessment and</u> <u>research structures</u>	 means structures for the investigation, identification and assessment of potential sites and energy sources for renewable electricity generation by existing and prospective generators and includes the following activities: Erecting an anemometer mast. Digging test pits, drilling boreholes, constructing investigation drives and removing samples to investigate geological conditions. Installing instruments into drill holes for monitoring groundwater levels and land movement. Erecting survey monuments and installing instruments to monitor land movement. Erecting telemetry stations for the transmission of instrument data. Installing microseismic stations to measure microseismic activity and ground noise. Erection of signs or notices giving warning of danger

Upgrading

as it applies to network utilities, upgrading means the replacement,

repair, renewal or improvement <u>or physical works that result in a</u>	<u>n 🖶</u> 🛛 Comment [GS60]: S2.21 KiwiRa	uil
increase in carrying capacity, operational efficiency, security or s	afety of	
existing network utilities but but excludes:	Comment [GS61]: Clause 20	
 <u>'maintenance' (as it relates to network utilities); and</u> 		
• 'minor upgrading': and		

- .
- any other activity specifically otherwise provided for under Rule 30.1; and
- any iner on, increase or change is specifically provided for and would comply with the applicable permitted activity standard specified in Rules 30.1, 30.5 and 30.6. Comment [GS62]: S

Utility

one or more of the following:

- The distribution or transmission by pipeline of natural or •____ manufactured gas, petroleum or geothermal energy;
- Networks for the purpose of telecommunication as defined in section 5 of the Telecommunications Act 2001 or radio-communication as defined in section 2(1) of the Radiocommunications Act 1989;
- The distribution or transmission of electricity by an electricity operator or electricity distributor as defined in section 2 of the Electricity Act 1992;
- The collection, treatment, distribution or transmission . by pipeline of water for supply (including irrigation);
- The development, operation and maintenance of a . drainage or sewerage system;
- The construction, operation and maintenance of a road or railway line;

and includes all structures in relation to the operation of the utility.